UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

SCHEDULE 13D

Under the Securities Exchange Act of 1934 (Amendment No. 2)*

BioScrip, Inc.

(Name of Issuer)

Common Stock, \$0.0001 par value

(Title of Class of Securities)

09069N108

(CUSIP Number)

Alison S. Ressler, Esq. Rita-Anne O'Neill, Esq. Sullivan & Cromwell LLP 1888 Century Park East, Suite 2100 Los Angeles, California 90067 (310) 712-6600

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

March 14, 2019

(Date of Event Which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§ 240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box. o

Note: Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See Rule 13d-7 for other parties to whom copies are to be sent.

* The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

Persons who respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB control number.

1.	Names of R ASSF IV A	Reporting Persons IV B Holdings, L.P.	
2.	Check the Appropriate Box if a Member of a Group (See Instructions)		
	(a)	o	
	(b)	x	
3.	SEC Use O	nly	
4.	Source of F OO	funds (See Instructions)	
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o	
6.	Citizenship or Place of Organization Delaware		
	7.	Sole Voting Power 0	
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Items 3, 4, 5 and 6)	
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0	
	10.	Shared Dispositive Power 7,188,615 (See Items 3, 4, 5 and 6)	
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Items 3, 4, 5 and 6)		
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o		
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)		

1.	Names of R ASSF Oper	Leporting Persons rating Manager IV, L.P.
2.	Check the A	Appropriate Box if a Member of a Group (See Instructions)
	(a)	0
	(b)	X
3.	SEC Use O	nly
4.	Source of F	unds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Names of R Ares Manag	Reporting Persons gement LLC	
2.	Check the Appropriate Box if a Member of a Group (See Instructions)		
	(a)	o	
	(b)	x	
	. ,		
3.	SEC Use O	nly	
4.	Source of F	unds (See Instructions)	
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o	
6.	Citizenship or Place of Organization Delaware		
	7.	Sole Voting Power 0	
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)	
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0	
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)	
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)		
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o		
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)		

1.	Names of R Ares Manag	Reporting Persons gement Holdings L.P.
2.	Check the A	Appropriate Box if a Member of a Group (See Instructions)
	(a)	0
	(b)	x
3.	SEC Use O	nly
4.	Source of F	funds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Names of R Ares Holdc	Deporting Persons to LLC
2.	Check the Appropriate Box if a Member of a Group (See Instructions)	
	(a)	o
	(b)	X
3.	SEC Use O	nly
4.	Source of F	runds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Names of R Ares Holdin	Reporting Persons ngs Inc.
2.		Appropriate Box if a Member of a Group (See Instructions)
	(a)	0
	(b)	X
3.	SEC Use O	nly
4.	Source of F	funds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Ares Manag	Reporting Persons gement Corporation
2.		Appropriate Box if a Member of a Group (See Instructions)
	(a)	0
	(b)	X
3.	SEC Use O	nly
4.	Source of Funds (See Instructions) OO	
5.	Check if Di	isclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Names of R Ares Voting	Reporting Persons g LLC
2.	Check the A	Appropriate Box if a Member of a Group (See Instructions)
	(a)	0
	(b)	x
3.	SEC Use O	nly
4.	Source of F	unds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Names of R Ares Manag	Reporting Persons gement GP LLC
2.	Check the Appropriate Box if a Member of a Group (See Instructions)	
	(a)	o
	(b)	X
3.	SEC Use O	nly
4.	Source of F	runds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

1.	Names of R Ares Partne	Reporting Persons ers Holdco LLC
2.		Appropriate Box if a Member of a Group (See Instructions)
	(a)	0
	(b)	X
3.	SEC Use O	nly
4.	Source of F	funds (See Instructions)
5.	Check if Di	sclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) o
6.	Citizenship or Place of Organization Delaware	
	7.	Sole Voting Power 0
Number of Shares Beneficially	8.	Shared Voting Power 7,188,615 (See Item 3, 4, 5 and 6)
Owned by Each Reporting Person With	9.	Sole Dispositive Power 0
	10.	Shared Dispositive Power 7,188,615 (See Item 3, 4, 5 and 6)
11.	Aggregate Amount Beneficially Owned by Each Reporting Person 7,188,615 (See Item 3, 4, 5 and 6)	
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) o	
13.	Percent of Class Represented by Amount in Row (11) 5.6% (See Item 3, 4, 5 and 6)	

Explanatory Note

This Amendment No. 2 to the statement on Schedule 13D amends and supplements the statement on Schedule 13D filed by certain of the Reporting Persons on July 10, 2017 (the "Original Schedule 13D"), as amended by Amendment No. 1 to the Original Schedule 13D filed by the Reporting Persons on February 25, 2019 (as so amended, the "13D Filing", and together with this Amendment No. 2, the "Schedule 13D"). Except as amended in this Amendment No. 2, the 13D Filing remains in full force and effect. Terms defined in the 13D Filing are used in this Amendment No. 2 as so defined, unless otherwise defined in this Amendment No. 2.

Item 2. Identity and Background

The last sentence of Item 2(a) of the 13D Filing is hereby amended and restated as follows:

The Reporting Persons have entered into a joint filing agreement, dated as of March 15, 2019, a copy of which is attached hereto as Exhibit 99.3.

Item 5. Interest in Securities of the Issuer

Item 5 of the 13D Filing is hereby amended and restated in its entirety as follows:

(a) Aggregate number and percentage of securities.

As of the date hereof, ASSF IV directly holds 7,188,615 shares of Common Stock. The Reporting Persons, as a result of the relationships described in Item 2, may be deemed to directly or indirectly beneficially own the shares of Common Stock held by ASSF IV, which are reported on the cover pages to this Schedule 13D for such Reporting Persons, as applicable. See also items 11 and 13 of the cover pages to, and Item 2 of, this Schedule 13D for the aggregate number of shares of Common Stock and percentage of Common Stock beneficially owned by each of the Reporting Persons. The applicable ownership percentages reported in this Schedule 13D are based on 128,155,291 shares of Common Stock outstanding as of March 7, 2019, as reported by the Issuer on its Annual Report on Form 10-K filed on March 15, 2019.

- (b) **Power to vote and dispose.** See items 7 through 10 of the cover pages to, and Item 2 of, this Schedule 13D for the number of shares of Common Stock beneficially owned by each of the Reporting Persons as to which there is sole or shared power to vote or to direct the vote, and sole or shared power to dispose or to direct the disposition.
- (c) **Transactions within the past 60 days.** The information set forth in Items 3, 4 and 6 in this Schedule 13D is incorporated herein by reference. Except as otherwise described in this Schedule 13D, none of the Reporting Persons has effected any transaction related to the Common Stock during the past 60 days.
- (d) Certain rights of other persons. Except as otherwise described in this Schedule 13D, no other person has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, securities covered by this Schedule 13D.
- (e) Date ceased to be a 5% owner. Not applicable.

Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the Issuer

Item 6 of the 13D Filing is hereby amended by adding the following immediately prior to the last paragraph thereof:

On March 14, 2019, the Issuer entered into an Agreement and Plan of Merger (the "Merger Agreement") with HC Group Holdings II, Inc., a Delaware corporation ("HC Group Holdings II") and certain other parties. The Merger Agreement provides for, among other things, the merger of a newly-formed subsidiary of the Issuer with and into HC Group Holdings II (the "First Merger"), and immediately following the First Merger, the merger of HC Group Holdings II with and into another newly-formed subsidiary of the Issuer (together with the First Merger, the "Mergers").

In connection with the Merger Agreement, ASSF IV and an account managed by ASSF Operating Manager IV (the "Managed Account") each entered into:

- a letter agreement with the Issuer (a "Letter Agreement");
- the Amended and Restated Warrant Agreement, among the Issuer, ASSF IV, the Managed Account and the other holders signatory thereto (the "Warrant Amendment"); and
- · an amendment to the Registration Rights Agreement, among the Issuer, ASSF IV, the Managed Account and the other holders signatory thereto (the "RRA Amendment").

In addition, in connection with the Mergers, Ares Management LLC, on behalf of one or more affiliated funds, investment vehicles and/or managed accounts, and one of its subsidiaries, Ares Capital Management LLC, on behalf of one or more affiliated funds, investment vehicles and/or managed accounts (collectively, the "Ares Parties"), entered into certain debt financing commitments (the "Debt Commitments" and, together with the Letter Agreements, Warrant Amendment and RRA Amendment, the "Transaction Documents").

Letter Agreements and Warrant Amendment

Under the terms of their respective Letter Agreements, each of ASSF IV and the Managed Account, as holders of the Warrants, and the Issuer have agreed to enter into the Warrant Amendment, conditioned upon the consummation of the First Merger. Under their respective Letter Agreements, in consideration of their execution and delivery of the Warrant Amendment, and conditioned upon the occurrence of the First Merger Effective Time (as defined in the Merger Agreement), the Issuer agrees to issue to ASSF IV and the Managed Account 1,012,226 and 337,408 shares, respectively, of Common Stock of the Issuer (the "Amendment Shares"), promptly following the occurrence of the First Merger Effective Time. Each of ASSF IV and the Managed Account agrees, pursuant to the terms of their respective Letter Agreements, not to sell their Warrants or exercise their exchange right under any Warrant prior to the earlier of the First Merger Effective Time or the termination of the Merger Agreement according to its terms.

The Warrant Amendment amends and restates the Warrant Agreement, in connection with, and conditioned upon, the consummation of the First Merger in accordance with the Merger Agreement. The Warrant Agreement is amended and restated so as to fix the aggregate number of shares of Common Stock issuable upon exercise of the Warrants at 8,287,317 shares. The number of shares of Common Stock issuable to ASSF IV and the Managed Account upon exercise of the Warrants following the effectiveness of the Warrant Amendment is fixed at 4,520,354 and 1,506,784, respectively. The Expiration Time and Exercise Price are unchanged from the Warrant Agreement. The Warrant Amendment has customary cash and cashless exercise provisions and anti-dilution provisions customary for a fixed-share warrant, including an anti-dilution adjustment in the event that the Issuer issues shares of Common Stock or Common Stock Equivalents (as defined in the Warrant Amendment) at an effective price per share less than the current market price of the Common Stock or the applicable exercise price of the Warrants, subject to certain exceptions. Other terms of the Warrant Agreement remain unchanged.

RRA Amendment

The RRA Amendment amends the Registration Rights Agreement to include the Amendment Shares under the definition of "Registrable Securities", and to specify the date after which registration rights are available with respect to the Amendment Shares. Other terms of the Registration Rights Agreement remain unchanged. The effectiveness of the RRA Amendment is conditioned upon the consummation of the First Merger.

Debt Commitments

Bank of America, N.A., Merrill Lynch, Pierce, Fenner & Smith Incorporated, Ares Capital Management LLC, on behalf of one or more affiliated funds, investment vehicles and/or managed accounts, Broad Street Loan Partners III, L.P., Broad Street Loan Partners III Offshore, L.P., Broad Street Loan Partners III Offshore, L.P., and Broad Street Senior Credit Partners II, L.P. (collectively, the "Lenders") provided a financing commitment letter dated March 14, 2019 (the "Commitment Letter") to HC Group Holdings II pursuant to which the Lenders, subject to the terms and conditions in the Commitment Letter, committed an aggregate amount of up to \$1,075 million toward senior secured credit facilities, consisting of a first lien term loan facility in an aggregate principal amount of up to \$925 million and an asset-based revolving credit facility in an aggregate principal amount of up to \$150 million, which will be used to fund a portion of the consideration payable in connection with the Mergers, to repay certain of the Issuer's existing indebtedness and certain of HC Group Holdings II's existing indebtedness, working capital and general corporate purposes and to pay fees and expenses associated with closing the financing of the Mergers.

Certain funds, investors, entities and accounts that are managed, sponsored or advised by Goldman Sachs & Co. LLC or its affiliates and the Ares Parties (collectively, the "Investors"), provided a financing commitment letter dated March 14, 2019 (the "Notes Commitment Letter") to HC Group Holdings II pursuant to which the Investors, subject to the terms and conditions in the Notes Commitment Letter, committed to purchase \$400 million of senior secured PIK toggle second lien floating rate notes (the "Notes"), the net proceeds of which will be used to fund a portion of the consideration payable in connection with the Mergers, to repay certain of the Issuer's existing indebtedness and certain of HC Group Holdings II's existing indebtedness and to pay fees and expenses associated with closing the financing of the Mergers.

The foregoing descriptions of the Transaction Documents do not purport to be complete and are qualified in their entirety by reference to the full text of the Transaction Documents, copies of which are filed as Exhibits 5.1 through 5.5 of this Schedule 13D and are incorporated by reference in this Schedule 13D.

Item 7. Material to be Filed as Exhibits

Item 7 of the 13D Filing is hereby amended by adding the following:

Exhibit 5.1	Letter Agreements, dated as of March 14, 2019, between BioScrip, Inc. and ASSF IV AIV B Holdings, L.P and between BioScrip, Inc. and Amari Investment Pte. Ltd.
Exhibit 5.2	Amended and Restated Warrant Agreement, dated as of March 14, 2019, by and among BioScrip, Inc., Amari Investment Pte. Ltd., ASSF IV AIV B Holdings, L.P., and the other holders signatory thereto.
Exhibit 5.3	Amendment No. 1 to Registration Rights Agreement, dated as of March 14, 2019, by and among BioScrip, Inc., Amari Investment Pte. Ltd., ASSF IV AIV B Holdings, L.P., and the other holders signatory thereto.
Exhibit 5.4	Commitment Letter, dated as of March 14, 2019, by and among HC Group Holdings II, Inc., Bank of America, N.A., Merrill Lynch, Pierce, Fenner & Smith Incorporated, Ares Capital Management LLC, Broad Street Loan Partners III, L.P., Broad Street Loan Partners III Offshore, L.P., Broad Street Loan Partners III Offshore, L.P., and Broad Street Senior Credit Partners II, L.P.
Exhibit 5.5	Notes Commitment Letter, dated as of March 14, 2019, by and among HC Group Holdings II, Inc., GS Mezzanine Partners VII, L.P., GS Mezzanine Partners VII Offshore Treaty, L.P., Broad Street Danish Credit Partners, L.P., Broad Street Credit Holdings LLC, Ares Management LLC and Ares Capital Management LLC.
Exhibit 99.3	Joint Filing Agreement, dated as of March 15, 2019, by and among the Reporting Persons.

SIGNATURE

After reasonable inquiry and to the best of my knowledge and belief, the undersigned certifies that the information set forth in this statement is true, complete and correct.

Date: March 15, 2019

ASSF IV AIV B HOLDINGS, L.P.

By: ASSF Operating Manager IV, L.P.,

Its: Manager

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ASSF OPERATING MANAGER IV, L.P.

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT HOLDINGS L.P.

By: ARES HOLDCO LLC

Its: General Partner

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES HOLDCO LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES HOLDINGS INC.

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT CORPORATION

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT GP LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES VOTING LLC

By: ARES PARTNERS HOLDCO LLC

Its: Sole Member

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES PARTNERS HOLDCO LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

EXHIBIT INDEX

Exhibit 5.1	Letter Agreements, dated as of March 14, 2019, between BioScrip, Inc. and ASSF IV AIV B Holdings, L.P and between BioScrip, Inc. and Amari Investment Pte. Ltd.
Exhibit 5.2	Amended and Restated Warrant Agreement, dated as of March 14, 2019, by and among BioScrip, Inc., Amari Investment Pte. Ltd., ASSF IV AIV B Holdings, L.P., and the other holders signatory thereto.
Exhibit 5.3	Amendment No. 1 to Registration Rights Agreement, dated as of March 14, 2019, by and among BioScrip, Inc., Amari Investment Pte. Ltd., ASSF IV AIV B Holdings, L.P., and the other holders signatory thereto.
Exhibit 5.4	Commitment Letter, dated as of March 14, 2019, by and among HC Group Holdings II, Inc., Bank of America, N.A., Merrill Lynch, Pierce, Fenner & Smith Incorporated, Ares Capital Management LLC, Broad Street Loan Partners III, L.P., Broad Street Loan Partners III Offshore, L.P., Broad Street Loan Partners III, L.P.
Exhibit 5.5	Notes Commitment Letter, dated as of March 14, 2019, by and among HC Group Holdings II, Inc., GS Mezzanine Partners VII, L.P., GS Mezzanine Partners VII Offshore, L.P., GS Mezzanine Partners VII Offshore Treaty, L.P., Broad Street Danish Credit Partners, L.P., Broad Street Credit Holdings LLC, Ares Management LLC and Ares Capital Management LLC.
Exhibit 99.3	Joint Filing Agreement, dated as of March 15, 2019, by and among the Reporting Persons.

Execution Version

BIOSCRIP, INC. 1600 Broadway, Suite 700 Denver, CO 80202

March 14, 2019

To the Holder listed on Exhibit A (the "Holder")

Re: Amended and Restated Warrant Agreement

Ladies and Gentlemen.

Reference is made to (i) that certain Warrant Agreement (the "<u>Warrant Agreement</u>"), dated June 29, 2017, by and among BioScrip, Inc., a Delaware corporation (the "<u>Company</u>") and the purchasers signatory thereto (collectively with their successors and assigns, the "<u>Original Purchasers</u>"); (ii) that certain Agreement and Plan of Merger (the "<u>Merger Agreement</u>"), dated as of the date hereof, by and among the Company, HC Group Holdings I, LLC ("<u>Omega Parent</u>"), HC Group Holdings II, Inc., Beta Sub, Inc., Beta Sub, LLC and HC Group Holdings III, Inc.; and (iii) that certain Amended and Restated Warrant Agreement, dated as of the date hereof, by and among the Company and the Holder and other Holders party thereto (the "<u>Warrant Amendment</u>"), attached hereto as Exhibit B. This letter agreement (this "<u>Letter Agreement</u>") confirms certain agreements and understandings between the Holder and the Company related to the Warrant Amendment (as defined below).

In consideration of the Holder's execution and delivery of the Warrant Amendment, and conditioned upon the occurrence of the First Merger Effective Time (as defined in the Merger Agreement), the Company hereby agrees to issue to the Holder 1,012,226 shares of Common Stock of the Company in the aggregate (the "Amendment Shares"), promptly following the occurrence of the First Merger Effective Time (as defined in the Merger Agreement). Furthermore, promptly following, and conditioned upon, the occurrence of the First Merger Effective Time, the Company and the Holder acknowledge and agree that the Registration Rights Agreement (the "Existing RRA"), entered into as of June 29, 2017, by and among the Company and the Original Purchasers, shall be amended and restated (the "RRA Amendment"), in the form of Exhibit C, with such RRA Amendment to be conditioned upon, and automatically effective as of, the occurrence of the Second Merger Effective Time, without any further action on the party of any party hereto. The Company and the Holder agree (the "RRA Interpretation") that, solely for purposes of the definition of "Registrable Securities" under the Existing RRA, during the period prior to the First Merger Effective Time, the Holder shall be deemed to beneficially own the Warrants underlying the Warrant Agreement, notwithstanding any agreement between the Holder and the Company not to exercise such Warrants. The execution and delivery of this Letter Agreement by the Holder shall be deemed a consent by the Holder, in its capacities as Holder under the Existing RRA, under Section 9 (Preservation of Rights) of the Existing RRA with respect to the registration rights to be granted to Omega Parent in connection with the Merger Agreement.

Following the execution and delivery of this Letter Agreement and until the Second Merger Effective Time, the Holder agrees not to, and to cause its controlled affiliates not to, (i) directly or indirectly

sell, transfer, assign, pledge, tender, convert, exchange or encumber or otherwise dispose of, any Warrant or (ii) exercise its exchange right under any Warrant.

The Company agrees that, to the extent an event occurs on or after the date hereof and prior to the Second Merger Effective Date which would cause an adjustment to the number of, or Exercise Price of, any Warrants under the Warrant Amendment were the Warrant Amendment then in effect (any such event, an "Adjustment Event"), then the Company shall apply, as of the Second Merger Effective Date, such adjustment to the number of, and Exercise Price of, Warrants that would otherwise be outstanding under the Warrant Amendment immediately after the Second Merger Effective Date (taking, solely for purposes of such adjustment, the number of shares of Common Stock outstanding under any adjustment calculation to be the number of shares of Common Stock outstanding on the date of such Adjustment Event, and giving cumulative effect to any prior adjustments due to other Adjustment Events).

The Company represents that, as of the First Merger Effective Time, upon issuance in accordance with, and payment pursuant to, the terms hereof, the Amendment Shares will be duly authorized, validly issued, fully paid and non-assessable. Upon issuance in accordance with, and payment pursuant to, the terms hereof, the Holder (or its affiliates, as applicable) will have good title to the Amendment Shares, free and clear of all liens, claims, encumbrances, charges, mortgages, options, pledges, security interests, hypothecations, easements, rights-of-way or encroachments of any nature whatsoever, whether voluntarily incurred or arising by operation of law ("Liens"), other than (a) transfer restrictions under federal and state securities laws and (b) Liens imposed solely due to the actions of the Holder. Upon issuance in accordance with, and payment pursuant to, the terms hereof, (i) the Amendment Shares will be issued in accordance with law (including applicable state Blue Sky laws) and the governing documents of the Company and will not be issued in violation of any preemptive or similar rights created by law or the governing documents of the Company or any other agreement to which the Company is bound and (ii) the Amendment Shares will not be required to be registered under the Securities Act.

The Company represents that, as of the First Merger Effective Time, the Warrants under the Warrant Amendment are duly authorized and validly issued and, when issued, the shares of Common Stock issuable pursuant to the Warrants under the Warrant Amendment (the "Underlying Shares") will be duly authorized, validly issued, fully paid and non-assessable. Upon issuance in accordance with, and payment pursuant to, the Warrant Amendment, the Underlying Shares will be issued in accordance with law (including applicable state Blue Sky laws) and the governing documents of the Company and will not be issued in violation of any preemptive or similar rights created by law or the governing documents of the Company or any other agreement to which the Company is bound and the Underlying Shares will not be required to be registered under the Securities Act.

The execution and delivery of this Agreement by the Holder shall constitute the express acknowledgment by and agreement of the Holder that the Holder shall not have any rights or entitlements under the Warrant Agreement, the Warrant Amendment or any Warrant (as defined in the Warrant Agreement) issued thereunder with respect to any antidilution or other adjustments to the Warrants issued thereunder and held by the Holder, including, without limitation, pursuant to Section 4 of the Warrant Agreement arising as a result of or related to the share issuances and other transactions contemplated by the Merger Agreement.

This Letter Agreement and the Warrant Amendment, embody the entire agreement and understanding

among the Company and the Holder, and supersede all prior agreements and understandings, relating to the subject matter hereof. The Company and the Holder acknowledge and agree that this Letter Agreement (other than the RRA Interpretation) and the Warrant Amendment shall automatically terminate and be null and void *ab initio* with no effect whatsoever on the parties hereto in the event that the Merger Agreement is terminated in accordance with its terms; provided, however, that such termination shall not relieve any party from liability for any willful and material breach of this Agreement prior to termination thereof. This Letter Agreement may be amended from time to time only upon the written consent by each party hereto. This Letter Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. This Letter Agreement may be executed in counterparts (including by PDF or electronic signature), each one of which shall be deemed an original and all of which together shall constitute one and the same document. This Letter Agreement, and any disputes arising hereunder or related hereto, shall be governed by, and construed in accordance with, the laws of the State of Delaware without regard to its conflict of laws that may lead to the application of the law as of any other jurisdiction.

[Remainder of page intentionally left blank.]

Respectfully,

BIOSCRIP, INC.

By: /s/ Daniel Greenleaf

Name: Daniel Greenleaf Title: President & CEO

Acknowledged and Agreed:

ASSF IV AIV B HOLDINGS, L.P.

By: ASSF Operating Manager IV, L.P., its manager

By:

/s/ Christopher Kerezsi Christopher Kerezsi Name: Title: Authorized Signatory

[Letter Agreement — 2017 Warrant Amendment]

Exhibit A

Holder

ASSF IV AIV B HOLDINGS, L.P.

ASSF IV AIV B HOLDINGS, L.P. ASSF IV AIV B Holdings, L.P. c/o Ares Management LLC 2000 Avenue of the Stars, 12th Floor Los Angeles, CA 90067

Exhibit B

[Amended and Restated Warrant Agreement]

See attached.

BIOSCRIP, INC.		
FORM OF AMI	ENDED AND RESTATED WARRANT A	AGREEMENT

Dated As Of March 14, 2019 Warrants to Purchase shares of Common Stock

BIOSCRIP, INC.

Amended and Restated Warrant Agreement

Warrants For Common Stock

AMENDED AND RESTATED WARRANT AGREEMENT, dated as March 14, 2019, among BioScrip, Inc., a Delaware corporation (together with its successors and assigns, the "<u>Company</u>") and the holders undersigned hereto (collectively and together with each of their respective successors and assigns, the "<u>Holders</u>"). Capitalized terms shall have the meaning specified in <u>Section 5.1</u> hereof.

RECITALS

WHEREAS, on the terms of the Warrant Purchase Agreement (the "Original Warrant Agreement"), dated as June 29, 2017, among the Company and the Original Purchasers party thereto, the Original Purchasers agreed to acquire from the Company, and the Company agreed to issue to the Original Purchasers, Warrants to purchase 4.99% of the Fully Diluted Common Stock (as defined in the Original Warrant Agreement) outstanding on the date of any exercise of the Warrants (less the percentage of the shares of Common Stock previously issued pursuant to the Warrants from time to time as a result of any partial exercise of the Warrants as set forth herein);

WHEREAS, the Company is party to an Agreement and Plan of Merger (the "Merger Agreement"), dated as of March 14, 2019, by and among the Company, HC Group Holdings II, Inc., a Delaware corporation ("Omega"), Beta Sub, LLC, a Delaware limited liability company ("Merger Sub LLC"), and certain other parties signatory thereto, pursuant to which Omega will be merged with and into a newly formed subsidiary of the Company with Omega continuing as the surviving entity (the "Surviving Corporation") (the "First Merger"), immediately following which the Surviving Corporation will be merged with and into Merger Sub LLC, with Merger Sub LLC continuing as the surviving entity (the "Second Merger");

WHEREAS, the undersigned Holders collectively own 100% of the Warrants issued pursuant to the Original Warrant Agreement;

WHEREAS, in connection with, and conditioned upon, the consummation of the First Merger in accordance with the Merger Agreement, the Company and the Holders wish to amend and restate the Original Warrant Agreement so as to fix the number of shares of Common Stock issuable upon exercise of the Warrants at 8,287,317 shares in the aggregate, with each Holder having a Warrant to purchase the number of shares of Common Stock set forth on Annex 1 hereto; and

WHEREAS, the Company and the Holders wish to enter into this Agreement to amend and restate the terms of the Warrants.

AGREEMENT

NOW, THEREFORE, in consideration of the premises and the mutual agreements set forth herein, the parties to this Agreement hereby agree to amend and restate the Original Warrant Agreement as follows:

1. FORM, EXECUTION AND TRANSFER OF WARRANT CERTIFICATES.

1.1 Form of Warrant Certificates.

The Warrant Certificates shall be in the form set forth in <u>Attachment A</u> hereto. The Warrant Certificates may have such letters, numbers or other marks of identification or designation as may be required to comply with any law or with any

rule or regulation of any governmental authority, stock exchange or self-regulatory organization made pursuant thereto ("<u>Law</u>"). Each Warrant Certificate shall be initially dated as of June 29, 2017 and subsequently dated the date of issuance thereof by the Company upon transfer or exchange. Each Warrant Certificate shall represent the right to purchase the number of shares of Common Stock set forth in such Warrant Certificate at a price per share of Common Stock equal to the Exercise Price applicable to that Warrant Certificate; *provided*, that the number of shares of Common Stock issuable upon exercise of the Warrants and the Exercise Price thereof shall be subject to adjustment as provided herein. The aggregate amount of shares of Common Stock issuable upon exercise of all Warrants as of the Second Merger Effective Time shall be 8,287,317, subject to adjustment as of the Second Merger Effective Time as provided herein, with each Holder having a Warrant to purchase the number of shares of Common Stock set forth on Annex 1 hereto.

1.2 Execution of Warrant Certificates; Registration Books.

- (a) Execution of Warrant Certificates. The Warrant Certificates shall be executed on behalf of the Company by an officer of the Company authorized by the Board of Directors. In case the officer of the Company who shall have signed any Warrant Certificate shall cease to be such an officer of the Company before issuance and delivery by the Company of such Warrant Certificate, such Warrant Certificate nevertheless may be issued and delivered with the same force and effect as though the individual who signed such Warrant Certificate had not ceased to be such an officer of the Company, and any Warrant Certificate may be signed on behalf of the Company by any individual who, at the actual date of the execution of such Warrant Certificate, shall be a proper officer of the Company to sign such Warrant Certificate, although at the date of the execution of this Agreement any such individual was not such an officer.
- (b) **Registration Books.** The Company will keep or cause to be kept at its office, maintained at the address of the Company referenced in Section 6.6, at the Company's transfer agent, or at such other office of the Company of which the Company shall have given notice to each holder of Warrant Certificates, books for registration and transfer of the Warrant Certificates issued hereunder. Such books shall show the names and addresses of the respective holders of the Warrant Certificates, the registration number and date of each of the Warrant Certificates and the Denomination thereof.

1.3 Transfer, Split Up, Combination and Exchange of Warrant Certificates; Lost or Stolen Warrant Certificates.

(a) Transfer, Split Up, etc.

- (i) **Transfer**. Subject to compliance with the Securities Act, any applicable state securities laws and the Company's organizational documents, any Warrant Certificate (or portion thereof), with or without other Warrant Certificates, may be transferred to any Person for a Warrant Certificate or Warrant Certificates in an aggregate like Denomination as the Warrant Certificate or Warrant Certificates (or portions thereof) surrendered then entitled such registered holder to purchase. Any registered holder desiring to transfer any Warrant Certificate shall make such request in writing delivered to the Company, which request shall include the identity of the Transferee and the aggregate Denomination of Warrants to be transferred, and shall surrender the Warrant Certificate or Warrant Certificates (or portions thereof) to be transferred at the office of the Company referenced in Section 6.6, whereupon the Company shall deliver promptly to such Transferee a Warrant Certificate or Warrant Certificates, as the case may be, as so requested, which Warrant Certificate or Warrant Certificates shall evidence, collectively, the same aggregate Denomination of Warrants as the Warrant Certificate or Warrant Certificates (or portions thereof) so surrendered for transfer and shall issue a new Warrant Certificate to the transferor representing the Warrants retained by the Transferor if such transfer involved less than the entire Denomination of Warrants held by such Transferor.
- (ii) <u>Split Up, Combination, Exchange, etc.</u> Any Warrant Certificate, with or without other Warrant Certificates, may be split up, combined or exchanged for another Warrant Certificate or Warrant Certificates, in an aggregate like Denomination as the Warrant Certificate or Warrant Certificate surrendered then entitle such registered holder to purchase. Any registered holder desiring to split up, combine or exchange any Warrant Certificate shall make such request in writing delivered to the Company, and shall surrender the Warrant Certificate or Warrant Certificates to be split up, combined or exchanged at the office of the Company referenced in <u>Section 6.6</u>, whereupon the Company shall deliver promptly to such registered holder a Warrant Certificate or Warrant Certificates, as the case may be, as so requested, which Warrant Certificate or Warrant Certificates shall evidence, collectively, the same aggregate Denomination as the Warrant Certificate or Warrant Certificates so surrendered for split-up, combination or exchange.
- (b) **Loss, Theft, etc.** Upon receipt by the Company of evidence reasonably satisfactory to it of the ownership of, and the loss, theft, destruction or mutilation of, any Warrant Certificate, and:
- (i) in the case of loss, theft or destruction, an affidavit of loss, together with a customary and reasonable indemnity reasonably satisfactory to the Company; or

(ii) in the case of mutilation, upon surrender and cancellation thereof;

the Company at its own expense will execute and deliver, in lieu thereof, a new Warrant Certificate, dated the date of such lost, stolen, destroyed or mutilated Warrant Certificate and of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant Certificate and evidencing the same Denomination as the Warrant Certificate so lost, stolen, destroyed or mutilated.

1.4 Subsequent Issuance of Warrant Certificates.

Subsequent to the original issuance, no Warrant Certificates shall be issued except:

- (a) Warrant Certificates issued upon any transfer, combination, split up or exchange of Warrants pursuant to Section 1.3(a);
- (b) Warrant Certificates issued in replacement of mutilated, destroyed, lost or stolen Warrant Certificates pursuant to Section 1.3(b);
- (c) Warrant Certificates issued pursuant to <u>Section 2.3</u> upon the partial exercise of any Warrant Certificate to evidence the unexercised portion of such Warrant Certificate; and
 - (d) Warrant Certificates to reflect any adjustments pursuant to Section 4.

1.5 Effect of Issuance in Registered Form.

Every holder of a Warrant Certificate by accepting the same consents and agrees with the Company and with every other holder of a Warrant Certificate that:

- (a) the Warrant Certificates, to the extent then currently transferable, are transferable only on the registry books of the Company if and when surrendered at the office of the Company referenced in <u>Section 6.5</u>, duly endorsed or accompanied by an instrument of transfer (in the form attached thereto) and payment of any applicable transfer, stamp or issue tax (a "Tax"); and
- (b) the Company may deem and treat the Person in whose name each Warrant Certificate is registered as the absolute owner thereof and of the Warrants evidenced thereby (notwithstanding any notations of ownership or writing on the Warrant Certificates made by anyone other than the Company) for all purposes whatsoever, and the Company shall not be affected by any notice to the contrary.

2. EXERCISE OF WARRANTS; PAYMENT OF EXERCISE PRICE.

2.1 Exercise of Warrants.

- (a) **Manner of Exercise**. At any time and from time to time prior to the Expiration Time, the holder of any Warrant Certificate may exercise the Warrants evidenced thereby, in whole or in any part, by surrender to the Company, at its office referenced in Section 6.6, of such Warrant Certificate, together with a duly executed election to purchase (a form of which is attached to each Warrant Certificate) and payment of the applicable Exercise Price for each share of Common Stock with respect to which the Warrants are then being exercised and an amount equal to any applicable Tax (if not payable by the Company as provided in Section 3.3). Such Exercise Price shall be payable in cash or by withholding of shares of Common Stock into which the Warrants are being exercised pursuant to Section 2.1(b).
- (b) **Payment in Cash or by Withholding of Common Stock**. Upon exercise of any Warrants, the holder of a Warrant Certificate may pay the Exercise Price by either (i) certified or official bank check payable to the order of the Company or by wire transfer of immediately available funds to the account of the Company designated in writing by the Company or (ii) instructing the Company to withhold that number of shares of Common Stock (or fraction thereof) then issuable upon such exercise with an aggregate Market Price as of such exercise date equal to the aggregate Exercise Price. The Company acknowledges that the provisions of this clause are intended, in part, to ensure that a full or partial exchange of a Warrant Certificate will qualify as a conversion, within the meaning of paragraph (d)(3)(ii) of Rule 144 under the Securities Act.
- (c) **Fractional shares of Common Stock**. The Company may, in accordance with <u>Section 2.6</u>, pay the exercising holder cash in lieu of issuing a fractional share in connection with an exercise of Warrants; provided that, if it does not issue a fractional share in such circumstances, it will make such cash payment.

2.2 Issuance of Common Stock.

Upon timely receipt of a Warrant Certificate, accompanied by the form of election to purchase duly executed, and payment of the Exercise Price for each of the shares of the Common Stock to be purchased and by an amount equal to any

applicable Tax (if not payable by the Company as provided in Section 3.3), the Company shall thereupon promptly cause certificates representing the number of whole shares of Common Stock then being purchased to be delivered to or upon the order of the registered holder of such Warrant Certificate, registered in such name or names as may be designated by such holder, and, promptly after such receipt deliver the cash, if any, to be paid in lieu of fractional shares pursuant to Section 2.6 to or upon the order of the registered holder of such Warrant Certificate.

2.3 Unexercised Warrants.

In the event that the registered holder of any Warrant Certificate shall exercise less than all the Warrants evidenced thereby, a new Warrant Certificate evidencing a number of unexercised Warrants shall be issued by the Company to the registered holder of such Warrant Certificate or to its duly authorized assigns.

2.4 Cancellation and Destruction of Warrant Certificates.

All Warrant Certificates surrendered to the Company for the purpose of exercise, payment of the Exercise Price, exchange, substitution or transfer shall be cancelled by it, and no Warrant Certificates shall be issued in lieu thereof except as expressly permitted by any of the provisions of this Agreement. The Company shall cancel and retire any other Warrant Certificates purchased or acquired by the Company.

2.5 Expiration.

All Warrants that have not been exercised or purchased in accordance with the provisions of this Agreement shall expire and all rights of holders of such Warrants hereunder shall terminate and cease at the Expiration Time.

2.6 Fractional shares of Common Stock.

The Company shall not be required to issue fractional shares of Common Stock upon the exercise of any Warrant. If fractional shares are not issued upon the exercise of any Warrant, there shall be paid to the holder thereof, in lieu of any fractional share of Common Stock resulting therefrom, an amount of cash equal to the product of:

- (a) the fractional amount of such share of Common Stock; and
- (b) the Market Price, as determined on the trading day immediately prior to the date of exercise of such Warrant.

2.7 RESERVED

3. AGREEMENTS OF THE COMPANY.

3.1 Reservation of Common Stock.

The Company covenants and agrees that it will at all times cause to be reserved and kept available out of its authorized and unissued shares of treasury shares of Common Stock such number of shares of Common Stock as will be sufficient to permit the exercise in full of all Warrants issued hereunder into Common Stock.

3.2 Common Stock to be Duly Authorized and Issued, Fully Paid and Nonassessable, etc; Compliance with Law

The Company covenants and agrees that it will take all such action as may be necessary to ensure that all shares of Common Stock delivered upon the exercise of any Warrant and the payment of the Exercise Price pursuant to Section 2.1 (in each case, at the time of delivery of the certificates representing such shares of Common Stock) shall (a) be duly and validly authorized and issued and fully paid and nonassessable, free of any preemptive rights in favor of any Person in respect of such issuance and free of any security interest, pledge, mortgage, lien, charge or other encumbrance created by, or arising out of actions of, the Company, in each case other than (i) resulting from the actions and circumstances of the holder of such shares of Common Stock, (ii) to the extent imposed by applicable laws or (iii) in the Company's organizational documents and (b) be issued without violation of any applicable Law.

3.3 Taxes.

The Company covenants and agrees that it will pay when due and payable any and all Taxes and charges that may be payable in respect of the initial issuance or delivery of:

- (a) each Warrant Certificate;
- (b) each Warrant Certificate issued in exchange for any other Warrant Certificate pursuant to Section 1.3, Section 2.3 or Section 4; and
- (c) each share of Common Stock issued upon the exercise of any Warrant.

The Company shall not, however, be required to:

- (i) pay any Tax that may be payable in respect of the transfer or delivery of Warrant Certificates in a name other than that of the registered holder of the Warrant Certificate surrendered for exercise, conversion, transfer or exchange (any such Tax being payable by the holder of such certificate at the time of surrender); or
- (ii) issue or deliver any such certificates referred to in the foregoing clause (i) until any such Tax referred to in the foregoing clause (i) shall have been paid.

3.4 Common Stock Record Date.

Each Person in whose name any certificate for shares of Common Stock is issued upon the exercise of Warrants shall for all purposes be deemed to have become the holder of record of the Common Stock represented thereby on, and such certificates (if any) shall be dated, the date upon which the Warrant Certificate evidencing such Warrants was duly surrendered with an election to purchase attached thereto duly executed and payment of the aggregate Exercise Price (and any applicable Taxes, if payable by such Person) was made.

3.5 Rights in Respect of Common Stock.

Prior to the exercise of the Warrants evidenced thereby, the holder of a Warrant Certificate shall not be entitled to any rights of a stockholder of the Company with respect to the Common Stock into which the Warrants shall be exercisable, including, without limitation, the right to vote in respect of any matter upon which the holders of Common Stock may vote, the right to receive any distributions of cash or property and, except as expressly set forth herein, the right to receive any notice of any proceedings of the Company. Prior to the exercise of the Warrants evidenced thereby, the holders of the Warrant Certificates shall not have as such any obligation in respect of any assessment or any other obligation or liability as a stockholder of the Company, whether such obligations or liabilities are asserted by the Company or by creditors of the Company.

3.6 Noncircumvention.

The Company hereby covenants and agrees that the Company will not, by amendment of its charter, bylaws or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issue or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant Agreement, and will at all times in good faith carry out all the provisions of this Warrant Agreement.

4. ANTI-DILUTION ADJUSTMENTS.

4.1 Adjustments.

The number of shares of Common Stock purchasable upon the exercise of each Warrant, and the Exercise Price, shall be subject to adjustment as set forth in this Section 4; provided that no adjustment of the Exercise Price shall result in an increase in the Exercise Price.

4.2 Stock Splits, Subdivisions, Reclassifications or Combinations.

If the Company shall (i) declare and pay a dividend or make a distribution on its Common Stock in shares of Common Stock, (ii) subdivide or reclassify the outstanding shares of Common Stock into a greater number of shares, or (iii) combine or reclassify the outstanding shares of Common Stock into a smaller number of shares, the Denomination of any Warrant Certificate and the Common Stock issuable upon exercise of any Warrants at the time of the record date for such dividend or effective date of such split, reverse split, subdivision, combination or reclassification shall be proportionately adjusted so that the holder of such Warrants after such date shall be entitled to purchase the number of shares of Common Stock which such holder would have owned or been entitled to receive in respect of the shares of Common Stock subject to such Warrants after such date had such Warrants been exercised immediately prior to such date. In such event the Exercise Price in effect at the time of the effective date of such split, reverse split, subdivision, combination or reclassification shall be adjusted to the number obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of such Warrants immediately before such adjustment and (2) the Exercise Price in effect immediately prior to the record or effective date, as the case may be, for the dividend, distribution, split, reverse split, subdivision, combination or reclassification giving rise to this adjustment by (y) the new number of shares of Common Stock issuable upon exercise of such Warrants determined pursuant to the immediately preceding sentence; provided that the Exercise Price shall not be adjusted to be less than the par value of the Common Stock; provided, further, that the adjustment to the Exercise Price in this Section 4.2 may result in an increase in the Exercise Price, notwithstanding Section 4.1.

4.3 Price Based Anti-Dilution

- (a) Without duplication of the adjustments set forth in Sections 4.2 or 4.4, if the Company shall issue after the Second Merger Effective Time or sell any shares of Common Stock (as actually issued or, pursuant to Section 4.3(b), deemed to be issued) for a consideration per share less than either the Market Price per share or the Exercise Price immediately prior to such issuance or sale, or if earlier, upon the execution of the definitive documentation with respect to such issuance or sale (the "Effective Time"), then immediately upon the Effective Time the Denomination of any Warrant Certificate and the number of shares of Common Stock issuable upon exercise of any Warrants at the time of the effective date shall be increased by multiplying such number of shares of Common Stock by a fraction, (i) the numerator of which shall be the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock so issued or sold (or deemed to be issued or sold pursuant to Section 4.3(b)), and (ii) the denominator of which shall be the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock which the aggregate consideration received by the Company for the total number of shares of Common Stock so issued or sold (or deemed to be issued or sold pursuant to Section 4.3(b)) would purchase if such shares were sold at the greater of the Market Price or the Exercise Price, and the Exercise Price shall be adjusted by multiplying the Exercise Price in effect immediately prior to such issuance or sale (or deemed issuance or sale) by the inverse of such fraction. For the purposes of this Section 4.3(a), none of the following issuances shall be considered the issuance or sale of Common Stock:
- (i) the issuance of Common Stock (a) upon the conversion or exercise of any then-outstanding Common Stock Equivalents; (b) to Persons in connection with joint ventures, strategic alliances or other commercial relationships with such Person relating to the operation of the Company's business and not for the primary purpose of raising equity capital, in connection with Business Combinations, or transactions in which the Company, directly or indirectly, acquires another business or the assets thereof or (c) in an offering for cash for account of the Company that is underwritten on a firm commitment basis and is registered with the Securities and Exchange Commission;
 - (ii) the issuance of any Common Stock or Common Stock Equivalents for which the adjustment provided in Section 4.2 applies; or
- (iii) the issuance of shares of Common Stock or Common Stock Equivalents to employees or directors of either the Company or any Company Subsidiary that is approved by the Board of Directors.
 - (b) For the purposes of Section 4.3(a), the following subparagraphs (i) to (iii), inclusive, shall also be applicable:
- (i) If the Company shall grant any rights to subscribe for, or any rights or options to purchase, Common Stock Equivalents, whether or not such rights or options or the right to convert or exchange any such Common Stock Equivalents are immediately exercisable, and the price per share for which Common Stock is issuable upon the exercise of such rights or options or upon conversion or exchange of such Common Stock Equivalents (determined by dividing (A) the total amount, if any, received or receivable by the Company as consideration for the granting of such rights or options, plus the minimum aggregate amount of additional consideration payable to the Company upon the exercise of such rights or options, plus, in the case of any such rights or options which relate to such Common Stock Equivalents, the minimum aggregate amount of additional consideration, if any, payable upon the issue or sale of such Common Stock Equivalents and upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the exercise of such rights or options or upon the conversion or exchange of all such Common Stock Equivalents issuable upon the exercise of such rights or options) shall be less than the Exercise Price or the Market Price per share of Common Stock immediately prior to the earlier of (x) the time of the granting of such rights or options and (y) the time of execution of definitive documentation with respect to such grant (such earlier time of (x) and (y) to be the "Effective Time" for purposes of Section 4.3(a)), then the total maximum number of shares of Common Stock issuable upon the exercise of such rights or options or upon conversion or exchange of the total maximum amount of such Common Stock Equivalents issuable upon the exercise of such rights or options shall (as of the date of granting of such rights or options or the date of execution of definitive documents, if earlier) be deemed to be outstanding and to have been issued for such price per share; provided that no further adjustment of the conversion price pursuant to this Section 4.3(a) or Section 4.3(b) (i) shall be made (x) upon the actual issuance or sale of such Common Stock Equivalents upon the exercise of any rights to subscribe for, or any rights or options to purchase, such Common Stock Equivalents or (y) upon the actual issuance or sale of such Common Stock upon the exercise of any such Common Stock Equivalents, including without limitation, in each case of clauses (x) and (y) with respect to shares of Common Stock Equivalents or Common Stock issued or issuable as a result of the effect of antidilution adjustments under any such security.

(ii) If the Company shall issue or sell any Common Stock Equivalents, whether or not the rights to exchange or convert thereunder are immediately exercisable, and the price per share for which Common Stock is issuable upon such conversion or exchange (determined by dividing (A) the total amount received or receivable by the Company as consideration for the issue or sale of such Common Stock Equivalents, plus the minimum aggregate amount of additional

consideration, if any, payable to the Company upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the conversion or exchange of all such Common Stock Equivalents) shall be less than the Exercise Price or the Market Price per share of Common Stock immediately prior to the Effective Time for such issuance or sale, then the total maximum number of shares of Common Stock issuable upon conversion or exchange of such Common Stock Equivalents shall (as of the Effective Time) be deemed to be outstanding and to have been issued for such price per share, provided that no further adjustment of the conversion price pursuant to this Section 4.3(a) or Section 4.3(b)(ii) shall be made upon the actual issuance or sale of such Common Stock upon the exercise of any such Common Stock Equivalents, including without limitation, in each case with respect to shares of Common Stock issued or issuable as a result of the effect of antidilution adjustments under any such security.

(iii) In case at any time any shares of Common Stock or Common Stock Equivalents or any rights or options to purchase any such Common Stock or Common Stock Equivalents shall be issued or sold for cash to a non-Affiliate of the Company, the consideration received therefor shall be deemed to be the amount received by the Company therefor. In case any shares of Common Stock or Common Stock Equivalents or any rights or options to purchase any such Common Stock or Common Stock Equivalents shall be issued or sold to an Affiliate of the Company or for a consideration other than or in addition to cash, the amount of the consideration received by the Company shall be deemed to be the Fair Market Value of solely such consideration received by the Company in respect of such purchase of Common Stock or Common Stock Equivalents.

4.4 Other Distributions. In case the Company shall fix a record date for the making of a dividend or distribution to holders of shares of its Common Stock of securities, evidences of indebtedness, assets, cash, rights or warrants (excluding dividends of its Common Stock and other dividends or distributions referred to in Section 4.2), in each such case, the Exercise Price in effect prior to such record date shall be reduced immediately thereafter to the price determined by multiplying the Exercise Price in effect immediately prior to the reduction by the quotient of (x) the Market Price of the Common Stock on the last trading day preceding the first date on which the Common Stock trades on the Exchange on which the Common Stock is listed or admitted to trading without the right to receive such distribution, minus the amount of cash and/or the Fair Market Value of the securities, evidences of indebtedness, assets, rights or warrants to be so distributed in respect of one share of Common Stock (such amount and/or Fair Market Value, the "Per Share Fair Market Value") divided by (y) such Market Price on such date specified in clause (x); such adjustment shall be made successively whenever such a record date is fixed. In obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of any Warrants shall be increased to the number obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of such Warrants before such adjustment, and (2) the Exercise Price in effect immediately prior to the distribution giving rise to this adjustment by (y) the new Exercise Price determined in accordance with the immediately preceding sentence. In the event that such distribution is not so made, the Exercise Price and the number of shares of Common Stock issuable upon exercise of such Warrants then in effect shall be readjusted, effective as of the date when the Board of Directors determines not to distribute such shares, evidences of indebt

4.5 Business Combinations.

In case of any Business Combination or reclassification of Common Stock (other than a reclassification of Common Stock referred to in Section 4.2), a holder's right to receive shares of Common Stock upon exercise of any Warrants shall be converted into the right to exercise such Warrant to acquire the number of shares of stock or other securities or property (including cash) which the Common Stock issuable (at the time of such Business Combination or reclassification) upon exercise of such Warrants immediately prior to such Business Combination or reclassification would have been entitled to receive upon consummation of such Business Combination or reclassification; and in any such case, if necessary, the provisions set forth herein with respect to the rights and interests thereafter of such holder shall be appropriately adjusted so as to be applicable, as nearly as may reasonably be, to such holder's right to exercise such Warrants in exchange for any shares of stock or other securities or property pursuant to this Section 4.5. In determining the kind and amount of stock, securities or the property receivable upon exercise of any Warrants following the consummation of such Business Combination, if the holders of Common Stock have the right to elect the kind or amount of consideration receivable upon consummation of such Business Combination. The Company shall not enter into or be party to any Business Combination unless the successor of the Company (if any), assumes in writing the obligation to deliver to each holder of Warrants hereunder in exchange for such Warrants a security of such successor evidenced by a written instrument substantially similar in form and substance to this Agreement.

4.6 Expiration of Rights or Options.

Upon the expiration of any rights or options to subscribe for, purchase or convert or exchange Common Stock or

Common Stock Equivalents in respect of the issuance, sale or grant of which adjustment was made pursuant to <u>Section 4.3</u>, without the exercise thereof, the Exercise Price and the number of shares of Common Stock purchasable upon the exercise of the Warrants shall, upon such expiration, be readjusted (and such readjustment may result in an increase in the Exercise Price, notwithstanding <u>Section 4.1</u>, not to exceed the Exercise Price prior to such original adjustment) and shall thereafter be such Exercise Price and such number of shares of Common Stock as would have been in effect had such Exercise Price and such number of shares of Common Stock not been originally adjusted (or had the original adjustment not been required, as the case may be), as if:

- (a) the only shares of Common Stock so issued were the shares of Common Stock, if any, actually issued or sold upon the exercise of such rights or options; and
- (b) such shares of Common Stock, if any, were issued or sold for the consideration actually received by the Company upon such exercise plus the aggregate consideration, if any, actually received by the Company for the issuance, sale or grant of all of such rights or options, whether or not exercised; provided that no such readjustment shall have the effect of increasing the Exercise Price by an amount in excess of the amount of the reduction initially made in respect of the issuance, sale, or grant of such rights or options.

4.7 Rounding of Calculations; Minimum Adjustments.

All calculations under this Section 4 shall be made to the nearest one-tenth (1/10th) of a cent or to the nearest one- hundredth (1/100th) of a share, as the case may be. Any provision of this Section 4 to the contrary notwithstanding, no adjustment in the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable shall be made if the amount of such adjustment would be less than \$0.01 or one-tenth (1/10th) of a share of Common Stock, but any such amount shall be carried forward and an adjustment with respect thereto shall be made at the earlier of (i) the time of any exercise of Warrants and (ii) the time of and together with any subsequent adjustment which, together with such amount and any other amount or amounts so carried forward, shall aggregate \$0.01 or 1/10th of a share of Common Stock, or more.

4.8 Timing of Issuance of Additional Common Stock Upon Certain Adjustments.

In any case in which the provisions of this <u>Section 4</u> shall require that an adjustment shall become effective immediately after a record date for an event, the Company may defer until the occurrence of such event (i) issuing to the holder of any Warrants exercised after such record date and before the occurrence of such event the additional shares of Common Stock issuable upon such exercise by reason of the adjustment required by such event over and above the shares of Common Stock issuable upon such exercise before giving effect to such adjustment and (ii) paying to such holder any amount of cash in lieu of a fractional share of Common Stock; *provided*, however, that the Company upon written request shall deliver to such holder a due bill or other appropriate instrument evidencing such holder's right to receive such additional shares, and such cash, upon the occurrence of the event requiring such adjustment.

4.9 Miscellaneous.

- (a) **Statement Regarding Adjustments**. Whenever after the Second Merger Effective Time the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable shall be adjusted as provided in <u>Section 4</u>, the Company shall promptly file at the principal office of the Company referenced in <u>Section 6.6</u> a statement showing in reasonable detail the facts requiring such adjustment and the Exercise Price that shall be in effect and the number of shares of Common Stock into which such Warrants shall be exercisable after such adjustment, and the Company shall also cause a copy of such statement to be sent by mail, first class postage prepaid, or email to each holder of Warrants at the address or email address appearing in the Company's records.
- (b) **Notice of Adjustment Event**. In the event that the Company shall propose after the Second Merger Effective Time to take any action of the type described in this Section 4 (but only if the action of the type described in this Section 4 would result in an adjustment in the Exercise Price or the number of shares of Common Stock into which Warrants are exercisable or a change in the type of securities or property to be delivered upon exercise of Warrants), the Company shall give notice to the holders of Warrants, in the manner set forth in Section 4.9(a), which notice shall specify the record date, if any, with respect to any such action and the approximate date on which such action is to take place. Such notice shall also set forth the facts with respect thereto as shall be reasonably necessary to indicate the effect on the Exercise Price and the number, kind or class of shares or other securities or property which shall be deliverable upon exercise of any Warrants. In the case of any action which would require the fixing of a record date, such notice shall be given at least 10 days prior to the date so fixed, and in case of all other action, such notice shall be given at least 15 days prior to the taking of such proposed action. Without limiting the foregoing, to the extent notice of any of the foregoing actions or events is given to the holders of the Common Stock, such notice shall be provided to the holders of the Warrants on or before such notice to the holders of Common Stock.
 - (c) Proceedings Prior to Any Action Requiring Adjustment. As a condition precedent to the taking of

any action which would require an adjustment pursuant to this Section 4, the Company shall take any action which may be necessary, including obtaining regulatory, New York Stock Exchange, NASDAQ Stock Market or other applicable national securities exchange (an "Exchange") or stockholder approvals or exemptions, in order that the Company may thereafter validly and legally issue as fully paid and nonassessable all shares of Common Stock that the holders are entitled to receive upon exercise of this any Warrants pursuant to this Section 4.

- (d) **Adjustment Rules**. Any adjustments pursuant to this Section 4 shall be made successively whenever an event referred to herein shall occur. If more than one subsection of this Section 4 is applicable to a single event, the subsection shall be applied that produces the largest adjustment and no single event shall cause an adjustment under more than one subsection of this Section 4 so as to result in duplication. If an adjustment in Exercise Price made hereunder would reduce the Exercise Price to an amount below par value of the Common Stock, then such adjustment in Exercise Price made hereunder shall reduce the Exercise Price to the par value of the Common Stock.
- (e) **Merger Agreement**. Notwithstanding anything contained herein to the contrary, the consummation of the transactions contemplated by the Merger Agreement (including the issuance of any shares of Common Stock by the Company contemplated therein) will not result in, or cause any adjustment in, the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable pursuant to this <u>Section 4</u>.

5. INTERPRETATION OF THIS AGREEMENT.

5.1 Certain Defined Terms.

For the purpose of this Agreement, the following terms shall have the meanings set forth below or set forth in the Section hereof following such term:

"Affiliate" means, with respect to any Person, (a) a director, officer or shareholder of such Person, (b) in the case of a natural person, a spouse, parent, sibling or descendant of such Person (or spouse, parent, sibling or descendant of any director or executive officer of such Person) and (c) any other Person that, directly or indirectly through one or more intermediaries, Controls, or is Controlled by, or is under common Control with, such Person, at such time; provided, however, that none of the Holders shall be deemed to be an "Affiliate" of the Company and no Person holding any one or more of the Warrants shall be deemed to be an "Affiliate" of the Company solely by virtue of the ownership thereof.

"Agreement" means this Amended and Restated Warrant Agreement as it may from time to time be amended, restated, modified or supplemented.

"Board of Directors" means the board of directors of the Company, including any duly authorized committee thereof.

"Beneficially Own" has the meaning has the meaning given to it in Section 13D of the Exchange Act and the rules promulgated thereunder.

"<u>Business Combination</u>" means any consolidation of the Company with, or merger of the Company with or into, another Person (other than a merger in which (a) the Company is the surviving corporation, (b) that does not result in any reclassification or change of shares of Common Stock outstanding immediately prior to such merger and (c) the holders of Common Stock are not entitled to receive any consideration therefrom), or any sale or conveyance to another Person of the assets of the Company substantially as an entirety.

"business day" means any day that is not a Saturday, Sunday or other day on which banking institutions in New York, New York are authorized or required by Law or executive order to close.

"Capital Stock" means (A) with respect to any Person that is a corporation or company, any and all shares, interests, participations or other equivalents (however designated) of capital or capital stock of such Person and (B) with respect to any Person that is not a corporation or company, any and all partnership or other equity interests of such Person.

"Charter" means, with respect to any Person, its certificate or articles of incorporation, articles of association, or similar organizational document.

"Common Stock" means the Company's common stock, par value \$.0001 per share.

"Common Stock Equivalents" means all options, derivatives, rights, warrants and convertible, exercisable or exchangeable securities or instruments (including, without limitation, awards or grants of such securities or instruments to directors, officers, employees or consultants of the Company or to other persons).

"Company" has the meaning set forth in the introductory paragraph hereof.

"Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause

the direction of the management and policies of such Person, whether through the ownership of voting Securities, by contract or otherwise.

"<u>Denomination</u>" means, in the case of any Warrant Certificate, the number of shares of Common Stock issuable upon exercise of the Warrants represented by such Warrant Certificate.

"Effective Date" shall mean the date on which the Second Merger is consummated in accordance with the terms of the Merger Agreement.

"Effective Time" has the meaning set forth in Section 4.3(a).

"Exchange" has the set forth in Section 4.9(c).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor statute, and the rules and regulations promulgated thereunder.

"Exercise Price" means, prior to any adjustment pursuant to Section 4 of this Agreement, the Initial Exercise Price; and thereafter, the Initial Exercise Price as successively adjusted and readjusted from time to time in accordance with the provisions of Section 4.

"Expiration Time" means 5:00 p.m., Eastern time, on the tenth (10th) year anniversary of June 29, 2017.

"Fair Market Value" means, with respect to any security or other property, the fair market value of such security or other property as determined by the Board of Directors, acting in good faith and transmitted to the holders of Warrants by written notice. The Required Warrantholders may object in writing to the Board of Director's calculation of Fair Market Value within 10 days of receipt of such written notice thereof. If the Required Warrantholders and the Board of Directors are unable to agree on Fair Market Value during the 10-day period following the delivery of the Required Warrantholders' objection, then the Board of Directors shall select and approve an appraiser of national recognition experienced in the business of evaluating or appraising the market value of securities (which appraiser shall be subject to approval by the Required Warrantholders, which approval shall not be unreasonably withheld, conditioned or delayed). The Fair Market Value established by such appraiser shall be conclusive and binding on the parties. The fees and expenses for such appraiser shall be borne 50% by the Company, on the one hand, and 50% by the holders of Warrants, on the other hand.

"Initial Exercise Price" means, with respect to each Warrant Certificate issued to a Holder on the Issue Date, \$2.00.

"Issue Date" means June 29, 2017.

"Law" has the set forth in Section 1.1.

"Market Price" means, with respect to a particular security, on any given day, the volume weighted average trading price for the period of ten consecutive trading days ending on the trading day prior to such given day as reported by Bloomberg Financial Markets, or any successor thereto, through its "Volume Weight Average Price" function, on the Exchange on which the applicable securities are listed or admitted to trading. "Market Price" shall be determined without reference to after hours or extended hours trading. If such security is not listed and traded in a manner that the quotations referred to above are available for the period required hereunder, the Market Price per share of Common Stock shall be deemed to be the fair market value per share of such security as determined in good faith by the Board of Directors in reliance on an opinion of a nationally recognized independent investment banking corporation retained by the Company for this purpose (which opinion shall be made available to the holders of Warrants); provided that the Required Warrantholders may object in writing to the Board of Director's calculation of fair market value within 10 days of receipt of written notice thereof. If the Required Warrantholders and the Board of Directors are unable to agree on fair market value during the 10-day period following the delivery of the Required Warrantholders' objection, then the Board of Directors shall select and approve an appraiser experienced in the business of evaluating or appraising the market value of securities (which appraiser shall be subject to approval by the Required Warrantholders, which approval shall not be unreasonably withheld, conditioned or delayed). The Market Price established by such appraiser shall be conclusive and binding on the parties. In the event the Market Price established by such appraiser is greater than the Market Price previously determined by the Board of Directors, the fees and expenses for such appraiser shall be borne by the Company. In the event the Market Price established by such appraiser is less than or equal to the Market Price previously determined by the Board of Directors, the fees and expenses for such appraiser shall be borne by the holders of Warrants. For the purposes of determining the Market Price of the Common Stock on the "trading day" preceding, on or following the occurrence of an event, (i) that trading day shall be deemed to commence immediately after the regular scheduled closing time of trading on the Nasdaq Stock Market or, if trading is closed at an earlier time, such earlier time and (ii) that trading day shall end at the next regular scheduled closing time, or if trading is closed at an earlier time, such earlier time (for the avoidance of doubt, and as an example, if the Market Price is to be determined as of the last trading day preceding a specified event and the closing time of trading on a particular day is 4:00 p.m. and the specified event occurs at 5:00 p.m. on that day, the Market Price would be determined by reference to such 4:00

p.m. closing price).

- "Original Warrant Agreement" has the meaning set forth in the recitals hereto.
- "Per Share Fair Market Value" has the meaning set forth in Section 4.4.
- "Person" has the meaning given to it in Section 3(a)(9) of the Exchange Act and as used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act.
- "Holders" has the meaning set forth in the introductory paragraph hereof.
- "Required Warrantholders" means, at any time, the holders of Warrants representing at least a majority of the Common Stock issuable upon exercise of the Warrants issued and outstanding hereunder and not previously exercised (exclusive of any Warrants directly or indirectly held by the Company or any Affiliate of the Company).
 - "Securities Act" means the Securities Act of 1933, as amended, or any successor statute, and the rules and regulations promulgated thereunder.
 - "Tax" or "Taxes" has the meaning set forth in Section 1.5(a).
- "trading day" means (A) if the shares of Common Stock are not traded on any national or regional securities exchange or association or over-the-counter market, a business day or (B) if the shares of Common Stock are traded on any national or regional securities exchange or association or over-the-counter market, a business day on which such relevant exchange or quotation system is scheduled to be open for business and on which the shares of Common Stock (i) are not suspended from trading on any national or regional securities exchange or association or over-the-counter market for any period or periods aggregating one half hour or longer; and (ii) have traded at least once on the national or regional securities exchange or association or over-the-counter market that is the primary market for the trading of the shares of Common Stock.
- "Transferee" means any registered transferee of all or any part of any one or more Warrant Certificates initially acquired by the Holders under this Agreement.
 - "Warrant" means a warrant to initially purchase one share of Common Stock issued pursuant to this Agreement.
 - "Warrant Certificate" means a certificate evidencing the Warrants in the form of Attachment A.
 - 5.2 Section Heading and Table of Contents and Construction.
- (a) Section Headings and Table of Contents, etc. The titles of the Sections of this Agreement and any Table of Contents of this Agreement appear as a matter of convenience only, do not constitute a part hereof and shall not affect the construction hereof. The words "herein," "hereof," "hereunder" and "hereto" refer to this Agreement as a whole and not to any particular Section or other subdivision. References to Sections are, unless otherwise specified, references to Annexes and Attachments are, unless otherwise specified, references to Annexes and Attachments attached to this Agreement.
- (b) **Independent Construction**. Each covenant contained herein shall be construed (absent an express contrary provision herein) as being independent of each other covenant contained herein, and compliance with any one covenant shall not (absent such an express contrary provision) be deemed to excuse compliance with one or more other covenants.

5.3 Directly or Indirectly.

Where any provision herein refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether such action is taken directly or indirectly by such Person, including actions taken by or on behalf of any partnership in which such Person is a general partner.

5.4 Governing Law.

THIS AGREEMENT AND THE WARRANT CERTIFICATES SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE PARTIES SHALL BE GOVERNED BY, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICT OF LAWS RULES THEREOF TO THE EXTENT THAT ANY SUCH RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION, EXCEPT TO THE EXTENT THAT THE DELAWARE GENERAL CORPORATION LAW SPECIFICALLY AND MANDATORILY APPLIES.

6. MISCELLANEOUS.

6.1 Condition to Effectiveness; Termination.

- (a) The parties agree and acknowledge that, notwithstanding anything to the contrary herein, the effectiveness of this Agreement is conditioned upon, and shall not be binding or effective until, the occurrence of the First Merger Effective Time (as such term is defined in the Merger Agreement) in accordance with the terms and conditions of the Merger Agreement.
- (b) In the event that the Merger Agreement is terminated in accordance with its terms, this Agreement shall automatically terminate and be null and void *ab initio* and of no effect whatsoever, and the Original Warrant Agreement shall remain in full force and effect and shall continue to be binding on and enforceable against the Company and the Holders.

6.2 Expenses.

Issuance of certificates for shares of Common Stock to a holder upon the exercise of any Warrants shall be made without charge to such holder for any Tax or stamp duty in respect of the issuance of such certificates, all of which Taxes and stamp duties shall be paid by the Company (for the avoidance of doubt, specifically excluding the Taxes not payable by the Company pursuant to Section 3.3).

6.3 Amendment and Waiver.

This Agreement may be amended, and the observance of any term of this Agreement may be waived, with and only with the written consent of the Company and the Required Warrantholders; provided, however, that no amendment or waiver of the provisions of Section 2.1, this Section 6.3, Section 4 or of any term defined in Section 5.1 to the extent used in such specifically referenced Sections, may be made without the prior written consent of all holders of Warrants then outstanding (excluding any Warrants directly or indirectly held by the Company or any Affiliate of the Company); and, provided, further, that:

- (a) no such amendment or waiver of any of the provisions of this Agreement pertaining to the Exercise Price or the number of shares or kind of Common Stock that may be purchased upon exercise of each Warrant; and
 - (b) no change accelerating the occurrence of the Expiration Time;

shall be effective as to a holder of Warrants unless consented to in writing by such holder.

6.4 Entire Agreement; Termination of Original Warrant Agreement.

This Agreement and the Warrant Certificates embody the entire agreement and understanding among the Company and the Holders, and supersede all prior agreements and understandings, relating to the subject matter hereof. The Company and the Holders hereby acknowledge and agree that as of the Effective Date, the Original Warrant Agreement and any Warrant Certificate issued thereunder are terminated and of no further force and effect and confirm, pursuant to Section 6.2 of the Original Warrant Agreement, that all provisions thereof are terminated and of no further force and effect and the Original Warrant Agreement and any Warrant Certificate issued thereunder are hereby superseded and replaced by this Agreement, which amends and restates the Original Agreement and any Warrant Certificate issued thereunder in their entirety. Each Holder shall, on or promptly following the Second Merger Effective Time, deliver and surrender to the Company any Warrant Certificate issued under the Original Agreement in such Holder's possession.

6.5 Successors and Assigns; Multiple Holders.

All covenants and other agreements in this Agreement made by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors and assigns of the parties hereto to the extent they become holders of Warrants (including, without limitation, any Transferee) whether so expressed or not. Notwithstanding the foregoing sentence, the Company may not assign any of its rights, duties or obligations hereunder or under the Warrant Certificates except pursuant to the terms of this Agreement without the prior written consent of the Required Warrantholders.

Each holder of a Warrant agrees that (i) no other holder of a Warrant will by virtue of this Warrant or exercise thereof be under any fiduciary or other duty to give or withhold any consent or approval under this Warrant or to take any other action or omit to take any action under this Warrant and (ii) each other holder of a Warrant may act or refrain from acting under this Warrant as such other holder may, in its discretion, elect.

6.6 Notices and Information.

All communications hereunder or under the Warrants shall be in writing and shall be delivered either by certified or registered mail, postage prepaid, return receipt requested, email or nationally recognized overnight courier, and shall be addressed to the following addresses:

(a) if to a Holder, at its address set forth on Annex 2 to this Agreement, or at such other address as such Holder shall have specified to the Company in writing;

- (b) if to any other holder of any Warrant Certificate, addressed to such other holder at such address as such other holder shall have specified to the Company in writing or, if any such other holder shall not have so specified an address to the Company, then addressed to such other holder in care of the last holder of such Warrant Certificate that shall have so specified an address to the Company; and
- (c) if to the Company, at the address set forth on Annex 3 to this Agreement, or at such other address as the Company shall have specified to each holder of Warrants in writing.

Any communication addressed and delivered as herein provided shall be deemed to be received when actually delivered to the address of the addressee (whether or not delivery is accepted) by a nationally recognized overnight delivery service which provides proof of delivery or on the date postmarked if sent by registered or certified mail or upon receipt by the recipient's email server if directed to the email address provided in the notice section hereof, as the case may be. Any communication not so addressed and delivered shall be ineffective unless actually received by the intended addressee. Notwithstanding the foregoing provisions of this Section 6.6, service of process in any suit, action or proceeding arising out of or relating to this Agreement or any document, agreement or transaction contemplated hereby shall be delivered in the manner provided in Section 6.9(c).

Upon exercise of any Warrant pursuant to the terms hereof, the Company will use commercially reasonable efforts to answer a limited number of reasonable questions and provide a limited amount of reasonable documentation regarding any non-confidential information supporting its calculation of Fully Diluted Common Stock and the related shares of Common Stock issuable for the Denomination of such Warrant being exercised.

6.7 Severability.

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

6.8 Execution in Counterpart.

This Agreement may be executed in one or more counterparts and shall be effective when at least one counterpart shall have been executed by each party hereto, and each set of counterparts that, collectively, show execution by each party hereto shall constitute one duplicate original.

6.9 Waiver of Jury Trial; Consent to Jurisdiction, Etc.

- (a) Waiver of Jury Trial. THE PARTIES HERETO VOLUNTARILY AND INTENTIONALLY WAIVE ANY RIGHT ANY OF THEM MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE WARRANTS OR ANY OF THE DOCUMENTS, AGREEMENTS OR TRANSACTIONS CONTEMPLATED HEREBY.
- (b) Consent to Jurisdiction. ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE WARRANTS, OR ANY OF THE DOCUMENTS, AGREEMENTS OR TRANSACTIONS CONTEMPLATED HEREBY (WHETHER IN TORT, CONTRACT OR OTHERWISE) OR ANY ACTION OR PROCEEDING TO EXECUTE OR OTHERWISE ENFORCE ANY JUDGMENT IN RESPECT OF ANY BREACH UNDER THIS AGREEMENT, THE WARRANTS OR ANY DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY SHALL BE BROUGHT BY SUCH PARTY IN ANY NEW YORK STATE COURT OR FEDERAL DISTRICT COURT LOCATED IN THE SOUTHERN DISTRICT OF NEW YORK AS SUCH PARTY MAY IN ITS SOLE DISCRETION ELECT, AND BY THE EXECUTION AND DELIVERY OF THIS AGREEMENT, THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMIT TO THE IN PERSONAM JURISDICTION OF EACH SUCH COURT, AND EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES AND AGREES NOT TO ASSERT IN ANY PROCEEDING BEFORE ANY TRIBUNAL, BY WAY OF MOTION, AS A DEFENSE OR OTHERWISE, ANY CLAIM THAT IT IS NOT SUBJECT TO THE IN PERSONAM JURISDICTION OF ANY SUCH COURT. IN ADDITION, EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE IN ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY DOCUMENT, AGREEMENT OR TRANSACTION CONTEMPLATED HEREBY BROUGHT IN ANY SUCH COURT, AND HEREBY IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.

(c) Service of Process. EACH PARTY HERETO IRREVOCABLY AGREES THAT PROCESS PERSONALLY SERVED OR SERVED BY U.S. REGISTERED MAIL AT THE ADDRESSES PROVIDED HEREIN

FOR NOTICES SHALL CONSTITUTE, TO THE EXTENT PERMITTED BY LAW, ADEQUATE SERVICE OF PROCESS IN ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE WARRANTS OR ANY DOCUMENT, AGREEMENT OR TRANSACTION CONTEMPLATED HEREBY, OR ANY ACTION OR PROCEEDING TO EXECUTE OR OTHERWISE ENFORCE ANY JUDGMENT IN RESPECT OF ANY BREACH HEREUNDER, UNDER THE WARRANTS OR UNDER ANY DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY. RECEIPT OF PROCESS SO SERVED SHALL BE CONCLUSIVELY PRESUMED AS EVIDENCED BY A DELIVERY RECEIPT FURNISHED BY THE UNITED STATES POSTAL SERVICE OR ANY COMMERCIAL DELIVERY SERVICE.

[Remainder of page intentionally left blank; next page is signature page]

IN WITNESS WHEREOF, the Parties have executed this Agreement on the date first above written.		
	COMPANY:	
	BioScrip, Inc.	
	By: Name: Title:	

HOLDERS:

ASSF IV AIV B HOLDINGS, L.P.

By: ASSF IV AIV B HOLDINGS GP LLC,

its general partner
By: ASSF IV AIV B, L.P.,

its sole member

By: ASSF MANAGEMENT IV, L.P.,

its general partner
By: ASSF MANAGEMENT IV GP LLC,

its general partner

Ву:	
Name:	
Title:	

J.P. Morg	an Securities LLC		
By:			

By:	
Name:	
Title:	

HOLDERS:

HOLDERS:

Goldman Sachs & Co. LLC

By: Name:	
Name:	
Title:	

HOLDERS:

Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc

By: Western Asset Management Company, as its Investment Manager and Agent

By: Name:	
Title:	

ATTACHMENT A [FORM OF WARRANT CERTIFICATE]

THE SECURITIES REPRESENTED HEREBY, AND THE SECURITIES ISSUABLE HEREBY, HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS AND NEITHER THE SECURITIES NOR ANY INTEREST THEREIN MAY BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT OR SUCH LAWS OR AN EXEMPTION FROM REGISTRATION UNDER SUCH ACT AND SUCH LAWS WHICH, IN THE OPINION OF COUNSEL FOR THE COMPANY, IS AVAILABLE. THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE TERMS OF A CERTAIN AMENDED AND RESTATED WARRANT AGREEMENT, DATED AS OF MARCH [], 2019, THE PROVISIONS OF WHICH ARE INCORPORATED HEREIN BY REFERENCE. A COPY OF SUCH AGREEMENT IS AVAILABLE FROM THE COMPANY UPON REQUEST.

WARRANT CERTIFICATE

BIOSCRIP, INC.

No. WRT- Warrants

Date: [], 20[]

This Warrant Certificate certifies that , or registered assigns, is the registered holder of Warrants entitling the owner thereof to purchase at any time on or after the date hereof and on or prior to the Expiration Time, [] fully paid and nonassessable shares of Common Stock, \$0.0001 par value per share (the "Common Stock"), of BIOSCRIP, INC., a Delaware corporation (together with its successors and assigns, the "Company"), at a purchase price (subject to adjustment as provided in the Warrant Agreement (as defined below), the "Exercise Price") of \$2.00 per share of Common Stock upon presentation and surrender of this Warrant Certificate to the Company with a duly executed election to purchase and payment of the Exercise Price (including by withholding of shares of Common Stock), all in the manner set forth in the Warrant Agreement (defined below). The Denomination of each Warrant and the Exercise Price are the Denomination and the Exercise Price as of the date hereof, and are subject to adjustment as referred to below.

The Warrants are issued pursuant to an Amended and Restated Warrant Agreement (as it may from time to time be amended or supplemented, the "Warrant Agreement"), dated as of March 14, 2019, among the Company and the Holders named therein, and are subject to all of the terms, provisions and conditions thereof, which Warrant Agreement is hereby incorporated herein by reference and made a part hereof and to which Warrant Agreement reference is hereby made for a full description of the rights, obligations, duties and immunities of the Company and the holders of the Warrant Certificates. Capitalized terms used, but not defined, herein have the respective meanings ascribed to them in the Warrant Agreement. In the event of any conflict between this Warrant Certificate and the Warrant Agreement, the Warrant Agreement shall control and govern.

As provided in the Warrant Agreement, the Exercise Price and the Denomination evidenced by this Warrant Certificate are, upon the happening of certain events, subject to modification and adjustment. Except as otherwise set forth in, and subject to, the Warrant Agreement, the Expiration Time of this Warrant Certificate is as set forth in the Warrant Agreement.

Subject to the limitations set forth in the Warrant Agreement, this Warrant Certificate shall be exercisable, at the election of the holder, at any time on or after the date hereof and on or prior to the Expiration Time either as an entirety or in part from time to time. If this Warrant Certificate shall be exercised in part, the holder shall be entitled to receive, upon surrender hereof, another Warrant Certificate or Warrant Certificates for the Warrants not exercised. This Warrant Certificate, with or without other Warrant Certificates, upon surrender in the manner set forth in the Warrant Agreement and subject to the conditions set forth in the Warrant Agreement, may be transferred or exchanged for another Warrant Certificate or Warrant Certificates of like tenor evidencing Warrants entitling the holder to a like number of shares of Common Stock as the Warrants evidenced by the Warrant Certificate or Warrant Certificates surrendered shall have entitled such holder to purchase.

Except as expressly set forth in the Warrant Agreement, no holder of this Warrant Certificate shall be entitled to vote or receive distributions or be deemed for any purpose the holder of shares of Common Stock or of any other securities of

the Company that may at any time be issued upon the exercise hereof, nor shall anything contained in the Warrant Agreement or herein be construed to confer upon the holder hereof, as such, any of the rights of a holder of a share of Common Stock in the Company or any right to vote upon any matter submitted to holders of shares of Common Stock at any meeting thereof, or to give or withhold consent to any corporate action of the Company (whether upon any recapitalization, issuance of stock, reclassification of Securities, change of par value, consolidation, merger, conveyance, or otherwise), or to receive dividends or subscription rights, or otherwise, until the Warrant or Warrants evidenced by this Warrant Certificate shall have been exercised as provided in the Warrant Agreement.

In the event of any inconsistency between this warrant Certificate and the Warrant agreement, the terms of the Warrant Agreement shall govern.

THIS WARRANT CERTIFICATE SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE COMPANY AND THE HOLDER HEREOF SHALL BE GOVERNED BY, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICT OF LAWS RULES THEREOF TO THE EXTENT THAT ANY SUCH RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION, EXCEPT TO THE EXTENT THAT THE DELAWARE GENERAL CORPORATION LAW SPECIFICALLY AND MANDATORILY APPLIES.

WITNESS the signature of a proper officer of the Company as of the date first above written.

BIOSCRIP, INC.

By: Name: Title:

[FORM OF ASSIGNMENT]

(To be executed by the registered holder if

such holder desires to transfer the Warrant Certificate)

FOR VALUE RECEIVED,

hereby sells, assigns and transfers unto (Please print name and address of transferee.)

the accompanying Warrant Certificate, together with all right, title and interest therein, and does hereby irrevocably constitute and appoint:

attorney, to transfer the accompanying Warrant Certificate on the books of the Company with full power of substitution. Dated: ,20.

[HOLDER]

By:

NOTICE

The signature to the foregoing Assignment must correspond to the name as written upon the face of the accompanying Warrant Certificate or any prior assignment thereof in every particular, without alteration or enlargement or any change whatsoever.

[FORM OF ELECTION TO PURCHASE]

(To be executed by the registered holder if such holder desires to exercise the Warrant Certificate)

To: BIOSCRIP, INC.

The undersigned hereby irrevocably elects to exercise	Warrants represented by the accompanying Warrant Certificate to purchase the
shares of Common Stock issuable upon the exercise of such Warn	ants, and requests that certificates for such shares be issued in the name of:

(Please print name and address.)

(Please insert social security or other identifying number.)

If such number of Warrants being exercised by the undersigned are not all the Warrants evidenced by the accompanying Warrant Certificate, a new Warrant Certificate for the balance remaining of such Warrants shall be registered in the name of and delivered to:

(Please print name and address.)

(Please insert social security or other identifying number.)

The undersigned is paying the Exercise Price for the Common Stock to be issued on exercise of the foregoing Warrants, unless payment of such Exercise Price has been waived by the Company, by

o certified or bank check by wire transfer, or

o withholding of shares of Common Stock into which the Warrants are being exercised.

Dated: , 20.

[HOLDER]

By:

NOTICE

The signature to the foregoing Election to Purchase must correspond to the name as written upon the face of the accompanying Warrant Certificate or any prior assignment thereof in every particular, without alteration or enlargement or any change whatsoever.

ANNEX 1 Warrants Held on the Effective Date by the Holders

Holder	Total Number of Warrants
ASSF IV AIV B Holdings, L.P.	4,520,354
Amari Investments Pte. Ltd.	1,506,784
J.P. Morgan Securities LLC	753,398
Goldman Sachs & Co. LLC	1,130,081
Western Asset Middle Market Debt Fund Inc.	113,016
Western Asset Middle Market Income Fund Inc	263,683

ANNEX 2 Address for Holders for Notices

ASSF IV AIV B HOLDINGS, L.P.

ASSF IV AIV B Holdings, L.P. c/o Ares Management LLC 2000 Avenue of the Stars, 12th Floor Los Angeles, CA 90067

J.P. MORGAN SECURITIES LLC

J.P. Morgan Securities LLC 4 New York Plaza, 15th Floor, Mail Code NY1-E054 New York, New York 10004 Attention: Jeffrey L. Panzo

Email: Jeffrey.L.Panzo@JPMorgan.com

Phone No.: (212) 499-1435

GOLDMAN SACHS & CO. LLC

Goldman Sachs & Co. LLC 200 West Street, 26th Floor Attn: Paul Burningham and Paige Cataruozolo New York, New York 10282 Email: fice-amssg-mo@gs.com

Phone No.: (917) 343-8393 (Paul Burningham),

(917) 343-3096 (Paige Cataruozolo)

WESTERN ASSET MIDDLE MARKET DEBT FUND INC. WESTERN ASSET MIDDLE MARKET INCOME FUND INC

Western Asset Management Company 385 East Colorado Boulevard Pasadena, California 91101 Attention: Legal Department

ANNEX 3 Address of Company for Notices

BioScrip, Inc. 1600 Broadway, Suite 700 Denver, CO 80202

Attn: Stephen Deitsch, Senior Vice President,

Chief Financial Officer & Treasurer Email: Stephen.Deitsch@bioscrip.com Telecopy Number: (720) 468-4040

With a copy (which shall not constitute notice) to:

Gibson, Dunn & Crutcher, LLP

200 Park Avenue

New York, New York 10036 Attention: Sean P. Griffiths Email: sgriffiths@gibsondunn.com Telecopy Number: (212) 251-5222

Exhibit C

[Amended and Restated Registration Rights Agreement]

See attached.

AMENDMENT NO. 1 TO REGISTRATION RIGHTS AGREEMENT

THIS AMENDMENT NO. 1 TO THE REGISTRATION RIGHTS AGREEMENT (this "Amendment"), effective as of March 14, 2019, is by and between BioScrip, Inc. (the "Company"), and the stockholders of the Company signatory hereto (each a "Stockholder" and collectively, the "Stockholders"). Capitalized but otherwise undefined terms herein have the meanings given to them in the Registration Rights Agreement (as defined below).

WHEREAS, the Company and the Stockholders are party to that certain Registration Rights Agreement, dated June 29, 2017 (the "Registration Rights Agreement"), governing the Company's obligations to register Registrable Securities of the Stockholders; and

WHEREAS, in connection with entering into that certain (i) Amended and Restated Warrant Agreement, dated as of the date hereof, by and between the Company and the Stockholders (the "Warrant Amendment"), (ii) those certain letter agreements, dated as of the date hereof, by and between the Company and each of the Stockholders (the "Letter Agreements"), pursuant to which the Company will issue an aggregate of 1,855,747 shares of Common Stock of the Company upon the closing of the transactions contemplated by the Merger Agreement (as defined below), and (iii) that certain Agreement and Plan of Merger, dated as of the date hereof, by and among the Company, HC Group Holdings II, Inc. and the other parties thereto, the Company and the Stockholders desire to amend the Registration Rights Agreement pursuant to Section 11(c) thereof and upon such terms as set forth herein.

NOW, **THEREFORE**, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- 1. Amendment to Registration Rights Agreement.
 - (a) Section 1(a) of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:

"Short-Form Registrations. At any time after June 29, 2018 twelve (12) months following the date hereof, with respect to the Purchased Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, or at any time after the twelve month anniversary of the First Merger Effective Time (as defined in the Merger Agreement), with respect to the Amendment Shares, each Holder may request registration under the Securities Act of all or any portion of its Registrable Securities on Form S-3 or any successor form (each, a "Short-Form Registration"), which may, if so requested, be a "shelf" registration under Rule 415 under the Securities Act. A registration shall not count as one of the permitted Short-Form Registrations unless and until a registration statement relating thereto has become effective under the Securities Act. Each request for a Short-Form Registration shall specify the number of Registrable Securities requested to be registered.

(b) Section 1(b) of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:

"Long-Form Registrations. At any time that a Holder is then eligible to request registration under the Securities Act of all or any portion of its Registrable Securities but where Short-Form Registration pursuant to Section 1(a) of this Agreement is not available to be used by the Company in respect of such proposed registration, but in no event earlier than twelve (12) months following the date hereof June 29, 2018, with respect to the Purchased Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, or the twelve month anniversary of the First Merger Effective Time (as defined in the Merger Agreement), with respect to the Amendment Shares, each Holder shall be entitled to request a registration on Form S-1 or any similar form (each, a "Long-Form Registration"). A registration shall not count as one of the permitted Long-Form Registrations unless and until a registration statement relating thereto has become effective under the Securities Act and each requesting Holder is able to register and sell at least thirty percent (30%) of its Registrable Securities thereunder.

(c) The definition of Registrable Securities in Section 10 of the Registration Rights Agreement is hereby amended and restated in its entirety to read as follows:

""Registrable Securities" means the Purchased Shares, the Amendment Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, together with any securities issued or issuable upon any stock

split, dividend or other distribution, recapitalization or similar event with respect to the foregoing. For purposes of this Agreement, a Person shall be deemed to be a holder of Registrable Securities whenever such Person has the right to acquire such Registrable Securities (upon conversion or exercise, in connection with a transfer of securities or otherwise, but disregarding any restrictions or limitations upon the exercise of such right), whether or not such acquisition has actually been effected. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (A) a Registration Statement covering such securities has been declared effective by the Securities and Exchange Commission and such securities have been disposed of pursuant to such effective Registration Statement, (B) such securities are sold under circumstances in which all of the applicable conditions of Rule 144 (or any similar provisions then in force) under the Securities Act are met, (C) the aggregate amount of such securities held by each Holder, together with all Affiliates of each such Holder and persons forming a "group" with each such Holder (within the meaning of Regulation 13D under the Securities Exchange Act of 1934, as amended from time to time, and the rules and regulations promulgated thereunder ("Regulation 13D")), consists of beneficial ownership (within the meaning of Regulation 13D) of less than 5.0% of the Common Stock of the Company and such securities are eligible for sale by each such Holder without registration pursuant to Rule 144 (or any similar provisions then in force) under the Securities Act without limitation thereunder on volume or manner of sale, (D) such securities are otherwise transferred and such securities may be resold without limitation or subsequent registration under the Securities Act, (E) such securities shall have ceased to be outstanding, or (F) the stock certificates or evidences of book-entry registration relating to such securities have had all restrictiv

(d) The following defined term shall be added to Section 10 of the Registration Rights Agreement:

""<u>Amendment Shares</u>" means the shares of Common Stock that the Company issued to the Stockholders pursuant to those certain letter agreements, dated as of March 14, 2019, by and between the Company and each of the Stockholders, in connection with the Amended and Restated Warrant Agreement, dated as of March 14, 2019, by and between the Company and the Stockholders."

- 2. <u>Condition to Effectiveness; Termination</u>. The Company and the Stockholders agree and acknowledge that, notwithstanding anything to the contrary herein, the effectiveness of this Amendment is conditioned upon, and shall not be binding or effective until, the occurrence of the First Merger Effective Time (as such term is defined in the Merger Agreement) in accordance with the terms and conditions of the Merger Agreement. In the event that the Merger Agreement is terminated in accordance with its terms, this Amendment shall automatically terminate and be null and void *ab initio* and of no effect whatsoever, and the Registration Rights Agreement shall remain in full force and effect and shall continue to be binding on and enforceable against the Company and the Purchasers.
- 3. <u>Confirmation</u>. Except as expressly modified by the terms and provisions of this Amendment, all of the terms and provisions of the Registration Rights Agreement are unchanged and continue in full force and effect and all rights, remedies, liabilities and obligations evidenced by the Registration Rights Agreement are hereby acknowledged by the Company and the Stockholders to be valid and in full force and effect.
- 4. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts, by facsimile if necessary, each of which shall be an original, but all of which together shall constitute one instrument.
- 5. Governing Law. This Amendment shall be governed by and construed in accordance with the internal laws of New York applicable to parties residing in New York, without regard applicable principles of conflicts of law. Each party irrevocably consents to the exclusive jurisdiction of any court located within New York County, New York, in connection with any matter based upon or arising out of this Amendment or the matters contemplated hereby and it agrees that process may be served upon it in any manner authorized by the laws of the State of New York for such persons and waives and covenants not to assert or plead any objection which it might otherwise have to such jurisdiction and such process.

[SIGNATURE PAGES FOLLOW]

BIOSCRIP, INC.
Name: Title:
STOCKHOLDERS:
ASSF IV AIV B HOLDINGS, L.P.
By: ASSF IV AIV B HOLDINGS GP LLC, its general partner By: ASSF IV AIV B, L.P., its sole member By: ASSF MANAGEMENT IV, L.P., its general partner By: ASSF MANAGEMENT IV GP LLC, its general partner
Name: Title:
J.P. Morgan Securities LLC
Name: Title:
Goldman Sachs & Co. LLC

Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc

By: Western Asset Management Company, as its Investment Manager and Agent

Name:

Title:

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

BIOSCRIP, INC. 1600 Broadway, Suite 700 Denver, CO 80202

March 14, 2019

To the Holder listed on Exhibit A (the "Holder")

Re: Amended and Restated Warrant Agreement

Ladies and Gentlemen,

Reference is made to (i) that certain Warrant Agreement (the "<u>Warrant Agreement</u>"), dated June 29, 2017, by and among BioScrip, Inc., a Delaware corporation (the "<u>Company</u>") and the purchasers signatory thereto (collectively with their successors and assigns, the "<u>Original Purchasers</u>"); (ii) that certain Agreement and Plan of Merger (the "<u>Merger Agreement</u>"), dated as of the date hereof, by and among the Company, HC Group Holdings I, LLC ("<u>Omega Parent</u>"), HC Group Holdings II, Inc., Beta Sub, Inc., Beta Sub, LLC and HC Group Holdings III, Inc.; and (iii) that certain Amended and Restated Warrant Agreement, dated as of the date hereof, by and among the Company and the Holder and other Holders party thereto (the "<u>Warrant Amendment</u>"), attached hereto as Exhibit B. This letter agreement (this "<u>Letter Agreement</u>") confirms certain agreements and understandings between the Holder and the Company related to the Warrant Amendment (as defined below).

In consideration of the Holder's execution and delivery of the Warrant Amendment, and conditioned upon the occurrence of the First Merger Effective Time (as defined in the Merger Agreement), the Company hereby agrees to issue to the Holder 337,408 shares of Common Stock of the Company in the aggregate (the "Amendment Shares"), promptly following the occurrence of the First Merger Effective Time (as defined in the Merger Agreement). Furthermore, promptly following, and conditioned upon, the occurrence of the First Merger Effective Time, the Company and the Holder acknowledge and agree that the Registration Rights Agreement (the "Existing RRA"), entered into as of June 29, 2017, by and among the Company and the Original Purchasers, shall be amended and restated (the "RRA Amendment"), in the form of Exhibit C, with such RRA Amendment to be conditioned upon, and automatically effective as of, the occurrence of the Second Merger Effective Time, without any further action on the party of any party hereto. The Company and the Holder agree (the "RRA Interpretation") that, solely for purposes of the definition of "Registrable Securities" under the Existing RRA, during the period prior to the First Merger Effective Time, the Holder shall be deemed to beneficially own the Warrants underlying the Warrant Agreement, notwithstanding any agreement between the Holder and the Company not to exercise such Warrants. The execution and delivery of this Letter Agreement by the Holder shall be deemed a consent by the Holder, in its capacities as Holder under the Existing RRA, under Section 9 (Preservation of Rights) of the Existing RRA with respect to the registration rights to be granted to Omega Parent in connection with the Merger Agreement.

Following the execution and delivery of this Letter Agreement and until the Second Merger Effective Time, the Holder agrees not to, and to cause its controlled affiliates not to, (i) directly or indirectly

sell, transfer, assign, pledge, tender, convert, exchange or encumber or otherwise dispose of, any Warrant or (ii) exercise its exchange right under any Warrant.

The Company agrees that, to the extent an event occurs on or after the date hereof and prior to the Second Merger Effective Date which would cause an adjustment to the number of, or Exercise Price of, any Warrants under the Warrant Amendment were the Warrant Amendment then in effect (any such event, an "Adjustment Event"), then the Company shall apply, as of the Second Merger Effective Date, such adjustment to the number of, and Exercise Price of, Warrants that would otherwise be outstanding under the Warrant Amendment immediately after the Second Merger Effective Date (taking, solely for purposes of such adjustment, the number of shares of Common Stock outstanding under any adjustment calculation to be the number of shares of Common Stock outstanding on the date of such Adjustment Event, and giving cumulative effect to any prior adjustments due to other Adjustment Events).

The Company represents that, as of the First Merger Effective Time, upon issuance in accordance with, and payment pursuant to, the terms hereof, the Amendment Shares will be duly authorized, validly issued, fully paid and non-assessable. Upon issuance in accordance with, and payment pursuant to, the terms hereof, the Holder (or its affiliates, as applicable) will have good title to the Amendment Shares, free and clear of all liens, claims, encumbrances, charges, mortgages, options, pledges, security interests, hypothecations, easements, rights-of-way or encroachments of any nature whatsoever, whether voluntarily incurred or arising by operation of law ("Liens"), other than (a) transfer restrictions under federal and state securities laws and (b) Liens imposed solely due to the actions of the Holder. Upon issuance in accordance with, and payment pursuant to, the terms hereof, (i) the Amendment Shares will be issued in accordance with law (including applicable state Blue Sky laws) and the governing documents of the Company and will not be issued in violation of any preemptive or similar rights created by law or the governing documents of the Company or any other agreement to which the Company is bound and (ii) the Amendment Shares will not be required to be registered under the Securities Act.

The Company represents that, as of the First Merger Effective Time, the Warrants under the Warrant Amendment are duly authorized and validly issued and, when issued, the shares of Common Stock issuable pursuant to the Warrants under the Warrant Amendment (the "Underlying Shares") will be duly authorized, validly issued, fully paid and non-assessable. Upon issuance in accordance with, and payment pursuant to, the Warrant Amendment, the Underlying Shares will be issued in accordance with law (including applicable state Blue Sky laws) and the governing documents of the Company and will not be issued in violation of any preemptive or similar rights created by law or the governing documents of the Company or any other agreement to which the Company is bound and the Underlying Shares will not be required to be registered under the Securities Act.

The execution and delivery of this Agreement by the Holder shall constitute the express acknowledgment by and agreement of the Holder that the Holder shall not have any rights or entitlements under the Warrant Agreement, the Warrant Amendment or any Warrant (as defined in the Warrant Agreement) issued thereunder with respect to any antidilution or other adjustments to the Warrants issued thereunder and held by the Holder, including, without limitation, pursuant to Section 4 of the Warrant Agreement arising as a result of or related to the share issuances and other transactions contemplated by the Merger Agreement.

This Letter Agreement and the Warrant Amendment, embody the entire agreement and understanding

among the Company and the Holder, and supersede all prior agreements and understandings, relating to the subject matter hereof. The Company and the Holder acknowledge and agree that this Letter Agreement (other than the RRA Interpretation) and the Warrant Amendment shall automatically terminate and be null and void *ab initio* with no effect whatsoever on the parties hereto in the event that the Merger Agreement is terminated in accordance with its terms; provided, however, that such termination shall not relieve any party from liability for any willful and material breach of this Agreement prior to termination thereof. This Letter Agreement may be amended from time to time only upon the written consent by each party hereto. This Letter Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. This Letter Agreement may be executed in counterparts (including by PDF or electronic signature), each one of which shall be deemed an original and all of which together shall constitute one and the same document. This Letter Agreement, and any disputes arising hereunder or related hereto, shall be governed by, and construed in accordance with, the laws of the State of Delaware without regard to its conflict of laws that may lead to the application of the law as of any other jurisdiction.

[Remainder of page intentionally left blank.]

Resi	pectfu	ıllv
res	Decui	шу

BIOSCRIP, INC.

By: /s/ Daniel Greenleaf

Name: Daniel Greenleaf

Title: President

Acknowledged and Agreed:

AMARI INVESTMENT PTE. LTD.

By: ASSF Operating Manager IV, L.P., its manager

By: /s/ Christopher Kerezsi

Name: Christopher Kerezsi
Title: Authorized Signatory

[Letter Agreement – 2017 Warrant Amendment]

Exhibit A

Holder

AMARI INVESTMENT PTE. LTD.

Amari Investment Pte. Ltd. c/o Ares Management LLC 2000 Avenue of the Stars, 12th Floor Los Angeles, CA 90067

Exhibit B

[Amended and Restated Warrant Agreement]

See attached.

	BIOSCRIP, INC.	
FORM OF AMEN	NDED AND RESTATED WARRAN	T AGREEMENT

Dated As Of March 14, 2019 Warrants to Purchase shares of Common Stock

BIOSCRIP, INC.

Amended and Restated Warrant Agreement

Warrants For Common Stock

AMENDED AND RESTATED WARRANT AGREEMENT, dated as March 14, 2019, among BioScrip, Inc., a Delaware corporation (together with its successors and assigns, the "<u>Company</u>") and the holders undersigned hereto (collectively and together with each of their respective successors and assigns, the "<u>Holders</u>"). Capitalized terms shall have the meaning specified in <u>Section 5.1</u> hereof.

RECITALS

WHEREAS, on the terms of the Warrant Purchase Agreement (the "Original Warrant Agreement"), dated as June 29, 2017, among the Company and the Original Purchasers party thereto, the Original Purchasers agreed to acquire from the Company, and the Company agreed to issue to the Original Purchasers, Warrants to purchase 4.99% of the Fully Diluted Common Stock (as defined in the Original Warrant Agreement) outstanding on the date of any exercise of the Warrants (less the percentage of the shares of Common Stock previously issued pursuant to the Warrants from time to time as a result of any partial exercise of the Warrants as set forth herein);

WHEREAS, the Company is party to an Agreement and Plan of Merger (the "Merger Agreement"), dated as of March 14, 2019, by and among the Company, HC Group Holdings II, Inc., a Delaware corporation ("Omega"), Beta Sub, LLC, a Delaware limited liability company ("Merger Sub LLC"), and certain other parties signatory thereto, pursuant to which Omega will be merged with and into a newly formed subsidiary of the Company with Omega continuing as the surviving entity (the "Surviving Corporation") (the "First Merger"), immediately following which the Surviving Corporation will be merged with and into Merger Sub LLC, with Merger Sub LLC continuing as the surviving entity (the "Second Merger");

WHEREAS, the undersigned Holders collectively own 100% of the Warrants issued pursuant to the Original Warrant Agreement;

WHEREAS, in connection with, and conditioned upon, the consummation of the First Merger in accordance with the Merger Agreement, the Company and the Holders wish to amend and restate the Original Warrant Agreement so as to fix the number of shares of Common Stock issuable upon exercise of the Warrants at 8,287,317 shares in the aggregate, with each Holder having a Warrant to purchase the number of shares of Common Stock set forth on Annex 1 hereto; and

WHEREAS, the Company and the Holders wish to enter into this Agreement to amend and restate the terms of the Warrants.

AGREEMENT

NOW, THEREFORE, in consideration of the premises and the mutual agreements set forth herein, the parties to this Agreement hereby agree to amend and restate the Original Warrant Agreement as follows:

1. FORM, EXECUTION AND TRANSFER OF WARRANT CERTIFICATES.

1.1 Form of Warrant Certificates.

The Warrant Certificates shall be in the form set forth in <u>Attachment A</u> hereto. The Warrant Certificates may have such letters, numbers or other marks of identification or designation as may be required to comply with any law or with any

rule or regulation of any governmental authority, stock exchange or self-regulatory organization made pursuant thereto ("Law"). Each Warrant Certificate shall be initially dated as of June 29, 2017 and subsequently dated the date of issuance thereof by the Company upon transfer or exchange. Each Warrant Certificate shall represent the right to purchase the number of shares of Common Stock set forth in such Warrant Certificate at a price per share of Common Stock equal to the Exercise Price applicable to that Warrant Certificate; *provided*, that the number of shares of Common Stock issuable upon exercise of the Warrants and the Exercise Price thereof shall be subject to adjustment as provided herein. The aggregate amount of shares of Common Stock issuable upon exercise of all Warrants as of the Second Merger Effective Time shall be 8,287,317, subject to adjustment as of the Second Merger Effective Time as provided herein, with each Holder having a Warrant to purchase the number of shares of Common Stock set forth on Annex 1 hereto.

1.2 Execution of Warrant Certificates; Registration Books.

- (a) Execution of Warrant Certificates. The Warrant Certificates shall be executed on behalf of the Company by an officer of the Company authorized by the Board of Directors. In case the officer of the Company who shall have signed any Warrant Certificate shall cease to be such an officer of the Company before issuance and delivery by the Company of such Warrant Certificate, such Warrant Certificate nevertheless may be issued and delivered with the same force and effect as though the individual who signed such Warrant Certificate had not ceased to be such an officer of the Company, and any Warrant Certificate may be signed on behalf of the Company by any individual who, at the actual date of the execution of such Warrant Certificate, shall be a proper officer of the Company to sign such Warrant Certificate, although at the date of the execution of this Agreement any such individual was not such an officer.
- (b) **Registration Books.** The Company will keep or cause to be kept at its office, maintained at the address of the Company referenced in Section 6.6, at the Company's transfer agent, or at such other office of the Company of which the Company shall have given notice to each holder of Warrant Certificates, books for registration and transfer of the Warrant Certificates issued hereunder. Such books shall show the names and addresses of the respective holders of the Warrant Certificates, the registration number and date of each of the Warrant Certificates and the Denomination thereof.

1.3 Transfer, Split Up, Combination and Exchange of Warrant Certificates; Lost or Stolen Warrant Certificates.

(a) Transfer, Split Up, etc.

- (i) **Transfer**. Subject to compliance with the Securities Act, any applicable state securities laws and the Company's organizational documents, any Warrant Certificate (or portion thereof), with or without other Warrant Certificates, may be transferred to any Person for a Warrant Certificate or Warrant Certificates in an aggregate like Denomination as the Warrant Certificate or Warrant Certificates (or portions thereof) surrendered then entitled such registered holder to purchase. Any registered holder desiring to transfer any Warrant Certificate shall make such request in writing delivered to the Company, which request shall include the identity of the Transferee and the aggregate Denomination of Warrants to be transferred, and shall surrender the Warrant Certificate or Warrant Certificates (or portions thereof) to be transferred at the office of the Company referenced in Section 6.6, whereupon the Company shall deliver promptly to such Transferee a Warrant Certificate or Warrant Certificates, as the case may be, as so requested, which Warrant Certificate or Warrant Certificates shall evidence, collectively, the same aggregate Denomination of Warrants as the Warrant Certificate or Warrant Certificates (or portions thereof) so surrendered for transfer and shall issue a new Warrant Certificate to the transferor representing the Warrants retained by the Transferor if such transfer involved less than the entire Denomination of Warrants held by such Transferor.
- (ii) <u>Split Up, Combination, Exchange, etc.</u> Any Warrant Certificate, with or without other Warrant Certificates, may be split up, combined or exchanged for another Warrant Certificate or Warrant Certificates, in an aggregate like Denomination as the Warrant Certificate or Warrant Certificate surrendered then entitle such registered holder to purchase. Any registered holder desiring to split up, combine or exchange any Warrant Certificate shall make such request in writing delivered to the Company, and shall surrender the Warrant Certificate or Warrant Certificates to be split up, combined or exchanged at the office of the Company referenced in <u>Section 6.6</u>, whereupon the Company shall deliver promptly to such registered holder a Warrant Certificate or Warrant Certificates, as the case may be, as so requested, which Warrant Certificate or Warrant Certificates shall evidence, collectively, the same aggregate Denomination as the Warrant Certificate or Warrant Certificates so surrendered for split-up, combination or exchange.
- (b) **Loss, Theft, etc.** Upon receipt by the Company of evidence reasonably satisfactory to it of the ownership of, and the loss, theft, destruction or mutilation of, any Warrant Certificate, and:
- (i) in the case of loss, theft or destruction, an affidavit of loss, together with a customary and reasonable indemnity reasonably satisfactory to the Company; or

(ii) in the case of mutilation, upon surrender and cancellation thereof;

the Company at its own expense will execute and deliver, in lieu thereof, a new Warrant Certificate, dated the date of such lost, stolen, destroyed or mutilated Warrant Certificate and of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant Certificate and evidencing the same Denomination as the Warrant Certificate so lost, stolen, destroyed or mutilated.

1.4 Subsequent Issuance of Warrant Certificates.

Subsequent to the original issuance, no Warrant Certificates shall be issued except:

- (a) Warrant Certificates issued upon any transfer, combination, split up or exchange of Warrants pursuant to Section 1.3(a);
- (b) Warrant Certificates issued in replacement of mutilated, destroyed, lost or stolen Warrant Certificates pursuant to Section 1.3(b);
- (c) Warrant Certificates issued pursuant to <u>Section 2.3</u> upon the partial exercise of any Warrant Certificate to evidence the unexercised portion of such Warrant Certificate; and
 - (d) Warrant Certificates to reflect any adjustments pursuant to Section 4.

1.5 Effect of Issuance in Registered Form.

Every holder of a Warrant Certificate by accepting the same consents and agrees with the Company and with every other holder of a Warrant Certificate that:

- (a) the Warrant Certificates, to the extent then currently transferable, are transferable only on the registry books of the Company if and when surrendered at the office of the Company referenced in <u>Section 6.5</u>, duly endorsed or accompanied by an instrument of transfer (in the form attached thereto) and payment of any applicable transfer, stamp or issue tax (a "<u>Tax</u>"); and
- (b) the Company may deem and treat the Person in whose name each Warrant Certificate is registered as the absolute owner thereof and of the Warrants evidenced thereby (notwithstanding any notations of ownership or writing on the Warrant Certificates made by anyone other than the Company) for all purposes whatsoever, and the Company shall not be affected by any notice to the contrary.

2. EXERCISE OF WARRANTS; PAYMENT OF EXERCISE PRICE.

2.1 Exercise of Warrants.

- (a) **Manner of Exercise**. At any time and from time to time prior to the Expiration Time, the holder of any Warrant Certificate may exercise the Warrants evidenced thereby, in whole or in any part, by surrender to the Company, at its office referenced in Section 6.6, of such Warrant Certificate, together with a duly executed election to purchase (a form of which is attached to each Warrant Certificate) and payment of the applicable Exercise Price for each share of Common Stock with respect to which the Warrants are then being exercised and an amount equal to any applicable Tax (if not payable by the Company as provided in Section 3.3). Such Exercise Price shall be payable in cash or by withholding of shares of Common Stock into which the Warrants are being exercised pursuant to Section 2.1(b).
- (b) **Payment in Cash or by Withholding of Common Stock**. Upon exercise of any Warrants, the holder of a Warrant Certificate may pay the Exercise Price by either (i) certified or official bank check payable to the order of the Company or by wire transfer of immediately available funds to the account of the Company designated in writing by the Company or (ii) instructing the Company to withhold that number of shares of Common Stock (or fraction thereof) then issuable upon such exercise with an aggregate Market Price as of such exercise date equal to the aggregate Exercise Price. The Company acknowledges that the provisions of this clause are intended, in part, to ensure that a full or partial exchange of a Warrant Certificate will qualify as a conversion, within the meaning of paragraph (d)(3)(ii) of Rule 144 under the Securities Act.
- (c) **Fractional shares of Common Stock**. The Company may, in accordance with <u>Section 2.6</u>, pay the exercising holder cash in lieu of issuing a fractional share in connection with an exercise of Warrants; provided that, if it does not issue a fractional share in such circumstances, it will make such cash payment.

2.2 Issuance of Common Stock.

Upon timely receipt of a Warrant Certificate, accompanied by the form of election to purchase duly executed, and payment of the Exercise Price for each of the shares of the Common Stock to be purchased and by an amount equal to any

applicable Tax (if not payable by the Company as provided in Section 3.3), the Company shall thereupon promptly cause certificates representing the number of whole shares of Common Stock then being purchased to be delivered to or upon the order of the registered holder of such Warrant Certificate, registered in such name or names as may be designated by such holder, and, promptly after such receipt deliver the cash, if any, to be paid in lieu of fractional shares pursuant to Section 2.6 to or upon the order of the registered holder of such Warrant Certificate.

2.3 Unexercised Warrants.

In the event that the registered holder of any Warrant Certificate shall exercise less than all the Warrants evidenced thereby, a new Warrant Certificate evidencing a number of unexercised Warrants shall be issued by the Company to the registered holder of such Warrant Certificate or to its duly authorized assigns.

2.4 Cancellation and Destruction of Warrant Certificates.

All Warrant Certificates surrendered to the Company for the purpose of exercise, payment of the Exercise Price, exchange, substitution or transfer shall be cancelled by it, and no Warrant Certificates shall be issued in lieu thereof except as expressly permitted by any of the provisions of this Agreement. The Company shall cancel and retire any other Warrant Certificates purchased or acquired by the Company.

2.5 Expiration.

All Warrants that have not been exercised or purchased in accordance with the provisions of this Agreement shall expire and all rights of holders of such Warrants hereunder shall terminate and cease at the Expiration Time.

2.6 Fractional shares of Common Stock.

The Company shall not be required to issue fractional shares of Common Stock upon the exercise of any Warrant. If fractional shares are not issued upon the exercise of any Warrant, there shall be paid to the holder thereof, in lieu of any fractional share of Common Stock resulting therefrom, an amount of cash equal to the product of:

- (a) the fractional amount of such share of Common Stock; and
- (b) the Market Price, as determined on the trading day immediately prior to the date of exercise of such Warrant.

2.7 RESERVED

3. AGREEMENTS OF THE COMPANY.

3.1 Reservation of Common Stock.

The Company covenants and agrees that it will at all times cause to be reserved and kept available out of its authorized and unissued shares of treasury shares of Common Stock such number of shares of Common Stock as will be sufficient to permit the exercise in full of all Warrants issued hereunder into Common Stock.

3.2 Common Stock to be Duly Authorized and Issued, Fully Paid and Nonassessable, etc; Compliance with

Law

The Company covenants and agrees that it will take all such action as may be necessary to ensure that all shares of Common Stock delivered upon the exercise of any Warrant and the payment of the Exercise Price pursuant to Section 2.1 (in each case, at the time of delivery of the certificates representing such shares of Common Stock) shall (a) be duly and validly authorized and issued and fully paid and nonassessable, free of any preemptive rights in favor of any Person in respect of such issuance and free of any security interest, pledge, mortgage, lien, charge or other encumbrance created by, or arising out of actions of, the Company, in each case other than (i) resulting from the actions and circumstances of the holder of such shares of Common Stock, (ii) to the extent imposed by applicable laws or (iii) in the Company's organizational documents and (b) be issued without violation of any applicable Law.

3.3 Taxes.

The Company covenants and agrees that it will pay when due and payable any and all Taxes and charges that may be payable in respect of the initial issuance or delivery of:

- (a) each Warrant Certificate;
- (b) each Warrant Certificate issued in exchange for any other Warrant Certificate pursuant to Section 1.3, Section 2.3 or Section 4; and
- (c) each share of Common Stock issued upon the exercise of any Warrant.

The Company shall not, however, be required to:

- (i) pay any Tax that may be payable in respect of the transfer or delivery of Warrant Certificates in a name other than that of the registered holder of the Warrant Certificate surrendered for exercise, conversion, transfer or exchange (any such Tax being payable by the holder of such certificate at the time of surrender); or
- (ii) issue or deliver any such certificates referred to in the foregoing clause (i) until any such Tax referred to in the foregoing clause (i) shall have been paid.

3.4 Common Stock Record Date.

Each Person in whose name any certificate for shares of Common Stock is issued upon the exercise of Warrants shall for all purposes be deemed to have become the holder of record of the Common Stock represented thereby on, and such certificates (if any) shall be dated, the date upon which the Warrant Certificate evidencing such Warrants was duly surrendered with an election to purchase attached thereto duly executed and payment of the aggregate Exercise Price (and any applicable Taxes, if payable by such Person) was made.

3.5 Rights in Respect of Common Stock.

Prior to the exercise of the Warrants evidenced thereby, the holder of a Warrant Certificate shall not be entitled to any rights of a stockholder of the Company with respect to the Common Stock into which the Warrants shall be exercisable, including, without limitation, the right to vote in respect of any matter upon which the holders of Common Stock may vote, the right to receive any distributions of cash or property and, except as expressly set forth herein, the right to receive any notice of any proceedings of the Company. Prior to the exercise of the Warrants evidenced thereby, the holders of the Warrant Certificates shall not have as such any obligation in respect of any assessment or any other obligation or liability as a stockholder of the Company, whether such obligations or liabilities are asserted by the Company or by creditors of the Company.

3.6 Noncircumvention.

The Company hereby covenants and agrees that the Company will not, by amendment of its charter, bylaws or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issue or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant Agreement, and will at all times in good faith carry out all the provisions of this Warrant Agreement.

4. ANTI-DILUTION ADJUSTMENTS.

4.1 Adjustments.

The number of shares of Common Stock purchasable upon the exercise of each Warrant, and the Exercise Price, shall be subject to adjustment as set forth in this Section 4; provided that no adjustment of the Exercise Price shall result in an increase in the Exercise Price.

4.2 Stock Splits, Subdivisions, Reclassifications or Combinations.

If the Company shall (i) declare and pay a dividend or make a distribution on its Common Stock in shares of Common Stock, (ii) subdivide or reclassify the outstanding shares of Common Stock into a greater number of shares, or (iii) combine or reclassify the outstanding shares of Common Stock into a smaller number of shares, the Denomination of any Warrant Certificate and the Common Stock issuable upon exercise of any Warrants at the time of the record date for such dividend or effective date of such split, reverse split, subdivision, combination or reclassification shall be proportionately adjusted so that the holder of such Warrants after such date shall be entitled to purchase the number of shares of Common Stock which such holder would have owned or been entitled to receive in respect of the shares of Common Stock subject to such Warrants after such date had such Warrants been exercised immediately prior to such date. In such event the Exercise Price in effect at the time of the effective date of such split, reverse split, subdivision, combination or reclassification shall be adjusted to the number obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of such Warrants immediately before such adjustment and (2) the Exercise Price in effect immediately prior to the record or effective date, as the case may be, for the dividend, distribution, split, reverse split, subdivision, combination or reclassification giving rise to this adjustment by (y) the new number of shares of Common Stock issuable upon exercise of such Warrants determined pursuant to the immediately preceding sentence; provided that the Exercise Price shall not be adjusted to be less than the par value of the Common Stock; provided, further, that the adjustment to the Exercise Price in this Section 4.2 may result in an increase in the Exercise Price, notwithstanding Section 4.1.

4.3 Price Based Anti-Dilution

- (a) Without duplication of the adjustments set forth in Sections 4.2 or 4.4, if the Company shall issue after the Second Merger Effective Time or sell any shares of Common Stock (as actually issued or, pursuant to Section 4.3(b), deemed to be issued) for a consideration per share less than either the Market Price per share or the Exercise Price immediately prior to such issuance or sale, or if earlier, upon the execution of the definitive documentation with respect to such issuance or sale (the "Effective Time"), then immediately upon the Effective Time the Denomination of any Warrant Certificate and the number of shares of Common Stock issuable upon exercise of any Warrants at the time of the effective date shall be increased by multiplying such number of shares of Common Stock by a fraction, (i) the numerator of which shall be the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock so issued or sold (or deemed to be issued or sold pursuant to Section 4.3(b)), and (ii) the denominator of which shall be the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock which the aggregate consideration received by the Company for the total number of shares of Common Stock so issued or sold (or deemed to be issued or sold pursuant to Section 4.3(b)) would purchase if such shares were sold at the greater of the Market Price or the Exercise Price, and the Exercise Price shall be adjusted by multiplying the Exercise Price in effect immediately prior to such issuance or sale (or deemed issuance or sale) by the inverse of such fraction. For the purposes of this Section 4.3(a), none of the following issuances shall be considered the issuance or sale of Common Stock:
- (i) the issuance of Common Stock (a) upon the conversion or exercise of any then-outstanding Common Stock Equivalents; (b) to Persons in connection with joint ventures, strategic alliances or other commercial relationships with such Person relating to the operation of the Company's business and not for the primary purpose of raising equity capital, in connection with Business Combinations, or transactions in which the Company, directly or indirectly, acquires another business or the assets thereof or (c) in an offering for cash for account of the Company that is underwritten on a firm commitment basis and is registered with the Securities and Exchange Commission;
 - (ii) the issuance of any Common Stock or Common Stock Equivalents for which the adjustment provided in Section 4.2 applies; or
- (iii) the issuance of shares of Common Stock or Common Stock Equivalents to employees or directors of either the Company or any Company Subsidiary that is approved by the Board of Directors.
 - (b) For the purposes of Section 4.3(a), the following subparagraphs (i) to (iii), inclusive, shall also be applicable:
- (i) If the Company shall grant any rights to subscribe for, or any rights or options to purchase, Common Stock Equivalents, whether or not such rights or options or the right to convert or exchange any such Common Stock Equivalents are immediately exercisable, and the price per share for which Common Stock is issuable upon the exercise of such rights or options or upon conversion or exchange of such Common Stock Equivalents (determined by dividing (A) the total amount, if any, received or receivable by the Company as consideration for the granting of such rights or options, plus the minimum aggregate amount of additional consideration payable to the Company upon the exercise of such rights or options, plus, in the case of any such rights or options which relate to such Common Stock Equivalents, the minimum aggregate amount of additional consideration, if any, payable upon the issue or sale of such Common Stock Equivalents and upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the exercise of such rights or options or upon the conversion or exchange of all such Common Stock Equivalents issuable upon the exercise of such rights or options) shall be less than the Exercise Price or the Market Price per share of Common Stock immediately prior to the earlier of (x) the time of the granting of such rights or options and (y) the time of execution of definitive documentation with respect to such grant (such earlier time of (x) and (y) to be the "Effective Time" for purposes of Section 4.3(a)), then the total maximum number of shares of Common Stock issuable upon the exercise of such rights or options or upon conversion or exchange of the total maximum amount of such Common Stock Equivalents issuable upon the exercise of such rights or options shall (as of the date of granting of such rights or options or the date of execution of definitive documents, if earlier) be deemed to be outstanding and to have been issued for such price per share; provided that no further adjustment of the conversion price pursuant to this Section 4.3(a) or Section 4.3(b) (i) shall be made (x) upon the actual issuance or sale of such Common Stock Equivalents upon the exercise of any rights to subscribe for, or any rights or options to purchase, such Common Stock Equivalents or (y) upon the actual issuance or sale of such Common Stock upon the exercise of any such Common Stock Equivalents, including without limitation, in each case of clauses (x) and (y) with respect to shares of Common Stock Equivalents or Common Stock issued or issuable as a result of the effect of antidilution adjustments under any such security.

(ii) If the Company shall issue or sell any Common Stock Equivalents, whether or not the rights to exchange or convert thereunder are immediately exercisable, and the price per share for which Common Stock is issuable upon such conversion or exchange (determined by dividing (A) the total amount received or receivable by the Company as consideration for the issue or sale of such Common Stock Equivalents, plus the minimum aggregate amount of additional

consideration, if any, payable to the Company upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the conversion or exchange of all such Common Stock Equivalents) shall be less than the Exercise Price or the Market Price per share of Common Stock immediately prior to the Effective Time for such issuance or sale, then the total maximum number of shares of Common Stock issuable upon conversion or exchange of such Common Stock Equivalents shall (as of the Effective Time) be deemed to be outstanding and to have been issued for such price per share, provided that no further adjustment of the conversion price pursuant to this Section 4.3(a) or Section 4.3(b)(ii) shall be made upon the actual issuance or sale of such Common Stock upon the exercise of any such Common Stock Equivalents, including without limitation, in each case with respect to shares of Common Stock issued or issuable as a result of the effect of antidilution adjustments under any such security.

(iii) In case at any time any shares of Common Stock or Common Stock Equivalents or any rights or options to purchase any such Common Stock or Common Stock Equivalents shall be issued or sold for cash to a non-Affiliate of the Company, the consideration received therefor shall be deemed to be the amount received by the Company therefor. In case any shares of Common Stock or Common Stock Equivalents or any rights or options to purchase any such Common Stock or Common Stock Equivalents shall be issued or sold to an Affiliate of the Company or for a consideration other than or in addition to cash, the amount of the consideration received by the Company shall be deemed to be the Fair Market Value of solely such consideration received by the Company in respect of such purchase of Common Stock or Common Stock Equivalents.

4.4 Other Distributions. In case the Company shall fix a record date for the making of a dividend or distribution to holders of shares of its Common Stock of securities, evidences of indebtedness, assets, cash, rights or warrants (excluding dividends of its Common Stock and other dividends or distributions referred to in Section 4.2), in each such case, the Exercise Price in effect prior to such record date shall be reduced immediately thereafter to the price determined by multiplying the Exercise Price in effect immediately prior to the reduction by the quotient of (x) the Market Price of the Common Stock on the last trading day preceding the first date on which the Common Stock trades on the Exchange on which the Common Stock is listed or admitted to trading without the right to receive such distribution, minus the amount of cash and/or the Fair Market Value of the securities, evidences of indebtedness, assets, rights or warrants to be so distributed in respect of one share of Common Stock (such amount and/or Fair Market Value, the "Per Share Fair Market Value") divided by (y) such Market Price on such date specified in clause (x); such adjustment shall be made successively whenever such a record date is fixed. In obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of any Warrants shall be increased to the number obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of such Warrants before such adjustment, and (2) the Exercise Price in effect immediately prior to the distribution giving rise to this adjustment by (y) the new Exercise Price determined in accordance with the immediately preceding sentence. In the event that such distribution is not so made, the Exercise Price and the number of shares of Common Stock issuable upon exercise of such Warrants then in effect shall be readjusted, effective as of the date when the Board of Directors determines not to distribute such shares, evidences of indebt

4.5 Business Combinations.

In case of any Business Combination or reclassification of Common Stock (other than a reclassification of Common Stock referred to in Section 4.2), a holder's right to receive shares of Common Stock upon exercise of any Warrants shall be converted into the right to exercise such Warrant to acquire the number of shares of stock or other securities or property (including cash) which the Common Stock issuable (at the time of such Business Combination or reclassification) upon exercise of such Warrants immediately prior to such Business Combination or reclassification would have been entitled to receive upon consummation of such Business Combination or reclassification; and in any such case, if necessary, the provisions set forth herein with respect to the rights and interests thereafter of such holder shall be appropriately adjusted so as to be applicable, as nearly as may reasonably be, to such holder's right to exercise such Warrants in exchange for any shares of stock or other securities or property pursuant to this Section 4.5. In determining the kind and amount of stock, securities or the property receivable upon exercise of any Warrants following the consummation of such Business Combination, if the holders of Common Stock have the right to elect the kind or amount of consideration receivable upon consummation of such Business Combination. The Company shall not enter into or be party to any Business Combination unless the successor of the Company (if any), assumes in writing the obligation to deliver to each holder of Warrants hereunder in exchange for such Warrants a security of such successor evidenced by a written instrument substantially similar in form and substance to this Agreement.

4.6 Expiration of Rights or Options.

Upon the expiration of any rights or options to subscribe for, purchase or convert or exchange Common Stock or

Common Stock Equivalents in respect of the issuance, sale or grant of which adjustment was made pursuant to Section 4.3, without the exercise thereof, the Exercise Price and the number of shares of Common Stock purchasable upon the exercise of the Warrants shall, upon such expiration, be readjusted (and such readjustment may result in an increase in the Exercise Price, notwithstanding Section 4.1, not to exceed the Exercise Price prior to such original adjustment) and shall thereafter be such Exercise Price and such number of shares of Common Stock as would have been in effect had such Exercise Price and such number of shares of Common Stock not been originally adjusted (or had the original adjustment not been required, as the case may be), as if:

- (a) the only shares of Common Stock so issued were the shares of Common Stock, if any, actually issued or sold upon the exercise of such rights or options; and
- (b) such shares of Common Stock, if any, were issued or sold for the consideration actually received by the Company upon such exercise plus the aggregate consideration, if any, actually received by the Company for the issuance, sale or grant of all of such rights or options, whether or not exercised; provided that no such readjustment shall have the effect of increasing the Exercise Price by an amount in excess of the amount of the reduction initially made in respect of the issuance, sale, or grant of such rights or options.

4.7 Rounding of Calculations; Minimum Adjustments.

All calculations under this Section 4 shall be made to the nearest one-tenth (1/10th) of a cent or to the nearest one-hundredth (1/100th) of a share, as the case may be. Any provision of this Section 4 to the contrary notwithstanding, no adjustment in the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable shall be made if the amount of such adjustment would be less than \$0.01 or one-tenth (1/10th) of a share of Common Stock, but any such amount shall be carried forward and an adjustment with respect thereto shall be made at the earlier of (i) the time of any exercise of Warrants and (ii) the time of and together with any subsequent adjustment which, together with such amount and any other amount or amounts so carried forward, shall aggregate \$0.01 or 1/10th of a share of Common Stock, or more.

4.8 Timing of Issuance of Additional Common Stock Upon Certain Adjustments.

In any case in which the provisions of this <u>Section 4</u> shall require that an adjustment shall become effective immediately after a record date for an event, the Company may defer until the occurrence of such event (i) issuing to the holder of any Warrants exercised after such record date and before the occurrence of such event the additional shares of Common Stock issuable upon such exercise by reason of the adjustment required by such event over and above the shares of Common Stock issuable upon such exercise before giving effect to such adjustment and (ii) paying to such holder any amount of cash in lieu of a fractional share of Common Stock; *provided*, however, that the Company upon written request shall deliver to such holder a due bill or other appropriate instrument evidencing such holder's right to receive such additional shares, and such cash, upon the occurrence of the event requiring such adjustment.

4.9 Miscellaneous.

- (a) **Statement Regarding Adjustments**. Whenever after the Second Merger Effective Time the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable shall be adjusted as provided in <u>Section 4</u>, the Company shall promptly file at the principal office of the Company referenced in <u>Section 6.6</u> a statement showing in reasonable detail the facts requiring such adjustment and the Exercise Price that shall be in effect and the number of shares of Common Stock into which such Warrants shall be exercisable after such adjustment, and the Company shall also cause a copy of such statement to be sent by mail, first class postage prepaid, or email to each holder of Warrants at the address or email address appearing in the Company's records.
- (b) **Notice of Adjustment Event**. In the event that the Company shall propose after the Second Merger Effective Time to take any action of the type described in this Section 4 (but only if the action of the type described in this Section 4 would result in an adjustment in the Exercise Price or the number of shares of Common Stock into which Warrants are exercisable or a change in the type of securities or property to be delivered upon exercise of Warrants), the Company shall give notice to the holders of Warrants, in the manner set forth in Section 4.9(a), which notice shall specify the record date, if any, with respect to any such action and the approximate date on which such action is to take place. Such notice shall also set forth the facts with respect thereto as shall be reasonably necessary to indicate the effect on the Exercise Price and the number, kind or class of shares or other securities or property which shall be deliverable upon exercise of any Warrants. In the case of any action which would require the fixing of a record date, such notice shall be given at least 10 days prior to the date so fixed, and in case of all other action, such notice shall be given at least 15 days prior to the taking of such proposed action. Without limiting the foregoing, to the extent notice of any of the foregoing actions or events is given to the holders of the Common Stock, such notice shall be provided to the holders of the Warrants on or before such notice to the holders of Common Stock.
 - (c) Proceedings Prior to Any Action Requiring Adjustment. As a condition precedent to the taking of

any action which would require an adjustment pursuant to this Section 4, the Company shall take any action which may be necessary, including obtaining regulatory, New York Stock Exchange, NASDAQ Stock Market or other applicable national securities exchange (an "Exchange") or stockholder approvals or exemptions, in order that the Company may thereafter validly and legally issue as fully paid and nonassessable all shares of Common Stock that the holders are entitled to receive upon exercise of this any Warrants pursuant to this Section 4.

- (d) **Adjustment Rules**. Any adjustments pursuant to this Section 4 shall be made successively whenever an event referred to herein shall occur. If more than one subsection of this Section 4 is applicable to a single event, the subsection shall be applied that produces the largest adjustment and no single event shall cause an adjustment under more than one subsection of this Section 4 so as to result in duplication. If an adjustment in Exercise Price made hereunder would reduce the Exercise Price to an amount below par value of the Common Stock, then such adjustment in Exercise Price made hereunder shall reduce the Exercise Price to the par value of the Common Stock.
- (e) **Merger Agreement**. Notwithstanding anything contained herein to the contrary, the consummation of the transactions contemplated by the Merger Agreement (including the issuance of any shares of Common Stock by the Company contemplated therein) will not result in, or cause any adjustment in, the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable pursuant to this Section 4.

5. INTERPRETATION OF THIS AGREEMENT.

5.1 Certain Defined Terms.

For the purpose of this Agreement, the following terms shall have the meanings set forth below or set forth in the Section hereof following such term:

"Affiliate" means, with respect to any Person, (a) a director, officer or shareholder of such Person, (b) in the case of a natural person, a spouse, parent, sibling or descendant of such Person (or spouse, parent, sibling or descendant of any director or executive officer of such Person) and (c) any other Person that, directly or indirectly through one or more intermediaries, Controls, or is Controlled by, or is under common Control with, such Person, at such time; provided, however, that none of the Holders shall be deemed to be an "Affiliate" of the Company and no Person holding any one or more of the Warrants shall be deemed to be an "Affiliate" of the company solely by virtue of the ownership thereof.

- "Agreement" means this Amended and Restated Warrant Agreement as it may from time to time be amended, restated, modified or supplemented.
- "Board of Directors" means the board of directors of the Company, including any duly authorized committee thereof.
- "Beneficially Own" has the meaning has the meaning given to it in Section 13D of the Exchange Act and the rules promulgated thereunder.
- "Business Combination" means any consolidation of the Company with, or merger of the Company with or into, another Person (other than a merger in which (a) the Company is the surviving corporation, (b) that does not result in any reclassification or change of shares of Common Stock outstanding immediately prior to such merger and (c) the holders of Common Stock are not entitled to receive any consideration therefrom), or any sale or conveyance to another Person of the assets of the Company substantially as an entirety.
- "business day" means any day that is not a Saturday, Sunday or other day on which banking institutions in New York, New York are authorized or required by Law or executive order to close.
- "Capital Stock" means (A) with respect to any Person that is a corporation or company, any and all shares, interests, participations or other equivalents (however designated) of capital or capital stock of such Person and (B) with respect to any Person that is not a corporation or company, any and all partnership or other equity interests of such Person.
 - "Charter" means, with respect to any Person, its certificate or articles of incorporation, articles of association, or similar organizational document.
 - "Common Stock" means the Company's common stock, par value \$.0001 per share.
- "Common Stock Equivalents" means all options, derivatives, rights, warrants and convertible, exercisable or exchangeable securities or instruments (including, without limitation, awards or grants of such securities or instruments to directors, officers, employees or consultants of the Company or to other persons).
 - "Company" has the meaning set forth in the introductory paragraph hereof.
 - "Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause

the direction of the management and policies of such Person, whether through the ownership of voting Securities, by contract or otherwise.

"<u>Denomination</u>" means, in the case of any Warrant Certificate, the number of shares of Common Stock issuable upon exercise of the Warrants represented by such Warrant Certificate.

"Effective Date" shall mean the date on which the Second Merger is consummated in accordance with the terms of the Merger Agreement.

"Effective Time" has the meaning set forth in Section 4.3(a).

"Exchange" has the set forth in Section 4.9(c).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor statute, and the rules and regulations promulgated thereunder.

"Exercise Price" means, prior to any adjustment pursuant to Section 4 of this Agreement, the Initial Exercise Price; and thereafter, the Initial Exercise Price as successively adjusted and readjusted from time to time in accordance with the provisions of Section 4.

"Expiration Time" means 5:00 p.m., Eastern time, on the tenth (10th) year anniversary of June 29, 2017.

"Fair Market Value" means, with respect to any security or other property, the fair market value of such security or other property as determined by the Board of Directors, acting in good faith and transmitted to the holders of Warrants by written notice. The Required Warrantholders may object in writing to the Board of Director's calculation of Fair Market Value within 10 days of receipt of such written notice thereof. If the Required Warrantholders and the Board of Directors are unable to agree on Fair Market Value during the 10-day period following the delivery of the Required Warrantholders' objection, then the Board of Directors shall select and approve an appraiser of national recognition experienced in the business of evaluating or appraising the market value of securities (which appraiser shall be subject to approval by the Required Warrantholders, which approval shall not be unreasonably withheld, conditioned or delayed). The Fair Market Value established by such appraiser shall be conclusive and binding on the parties. The fees and expenses for such appraiser shall be borne 50% by the Company, on the one hand, and 50% by the holders of Warrants, on the other hand.

"Initial Exercise Price" means, with respect to each Warrant Certificate issued to a Holder on the Issue Date, \$2.00.

"Issue Date" means June 29, 2017.

"Law" has the set forth in Section 1.1.

"Market Price" means, with respect to a particular security, on any given day, the volume weighted average trading price for the period of ten consecutive trading days ending on the trading day prior to such given day as reported by Bloomberg Financial Markets, or any successor thereto, through its "Volume Weight Average Price" function, on the Exchange on which the applicable securities are listed or admitted to trading. "Market Price" shall be determined without reference to after hours or extended hours trading. If such security is not listed and traded in a manner that the quotations referred to above are available for the period required hereunder, the Market Price per share of Common Stock shall be deemed to be the fair market value per share of such security as determined in good faith by the Board of Directors in reliance on an opinion of a nationally recognized independent investment banking corporation retained by the Company for this purpose (which opinion shall be made available to the holders of Warrants); provided that the Required Warrantholders may object in writing to the Board of Director's calculation of fair market value within 10 days of receipt of written notice thereof. If the Required Warrantholders and the Board of Directors are unable to agree on fair market value during the 10-day period following the delivery of the Required Warrantholders' objection, then the Board of Directors shall select and approve an appraiser experienced in the business of evaluating or appraising the market value of securities (which appraiser shall be subject to approval by the Required Warrantholders, which approval shall not be unreasonably withheld, conditioned or delayed). The Market Price established by such appraiser shall be conclusive and binding on the parties. In the event the Market Price established by such appraiser is greater than the Market Price previously determined by the Board of Directors, the fees and expenses for such appraiser shall be borne by the Company. In the event the Market Price established by such appraiser is less than or equal to the Market Price previously determined by the Board of Directors, the fees and expenses for such appraiser shall be borne by the holders of Warrants. For the purposes of determining the Market Price of the Common Stock on the "trading day" preceding, on or following the occurrence of an event, (i) that trading day shall be deemed to commence immediately after the regular scheduled closing time of trading on the Nasdaq Stock Market or, if trading is closed at an earlier time, such earlier time and (ii) that trading day shall end at the next regular scheduled closing time, or if trading is closed at an earlier time, such earlier time (for the avoidance of doubt, and as an example, if the Market Price is to be determined as of the last trading day preceding a specified event and the closing time of trading on a particular day is 4:00 p.m. and the specified event occurs at 5:00 p.m. on that day, the Market Price would be determined by reference to such 4:00

p.m. closing price).

- "Original Warrant Agreement" has the meaning set forth in the recitals hereto.
- "Per Share Fair Market Value" has the meaning set forth in Section 4.4.
- "Person" has the meaning given to it in Section 3(a)(9) of the Exchange Act and as used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act.
- "Holders" has the meaning set forth in the introductory paragraph hereof.
- "Required Warrantholders" means, at any time, the holders of Warrants representing at least a majority of the Common Stock issuable upon exercise of the Warrants issued and outstanding hereunder and not previously exercised (exclusive of any Warrants directly or indirectly held by the Company or any Affiliate of the Company).
 - "Securities Act" means the Securities Act of 1933, as amended, or any successor statute, and the rules and regulations promulgated thereunder.
 - "Tax" or "Taxes" has the meaning set forth in Section 1.5(a).
- "trading day" means (A) if the shares of Common Stock are not traded on any national or regional securities exchange or association or over-the-counter market, a business day or (B) if the shares of Common Stock are traded on any national or regional securities exchange or association or over-the-counter market, a business day on which such relevant exchange or quotation system is scheduled to be open for business and on which the shares of Common Stock (i) are not suspended from trading on any national or regional securities exchange or association or over-the-counter market for any period or periods aggregating one half hour or longer; and (ii) have traded at least once on the national or regional securities exchange or association or over-the-counter market that is the primary market for the trading of the shares of Common Stock.
- "Transferee" means any registered transferee of all or any part of any one or more Warrant Certificates initially acquired by the Holders under this Agreement.
 - "Warrant" means a warrant to initially purchase one share of Common Stock issued pursuant to this Agreement.
 - "Warrant Certificate" means a certificate evidencing the Warrants in the form of Attachment A.
 - 5.2 Section Heading and Table of Contents and Construction.
- (a) Section Headings and Table of Contents, etc. The titles of the Sections of this Agreement and any Table of Contents of this Agreement appear as a matter of convenience only, do not constitute a part hereof and shall not affect the construction hereof. The words "herein," "hereof," "hereunder" and "hereto" refer to this Agreement as a whole and not to any particular Section or other subdivision. References to Sections are, unless otherwise specified, references to Sections of this Agreement. References to Annexes and Attachments are, unless otherwise specified, references to Annexes and Attachments attached to this Agreement.
- (b) **Independent Construction**. Each covenant contained herein shall be construed (absent an express contrary provision herein) as being independent of each other covenant contained herein, and compliance with any one covenant shall not (absent such an express contrary provision) be deemed to excuse compliance with one or more other covenants.

5.3 Directly or Indirectly.

Where any provision herein refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether such action is taken directly or indirectly by such Person, including actions taken by or on behalf of any partnership in which such Person is a general partner.

5.4 Governing Law.

THIS AGREEMENT AND THE WARRANT CERTIFICATES SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE PARTIES SHALL BE GOVERNED BY, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICT OF LAWS RULES THEREOF TO THE EXTENT THAT ANY SUCH RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION, EXCEPT TO THE EXTENT THAT THE DELAWARE GENERAL CORPORATION LAW SPECIFICALLY AND MANDATORILY APPLIES.

6. MISCELLANEOUS.

6.1 Condition to Effectiveness; Termination.

- (a) The parties agree and acknowledge that, notwithstanding anything to the contrary herein, the effectiveness of this Agreement is conditioned upon, and shall not be binding or effective until, the occurrence of the First Merger Effective Time (as such term is defined in the Merger Agreement) in accordance with the terms and conditions of the Merger Agreement.
- (b) In the event that the Merger Agreement is terminated in accordance with its terms, this Agreement shall automatically terminate and be null and void *ab initio* and of no effect whatsoever, and the Original Warrant Agreement shall remain in full force and effect and shall continue to be binding on and enforceable against the Company and the Holders.

6.2 Expenses.

Issuance of certificates for shares of Common Stock to a holder upon the exercise of any Warrants shall be made without charge to such holder for any Tax or stamp duty in respect of the issuance of such certificates, all of which Taxes and stamp duties shall be paid by the Company (for the avoidance of doubt, specifically excluding the Taxes not payable by the Company pursuant to Section 3.3).

6.3 Amendment and Waiver.

This Agreement may be amended, and the observance of any term of this Agreement may be waived, with and only with the written consent of the Company and the Required Warrantholders; provided, however, that no amendment or waiver of the provisions of Section 2.1, this Section 6.3, Section 4 or of any term defined in Section 5.1 to the extent used in such specifically referenced Sections, may be made without the prior written consent of all holders of Warrants then outstanding (excluding any Warrants directly or indirectly held by the Company or any Affiliate of the Company); and, provided, further, that:

- (a) no such amendment or waiver of any of the provisions of this Agreement pertaining to the Exercise Price or the number of shares or kind of Common Stock that may be purchased upon exercise of each Warrant; and
 - (b) no change accelerating the occurrence of the Expiration Time;

shall be effective as to a holder of Warrants unless consented to in writing by such holder.

6.4 Entire Agreement; Termination of Original Warrant Agreement.

This Agreement and the Warrant Certificates embody the entire agreement and understanding among the Company and the Holders, and supersede all prior agreements and understandings, relating to the subject matter hereof. The Company and the Holders hereby acknowledge and agree that as of the Effective Date, the Original Warrant Agreement and any Warrant Certificate issued thereunder are terminated and of no further force and effect and confirm, pursuant to Section 6.2 of the Original Warrant Agreement, that all provisions thereof are terminated and of no further force and effect and the Original Warrant Agreement and any Warrant Certificate issued thereunder are hereby superseded and replaced by this Agreement, which amends and restates the Original Agreement and any Warrant Certificate issued thereunder in their entirety. Each Holder shall, on or promptly following the Second Merger Effective Time, deliver and surrender to the Company any Warrant Certificate issued under the Original Agreement in such Holder's possession.

6.5 Successors and Assigns; Multiple Holders.

All covenants and other agreements in this Agreement made by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors and assigns of the parties hereto to the extent they become holders of Warrants (including, without limitation, any Transferee) whether so expressed or not. Notwithstanding the foregoing sentence, the Company may not assign any of its rights, duties or obligations hereunder or under the Warrant Certificates except pursuant to the terms of this Agreement without the prior written consent of the Required Warrantholders.

Each holder of a Warrant agrees that (i) no other holder of a Warrant will by virtue of this Warrant or exercise thereof be under any fiduciary or other duty to give or withhold any consent or approval under this Warrant or to take any other action or omit to take any action under this Warrant and (ii) each other holder of a Warrant may act or refrain from acting under this Warrant as such other holder may, in its discretion, elect.

6.6 Notices and Information.

All communications hereunder or under the Warrants shall be in writing and shall be delivered either by certified or registered mail, postage prepaid, return receipt requested, email or nationally recognized overnight courier, and shall be addressed to the following addresses:

(a) if to a Holder, at its address set forth on Annex 2 to this Agreement, or at such other address as such Holder shall have specified to the Company in writing;

- (b) if to any other holder of any Warrant Certificate, addressed to such other holder at such address as such other holder shall have specified to the Company in writing or, if any such other holder shall not have so specified an address to the Company, then addressed to such other holder in care of the last holder of such Warrant Certificate that shall have so specified an address to the Company; and
- (c) if to the Company, at the address set forth on Annex 3 to this Agreement, or at such other address as the Company shall have specified to each holder of Warrants in writing.

Any communication addressed and delivered as herein provided shall be deemed to be received when actually delivered to the address of the addressee (whether or not delivery is accepted) by a nationally recognized overnight delivery service which provides proof of delivery or on the date postmarked if sent by registered or certified mail or upon receipt by the recipient's email server if directed to the email address provided in the notice section hereof, as the case may be. Any communication not so addressed and delivered shall be ineffective unless actually received by the intended addressee. Notwithstanding the foregoing provisions of this Section 6.6, service of process in any suit, action or proceeding arising out of or relating to this Agreement or any document, agreement or transaction contemplated hereby shall be delivered in the manner provided in Section 6.9(c).

Upon exercise of any Warrant pursuant to the terms hereof, the Company will use commercially reasonable efforts to answer a limited number of reasonable questions and provide a limited amount of reasonable documentation regarding any non-confidential information supporting its calculation of Fully Diluted Common Stock and the related shares of Common Stock issuable for the Denomination of such Warrant being exercised.

6.7 Severability.

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

6.8 Execution in Counterpart.

This Agreement may be executed in one or more counterparts and shall be effective when at least one counterpart shall have been executed by each party hereto, and each set of counterparts that, collectively, show execution by each party hereto shall constitute one duplicate original.

6.9 Waiver of Jury Trial; Consent to Jurisdiction, Etc.

- (a) Waiver of Jury Trial. THE PARTIES HERETO VOLUNTARILY AND INTENTIONALLY WAIVE ANY RIGHT ANY OF THEM MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE WARRANTS OR ANY OF THE DOCUMENTS, AGREEMENTS OR TRANSACTIONS CONTEMPLATED HEREBY.
- (b) Consent to Jurisdiction. ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE WARRANTS, OR ANY OF THE DOCUMENTS, AGREEMENTS OR TRANSACTIONS CONTEMPLATED HEREBY (WHETHER IN TORT, CONTRACT OR OTHERWISE) OR ANY ACTION OR PROCEEDING TO EXECUTE OR OTHERWISE ENFORCE ANY JUDGMENT IN RESPECT OF ANY BREACH UNDER THIS AGREEMENT, THE WARRANTS OR ANY DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY SHALL BE BROUGHT BY SUCH PARTY IN ANY NEW YORK STATE COURT OR FEDERAL DISTRICT COURT LOCATED IN THE SOUTHERN DISTRICT OF NEW YORK AS SUCH PARTY MAY IN ITS SOLE DISCRETION ELECT, AND BY THE EXECUTION AND DELIVERY OF THIS AGREEMENT, THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMIT TO THE IN PERSONAM JURISDICTION OF EACH SUCH COURT, AND EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES AND AGREES NOT TO ASSERT IN ANY PROCEEDING BEFORE ANY TRIBUNAL, BY WAY OF MOTION, AS A DEFENSE OR OTHERWISE, ANY CLAIM THAT IT IS NOT SUBJECT TO THE IN PERSONAM JURISDICTION OF ANY SUCH COURT. IN ADDITION, EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE IN ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY DOCUMENT, AGREEMENT OR TRANSACTION CONTEMPLATED HEREBY BROUGHT IN ANY SUCH COURT, AND HEREBY IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.
- (c) Service of Process. EACH PARTY HERETO IRREVOCABLY AGREES THAT PROCESS PERSONALLY SERVED OR SERVED BY U.S. REGISTERED MAIL AT THE ADDRESSES PROVIDED HEREIN

FOR NOTICES SHALL CONSTITUTE, TO THE EXTENT PERMITTED BY LAW, ADEQUATE SERVICE OF PROCESS IN ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE WARRANTS OR ANY DOCUMENT, AGREEMENT OR TRANSACTION CONTEMPLATED HEREBY, OR ANY ACTION OR PROCEEDING TO EXECUTE OR OTHERWISE ENFORCE ANY JUDGMENT IN RESPECT OF ANY BREACH HEREUNDER, UNDER THE WARRANTS OR UNDER ANY DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY. RECEIPT OF PROCESS SO SERVED SHALL BE CONCLUSIVELY PRESUMED AS EVIDENCED BY A DELIVERY RECEIPT FURNISHED BY THE UNITED STATES POSTAL SERVICE OR ANY COMMERCIAL DELIVERY SERVICE.

[Remainder of page intentionally left blank; next page is signature page]

IN WITNESS WHEREOF, the Parties have executed this Agreement or	the date first above written.
<u>C</u>	COMPANY:
В	ioScrip, Inc.
N	y: ame: itle:

HOLDERS:

ASSF IV AIV B HOLDINGS, L.P.

By: ASSF IV AIV B HOLDINGS GP LLC,

its general partner
By: ASSF IV AIV B, L.P.,

its sole member

By: ASSF MANAGEMENT IV, L.P.,

its general partner
By: ASSF MANAGEMENT IV GP LLC,

its general partner

By:			
Name:			
Title:			
	-		

J.P. Morgan Securities LLC
By:
Name:
Title:

HOLDERS:

HOLDERS:

Goldman Sachs & Co. LLC

By:	
Name:	
Title:	

HOLDERS:

Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc

By: Western Asset Management Company, as its Investment Manager and Agent

By:	
Name:	
Title:	

ATTACHMENT A [FORM OF WARRANT CERTIFICATE]

THE SECURITIES REPRESENTED HEREBY, AND THE SECURITIES ISSUABLE HEREBY, HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS AND NEITHER THE SECURITIES NOR ANY INTEREST THEREIN MAY BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT OR SUCH LAWS OR AN EXEMPTION FROM REGISTRATION UNDER SUCH ACT AND SUCH LAWS WHICH, IN THE OPINION OF COUNSEL FOR THE COMPANY, IS AVAILABLE. THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE TERMS OF A CERTAIN AMENDED AND RESTATED WARRANT AGREEMENT, DATED AS OF MARCH [], 2019, THE PROVISIONS OF WHICH ARE INCORPORATED HEREIN BY REFERENCE. A COPY OF SUCH AGREEMENT IS AVAILABLE FROM THE COMPANY UPON REQUEST.

WARRANT CERTIFICATE

BIOSCRIP, INC.

No. WRT- Warrants

Date: [], 20[]

This Warrant Certificate certifies that , or registered assigns, is the registered holder of Warrants entitling the owner thereof to purchase at any time on or after the date hereof and on or prior to the Expiration Time, [] fully paid and nonassessable shares of Common Stock, \$0.0001 par value per share (the "Common Stock"), of BIOSCRIP, INC., a Delaware corporation (together with its successors and assigns, the "Company"), at a purchase price (subject to adjustment as provided in the Warrant Agreement (as defined below), the "Exercise Price") of \$2.00 per share of Common Stock upon presentation and surrender of this Warrant Certificate to the Company with a duly executed election to purchase and payment of the Exercise Price (including by withholding of shares of Common Stock), all in the manner set forth in the Warrant Agreement (defined below). The Denomination of each Warrant and the Exercise Price are the Denomination and the Exercise Price as of the date hereof, and are subject to adjustment as referred to below.

The Warrants are issued pursuant to an Amended and Restated Warrant Agreement (as it may from time to time be amended or supplemented, the "Warrant Agreement"), dated as of March 14, 2019, among the Company and the Holders named therein, and are subject to all of the terms, provisions and conditions thereof, which Warrant Agreement is hereby incorporated herein by reference and made a part hereof and to which Warrant Agreement reference is hereby made for a full description of the rights, obligations, duties and immunities of the Company and the holders of the Warrant Certificates. Capitalized terms used, but not defined, herein have the respective meanings ascribed to them in the Warrant Agreement. In the event of any conflict between this Warrant Certificate and the Warrant Agreement, the Warrant Agreement shall control and govern.

As provided in the Warrant Agreement, the Exercise Price and the Denomination evidenced by this Warrant Certificate are, upon the happening of certain events, subject to modification and adjustment. Except as otherwise set forth in, and subject to, the Warrant Agreement, the Expiration Time of this Warrant Certificate is as set forth in the Warrant Agreement.

Subject to the limitations set forth in the Warrant Agreement, this Warrant Certificate shall be exercisable, at the election of the holder, at any time on or after the date hereof and on or prior to the Expiration Time either as an entirety or in part from time to time. If this Warrant Certificate shall be exercised in part, the holder shall be entitled to receive, upon surrender hereof, another Warrant Certificate or Warrant Certificates for the Warrants not exercised. This Warrant Certificate, with or without other Warrant Certificates, upon surrender in the manner set forth in the Warrant Agreement and subject to the conditions set forth in the Warrant Agreement, may be transferred or exchanged for another Warrant Certificate or Warrant Certificates of like tenor evidencing Warrants entitling the holder to a like number of shares of Common Stock as the Warrants evidenced by the Warrant Certificate or Warrant Certificates surrendered shall have entitled such holder to purchase.

Except as expressly set forth in the Warrant Agreement, no holder of this Warrant Certificate shall be entitled to vote or receive distributions or be deemed for any purpose the holder of shares of Common Stock or of any other securities of

the Company that may at any time be issued upon the exercise hereof, nor shall anything contained in the Warrant Agreement or herein be construed to confer upon the holder hereof, as such, any of the rights of a holder of a share of Common Stock in the Company or any right to vote upon any matter submitted to holders of shares of Common Stock at any meeting thereof, or to give or withhold consent to any corporate action of the Company (whether upon any recapitalization, issuance of stock, reclassification of Securities, change of par value, consolidation, merger, conveyance, or otherwise), or to receive dividends or subscription rights, or otherwise, until the Warrant or Warrants evidenced by this Warrant Certificate shall have been exercised as provided in the Warrant Agreement.

In the event of any inconsistency between this warrant Certificate and the Warrant agreement, the terms of the Warrant Agreement shall govern.

THIS WARRANT CERTIFICATE SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE COMPANY AND THE HOLDER HEREOF SHALL BE GOVERNED BY, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICT OF LAWS RULES THEREOF TO THE EXTENT THAT ANY SUCH RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION, EXCEPT TO THE EXTENT THAT THE DELAWARE GENERAL CORPORATION LAW SPECIFICALLY AND MANDATORILY APPLIES.

WITNESS the signature of a proper officer of the Company as of the date first above written.

BIOSCRIP, INC.

By: Name: Title:

[FORM OF ASSIGNMENT]

(To be executed by the registered holder if

such holder desires to transfer the Warrant Certificate)

FOR VALUE RECEIVED,

hereby sells, assigns and transfers unto

(Please print name and address of transferee.)

the accompanying Warrant Certificate, together with all right, title and interest therein, and does hereby irrevocably constitute and appoint:

attorney, to transfer the accompanying Warrant Certificate on the books of the Company with full power of substitution. Dated: ,20.

[HOLDER]

By:

NOTICE

The signature to the foregoing Assignment must correspond to the name as written upon the face of the accompanying Warrant Certificate or any prior assignment thereof in every particular, without alteration or enlargement or any change whatsoever.

[FORM OF ELECTION TO PURCHASE]

(To be executed by the registered holder if such holder desires to exercise the Warrant Certificate)

To: BIOSCRIP, INC.

The undersigned hereby irrevocably elects to exercise	Warrants represented by the accompanying Warrant Certificate to purchase
the shares of Common Stock issuable upon the exercise of such V	Warrants, and requests that certificates for such shares be issued in the name of:

(Please print name and address.)

(Please insert social security or other identifying number.)

If such number of Warrants being exercised by the undersigned are not all the Warrants evidenced by the accompanying Warrant Certificate, a new Warrant Certificate for the balance remaining of such Warrants shall be registered in the name of and delivered to:

(Please print name and address.)

(Please insert social security or other identifying number.)

The undersigned is paying the Exercise Price for the Common Stock to be issued on exercise of the foregoing Warrants, unless payment of such Exercise Price has been waived by the Company, by

o certified or bank check by wire transfer, or

o withholding of shares of Common Stock into which the Warrants are being exercised.

Dated: , 20.

[HOLDER]

By:

NOTICE

The signature to the foregoing Election to Purchase must correspond to the name as written upon the face of the accompanying Warrant Certificate or any prior assignment thereof in every particular, without alteration or enlargement or any change whatsoever.

ANNEX 1 Warrants Held on the Effective Date by the Holders

Holder	Total Number of Warrants
ASSF IV AIV B Holdings, L.P.	4,520,354
Amari Investments Pte. Ltd.	1,506,784
J.P. Morgan Securities LLC	753,398
Goldman Sachs & Co. LLC	1,130,081
Western Asset Middle Market Debt Fund Inc.	113,016
Western Asset Middle Market Income Fund Inc	263,683

ANNEX 2 Address for Holders for Notices

ASSF IV AIV B HOLDINGS, L.P.

ASSF IV AIV B Holdings, L.P. c/o Ares Management LLC 2000 Avenue of the Stars, 12th Floor Los Angeles, CA 90067

J.P. MORGAN SECURITIES LLC

J.P. Morgan Securities LLC 4 New York Plaza, 1 5th Floor, Mail Code NY1 -E054 New York, New York 10004 Attention: Jeffrey L. Panzo

Email: Jeffrey.L.Panzo@JPMorgan.com

Phone No.: (212) 499-1435

GOLDMAN SACHS & CO. LLC

Goldman Sachs & Co. LLC 200 West Street, 26th Floor Attn: Paul Burningham and Paige Cataruozolo New York, New York 10282 Email: fice-amssg-mo@gs.com

Phone No.: (917) 343-8393 (Paul Burningham),

(917) 343-3096 (Paige Cataruozolo)

WESTERN ASSET MIDDLE MARKET DEBT FUND INC. WESTERN ASSET MIDDLE MARKET INCOME FUND INC

Western Asset Management Company 385 East Colorado Boulevard Pasadena, California 91101 Attention: Legal Department

ANNEX 3 Address of Company for Notices

BioScrip, Inc. 1600 Broadway, Suite 700 Denver, CO 80202

Attn: Stephen Deitsch, Senior Vice President,

Chief Financial Officer & Treasurer Email: Stephen.Deitsch@bioscrip.com Telecopy Number: (720) 468-4040

With a copy (which shall not constitute notice) to:

Gibson, Dunn & Crutcher, LLP

200 Park Avenue

New York, New York 10036 Attention: Sean P. Griffiths Email: sgriffiths@gibsondunn.com Telecopy Number: (212) 251-5222

Exhibit C

[Amended and Restated Registration Rights Agreement]

See attached.

AMENDMENT NO. 1 TO REGISTRATION RIGHTS AGREEMENT

THIS AMENDMENT NO. 1 TO THE REGISTRATION RIGHTS AGREEMENT (this "Amendment"), effective as of March 14, 2019, is by and between BioScrip, Inc. (the "Company"), and the stockholders of the Company signatory hereto (each a "Stockholder" and collectively, the "Stockholders"). Capitalized but otherwise undefined terms herein have the meanings given to them in the Registration Rights Agreement (as defined below).

WHEREAS, the Company and the Stockholders are party to that certain Registration Rights Agreement, dated June 29, 2017 (the "Registration Rights Agreement"), governing the Company's obligations to register Registrable Securities of the Stockholders; and

WHEREAS, in connection with entering into that certain (i) Amended and Restated Warrant Agreement, dated as of the date hereof, by and between the Company and the Stockholders (the "Warrant Amendment"), (ii) those certain letter agreements, dated as of the date hereof, by and between the Company and each of the Stockholders (the "Letter Agreements"), pursuant to which the Company will issue an aggregate of 1,855,747 shares of Common Stock of the Company upon the closing of the transactions contemplated by the Merger Agreement (as defined below), and (iii) that certain Agreement and Plan of Merger, dated as of the date hereof, by and among the Company, HC Group Holdings II, Inc. and the other parties thereto, the Company and the Stockholders desire to amend the Registration Rights Agreement pursuant to Section 11(c) thereof and upon such terms as set forth herein.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- 1. Amendment to Registration Rights Agreement.
 - (a) Section 1(a) of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:

"Short-Form Registrations. At any time after June 29, 2018 twelve (12) months following the date hereof, with respect to the Purchased Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, or at any time after the twelve month anniversary of the First Merger Effective Time (as defined in the Merger Agreement), with respect to the Amendment Shares, each Holder may request registration under the Securities Act of all or any portion of its Registrable Securities on Form S-3 or any successor form (each, a "Short-Form Registration"), which may, if so requested, be a "shelf" registration under Rule 415 under the Securities Act. A registration shall not count as one of the permitted Short-Form Registrations unless and until a registration statement relating thereto has become effective under the Securities Act. Each request for a Short-Form Registration shall specify the number of Registrable Securities requested to be registered.

(b) Section 1(b) of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:

"Long-Form Registrations. At any time that a Holder is then eligible to request registration under the Securities Act of all or any portion of its Registrable Securities but where Short-Form Registration pursuant to Section 1(a) of this Agreement is not available to be used by the Company in respect of such proposed registration, but in no event earlier than twelve (12) months following the date hereof June 29, 2018, with respect to the Purchased Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, or the twelve month anniversary of the First Merger Effective Time (as defined in the Merger Agreement), with respect to the Amendment Shares, each Holder shall be entitled to request a registration on Form S-1 or any similar form (each, a "Long-Form Registration"). A registration shall not count as one of the permitted Long-Form Registrations unless and until a registration statement relating thereto has become effective under the Securities Act and each requesting Holder is able to register and sell at least thirty percent (30%) of its Registrable Securities thereunder.

(c) The definition of Registrable Securities in Section 10 of the Registration Rights Agreement is hereby amended and restated in its entirety to read as follows:

""Registrable Securities" means the Purchased Shares, the Amendment Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, together with any securities issued or issuable upon any stock

split, dividend or other distribution, recapitalization or similar event with respect to the foregoing. For purposes of this Agreement, a Person shall be deemed to be a holder of Registrable Securities whenever such Person has the right to acquire such Registrable Securities (upon conversion or exercise, in connection with a transfer of securities or otherwise, but disregarding any restrictions or limitations upon the exercise of such right), whether or not such acquisition has actually been effected. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (A) a Registration Statement covering such securities has been declared effective by the Securities and Exchange Commission and such securities have been disposed of pursuant to such effective Registration Statement, (B) such securities are sold under circumstances in which all of the applicable conditions of Rule 144 (or any similar provisions then in force) under the Securities Act are met, (C) the aggregate amount of such securities held by each Holder, together with all Affiliates of each such Holder and persons forming a "group" with each such Holder (within the meaning of Regulation 13D under the Securities Exchange Act of 1934, as amended from time to time, and the rules and regulations promulgated thereunder ("Regulation 1 3D")), consists of beneficial ownership (within the meaning of Regulation 13D) of less than 5.0% of the Common Stock of the Company and such securities are eligible for sale by each such Holder without registration pursuant to Rule 144 (or any similar provisions then in force) under the Securities Act without limitation thereunder on volume or manner of sale, (D) such securities are otherwise transferred and such securities may be resold without limitation or subsequent registration under the Securities Act, (E) such securities shall have ceased to be outstanding, or (F) the stock certificates or evidences of book-entry registration relating to such securities have had all restricti

- (d) The following defined term shall be added to Section 10 of the Registration Rights Agreement:
- ""<u>Amendment Shares</u>" means the shares of Common Stock that the Company issued to the Stockholders pursuant to those certain letter agreements, dated as of March 14, 2019, by and between the Company and each of the Stockholders, in connection with the Amended and Restated Warrant Agreement, dated as of March 14, 2019, by and between the Company and the Stockholders."
- 2. <u>Condition to Effectiveness; Termination</u>. The Company and the Stockholders agree and acknowledge that, notwithstanding anything to the contrary herein, the effectiveness of this Amendment is conditioned upon, and shall not be binding or effective until, the occurrence of the First Merger Effective Time (as such term is defined in the Merger Agreement) in accordance with the terms and conditions of the Merger Agreement. In the event that the Merger Agreement is terminated in accordance with its terms, this Amendment shall automatically terminate and be null and void *ab initio* and of no effect whatsoever, and the Registration Rights Agreement shall remain in full force and effect and shall continue to be binding on and enforceable against the Company and the Purchasers.
- 3. <u>Confirmation</u>. Except as expressly modified by the terms and provisions of this Amendment, all of the terms and provisions of the Registration Rights Agreement are unchanged and continue in full force and effect and all rights, remedies, liabilities and obligations evidenced by the Registration Rights Agreement are hereby acknowledged by the Company and the Stockholders to be valid and in full force and effect.
- 4. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts, by facsimile if necessary, each of which shall be an original, but all of which together shall constitute one instrument.
- 5. <u>Governing Law</u>. This Amendment shall be governed by and construed in accordance with the internal laws of New York applicable to parties residing in New York, without regard applicable principles of conflicts of law. Each party irrevocably consents to the exclusive jurisdiction of any court located within New York County, New York, in connection with any matter based upon or arising out of this Amendment or the matters contemplated hereby and it agrees that process may be served upon it in any manner authorized by the laws of the State of New York for such persons and waives and covenants not to assert or plead any objection which it might otherwise have to such jurisdiction and such process.

[SIGNATURE PAGES FOLLOW]

COMPANY:
BIOSCRIP, INC.
Name: Title:
Title.
STOCKHOLDERS:
ASSF IV AIV B HOLDINGS, L.P.
By: ASSF IV AIV B HOLDINGS GP LLC,
its general partner By: ASSF IV AIV B, L.P.,
its sole member
By: ASSF MANAGEMENT IV, L.P., its general partner
By: ASSF MANAGEMENT IV GP LLC,
its general partner
Name:
Title:
J.P. Morgan Securities LLC
Name: Title:
Goldman Sachs & Co. LLC
N.
Name: Title:

Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc

By: Western Asset Management Company, as its Investment Manager and Agent

Name:

Title:

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

	BIOSCRIP, INC	
_		
FORM OF AMEN	IDED AND RESTATED	WARRANT AGREEMENT

Dated As Of March 14, 2019 Warrants to Purchase shares of Common Stock

BIOSCRIP, INC.

Amended and Restated Warrant Agreement

Warrants For Common Stock

AMENDED AND RESTATED WARRANT AGREEMENT, dated as March 14, 2019, among BioScrip, Inc., a Delaware corporation (together with its successors and assigns, the "<u>Company</u>") and the holders undersigned hereto (collectively and together with each of their respective successors and assigns, the "<u>Holders</u>"). Capitalized terms shall have the meaning specified in <u>Section 5.1</u> hereof.

RECITALS

WHEREAS, on the terms of the Warrant Purchase Agreement (the "Original Warrant Agreement"), dated as June 29, 2017, among the Company and the Original Purchasers party thereto, the Original Purchasers agreed to acquire from the Company, and the Company agreed to issue to the Original Purchasers, Warrants to purchase 4.99% of the Fully Diluted Common Stock (as defined in the Original Warrant Agreement) outstanding on the date of any exercise of the Warrants (less the percentage of the shares of Common Stock previously issued pursuant to the Warrants from time to time as a result of any partial exercise of the Warrants as set forth herein);

WHEREAS, the Company is party to an Agreement and Plan of Merger (the "Merger Agreement"), dated as of March 14, 2019, by and among the Company, HC Group Holdings II, Inc., a Delaware corporation ("Omega"), Beta Sub, LLC, a Delaware limited liability company ("Merger Sub LLC"), and certain other parties signatory thereto, pursuant to which Omega will be merged with and into a newly formed subsidiary of the Company with Omega continuing as the surviving entity (the "Surviving Corporation") (the "First Merger"), immediately following which the Surviving Corporation will be merged with and into Merger Sub LLC, with Merger Sub LLC continuing as the surviving entity (the "Second Merger");

WHEREAS, the undersigned Holders collectively own 100% of the Warrants issued pursuant to the Original Warrant Agreement;

WHEREAS, in connection with, and conditioned upon, the consummation of the First Merger in accordance with the Merger Agreement, the Company and the Holders wish to amend and restate the Original Warrant Agreement so as to fix the number of shares of Common Stock issuable upon exercise of the Warrants at 8,287,317 shares in the aggregate, with each Holder having a Warrant to purchase the number of shares of Common Stock set forth on Annex 1 hereto; and

WHEREAS, the Company and the Holders wish to enter into this Agreement to amend and restate the terms of the Warrants.

AGREEMENT

NOW, THEREFORE, in consideration of the premises and the mutual agreements set forth herein, the parties to this Agreement hereby agree to amend and restate the Original Warrant Agreement as follows:

1. FORM, EXECUTION AND TRANSFER OF WARRANT CERTIFICATES.

1.1 Form of Warrant Certificates.

The Warrant Certificates shall be in the form set forth in <u>Attachment A</u> hereto. The Warrant Certificates may have such letters, numbers or other marks of identification or designation as may be required to comply with any law or with any rule or regulation of any governmental authority, stock exchange or self-regulatory organization made pursuant thereto

("<u>Law</u>"). Each Warrant Certificate shall be initially dated as of June 29, 2017 and subsequently dated the date of issuance thereof by the Company upon transfer or exchange. Each Warrant Certificate shall represent the right to purchase the number of shares of Common Stock set forth in such Warrant Certificate at a price per share of Common Stock equal to the Exercise Price applicable to that Warrant Certificate; *provided*, that the number of shares of Common Stock issuable upon exercise of the Warrants and the Exercise Price thereof shall be subject to adjustment as provided herein. The aggregate amount of shares of Common Stock issuable upon exercise of all Warrants as of the Second Merger Effective Time shall be 8,287,317, subject to adjustment as of the Second Merger Effective Time as provided herein, with each Holder having a Warrant to purchase the number of shares of Common Stock set forth on <u>Annex</u> 1 hereto.

1.2 Execution of Warrant Certificates; Registration Books.

- (a) Execution of Warrant Certificates. The Warrant Certificates shall be executed on behalf of the Company by an officer of the Company authorized by the Board of Directors. In case the officer of the Company who shall have signed any Warrant Certificate shall cease to be such an officer of the Company before issuance and delivery by the Company of such Warrant Certificate, such Warrant Certificate nevertheless may be issued and delivered with the same force and effect as though the individual who signed such Warrant Certificate had not ceased to be such an officer of the Company, and any Warrant Certificate may be signed on behalf of the Company by any individual who, at the actual date of the execution of such Warrant Certificate, shall be a proper officer of the Company to sign such Warrant Certificate, although at the date of the execution of this Agreement any such individual was not such an officer.
- (b) **Registration Books.** The Company will keep or cause to be kept at its office, maintained at the address of the Company referenced in Section 6.6, at the Company's transfer agent, or at such other office of the Company of which the Company shall have given notice to each holder of Warrant Certificates, books for registration and transfer of the Warrant Certificates issued hereunder. Such books shall show the names and addresses of the respective holders of the Warrant Certificates, the registration number and date of each of the Warrant Certificates and the Denomination thereof.

1.3 Transfer, Split Up, Combination and Exchange of Warrant Certificates; Lost or Stolen Warrant Certificates.

(a) Transfer, Split Up, etc.

- (i) **Transfer**. Subject to compliance with the Securities Act, any applicable state securities laws and the Company's organizational documents, any Warrant Certificate (or portion thereof), with or without other Warrant Certificates, may be transferred to any Person for a Warrant Certificate or Warrant Certificates in an aggregate like Denomination as the Warrant Certificate or Warrant Certificates (or portions thereof) surrendered then entitled such registered holder to purchase. Any registered holder desiring to transfer any Warrant Certificate shall make such request in writing delivered to the Company, which request shall include the identity of the Transferee and the aggregate Denomination of Warrants to be transferred, and shall surrender the Warrant Certificate or Warrant Certificates (or portions thereof) to be transferred at the office of the Company referenced in Section 6.6, whereupon the Company shall deliver promptly to such Transferee a Warrant Certificate or Warrant Certificates, as the case may be, as so requested, which Warrant Certificate or Warrant Certificates shall evidence, collectively, the same aggregate Denomination of Warrants as the Warrant Certificate or Warrant Certificates (or portions thereof) so surrendered for transfer and shall issue a new Warrant Certificate to the transferor representing the Warrants retained by the Transferor if such transfer involved less than the entire Denomination of Warrants held by such Transferor.
- (ii) <u>Split Up, Combination, Exchange, etc.</u> Any Warrant Certificate, with or without other Warrant Certificates, may be split up, combined or exchanged for another Warrant Certificate or Warrant Certificates, in an aggregate like Denomination as the Warrant Certificate or Warrant Certificates surrendered then entitle such registered holder to purchase. Any registered holder desiring to split up, combine or exchange any Warrant Certificate shall make such request in writing delivered to the Company, and shall surrender the Warrant Certificate or Warrant Certificates to be split up, combined or exchanged at the office of the Company referenced in <u>Section 6.6</u>, whereupon the Company shall deliver promptly to such registered holder a Warrant Certificate or Warrant Certificates, as the case may be, as so requested, which Warrant Certificate or Warrant Certificates shall evidence, collectively, the same aggregate Denomination as the Warrant Certificate or Warrant Certificates so surrendered for split-up, combination or exchange.
- (b) **Loss, Theft, etc.** Upon receipt by the Company of evidence reasonably satisfactory to it of the ownership of, and the loss, theft, destruction or mutilation of, any Warrant Certificate, and:
- (i) in the case of loss, theft or destruction, an affidavit of loss, together with a customary and reasonable indemnity reasonably satisfactory to the Company; or
 - (ii) in the case of mutilation, upon surrender and cancellation thereof;

the Company at its own expense will execute and deliver, in lieu thereof, a new Warrant Certificate, dated the date of such lost, stolen, destroyed or mutilated Warrant Certificate and of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant Certificate and evidencing the same Denomination as the Warrant Certificate so lost, stolen, destroyed or mutilated.

1.4 Subsequent Issuance of Warrant Certificates.

Subsequent to the original issuance, no Warrant Certificates shall be issued except:

- (a) Warrant Certificates issued upon any transfer, combination, split up or exchange of Warrants pursuant to Section 1.3(a);
- (b) Warrant Certificates issued in replacement of mutilated, destroyed, lost or stolen Warrant Certificates pursuant to Section 1.3(b);
- (c) Warrant Certificates issued pursuant to <u>Section 2.3</u> upon the partial exercise of any Warrant Certificate to evidence the unexercised portion of such Warrant Certificate; and
 - (d) Warrant Certificates to reflect any adjustments pursuant to Section 4.

1.5 Effect of Issuance in Registered Form.

Every holder of a Warrant Certificate by accepting the same consents and agrees with the Company and with every other holder of a Warrant Certificate that:

- (a) the Warrant Certificates, to the extent then currently transferable, are transferable only on the registry books of the Company if and when surrendered at the office of the Company referenced in <u>Section 6.5</u>, duly endorsed or accompanied by an instrument of transfer (in the form attached thereto) and payment of any applicable transfer, stamp or issue tax (a "<u>Tax</u>"); and
- (b) the Company may deem and treat the Person in whose name each Warrant Certificate is registered as the absolute owner thereof and of the Warrants evidenced thereby (notwithstanding any notations of ownership or writing on the Warrant Certificates made by anyone other than the Company) for all purposes whatsoever, and the Company shall not be affected by any notice to the contrary.

2. EXERCISE OF WARRANTS; PAYMENT OF EXERCISE PRICE.

2.1 Exercise of Warrants.

- (a) **Manner of Exercise**. At any time and from time to time prior to the Expiration Time, the holder of any Warrant Certificate may exercise the Warrants evidenced thereby, in whole or in any part, by surrender to the Company, at its office referenced in Section 6.6, of such Warrant Certificate, together with a duly executed election to purchase (a form of which is attached to each Warrant Certificate) and payment of the applicable Exercise Price for each share of Common Stock with respect to which the Warrants are then being exercised and an amount equal to any applicable Tax (if not payable by the Company as provided in Section 3.3). Such Exercise Price shall be payable in cash or by withholding of shares of Common Stock into which the Warrants are being exercised pursuant to Section 2.1(b).
- (b) **Payment in Cash or by Withholding of Common Stock**. Upon exercise of any Warrants, the holder of a Warrant Certificate may pay the Exercise Price by either (i) certified or official bank check payable to the order of the Company or by wire transfer of immediately available funds to the account of the Company designated in writing by the Company or (ii) instructing the Company to withhold that number of shares of Common Stock (or fraction thereof) then issuable upon such exercise with an aggregate Market Price as of such exercise date equal to the aggregate Exercise Price. The Company acknowledges that the provisions of this clause are intended, in part, to ensure that a full or partial exchange of a Warrant Certificate will qualify as a conversion, within the meaning of paragraph (d)(3)(ii) of Rule 144 under the Securities Act.
- (c) **Fractional shares of Common Stock**. The Company may, in accordance with <u>Section 2.6</u>, pay the exercising holder cash in lieu of issuing a fractional share in connection with an exercise of Warrants; provided that, if it does not issue a fractional share in such circumstances, it will make such cash payment.

2.2 Issuance of Common Stock.

Upon timely receipt of a Warrant Certificate, accompanied by the form of election to purchase duly executed, and payment of the Exercise Price for each of the shares of the Common Stock to be purchased and by an amount equal to any applicable Tax (if not payable by the Company as provided in Section 3.3), the Company shall thereupon promptly cause

certificates representing the number of whole shares of Common Stock then being purchased to be delivered to or upon the order of the registered holder of such Warrant Certificate, registered in such name or names as may be designated by such holder, and, promptly after such receipt deliver the cash, if any, to be paid in lieu of fractional shares pursuant to Section 2.6 to or upon the order of the registered holder of such Warrant Certificate.

2.3 Unexercised Warrants.

In the event that the registered holder of any Warrant Certificate shall exercise less than all the Warrants evidenced thereby, a new Warrant Certificate evidencing a number of unexercised Warrants shall be issued by the Company to the registered holder of such Warrant Certificate or to its duly authorized assigns.

2.4 Cancellation and Destruction of Warrant Certificates.

All Warrant Certificates surrendered to the Company for the purpose of exercise, payment of the Exercise Price, exchange, substitution or transfer shall be cancelled by it, and no Warrant Certificates shall be issued in lieu thereof except as expressly permitted by any of the provisions of this Agreement. The Company shall cancel and retire any other Warrant Certificates purchased or acquired by the Company.

2.5 Expiration.

All Warrants that have not been exercised or purchased in accordance with the provisions of this Agreement shall expire and all rights of holders of such Warrants hereunder shall terminate and cease at the Expiration Time.

2.6 Fractional shares of Common Stock.

The Company shall not be required to issue fractional shares of Common Stock upon the exercise of any Warrant. If fractional shares are not issued upon the exercise of any Warrant, there shall be paid to the holder thereof, in lieu of any fractional share of Common Stock resulting therefrom, an amount of cash equal to the product of:

- (a) the fractional amount of such share of Common Stock; and
- (b) the Market Price, as determined on the trading day immediately prior to the date of exercise of such Warrant.

2.7 RESERVED

3. AGREEMENTS OF THE COMPANY.

3.1 Reservation of Common Stock.

The Company covenants and agrees that it will at all times cause to be reserved and kept available out of its authorized and unissued shares of treasury shares of Common Stock such number of shares of Common Stock as will be sufficient to permit the exercise in full of all Warrants issued hereunder into Common Stock.

3.2 Common Stock to be Duly Authorized and Issued, Fully Paid and Nonassessable, etc; Compliance with Law

The Company covenants and agrees that it will take all such action as may be necessary to ensure that all shares of Common Stock delivered upon the exercise of any Warrant and the payment of the Exercise Price pursuant to Section 2.1 (in each case, at the time of delivery of the certificates representing such shares of Common Stock) shall (a) be duly and validly authorized and issued and fully paid and nonassessable, free of any preemptive rights in favor of any Person in respect of such issuance and free of any security interest, pledge, mortgage, lien, charge or other encumbrance created by, or arising out of actions of, the Company, in each case other than (i) resulting from the actions and circumstances of the holder of such shares of Common Stock, (ii) to the extent imposed by applicable laws or (iii) in the Company's organizational documents and (b) be issued without violation of any applicable Law.

3.3 Taxes.

The Company covenants and agrees that it will pay when due and payable any and all Taxes and charges that may be payable in respect of the initial issuance or delivery of:

- (a) each Warrant Certificate;
- (b) each Warrant Certificate issued in exchange for any other Warrant Certificate pursuant to Section 1.3, Section 2.3 or Section 4; and
- (c) each share of Common Stock issued upon the exercise of any Warrant.

The Company shall not, however, be required to:

- (i) pay any Tax that may be payable in respect of the transfer or delivery of Warrant Certificates in a name other than that of the registered holder of the Warrant Certificate surrendered for exercise, conversion, transfer or exchange (any such Tax being payable by the holder of such certificate at the time of surrender); or
- (ii) issue or deliver any such certificates referred to in the foregoing clause (i) until any such Tax referred to in the foregoing clause (i) shall have been paid.

3.4 Common Stock Record Date.

Each Person in whose name any certificate for shares of Common Stock is issued upon the exercise of Warrants shall for all purposes be deemed to have become the holder of record of the Common Stock represented thereby on, and such certificates (if any) shall be dated, the date upon which the Warrant Certificate evidencing such Warrants was duly surrendered with an election to purchase attached thereto duly executed and payment of the aggregate Exercise Price (and any applicable Taxes, if payable by such Person) was made.

3.5 Rights in Respect of Common Stock.

Prior to the exercise of the Warrants evidenced thereby, the holder of a Warrant Certificate shall not be entitled to any rights of a stockholder of the Company with respect to the Common Stock into which the Warrants shall be exercisable, including, without limitation, the right to vote in respect of any matter upon which the holders of Common Stock may vote, the right to receive any distributions of cash or property and, except as expressly set forth herein, the right to receive any notice of any proceedings of the Company. Prior to the exercise of the Warrants evidenced thereby, the holders of the Warrant Certificates shall not have as such any obligation in respect of any assessment or any other obligation or liability as a stockholder of the Company, whether such obligations or liabilities are asserted by the Company or by creditors of the Company.

3.6 Noncircumvention.

The Company hereby covenants and agrees that the Company will not, by amendment of its charter, bylaws or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issue or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant Agreement, and will at all times in good faith carry out all the provisions of this Warrant Agreement.

4. ANTI-DILUTION ADJUSTMENTS.

4.1 Adjustments.

The number of shares of Common Stock purchasable upon the exercise of each Warrant, and the Exercise Price, shall be subject to adjustment as set forth in this Section 4; provided that no adjustment of the Exercise Price shall result in an increase in the Exercise Price.

4.2 Stock Splits, Subdivisions, Reclassifications or Combinations.

If the Company shall (i) declare and pay a dividend or make a distribution on its Common Stock in shares of Common Stock, (ii) subdivide or reclassify the outstanding shares of Common Stock into a greater number of shares, or (iii) combine or reclassify the outstanding shares of Common Stock into a smaller number of shares, the Denomination of any Warrant Certificate and the Common Stock issuable upon exercise of any Warrants at the time of the record date for such dividend or effective date of such split, reverse split, subdivision, combination or reclassification shall be proportionately adjusted so that the holder of such Warrants after such date shall be entitled to purchase the number of shares of Common Stock which such holder would have owned or been entitled to receive in respect of the shares of Common Stock subject to such Warrants after such date had such Warrants been exercised immediately prior to such date. In such event the Exercise Price in effect at the time of the effective date of such split, reverse split, subdivision, combination or reclassification shall be adjusted to the number obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of such Warrants immediately before such adjustment and (2) the Exercise Price in effect immediately prior to the record or effective date, as the case may be, for the dividend, distribution, split, reverse split, subdivision, combination or reclassification giving rise to this adjustment by (y) the new number of shares of Common Stock issuable upon exercise of such Warrants determined pursuant to the immediately preceding sentence; provided that the Exercise Price shall not be adjusted to be less than the par value of the Common Stock; provided, further, that the adjustment to the Exercise Price in this Section 4.2 may result in an increase in the Exercise Price, notwithstanding Section 4.1.

4.3 Price Based Anti-Dilution

- (a) Without duplication of the adjustments set forth in Sections 4.2 or 4.4, if the Company shall issue after the Second Merger Effective Time or sell any shares of Common Stock (as actually issued or, pursuant to Section 4.3(b), deemed to be issued) for a consideration per share less than either the Market Price per share or the Exercise Price immediately prior to such issuance or sale, or if earlier, upon the execution of the definitive documentation with respect to such issuance or sale (the "Effective Time"), then immediately upon the Effective Time the Denomination of any Warrant Certificate and the number of shares of Common Stock issuable upon exercise of any Warrants at the time of the effective date shall be increased by multiplying such number of shares of Common Stock by a fraction, (i) the numerator of which shall be the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock so issued or sold (or deemed to be issued or sold pursuant to Section 4.3(b)), and (ii) the denominator of which shall be the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock actually outstanding immediately prior to the Effective Time plus the number of shares of Common Stock which the aggregate consideration received by the Company for the total number of shares of Common Stock so issued or sold (or deemed to be issued or sold pursuant to Section 4.3(b)) would purchase if such shares were sold at the greater of the Market Price or the Exercise Price, and the Exercise Price shall be adjusted by multiplying the Exercise Price in effect immediately prior to such issuance or sale (or deemed issuance or sale) by the inverse of such fraction. For the purposes of this Section 4.3(a), none of the following issuances shall be considered the issuance or sale of Common Stock:
- (i) the issuance of Common Stock (a) upon the conversion or exercise of any then-outstanding Common Stock Equivalents; (b) to Persons in connection with joint ventures, strategic alliances or other commercial relationships with such Person relating to the operation of the Company's business and not for the primary purpose of raising equity capital, in connection with Business Combinations, or transactions in which the Company, directly or indirectly, acquires another business or the assets thereof or (c) in an offering for cash for account of the Company that is underwritten on a firm commitment basis and is registered with the Securities and Exchange Commission;
- (ii) the issuance of any Common Stock or Common Stock Equivalents for which the adjustment provided in <u>Section 4.2</u> applies; or
- (iii) the issuance of shares of Common Stock or Common Stock Equivalents to employees or directors of either the Company or any Company Subsidiary that is approved by the Board of Directors.
 - (b) For the purposes of Section 4.3(a), the following subparagraphs (i) to (iii), inclusive, shall also be applicable:
- (i) If the Company shall grant any rights to subscribe for, or any rights or options to purchase, Common Stock Equivalents, whether or not such rights or options or the right to convert or exchange any such Common Stock Equivalents are immediately exercisable, and the price per share for which Common Stock is issuable upon the exercise of such rights or options or upon conversion or exchange of such Common Stock Equivalents (determined by dividing (A) the total amount, if any, received or receivable by the Company as consideration for the granting of such rights or options, plus the minimum aggregate amount of additional consideration payable to the Company upon the exercise of such rights or options, plus, in the case of any such rights or options which relate to such Common Stock Equivalents, the minimum aggregate amount of additional consideration, if any, payable upon the issue or sale of such Common Stock Equivalents and upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the exercise of such rights or options or upon the conversion or exchange of all such Common Stock Equivalents issuable upon the exercise of such rights or options) shall be less than the Exercise Price or the Market Price per share of Common Stock immediately prior to the earlier of (x) the time of the granting of such rights or options and (y) the time of execution of definitive documentation with respect to such grant (such earlier time of (x) and (y) to be the "Effective Time" for purposes of Section 4.3(a)), then the total maximum number of shares of Common Stock issuable upon the exercise of such rights or options or upon conversion or exchange of the total maximum amount of such Common Stock Equivalents issuable upon the exercise of such rights or options shall (as of the date of granting of such rights or options or the date of execution of definitive documents, if earlier) be deemed to be outstanding and to have been issued for such price per share; provided that no further adjustment of the conversion price pursuant to this Section 4.3(a) or Section 4.3(b) (i) shall be made (x) upon the actual issuance or sale of such Common Stock Equivalents upon the exercise of any rights to subscribe for, or any rights or options to purchase, such Common Stock Equivalents or (y) upon the actual issuance or sale of such Common Stock upon the exercise of any such Common Stock Equivalents, including without limitation, in each case of clauses (x) and (y) with respect to shares of Common Stock Equivalents or Common Stock issued or issuable as a result of the effect of antidilution adjustments under any such security.

(ii) If the Company shall issue or sell any Common Stock Equivalents, whether or not the rights to exchange or convert thereunder are immediately exercisable, and the price per share for which Common Stock is issuable upon such conversion or exchange (determined by dividing (A) the total amount received or receivable by the Company as consideration for the issue or sale of such Common Stock Equivalents, plus the minimum aggregate amount of additional

consideration, if any, payable to the Company upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the conversion or exchange of all such Common Stock Equivalents) shall be less than the Exercise Price or the Market Price per share of Common Stock immediately prior to the Effective Time for such issuance or sale, then the total maximum number of shares of Common Stock issuable upon conversion or exchange of such Common Stock Equivalents shall (as of the Effective Time) be deemed to be outstanding and to have been issued for such price per share, provided that no further adjustment of the conversion price pursuant to this Section 4.3(a) or Section 4.3(b)(ii) shall be made upon the actual issuance or sale of such Common Stock upon the exercise of any such Common Stock Equivalents, including without limitation, in each case with respect to shares of Common Stock issued or issuable as a result of the effect of antidilution adjustments under any such security.

(iii) In case at any time any shares of Common Stock or Common Stock Equivalents or any rights or options to purchase any such Common Stock or Common Stock Equivalents shall be issued or sold for cash to a non-Affiliate of the Company, the consideration received therefor shall be deemed to be the amount received by the Company therefor. In case any shares of Common Stock or Common Stock Equivalents or any rights or options to purchase any such Common Stock or Common Stock Equivalents shall be issued or sold to an Affiliate of the Company or for a consideration other than or in addition to cash, the amount of the consideration received by the Company shall be deemed to be the Fair Market Value of solely such consideration received by the Company in respect of such purchase of Common Stock or Common Stock Equivalents.

4.4 Other Distributions. In case the Company shall fix a record date for the making of a dividend or distribution to holders of shares of its Common Stock of securities, evidences of indebtedness, assets, cash, rights or warrants (excluding dividends of its Common Stock and other dividends or distributions referred to in Section 4.2), in each such case, the Exercise Price in effect prior to such record date shall be reduced immediately thereafter to the price determined by multiplying the Exercise Price in effect immediately prior to the reduction by the quotient of (x) the Market Price of the Common Stock on the last trading day preceding the first date on which the Common Stock trades on the Exchange on which the Common Stock is listed or admitted to trading without the right to receive such distribution, minus the amount of cash and/or the Fair Market Value of the securities, evidences of indebtedness, assets, rights or warrants to be so distributed in respect of one share of Common Stock (such amount and/or Fair Market Value, the "Per Share Fair Market Value") divided by (y) such Market Price on such date specified in clause (x); such adjustment shall be made successively whenever such a record date is fixed. In divided by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of any Warrants shall be increased to the number obtained by dividing (x) the product of (1) the number of shares of Common Stock issuable upon the exercise of such Warrants before such adjustment, and (2) the Exercise Price in effect immediately prior to the distribution giving rise to this adjustment by (y) the new Exercise Price determined in accordance with the immediately preceding sentence. In the event that such distribution is not so made, the Exercise Price and the number of shares of Common Stock issuable upon exercise of such Warrants then in effect shall be readjusted, effective as of the date when the Board of Directors determines not to distribute such shares, evidences of indebte

4.5 Business Combinations.

In case of any Business Combination or reclassification of Common Stock (other than a reclassification of Common Stock referred to in Section 4.2), a holder's right to receive shares of Common Stock upon exercise of any Warrants shall be converted into the right to exercise such Warrant to acquire the number of shares of stock or other securities or property (including cash) which the Common Stock issuable (at the time of such Business Combination or reclassification) upon exercise of such Warrants immediately prior to such Business Combination or reclassification would have been entitled to receive upon consummation of such Business Combination or reclassification; and in any such case, if necessary, the provisions set forth herein with respect to the rights and interests thereafter of such holder shall be appropriately adjusted so as to be applicable, as nearly as may reasonably be, to such holder's right to exercise such Warrants in exchange for any shares of stock or other securities or property pursuant to this Section 4.5. In determining the kind and amount of stock, securities or the property receivable upon exercise of any Warrants following the consummation of such Business Combination, if the holders of Common Stock have the right to elect the kind or amount of consideration receivable upon consummation of such Business Combination. The Company shall not enter into or be party to any Business Combination unless the successor of the Company (if any), assumes in writing the obligation to deliver to each holder of Warrants hereunder in exchange for such Warrants a security of such successor evidenced by a written instrument substantially similar in form and substance to this Agreement.

4.6 Expiration of Rights or Options.

Upon the expiration of any rights or options to subscribe for, purchase or convert or exchange Common Stock or

Common Stock Equivalents in respect of the issuance, sale or grant of which adjustment was made pursuant to <u>Section 4.3</u>, without the exercise thereof, the Exercise Price and the number of shares of Common Stock purchasable upon the exercise of the Warrants shall, upon such expiration, be readjusted (and such readjustment may result in an increase in the Exercise Price, notwithstanding <u>Section 4.1</u>, not to exceed the Exercise Price prior to such original adjustment) and shall thereafter be such Exercise Price and such number of shares of Common Stock as would have been in effect had such Exercise Price and such number of shares of Common Stock not been originally adjusted (or had the original adjustment not been required, as the case may be), as if:

- (a) the only shares of Common Stock so issued were the shares of Common Stock, if any, actually issued or sold upon the exercise of such rights or options; and
- (b) such shares of Common Stock, if any, were issued or sold for the consideration actually received by the Company upon such exercise plus the aggregate consideration, if any, actually received by the Company for the issuance, sale or grant of all of such rights or options, whether or not exercised; provided that no such readjustment shall have the effect of increasing the Exercise Price by an amount in excess of the amount of the reduction initially made in respect of the issuance, sale, or grant of such rights or options.

4.7 Rounding of Calculations; Minimum Adjustments.

All calculations under this Section 4 shall be made to the nearest one-tenth (1/10th) of a cent or to the nearest one-hundredth (1/100th) of a share, as the case may be. Any provision of this Section 4 to the contrary notwithstanding, no adjustment in the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable shall be made if the amount of such adjustment would be less than \$0.01 or one-tenth (1/10th) of a share of Common Stock, but any such amount shall be carried forward and an adjustment with respect thereto shall be made at the earlier of (i) the time of any exercise of Warrants and (ii) the time of and together with any subsequent adjustment which, together with such amount and any other amount or amounts so carried forward, shall aggregate \$0.01 or 1/10th of a share of Common Stock, or more.

4.8 Timing of Issuance of Additional Common Stock Upon Certain Adjustments.

In any case in which the provisions of this <u>Section 4</u> shall require that an adjustment shall become effective immediately after a record date for an event, the Company may defer until the occurrence of such event (i) issuing to the holder of any Warrants exercised after such record date and before the occurrence of such event the additional shares of Common Stock issuable upon such exercise by reason of the adjustment required by such event over and above the shares of Common Stock issuable upon such exercise before giving effect to such adjustment and (ii) paying to such holder any amount of cash in lieu of a fractional share of Common Stock; *provided*, however, that the Company upon written request shall deliver to such holder a due bill or other appropriate instrument evidencing such holder's right to receive such additional shares, and such cash, upon the occurrence of the event requiring such adjustment.

4.9 Miscellaneous.

- (a) **Statement Regarding Adjustments**. Whenever after the Second Merger Effective Time the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable shall be adjusted as provided in <u>Section 4</u>, the Company shall promptly file at the principal office of the Company referenced in <u>Section 6.6</u> a statement showing in reasonable detail the facts requiring such adjustment and the Exercise Price that shall be in effect and the number of shares of Common Stock into which such Warrants shall be exercisable after such adjustment, and the Company shall also cause a copy of such statement to be sent by mail, first class postage prepaid, or email to each holder of Warrants at the address or email address appearing in the Company's records.
- (b) **Notice of Adjustment Event**. In the event that the Company shall propose after the Second Merger Effective Time to take any action of the type described in this Section 4 (but only if the action of the type described in this Section 4 would result in an adjustment in the Exercise Price or the number of shares of Common Stock into which Warrants are exercisable or a change in the type of securities or property to be delivered upon exercise of Warrants), the Company shall give notice to the holders of Warrants, in the manner set forth in Section 4.9(a), which notice shall specify the record date, if any, with respect to any such action and the approximate date on which such action is to take place. Such notice shall also set forth the facts with respect thereto as shall be reasonably necessary to indicate the effect on the Exercise Price and the number, kind or class of shares or other securities or property which shall be deliverable upon exercise of any Warrants. In the case of any action which would require the fixing of a record date, such notice shall be given at least 10 days prior to the date so fixed, and in case of all other action, such notice shall be given at least 15 days prior to the taking of such proposed action. Without limiting the foregoing, to the extent notice of any of the foregoing actions or events is given to the holders of the Common Stock, such notice shall be provided to the holders of the Warrants on or before such notice to the holders of Common Stock.
 - (c) Proceedings Prior to Any Action Requiring Adjustment. As a condition precedent to the taking of

any action which would require an adjustment pursuant to this Section 4, the Company shall take any action which may be necessary, including obtaining regulatory, New York Stock Exchange, NASDAQ Stock Market or other applicable national securities exchange (an "Exchange") or stockholder approvals or exemptions, in order that the Company may thereafter validly and legally issue as fully paid and nonassessable all shares of Common Stock that the holders are entitled to receive upon exercise of this any Warrants pursuant to this Section 4.

- (d) **Adjustment Rules**. Any adjustments pursuant to this Section 4 shall be made successively whenever an event referred to herein shall occur. If more than one subsection of this Section 4 is applicable to a single event, the subsection shall be applied that produces the largest adjustment and no single event shall cause an adjustment under more than one subsection of this Section 4 so as to result in duplication. If an adjustment in Exercise Price made hereunder would reduce the Exercise Price to an amount below par value of the Common Stock, then such adjustment in Exercise Price made hereunder shall reduce the Exercise Price to the par value of the Common Stock.
- (e) **Merger Agreement**. Notwithstanding anything contained herein to the contrary, the consummation of the transactions contemplated by the Merger Agreement (including the issuance of any shares of Common Stock by the Company contemplated therein) will not result in, or cause any adjustment in, the Exercise Price or the number of shares of Common Stock into which any Warrants are exercisable pursuant to this <u>Section 4</u>.

5. INTERPRETATION OF THIS AGREEMENT.

5.1 Certain Defined Terms.

For the purpose of this Agreement, the following terms shall have the meanings set forth below or set forth in the Section hereof following such term:

"Affiliate" means, with respect to any Person, (a) a director, officer or shareholder of such Person, (b) in the case of a natural person, a spouse, parent, sibling or descendant of such Person (or spouse, parent, sibling or descendant of any director or executive officer of such Person) and (c) any other Person that, directly or indirectly through one or more intermediaries, Controls, or is Controlled by, or is under common Control with, such Person, at such time; provided, however, that none of the Holders shall be deemed to be an "Affiliate" of the Company and no Person holding any one or more of the Warrants shall be deemed to be an "Affiliate" of the Company solely by virtue of the ownership thereof.

- "Agreement" means this Amended and Restated Warrant Agreement as it may from time to time be amended, restated, modified or supplemented.
- "Board of Directors" means the board of directors of the Company, including any duly authorized committee thereof.
- "Beneficially Own" has the meaning has the meaning given to it in Section 13D of the Exchange Act and the rules promulgated thereunder.
- "<u>Business Combination</u>" means any consolidation of the Company with, or merger of the Company with or into, another Person (other than a merger in which (a) the Company is the surviving corporation, (b) that does not result in any reclassification or change of shares of Common Stock outstanding immediately prior to such merger and (c) the holders of Common Stock are not entitled to receive any consideration therefrom), or any sale or conveyance to another Person of the assets of the Company substantially as an entirety.
- "business day" means any day that is not a Saturday, Sunday or other day on which banking institutions in New York, New York are authorized or required by Law or executive order to close.
- "Capital Stock" means (A) with respect to any Person that is a corporation or company, any and all shares, interests, participations or other equivalents (however designated) of capital or capital stock of such Person and (B) with respect to any Person that is not a corporation or company, any and all partnership or other equity interests of such Person.
 - "Charter" means, with respect to any Person, its certificate or articles of incorporation, articles of association, or similar organizational document.
 - "Common Stock" means the Company's common stock, par value \$.0001 per share.
- "Common Stock Equivalents" means all options, derivatives, rights, warrants and convertible, exercisable or exchangeable securities or instruments (including, without limitation, awards or grants of such securities or instruments to directors, officers, employees or consultants of the Company or to other persons).
 - "Company" has the meaning set forth in the introductory paragraph hereof.
 - "Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause

the direction of the management and policies of such Person, whether through the ownership of voting Securities, by contract or otherwise.

"<u>Denomination</u>" means, in the case of any Warrant Certificate, the number of shares of Common Stock issuable upon exercise of the Warrants represented by such Warrant Certificate.

"Effective Date" shall mean the date on which the Second Merger is consummated in accordance with the terms of the Merger Agreement.

"Effective Time" has the meaning set forth in Section 4.3(a).

"Exchange" has the set forth in Section 4.9(c).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor statute, and the rules and regulations promulgated thereunder.

"Exercise Price" means, prior to any adjustment pursuant to Section 4 of this Agreement, the Initial Exercise Price; and thereafter, the Initial Exercise Price as successively adjusted and readjusted from time to time in accordance with the provisions of Section 4.

"Expiration Time" means 5:00 p.m., Eastern time, on the tenth (10th) year anniversary of June 29, 2017.

"Fair Market Value" means, with respect to any security or other property, the fair market value of such security or other property as determined by the Board of Directors, acting in good faith and transmitted to the holders of Warrants by written notice. The Required Warrantholders may object in writing to the Board of Director's calculation of Fair Market Value within 10 days of receipt of such written notice thereof. If the Required Warrantholders and the Board of Directors are unable to agree on Fair Market Value during the 10-day period following the delivery of the Required Warrantholders' objection, then the Board of Directors shall select and approve an appraiser of national recognition experienced in the business of evaluating or appraising the market value of securities (which appraiser shall be subject to approval by the Required Warrantholders, which approval shall not be unreasonably withheld, conditioned or delayed). The Fair Market Value established by such appraiser shall be conclusive and binding on the parties. The fees and expenses for such appraiser shall be borne 50% by the Company, on the one hand, and 50% by the holders of Warrants, on the other hand.

"Initial Exercise Price" means, with respect to each Warrant Certificate issued to a Holder on the Issue Date, \$2.00.

"Issue Date" means June 29, 2017.

"Law" has the set forth in Section 1.1.

"Market Price" means, with respect to a particular security, on any given day, the volume weighted average trading price for the period of ten consecutive trading days ending on the trading day prior to such given day as reported by Bloomberg Financial Markets, or any successor thereto, through its "Volume Weight Average Price" function, on the Exchange on which the applicable securities are listed or admitted to trading. "Market Price" shall be determined without reference to after hours or extended hours trading. If such security is not listed and traded in a manner that the quotations referred to above are available for the period required hereunder, the Market Price per share of Common Stock shall be deemed to be the fair market value per share of such security as determined in good faith by the Board of Directors in reliance on an opinion of a nationally recognized independent investment banking corporation retained by the Company for this purpose (which opinion shall be made available to the holders of Warrants); provided that the Required Warrantholders may object in writing to the Board of Director's calculation of fair market value within 10 days of receipt of written notice thereof. If the Required Warrantholders and the Board of Directors are unable to agree on fair market value during the 10-day period following the delivery of the Required Warrantholders' objection, then the Board of Directors shall select and approve an appraiser experienced in the business of evaluating or appraising the market value of securities (which appraiser shall be subject to approval by the Required Warrantholders, which approval shall not be unreasonably withheld, conditioned or delayed). The Market Price established by such appraiser shall be conclusive and binding on the parties. In the event the Market Price established by such appraiser is greater than the Market Price previously determined by the Board of Directors, the fees and expenses for such appraiser shall be borne by the Company. In the event the Market Price established by such appraiser is less than or equal to the Market Price previously determined by the Board of Directors, the fees and expenses for such appraiser shall be borne by the holders of Warrants. For the purposes of determining the Market Price of the Common Stock on the "trading day" preceding, on or following the occurrence of an event, (i) that trading day shall be deemed to commence immediately after the regular scheduled closing time of trading on the Nasdaq Stock Market or, if trading is closed at an earlier time, such earlier time and (ii) that trading day shall end at the next regular scheduled closing time, or if trading is closed at an earlier time, such earlier time (for the avoidance of doubt, and as an example, if the Market Price is to be determined as of the last trading day preceding a specified event and the closing time of trading on a particular day is 4:00 p.m. and the specified event occurs at 5:00 p.m. on that day, the Market Price would be determined by reference to such 4:00

p.m. closing price).

- "Original Warrant Agreement" has the meaning set forth in the recitals hereto.
- "Per Share Fair Market Value" has the meaning set forth in Section 4.4.
- "Person" has the meaning given to it in Section 3(a)(9) of the Exchange Act and as used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act.
- "Holders" has the meaning set forth in the introductory paragraph hereof.
- "Required Warrantholders" means, at any time, the holders of Warrants representing at least a majority of the Common Stock issuable upon exercise of the Warrants issued and outstanding hereunder and not previously exercised (exclusive of any Warrants directly or indirectly held by the Company or any Affiliate of the Company).
 - "Securities Act" means the Securities Act of 1933, as amended, or any successor statute, and the rules and regulations promulgated thereunder.
 - "Tax" or "Taxes" has the meaning set forth in Section 1.5(a).
- "trading day" means (A) if the shares of Common Stock are not traded on any national or regional securities exchange or association or over-the-counter market, a business day or (B) if the shares of Common Stock are traded on any national or regional securities exchange or association or over-the-counter market, a business day on which such relevant exchange or quotation system is scheduled to be open for business and on which the shares of Common Stock (i) are not suspended from trading on any national or regional securities exchange or association or over-the-counter market for any period or periods aggregating one half hour or longer; and (ii) have traded at least once on the national or regional securities exchange or association or over-the-counter market that is the primary market for the trading of the shares of Common Stock.
- "Transferee" means any registered transferee of all or any part of any one or more Warrant Certificates initially acquired by the Holders under this Agreement.
 - "Warrant" means a warrant to initially purchase one share of Common Stock issued pursuant to this Agreement.
 - "Warrant Certificate" means a certificate evidencing the Warrants in the form of Attachment A.
 - 5.2 Section Heading and Table of Contents and Construction.
- (a) Section Headings and Table of Contents, etc. The titles of the Sections of this Agreement and any Table of Contents of this Agreement appear as a matter of convenience only, do not constitute a part hereof and shall not affect the construction hereof. The words "herein," "hereof," "hereunder" and "hereto" refer to this Agreement as a whole and not to any particular Section or other subdivision. References to Sections are, unless otherwise specified, references to Sections of this Agreement. References to Annexes and Attachments are, unless otherwise specified, references to Annexes and Attachments attached to this Agreement.
- (b) **Independent Construction**. Each covenant contained herein shall be construed (absent an express contrary provision herein) as being independent of each other covenant contained herein, and compliance with any one covenant shall not (absent such an express contrary provision) be deemed to excuse compliance with one or more other covenants.

5.3 Directly or Indirectly.

Where any provision herein refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether such action is taken directly or indirectly by such Person, including actions taken by or on behalf of any partnership in which such Person is a general partner.

5.4 Governing Law.

THIS AGREEMENT AND THE WARRANT CERTIFICATES SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE PARTIES SHALL BE GOVERNED BY, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICT OF LAWS RULES THEREOF TO THE EXTENT THAT ANY SUCH RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION, EXCEPT TO THE EXTENT THAT THE DELAWARE GENERAL CORPORATION LAW SPECIFICALLY AND MANDATORILY APPLIES.

6. MISCELLANEOUS.

6.1 Condition to Effectiveness; Termination.

- (a) The parties agree and acknowledge that, notwithstanding anything to the contrary herein, the effectiveness of this Agreement is conditioned upon, and shall not be binding or effective until, the occurrence of the First Merger Effective Time (as such term is defined in the Merger Agreement) in accordance with the terms and conditions of the Merger Agreement.
- (b) In the event that the Merger Agreement is terminated in accordance with its terms, this Agreement shall automatically terminate and be null and void *ab initio* and of no effect whatsoever, and the Original Warrant Agreement shall remain in full force and effect and shall continue to be binding on and enforceable against the Company and the Holders.

6.2 Expenses.

Issuance of certificates for shares of Common Stock to a holder upon the exercise of any Warrants shall be made without charge to such holder for any Tax or stamp duty in respect of the issuance of such certificates, all of which Taxes and stamp duties shall be paid by the Company (for the avoidance of doubt, specifically excluding the Taxes not payable by the Company pursuant to Section 3.3).

6.3 Amendment and Waiver.

This Agreement may be amended, and the observance of any term of this Agreement may be waived, with and only with the written consent of the Company and the Required Warrantholders; provided, however, that no amendment or waiver of the provisions of Section 2.1, this Section 6.3, Section 4 or of any term defined in Section 5.1 to the extent used in such specifically referenced Sections, may be made without the prior written consent of all holders of Warrants then outstanding (excluding any Warrants directly or indirectly held by the Company or any Affiliate of the Company); and, provided, further, that:

- (a) no such amendment or waiver of any of the provisions of this Agreement pertaining to the Exercise Price or the number of shares or kind of Common Stock that may be purchased upon exercise of each Warrant; and
 - (b) no change accelerating the occurrence of the Expiration Time;

shall be effective as to a holder of Warrants unless consented to in writing by such holder.

6.4 Entire Agreement; Termination of Original Warrant Agreement.

This Agreement and the Warrant Certificates embody the entire agreement and understanding among the Company and the Holders, and supersede all prior agreements and understandings, relating to the subject matter hereof. The Company and the Holders hereby acknowledge and agree that as of the Effective Date, the Original Warrant Agreement and any Warrant Certificate issued thereunder are terminated and of no further force and effect and confirm, pursuant to Section 6.2 of the Original Warrant Agreement, that all provisions thereof are terminated and of no further force and effect and the Original Warrant Agreement and any Warrant Certificate issued thereunder are hereby superseded and replaced by this Agreement, which amends and restates the Original Agreement and any Warrant Certificate issued thereunder in their entirety. Each Holder shall, on or promptly following the Second Merger Effective Time, deliver and surrender to the Company any Warrant Certificate issued under the Original Agreement in such Holder's possession.

6.5 Successors and Assigns; Multiple Holders.

All covenants and other agreements in this Agreement made by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors and assigns of the parties hereto to the extent they become holders of Warrants (including, without limitation, any Transferee) whether so expressed or not. Notwithstanding the foregoing sentence, the Company may not assign any of its rights, duties or obligations hereunder or under the Warrant Certificates except pursuant to the terms of this Agreement without the prior written consent of the Required Warrantholders.

Each holder of a Warrant agrees that (i) no other holder of a Warrant will by virtue of this Warrant or exercise thereof be under any fiduciary or other duty to give or withhold any consent or approval under this Warrant or to take any other action or omit to take any action under this Warrant and (ii) each other holder of a Warrant may act or refrain from acting under this Warrant as such other holder may, in its discretion, elect.

6.6 Notices and Information.

All communications hereunder or under the Warrants shall be in writing and shall be delivered either by certified or registered mail, postage prepaid, return receipt requested, email or nationally recognized overnight courier, and shall be addressed to the following addresses:

(a) if to a Holder, at its address set forth on Annex 2 to this Agreement, or at such other address as such Holder shall have specified to the Company in writing;

- (b) if to any other holder of any Warrant Certificate, addressed to such other holder at such address as such other holder shall have specified to the Company in writing or, if any such other holder shall not have so specified an address to the Company, then addressed to such other holder in care of the last holder of such Warrant Certificate that shall have so specified an address to the Company; and
- (c) if to the Company, at the address set forth on Annex 3 to this Agreement, or at such other address as the Company shall have specified to each holder of Warrants in writing.

Any communication addressed and delivered as herein provided shall be deemed to be received when actually delivered to the address of the addressee (whether or not delivery is accepted) by a nationally recognized overnight delivery service which provides proof of delivery or on the date postmarked if sent by registered or certified mail or upon receipt by the recipient's email server if directed to the email address provided in the notice section hereof, as the case may be. Any communication not so addressed and delivered shall be ineffective unless actually received by the intended addressee. Notwithstanding the foregoing provisions of this Section 6.6, service of process in any suit, action or proceeding arising out of or relating to this Agreement or any document, agreement or transaction contemplated hereby shall be delivered in the manner provided in Section 6.9(c).

Upon exercise of any Warrant pursuant to the terms hereof, the Company will use commercially reasonable efforts to answer a limited number of reasonable questions and provide a limited amount of reasonable documentation regarding any non-confidential information supporting its calculation of Fully Diluted Common Stock and the related shares of Common Stock issuable for the Denomination of such Warrant being exercised.

6.7 Severability.

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

6.8 Execution in Counterpart.

This Agreement may be executed in one or more counterparts and shall be effective when at least one counterpart shall have been executed by each party hereto, and each set of counterparts that, collectively, show execution by each party hereto shall constitute one duplicate original.

6.9 Waiver of Jury Trial; Consent to Jurisdiction, Etc.

- (a) Waiver of Jury Trial. THE PARTIES HERETO VOLUNTARILY AND INTENTIONALLY WAIVE ANY RIGHT ANY OF THEM MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE WARRANTS OR ANY OF THE DOCUMENTS, AGREEMENTS OR TRANSACTIONS CONTEMPLATED HEREBY.
- (b) Consent to Jurisdiction. ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE WARRANTS, OR ANY OF THE DOCUMENTS, AGREEMENTS OR TRANSACTIONS CONTEMPLATED HEREBY (WHETHER IN TORT, CONTRACT OR OTHERWISE) OR ANY ACTION OR PROCEEDING TO EXECUTE OR OTHERWISE ENFORCE ANY JUDGMENT IN RESPECT OF ANY BREACH UNDER THIS AGREEMENT, THE WARRANTS OR ANY DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY SHALL BE BROUGHT BY SUCH PARTY IN ANY NEW YORK STATE COURT OR FEDERAL DISTRICT COURT LOCATED IN THE SOUTHERN DISTRICT OF NEW YORK AS SUCH PARTY MAY IN ITS SOLE DISCRETION ELECT, AND BY THE EXECUTION AND DELIVERY OF THIS AGREEMENT, THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMIT TO THE IN PERSONAM JURISDICTION OF EACH SUCH COURT, AND EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES AND AGREES NOT TO ASSERT IN ANY PROCEEDING BEFORE ANY TRIBUNAL, BY WAY OF MOTION, AS A DEFENSE OR OTHERWISE, ANY CLAIM THAT IT IS NOT SUBJECT TO THE IN PERSONAM JURISDICTION OF ANY SUCH COURT. IN ADDITION, EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE IN ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY DOCUMENT, AGREEMENT OR TRANSACTION CONTEMPLATED HEREBY BROUGHT IN ANY SUCH COURT, AND HEREBY IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.
- (c) Service of Process. EACH PARTY HERETO IRREVOCABLY AGREES THAT PROCESS PERSONALLY SERVED OR SERVED BY U.S. REGISTERED MAIL AT THE ADDRESSES PROVIDED HEREIN

FOR NOTICES SHALL CONSTITUTE, TO THE EXTENT PERMITTED BY LAW, ADEQUATE SERVICE OF PROCESS IN ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE WARRANTS OR ANY DOCUMENT, AGREEMENT OR TRANSACTION CONTEMPLATED HEREBY, OR ANY ACTION OR PROCEEDING TO EXECUTE OR OTHERWISE ENFORCE ANY JUDGMENT IN RESPECT OF ANY BREACH HEREUNDER, UNDER THE WARRANTS OR UNDER ANY DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY. RECEIPT OF PROCESS SO SERVED SHALL BE CONCLUSIVELY PRESUMED AS EVIDENCED BY A DELIVERY RECEIPT FURNISHED BY THE UNITED STATES POSTAL SERVICE OR ANY COMMERCIAL DELIVERY SERVICE.

[Remainder of page intentionally left blank; next page is signature page]

IN WITNESS WHEREOF, the Parties have executed this Agreement on the date first above written.

COMPANY:

BioScrip, Inc.

By: /s/ Daniel Greenleaf
Name: Daniel Greenleaf

Title: President & CEO

[Signature Page to Amended and Restated Warrant Agreement]

HOLDERS:

AMARI INVESTMENT PTE. LTD.

By: ASSF OPERATING MANAGER IV, L.P., its manager

By: /s/ Christopher Kerezsi
Name: Christopher Kerezsi

Title: Authorized Signatory

[Signature page to Amended and Restated Warrant Agreement]

HOLDERS:

ASSF IV AIV B HOLDINGS, L.P.

By: ASSF OPERATING MANAGER IV, L.P., its manager

By: /s/ Christopher Kerezsi
Name: Christopher Kerezsi

Name: Christopher Kerezsi
Title: Authorized Signatory

[Signature page to Amended and Restated Warrant Agreement]

PURCHASERS:

J.P. Morgan Securities LLC

Brian M. Ercolani, Asc	
Attorney-in-Fact	
	,

HOLDERS:

Goldman Sachs & Co. LLC

By: /s/ Daniel Oneglia
Name: Daniel Oneglia
Title: Managing Director

PURCHASERS:

Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc

By: Western Asset Management Company, LLC, as its Investment Manager and Agent

By: /s/ Adam Wright

Name: Adam Wright

Title: Manager, U.S. Legal Affairs

ATTACHMENT A [FORM OF WARRANT CERTIFICATE]

THE SECURITIES REPRESENTED HEREBY, AND THE SECURITIES ISSUABLE HEREBY, HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS AND NEITHER THE SECURITIES NOR ANY INTEREST THEREIN MAY BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT OR SUCH LAWS OR AN EXEMPTION FROM REGISTRATION UNDER SUCH ACT AND SUCH LAWS WHICH, IN THE OPINION OF COUNSEL FOR THE COMPANY, IS AVAILABLE. THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE TERMS OF A CERTAIN AMENDED AND RESTATED WARRANT AGREEMENT, DATED AS OF MARCH [.], 2019, THE PROVISIONS OF WHICH ARE INCORPORATED HEREIN BY REFERENCE. A COPY OF SUCH AGREEMENT IS AVAILABLE FROM THE COMPANY UPON REQUEST.

WARRANT CERTIFICATE

BIOSCRIP, INC.

No. WRT- Warrants

Date: [], 20[]

This Warrant Certificate certifies that , or registered assigns, is the registered holder of Warrants entitling the owner thereof to purchase at any time on or after the date hereof and on or prior to the Expiration Time, [] fully paid and nonassessable shares of Common Stock, \$0.0001 par value per share (the "Common Stock"), of BIOSCRIP, INC., a Delaware corporation (together with its successors and assigns, the "Company"), at a purchase price (subject to adjustment as provided in the Warrant Agreement (as defined below), the "Exercise Price") of \$2.00 per share of Common Stock upon presentation and surrender of this Warrant Certificate to the Company with a duly executed election to purchase and payment of the Exercise Price (including by withholding of shares of Common Stock), all in the manner set forth in the Warrant Agreement (defined below). The Denomination of each Warrant and the Exercise Price are the Denomination and the Exercise Price as of the date hereof, and are subject to adjustment as referred to below.

The Warrants are issued pursuant to an Amended and Restated Warrant Agreement (as it may from time to time be amended or supplemented, the "Warrant Agreement"), dated as of March 14, 2019, among the Company and the Holders named therein, and are subject to all of the terms, provisions and conditions thereof, which Warrant Agreement is hereby incorporated herein by reference and made a part hereof and to which Warrant Agreement reference is hereby made for a full description of the rights, obligations, duties and immunities of the Company and the holders of the Warrant Certificates. Capitalized terms used, but not defined, herein have the respective meanings ascribed to them in the Warrant Agreement. In the event of any conflict between this Warrant Certificate and the Warrant Agreement, the Warrant Agreement shall control and govern.

As provided in the Warrant Agreement, the Exercise Price and the Denomination evidenced by this Warrant Certificate are, upon the happening of certain events, subject to modification and adjustment. Except as otherwise set forth in, and subject to, the Warrant Agreement, the Expiration Time of this Warrant Certificate is as set forth in the Warrant Agreement.

Subject to the limitations set forth in the Warrant Agreement, this Warrant Certificate shall be exercisable, at the election of the holder, at any time on or after the date hereof and on or prior to the Expiration Time either as an entirety or in part from time to time. If this Warrant Certificate shall be exercised in part, the holder shall be entitled to receive, upon surrender hereof, another Warrant Certificate or Warrant Certificates for the Warrants not exercised. This Warrant Certificate, with or without other Warrant Certificates, upon surrender in the manner set forth in the Warrant Agreement and subject to the conditions set forth in the Warrant Agreement, may be transferred or exchanged for another Warrant Certificate or Warrant Certificates of like tenor evidencing Warrants entitling the holder to a like number of shares of Common Stock as the Warrants evidenced by the Warrant Certificate or Warrant Certificates surrendered shall have entitled such holder to purchase.

Except as expressly set forth in the Warrant Agreement, no holder of this Warrant Certificate shall be entitled to vote or receive distributions or be deemed for any purpose the holder of shares of Common Stock or of any other securities of

the Company that may at any time be issued upon the exercise hereof, nor shall anything contained in the Warrant Agreement or herein be construed to confer upon the holder hereof, as such, any of the rights of a holder of a share of Common Stock in the Company or any right to vote upon any matter submitted to holders of shares of Common Stock at any meeting thereof, or to give or withhold consent to any corporate action of the Company (whether upon any recapitalization, issuance of stock, reclassification of Securities, change of par value, consolidation, merger, conveyance, or otherwise), or to receive dividends or subscription rights, or otherwise, until the Warrant or Warrants evidenced by this Warrant Certificate shall have been exercised as provided in the Warrant Agreement.

In the event of any inconsistency between this warrant Certificate and the Warrant agreement, the terms of the Warrant Agreement shall govern.

THIS WARRANT CERTIFICATE SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE COMPANY AND THE HOLDER HEREOF SHALL BE GOVERNED BY, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICT OF LAWS RULES THEREOF TO THE EXTENT THAT ANY SUCH RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION, EXCEPT TO THE EXTENT THAT THE DELAWARE GENERAL CORPORATION LAW SPECIFICALLY AND MANDATORILY APPLIES.

WITNESS the signature of a proper officer of the Company as of the date first above written.

BIOSCRIP, INC.

By: Name: Title:

[FORM OF ASSIGNMENT]

(To be executed by the registered holder if

such holder desires to transfer the Warrant Certificate)

FOR VALUE RECEIVED,

hereby sells, assigns and transfers unto (Please print name and address of transferee.)

the accompanying Warrant Certificate, together with all right, title and interest therein, and does hereby irrevocably constitute and appoint:

attorney, to transfer the accompanying Warrant Certificate on the books of the Company with full power of substitution.

Dated: , 20.

[HOLDER]

By:

NOTICE

The signature to the foregoing Assignment must correspond to the name as written upon the face of the accompanying Warrant Certificate or any prior assignment thereof in every particular, without alteration or enlargement or any change whatsoever.

[FORM OF ELECTION TO PURCHASE]

(To be executed by the registered holder if such holder desires to exercise the Warrant Certificate)

To: BIOSCRIP, INC.

The undersigned hereby irrevocably elects to exercise	Warrants represented by the accompanying Warrant Certificate to purchase the
shares of Common Stock issuable upon the exercise of such Warrants, an	d requests that certificates for such shares be issued in the name of:

(Please print name and address.)

(Please insert social security or other identifying number.)

If such number of Warrants being exercised by the undersigned are not all the Warrants evidenced by the accompanying Warrant Certificate, a new Warrant Certificate for the balance remaining of such Warrants shall be registered in the name of and delivered to:

(Please print name and address.)

(Please insert social security or other identifying number.)

The undersigned is paying the Exercise Price for the Common Stock to be issued on exercise of the foregoing Warrants, unless payment of such Exercise Price has been waived by the Company, by

o certified or bank check by wire transfer, or

o withholding of shares of Common Stock into which the Warrants are being exercised.

Dated: , 20.

[HOLDER]

By:

NOTICE

The signature to the foregoing Election to Purchase must correspond to the name as written upon the face of the accompanying Warrant Certificate or any prior assignment thereof in every particular, without alteration or enlargement or any change whatsoever.

ANNEX 1 Warrants Held on the Effective Date by the Holders

Holder	Total Number of Warrants
ASSF IV AIV B Holdings, L.P.	4,520,354
Amari Investments Pte. Ltd.	1,506,784
J.P. Morgan Securities LLC	753,398
Goldman Sachs & Co. LLC	1,130,081
Western Asset Middle Market Debt Fund Inc.	113,016
Western Asset Middle Market Income Fund Inc	263,683

ANNEX 2 Address for Holders for Notices

ASSF IV AIV B HOLDINGS, L.P.

ASSF IV AIV B Holdings, L.P. c/o Ares Management LLC 2000 Avenue of the Stars, 12th Floor Los Angeles, CA 90067

J.P. MORGAN SECURITIES LLC

J.P. Morgan Securities LLC 4 New York Plaza, 15th Floor, Mail Code NY1-E054 New York, New York 10004

Attention: Jeffrey L. Panzo Email: Jeffrey.L.Panzo@JPMorgan.com

Phone No.: (212) 499-1435

GOLDMAN SACHS & CO. LLC

Goldman Sachs & Co. LLC 200 West Street, 26th Floor Attn: Paul Burningham and Paige Cataruozolo New York, New York 10282 Email: fice-amssg-mo@gs.com

Phone No.: (917) 343-8393 (Paul Burningham),

(917) 343-3096 (Paige Cataruozolo)

WESTERN ASSET MIDDLE MARKET DEBT FUND INC. WESTERN ASSET MIDDLE MARKET INCOME FUND INC

Western Asset Management Company 385 East Colorado Boulevard Pasadena, California 91101 Attention: Legal Department

ANNEX 3 Address of Company for Notices

BioScrip, Inc. 1600 Broadway, Suite 700 Denver, CO 80202

Attn: Stephen Deitsch, Senior Vice President,

Chief Financial Officer & Treasurer Email: Stephen.Deitsch@bioscrip.com Telecopy Number: (720) 468-4040

With a copy (which shall not constitute notice) to:

Gibson, Dunn & Crutcher, LLP

200 Park Avenue

New York, New York 10036 Attention: Sean P. Griffiths Email: sgriffiths@gibsondunn.com Telecopy Number: (212) 251-5222

AMENDMENT NO. 1 TO REGISTRATION RIGHTS AGREEMENT

THIS AMENDMENT NO. 1 TO THE REGISTRATION RIGHTS AGREEMENT (this "Amendment"), effective as of March 14, 2019, is by and between BioScrip, Inc. (the "Company"), and the stockholders of the Company signatory hereto (each a "Stockholder" and collectively, the "Stockholders"). Capitalized but otherwise undefined terms herein have the meanings given to them in the Registration Rights Agreement (as defined below).

WHEREAS, the Company and the Stockholders are party to that certain Registration Rights Agreement, dated June 29, 2017 (the "Registration Rights Agreement"), governing the Company's obligations to register Registrable Securities of the Stockholders; and

WHEREAS, in connection with entering into that certain (i) Amended and Restated Warrant Agreement, dated as of the date hereof, by and between the Company and the Stockholders (the "Warrant Amendment"), (ii) those certain letter agreements, dated as of the date hereof, by and between the Company and each of the Stockholders (the "Letter Agreements"), pursuant to which the Company will issue an aggregate of 1,855,747 shares of Common Stock of the Company upon the closing of the transactions contemplated by the Merger Agreement (as defined below), and (iii) that certain Agreement and Plan of Merger, dated as of the date hereof, by and among the Company, HC Group Holdings II, Inc. and the other parties thereto, the Company and the Stockholders desire to amend the Registration Rights Agreement pursuant to Section 11(c) thereof and upon such terms as set forth herein.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- 1. Amendment to Registration Rights Agreement.
 - (a) Section 1(a) of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:

"Short-Form Registrations. At any time after June 29, 2018 twelve (12) months following the date hereof, with respect to the Purchased Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, or at any time after the twelve month anniversary of the First Merger Effective Time (as defined in the Merger Agreement), with respect to the Amendment Shares, each Holder may request registration under the Securities Act of all or any portion of its Registrable Securities on Form S-3 or any successor form (each, a "Short-Form Registration"), which may, if so requested, be a "shelf" registration under Rule 415 under the Securities Act. A registration shall not count as one of the permitted Short-Form Registrations unless and until a registration statement relating thereto has become effective under the Securities Act. Each request for a Short-Form Registration shall specify the number of Registrable Securities requested to be registered.

(b) Section 1(b) of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:

"Long-Form Registrations. At any time that a Holder is then eligible to request registration under the Securities Act of all or any portion of its Registrable Securities but where Short-Form Registration pursuant to Section 1(a) of this Agreement is not available to be used by the Company in respect of such proposed registration, but in no event earlier than twelve (12) months following the date hereof June 29, 2018, with respect to the Purchased Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, or the twelve month anniversary of the First Merger Effective Time (as defined in the Merger Agreement), with respect to the Amendment Shares, each Holder shall be entitled to request a registration on Form S-1 or any similar form (each, a "Long-Form Registration"). A registration shall not count as one of the permitted Long-Form Registrations unless and until a registration statement relating thereto has become effective under the Securities Act and each requesting Holder is able to register and sell at least thirty percent (30%) of its Registrable Securities thereunder.

(c) The definition of Registrable Securities in Section 10 of the Registration Rights Agreement is hereby amended and restated in its entirety to read as follows:

""Registrable Securities" means the Purchased Shares, the Amendment Shares and the Common Stock that has been or will be issued upon exercise of the Warrants, together with any securities issued or issuable upon any stock split, dividend or other distribution, recapitalization or similar event with respect to the foregoing. For purposes of

this Agreement, a Person shall be deemed to be a holder of Registrable Securities whenever such Person has the right to acquire such Registrable Securities (upon conversion or exercise, in connection with a transfer of securities or otherwise, but disregarding any restrictions or limitations upon the exercise of such right), whether or not such acquisition has actually been effected. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (A) a Registration Statement covering such securities has been declared effective by the Securities and Exchange Commission and such securities have been disposed of pursuant to such effective Registration Statement, (B) such securities are sold under circumstances in which all of the applicable conditions of Rule 144 (or any similar provisions then in force) under the Securities Act are met, (C) the aggregate amount of such securities held by each Holder, together with all Affiliates of each such Holder and persons forming a "group" with each such Holder (within the meaning of Regulation 13D under the Securities Exchange Act of 1934, as amended from time to time, and the rules and regulations promulgated thereunder ("Regulation 13D")), consists of beneficial ownership (within the meaning of Regulation 13D) of less than 5.0% of the Common Stock of the Company and such securities are eligible for sale by each such Holder without registration pursuant to Rule 144 (or any similar provisions then in force) under the Securities Act without limitation thereunder on volume or manner of sale, (D) such securities are otherwise transferred and such securities may be resold without limitation registration relating to such securities have had all restrictive legends removed."

(d) The following defined term shall be added to Section 10 of the Registration Rights Agreement:

""<u>Amendment Shares</u>" means the shares of Common Stock that the Company issued to the Stockholders pursuant to those certain letter agreements, dated as of March 14, 2019, by and between the Company and each of the Stockholders, in connection with the Amended and Restated Warrant Agreement, dated as of March 14, 2019, by and between the Company and the Stockholders."

- 2. <u>Condition to Effectiveness; Termination</u>. The Company and the Stockholders agree and acknowledge that, notwithstanding anything to the contrary herein, the effectiveness of this Amendment is conditioned upon, and shall not be binding or effective until, the occurrence of the First Merger Effective Time (as such term is defined in the Merger Agreement) in accordance with the terms and conditions of the Merger Agreement. In the event that the Merger Agreement is terminated in accordance with its terms, this Amendment shall automatically terminate and be null and void *ab initio* and of no effect whatsoever, and the Registration Rights Agreement shall remain in full force and effect and shall continue to be binding on and enforceable against the Company and the Purchasers.
- 3. <u>Confirmation</u>. Except as expressly modified by the terms and provisions of this Amendment, all of the terms and provisions of the Registration Rights Agreement are unchanged and continue in full force and effect and all rights, remedies, liabilities and obligations evidenced by the Registration Rights Agreement are hereby acknowledged by the Company and the Stockholders to be valid and in full force and effect.
- 4. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts, by facsimile if necessary, each of which shall be an original, but all of which together shall constitute one instrument.
- 5. Governing Law. This Amendment shall be governed by and construed in accordance with the internal laws of New York applicable to parties residing in New York, without regard applicable principles of conflicts of law. Each party irrevocably consents to the exclusive jurisdiction of any court located within New York County, New York, in connection with any matter based upon or arising out of this Amendment or the matters contemplated hereby and it agrees that process may be served upon it in any manner authorized by the laws of the State of New York for such persons and waives and covenants not to assert or plead any objection which it might otherwise have to such jurisdiction and such process.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered as of the date first above written.

COMPANY:

BIOSCRIP, INC.

/s/ Daniel Greenleaf

Name: Daniel Greenleaf Title: President & CEO

[Signature Page to Amended and Restated Registration Rights Agreement]

IN WITNESS WHEREOF, the parties hereto have caused this Ame	endment to be executed and delivered as of the date first above written.
	COMPANY:
	BIOSCRIP, INC.
	Name: Title:
	STOCKHOLDERS:
	AMARI INVESTMENT PTE. LTD.
	By: ASSF OPERATING MANAGER IV, L.P., its manager
	/s/ Christopher Kerezsi
	Name: Christopher Kerezsi Title: Authorized Signatory
	J.P. Morgan Securities LLC
	Name: Title:
	Goldman Sachs & Co. LLC
	Name: Title:
	Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc By: Western Asset Management Company, as its Investment Manager and Agent
	Name: Title:
[Signature Page to Amendment No	. 1 to Registration Rights Agreement]

	COMPANY:
	BIOSCRIP, INC.
	Name: Title:
	STOCKHOLDERS:
	ASSF IV AIV B HOLDINGS, L.P.
	By: ASSF OPERATING MANAGER IV, L.P., its manager
	/s/ Christopher Kerezsi
	Name: Christopher Kerezsi Title: Authorized Signatory
	J.P. Morgan Securities LLC
	Name: Title:
	Goldman Sachs & Co. LLC
	Name: Title:
	Western Asset Middle Market Debt Fund Inc. Western Asset Middle Market Income Fund Inc By: Western Asset Management Company, as its Investment Manager and Agent
	Name: Title:
[Signature Page to Amendment No	o. 1 to Registration Rights Agreement]

COMPANY:
BIOSCRIP, INC.
Na
Name: Title:
STOCKHOLDERS:
ASSF IV AIV B HOLDINGS, L.P.
By: ASSF IV AIV B HOLDINGS GP LLC, its general partner
By: ASSF IV AIV B, L.P.,
its sole member By: ASSF MANAGEMENT IV, L.P.,
its general partner
By: ASSF MANAGEMENT IV GP LLC, its general partner
Name
Name: Title:
J.P. Morgan Securities LLC
/s/ Brian M. Ercolani, Asc
Name: Brian M. Ercolani, Asc Title: Attorney-in-Fact
Goldman Sachs & Co. LLC
Name:
Title:
Western Asset Middle Market Debt Fund Inc.
Western Asset Middle Market Income Fund Inc By: Western Asset Management Company,

Name:

Title:

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

COMPANY:
BIOSCRIP, INC.
Name:
Title:
STOCKHOLDERS:
ASSF IV AIV B HOLDINGS, L.P.
By: ASSF IV AIV B HOLDINGS GP LLC,
its general partner By: ASSF IV AIV B, L.P.,
its sole member
By: ASSF MANAGEMENT IV, L.P.,
its general partner
By: ASSF MANAGEMENT IV GP LLC,
its general partner
Name:
Title:
J.P. Morgan Securities LLC
Name:
Title:
Goldman Sachs & Co. LLC
/s/ Daniel Onglia
Name: Daniel Onglia
Title: Managing Director
Western Asset Middle Market Debt Fund Inc.
Western Asset Middle Market Income Fund Inc
D. W. M.

By: Western Asset Management Company, as its Investment Manager and Agent

Name:

Title:

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

COMPANY:
BIOSCRIP, INC.
Name:
Title:
STOCKHOLDERS:
ASSF IV AIV B HOLDINGS, L.P.
By: ASSF IV AIV B HOLDINGS GP LLC,
its general partner By: ASSF IV AIV B, L.P.,
its sole member
By: ASSF MANAGEMENT IV, L.P.,
its general partner By: ASSF MANAGEMENT IV GP LLC,
its general partner
Name:
Title:
J.P. Morgan Securities LLC
Name:
Title:
Goldman Sachs & Co. LLC
Name:
Name. Title:

Western Asset Middle Market Debt Fund Inc. **Western Asset Middle Market Income Fund Inc**

By: Western Asset Management Company, LLC, as its Investment Manager and Agent

/s/ Adam Wright Name: Adam Wright Title: Manager, U.S. Legal Affairs

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

EXECUTION VERSION

BANK OF AMERICA, N.A. MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED

One Bryant Park New York, New York 10036 ARES CAPITAL MANAGEMENT LLC

245 Park Avenue, 44th Floor New York, New York 10167 BROAD STREET LOAN PARTNERS III, L.P.
BROAD STREET LOAN PARTNERS III OFFSHORE, L.P.
BROAD STREET LOAN PARTNERS III OFFSHORE-UNLEVERED, L.P.
BROAD STREET SENIOR CREDIT PARTNERS II, L.P.

200 West Street New York, New York 10282

March 14, 2019

HC Group Holdings II, Inc. c/o Option Care Infusion Services, Inc. 3000 Lakeside Dr. #300N Bannockburn, IL 60015 Attention: Mike Shapiro

> Project Opal Senior Credit Facilities Commitment Letter

Ladies and Gentlemen:

You have advised Bank of America, N.A. (together with any of its designated affiliates and branches, "Bank of America"), Merrill Lynch, Pierce, Fenner & Smith Incorporated (together with any of its designated affiliates, "MLPFS"), Ares Capital Management LLC (on behalf of one or more affiliated funds, investment vehicles and/or managed accounts, "Ares"), and the entities listed on Schedule I hereto (the "Initial MBD Lenders" and, together with any Other MBD Lenders (as defined below) to which the Initial MBD Lenders have reallocated, sold, resold, assigned and/or transferred commitments pursuant to, and in accordance with, this Commitment Letter, the "MBD Lenders", and collectively with Bank of America, MLPFS, Ares and any Additional Agent (as defined below) appointed in accordance with Section 2 below, each a "Commitment Party," and together, the "Commitment Parties," "us" or "we"), that you intend to consummate the transactions described in Exhibit A hereto. Capitalized terms used but not defined herein are used with the meanings assigned to them on the exhibits attached hereto (such exhibits, together with this letter, collectively, the "Commitment Letter").

1. <u>Commitments</u>

In connection with the Transactions, (a)(i) Bank of America is pleased to advise you of its commitment to provide, and hereby agrees to provide, 65% of the principal amount of the First Lien Term Facility (as defined below), (ii) Ares is pleased to advise you of its commitment to provide, and hereby agrees to provide, 17.5% of the principal amount of the First Lien Term Facility and (iii) the Initial MBD Lenders (together with Bank of America, Ares, and any other Initial First Lien Lender appointed in accordance with Section 2 below on a several (and not joint) basis, if applicable, each an "Initial First Lien Lender," and collectively, the "Initial First Lien Lenders") are pleased to advise you of their several, but not joint, commitments to provide, in the aggregate for all the applicable entities, 17.5% of the principal amount of the First Lien Term Facility, which commitments will be allocated among the Initial MBD Lenders in accordance with the commitments set forth on Schedule I hereto and (b) Bank of America (together with any other Initial ABL Lender appointed in accordance with Section 2 below on a several (and not joint) basis, if applicable, collectively, the "Initial ABL Lenders"; the Initial ABL Lenders and the Initial First Lien Lenders are collectively referred to as the "Initial Senior Lenders") is pleased to advise you of its commitment to provide, and hereby agrees to provide, 100% of the principal amount of the ABL Facility (as defined below), in each case upon the terms expressly set forth in this Commitment Letter (including, without limitation, in each of the Summary of Terms and Conditions attached hereto as Exhibits B and C, respectively (each, a "Term Sheet" and collectively, the "Term Sheets")) and subject solely to the Exclusive Funding Conditions (as defined below). For the avoidance of doubt, the commitments of the Initial Senior Lenders hereunder are several and not joint.

Notwithstanding anything to the contrary contained herein or in the Fee Letter (as defined below), each of Ares and each MBD Lender is providing its respective commitment with respect to the First Lien Term Facility solely as principal and in so doing is not acting as an underwriter, arranger or agent with respect to the Senior Credit Facilities or otherwise providing any services hereunder.

2. <u>Titles and Roles</u>

You hereby appoint (a)(i) MLPFS, together with any other lead arrangers or bookrunners appointed as contemplated in the immediately succeeding paragraph, to act as joint lead arrangers and joint bookrunners (in such capacity, the "First Lien Lead Arrangers") and (ii) Bank of America to act as sole administrative agent (in such capacity, the "First Lien Administrative Agent") for the First Lien Term Facility and (b) (i) MLPFS, together with any other lead arrangers or bookrunners appointed as contemplated in the immediately succeeding paragraph, to act as joint lead arrangers and joint bookrunners (in such capacity, the "ABL Lead Arrangers" and, together with the First Lien Lead Arrangers, the "Senior Lead Arrangers") and (ii) Bank of America to act as sole administrative agent (in such capacity, the "ABL Administrative Agent" and, together with the First Lien Administrative Agent, the "Administrative Agent") for the ABL Facility.

It is further agreed that MLPFS will have "left" placement on any marketing materials or other documentation used in connection with any of the Senior Credit Facilities and shall hold the leading role and responsibility associated with such "top left" placement. You agree that no

other agents or arrangers will be appointed, no other titles will be awarded and no compensation (other than that compensation expressly contemplated by this Commitment Letter and the Fee Letter referred to below) will be paid in connection with obtaining commitments under the Senior Credit Facilities unless you and we shall so agree; provided that, on or prior to the date which is fifteen (15) business days after the date you execute and deliver this Commitment Letter, you may appoint, with up to (i) 15.00% in the aggregate of the economics and commitment amounts for the First Lien Term Facility; and (ii) 66.67% in the aggregate of the economics and commitment amounts for the ABL Facility, in each case, additional agents, co-agents, lead arrangers, bookrunners, managers or arrangers (any such agent, co-agent, lead arranger, bookrunner, manager or arranger, an "Additional Agent") (it being understood that (a) such Additional Agents (or their affiliates) shall assume a proportion of the commitments with respect to the applicable Senior Credit Facility that is equal to the proportion of the economics allocated to such Additional Agents (or their affiliates), (b) no Additional Agent (nor any affiliate thereof) shall receive greater economics (measured as a percentage of its commitment) in respect of any Senior Credit Facility than that received by MLPFS and (c) to the extent you appoint Additional Agents and/or confer additional titles in respect of any Senior Credit Facility on the Additional Agents, the economics allocated to, and the commitment amounts of, MLPFS (but not any other Initial Senior Lender) in respect of such Senior Credit Facility will be proportionately reduced by the amount of the economics allocated to, and the commitment amount of, such Additional Agents (or their affiliates (other than Excluded Affiliates)), in each case upon the execution and delivery by such Additional Agents and you of customary joinder documentation or an amendment to this Commitment Letter and, thereafter, each such Additional Agent shall constitute a "Commitment Party," "Initial Senior Lender" and/or "Senior Lead Arranger," as applicable, under this Commitment Letter and under the Fee Letter The parties hereto agree that Excluded Affiliates (as defined below) of such Additional Agents shall be treated in the same manner as Excluded Affiliates of the Initial Senior Lender and Senior Lead Arrangers hereunder. For the avoidance of doubt, your obligations under this paragraph shall automatically terminate upon the date on which this Commitment Letter terminates pursuant to Section 10.

Syndication

The Senior Lead Arrangers intend to syndicate the First Lien Term Facility and may syndicate the ABL Facility, in each case, to a group of banks, financial institutions and other institutional lenders, reasonably acceptable to you, such consent not to be unreasonably withheld or delayed (together with the Initial Senior Lenders but excluding Disqualified Institutions, the "Lenders"); provided that, the Senior Lead Arrangers will not syndicate, assign or participate (i) to (a) those persons that are competitors of you, the Company and its subsidiaries, Walgreens Co. ("Walgreens") or Sponsor (as defined below) to the extent separately identified by you or the Sponsor to us (or, after the Closing Date, by the Borrower or the Sponsor to the Administrative Agent) in writing from time to time, in each case including any such entity's affiliates that are identified as such pursuant to this clause (a) or those that are clearly identifiable on the basis of their name (in each case, other than bona fide diversified debt funds) (other than those excluded pursuant to clauses (b) or (c) of this paragraph)), (b) those banks, financial institutions and other persons to the extent separately identified by you or the Sponsor to us in writing prior to the date hereof or as you and we shall mutually agree on or after such date, in each case including any such entity's affiliates that are identified as such pursuant to this clause (b) or are clearly

identifiable as such on the basis of their name or (c) Excluded Affiliates (such persons or entities in clause (a), (b) or (c), collectively, the "Disqualified Institutions") or (ii) the commitments of the MBD Lenders and of Ares in respect of the First Lien Term Facility, provided that any additional designation permitted by the foregoing shall not apply retroactively to any prior allocation, assignment or participation, as applicable. Notwithstanding any other provision of this Commitment Letter to the contrary and notwithstanding any assignment, syndication or participation by the Initial Senior Lenders (except in the case of an assignment to an Additional Agent), (i) the Initial Senior Lenders shall not be released, relieved or novated from their respective obligations hereunder (including their respective obligation to fund the Senior Credit Facilities on the Closing Date) in connection with any syndication, assignment or participation of the Senior Credit Facilities, including their respective commitments in respect thereof, until the initial funding of the Senior Credit Facilities on the Closing Date, (ii) no assignment or novation shall become effective with respect to all or any portion of the Initial Senior Lenders' respective commitments in respect of the Senior Credit Facilities until the initial funding of the Senior Credit Facilities, including all retain exclusive control over all rights and obligations with respect to their respective commitments in respect of the Senior Credit Facilities, including all rights with respect to consents, modifications, supplements and amendments, until the initial funding of the Senior Credit Facilities on the Closing Date has occurred. All commitments with respect to the First Lien Term Facility shall be allocated first to reduce the commitments held by Bank of America until Bank of America holds commitments with respect to the First Lien Term Facility of not more than \$0.

The Senior Lead Arrangers intend to commence syndication efforts with respect to the Senior Credit Facilities on a date to be reasonably mutually agreed between you and the Senior Lead Arrangers and, until the earlier to occur of (a) a Successful Syndication (as defined in the Fee Letter) and (b) 30 days after the Closing Date (such period, the "Syndication Period"), you agree to use commercially reasonable efforts to assist (and to use commercially reasonable efforts to cause the Company to assist) the Senior Lead Arrangers in completing a syndication reasonably satisfactory to the Senior Lead Arrangers and you. Such assistance shall include using your commercially reasonable efforts to (i) ensure that the syndication efforts benefit from your and the Sponsor's existing banking relationships and, to the extent reasonably practical and appropriate, the Company's existing lending and investment banking relationships; (ii) direct contact between your and the Sponsor's senior management and the proposed Lenders (and using your commercially reasonable efforts to obtain such contact between the senior management of the Company and the proposed Lenders) at times and locations to be mutually agreed; (iii) your and the Sponsor's assistance (and using your commercially reasonable efforts to cause the senior management of the Company to assist) in the preparation of confidential information memoranda customary for transactions of this type (the "Confidential Information Memorandum") and other customary and readily available marketing materials to be used in connection with the syndication of the Senior Credit Facilities (all of which shall be in form and substance consistent with confidential information memoranda used in recent loan transactions sponsored by the Sponsor), including by providing the projections (such projections, including financial estimates, budgets and forecasts, the "Projections"); (iv) the hosting, with the Senior Lead Arrangers, of one formal bank meeting of prospective Lenders at a time and locati

(and using your commercially reasonable efforts to cause the senior management of the Company to be available for such meeting and, to the extent we, in consultation with you, reasonably determine to be necessary, such conference calls), (v) your ensuring (and in the case of the Company using commercially reasonable efforts to ensure) that, until the end of the Syndication Period, there shall be no other issues, offerings, placements or arrangements of competing debt securities or credit facilities by or on behalf of you or the Company being offered, placed or arranged (other than the Senior Credit Facilities, the Second Lien Notes, indebtedness to be agreed to remain outstanding after the Closing Date and indebtedness permitted under the Merger Agreement), in each case, which would reasonably be expected to have a materially adverse impact on the primary syndication of the Senior Credit Facilities and (vi) using commercially reasonable efforts to obtain prior to the launch of general syndication (x) public corporate/family ratings for the Borrower (after giving effect to the Transactions) and (y) public ratings for the First Lien Term Facility from Moody's Investors Services, Inc. and Standard & Poor's Financial Services LLC, a subsidiary of S&P Global Inc. You further agree, until the Closing Date, to provide to the MBD Lenders and Ares all information with respect to the business, operations, assets, liabilities, financial projections and prospects of the Borrower, the Company and their respective subsidiaries and the Transactions, and such other information, in each case, as is generally provided to the First Lien Lead Arrangers. For the avoidance of doubt, (1) you will not be required to provide any information (x) to the extent that the provision thereof would violate any attorney-client privilege, law, rule or regulation, or any fiduciary duty or obligation of confidentiality (not created in contemplation hereof) binding on you, the Company or your or its respective affiliates (provided that in the case of any confidentiality obligation binding on you or your affiliates, you shall use commercially reasonable efforts to notify us if any such information that we have specifically identified and requested is being withheld as a result of any such obligation of confidentiality) or (y) that consists of customer-specific data or competitively sensitive information of you or the Company that is not required to be provided pursuant to the Merger Agreement, (2) your commercially reasonable efforts to cause the Company or its management to provide information or to do or assist with any of the provisions of this paragraph shall not include actions or assistance to the extent the same would be in contravention of the Merger Agreement and shall be subject to the limitations set forth in the Merger Agreement and otherwise limited to actions or assistance to the extent practical and appropriate and (3) the only financial statements that shall be required to be provided to the Senior Lead Arrangers in connection with the syndication of the Senior Credit Facilities shall be those required to be delivered pursuant to paragraph 5 of Exhibit D hereto (as and when required pursuant to the terms thereof). Notwithstanding anything to the contrary contained in this Commitment Letter or the Fee Letter or any other letter agreement or undertaking concerning the financing of the Transactions to the contrary, neither the commencement nor the completion of the syndication of the Senior Credit Facilities, the obtaining of any ratings nor the compliance with any of the other provisions set forth in any provision of this or any other paragraph of this Section 3 shall constitute a condition to the commitments hereunder or the funding of the Senior Credit Facilities on the Closing Date or at any time thereafter.

The Senior Lead Arrangers, in their capacity as such, will manage, in consultation with you (subject to your rights set forth in the preceding paragraphs of this Section 3 and your rights to appoint Additional Agents in Section 2), all aspects of the syndication, including decisions as to the selection of institutions (other than Disqualified Institutions) to be approached and when they will be approached, when the Lenders' commitments will be accepted, which Lenders

(other than Disqualified Institutions) will participate, the allocation of the commitments among the Lenders and the amount and distribution of fees among the Lenders.

You acknowledge that (a) the Senior Lead Arrangers will make available an information package and presentation to the proposed syndicate of Lenders by posting the information package and presentation on IntraLinks or another similar electronic system and (b) certain prospective Lenders may be "public side" Lenders (i.e., Lenders that have personnel that do not wish to receive material non-public information (within the meaning of the United States federal and state securities laws, "MNPI") with respect to you, the Company, the respective securities of of any of the foregoing or the Merger and who may be engaged in investment and other market-related activities with respect to such entities' securities). At the reasonable request of the Senior Lead Arrangers, you agree to assist (and to use commercially reasonable efforts to cause the Company to assist) in the preparation of a version of the information package and presentation consisting exclusively of information and documentation that is (a) publicly available (or customarily contained in any "public side" confidential information memorandum for senior secured bank facilities) or (b) not material with respect to you, the Company or your or any of their respective subsidiaries or any of your or their respective securities for purposes of United States Federal and state securities laws assuming such laws are applicable to you, the Company or your or their respective subsidiaries (all such information and documentation being "Public Lender Information" and with any information and documentation that is not Public Lender Information being referred to herein as "Private Lender Information"). It is understood that in connection with your assistance described above, customary authorization letters will be included in the Confidential Information Memorandum that authorizes the distribution of such information to prospective Lenders and confirms to the Senior Lead Arrangers that the public version of the Confidential Information Memorandum does not include Private Lender Information about you, the Company or your or their respective securities and will contain customary language exculpating us, our affiliates, you, the Sponsor, Walgreens, the Company and your and their affiliates with respect to any liability related to the use of the contents of such marketing material by the recipients thereof in violation of applicable securities laws. You acknowledge and agree that the following documents may be distributed to prospective Lenders wishing to receive only the Public Lender Information (unless you promptly notify us otherwise and provided that you have been given a reasonable opportunity to review such documents); (i) drafts and final definitive documentation with respect to the Senior Credit Facilities (excluding, if applicable, any specifically identified schedules thereof); (ii) administrative materials prepared by the Senior Lead Arrangers for prospective Lenders (such as a Lender meeting invitation, allocations and funding and closing memoranda (but excluding any Projections)); and (iii) term sheets and notification of changes in the terms of the Senior Credit Facilities. You also agree to identify that portion of any other Information (as defined below) as relating to you or the Company (the "Borrower Materials") to be distributed to "public side" Lenders and that you will clearly and conspicuously mark such materials "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof. By marking Borrower Materials "PUBLIC," you shall be deemed to have authorized the Senior Lead Arrangers and the proposed Lenders to treat such Borrower Materials as not containing any Private Lender Information (it being understood that you shall not be under any obligation to mark the Information Materials "PUBLIC"). You agree that, unless expressly identified as Public Lender Information, each document to be disseminated by the Senior Lead Arrangers to any Lender in connection with the Senior Credit Facilities will be deemed to

contain Private Lender Information (except with respect to those documents described in clauses (i), (ii) and (iii) of the second preceding sentence).

Information

You hereby represent and warrant (with respect to Information (as defined below) relating to the Company and its subsidiaries and businesses, to your knowledge) that (a) all written information, other than the Projections, forward looking statements and information of a general economic or industryspecific nature, concerning you, the Borrower, the Company and your and their respective subsidiaries, the Merger and the other Transactions contemplated hereby (the "Information"), that has been or will be made available to us by you or your representatives in connection with the transactions contemplated hereby, taken as a whole and as supplemented, does not contain (or, in the case of Information furnished after the date hereof, will not contain), as of the time it was (or hereafter is) furnished, any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statements are made, as supplemented and updated as provided below and (b) the Projections that have been or will be made available to us by or on behalf of you in connection with the transactions contemplated hereby have been or will be prepared in good faith based upon assumptions believed by you to be reasonable at the time furnished (it being recognized by the Commitment Parties that (i) such Projections are not to be viewed as facts or a guarantee of performance and are subject to significant uncertainties and contingencies many of which are beyond your control and (ii) no assurance can be given that any particular financial projections will be realized, and that actual results during the period or periods covered by any such Projections may differ from the projected results, and such differences may be material). You agree that if, at any time prior to the later of the Closing Date and the end of the Syndication Period, you become aware that any of the representations and warranties in the preceding sentence would be (with respect to Information relating to the Company and its subsidiaries and businesses, to your knowledge) incorrect in any material respect if the Information or Projections were being furnished and such representations and warranties were being made at such time, then you will use commercially reasonable efforts to promptly supplement the Information and the Projections so that (with respect to Information relating to the Company and its businesses, to your knowledge) such representations are correct, in all material respects, under those circumstances. The making or accuracy of the foregoing representations and warranties, whether or not cured, shall not be a condition to the obligations of the Commitment Parties. You understand that the Commitment Parties may use and rely on the Information and the Projections without independent verification thereof, and we do not assume responsibility for the accuracy or completeness of the Information or the Projections, including in the case of the Senior Lead Arrangers, in connection with arranging and syndicating the First Lien Term Facility and the ABL Facility.

5. Fees and Closing Payments

As consideration for the commitments and agreements of the Commitment Parties hereunder, you agree to pay or cause to be paid the nonrefundable compensation and closing payments described in the separate fee and closing payment letter dated the date hereof and delivered herewith (the "Fee Letter") on the terms and subject to the conditions expressly set forth therein.

6. Conditions

Each Initial Senior Lender's commitments hereunder are subject solely to the satisfaction (or waiver) of the Exclusive Funding Conditions, and upon the satisfaction (or waiver by the Initial Senior Lenders with respect to the applicable Senior Credit Facility) of the Exclusive Funding Conditions, the initial funding of the Senior Credit Facilities shall occur.

Notwithstanding anything in this Commitment Letter, the Fee Letter, the Senior Credit Documentation or any other letter agreement or other undertaking concerning the financing of the transactions contemplated hereby to the contrary, (a) the only representations the accuracy of which shall be a condition to the availability of the Senior Credit Facilities on the Closing Date shall be (i) such of the representations made by or with respect to the Company in the Merger Agreement as are material to the interests of the Lenders, but only to the extent that you or your applicable affiliates have the right (determined without regard to any notice provisions but taking into account any applicable cure provisions) to terminate your (or their) obligations under the Merger Agreement or decline to consummate the Merger as a result of a breach of such representations in the Merger Agreement (the "Specified Merger Agreement Representations") and (ii) the Specified Representations (as defined below) and (b) the terms of the Senior Credit Documentation shall be in a form such that they do not impair the availability of the Senior Credit Facilities on the Closing Date if the Exclusive Funding Conditions are satisfied (or waived by the Initial Senior Lenders), it being understood that, to the extent any lien search, insurance certificate or endorsement or Collateral (including the creation or perfection of any security interest) is not or cannot be provided on the Closing Date (other than the pledge and perfection of Collateral of the Borrower and the Guarantors with respect to which a lien may be perfected solely by (A) the filing of financing statements under the Uniform Commercial Code ("UCC") and (B) the delivery of stock certificates or other certificates, if any, representing equity interests of the Borrower or the Guarantors required to be pledged pursuant to the Term Sheets to the extent (x) possession of such certificates perfects a security interest therein and (y) other than in the case of stock certificates or other certificates representing equity interests of the Initial Borrower, such stock certificates have been received from the Company after your use of commercially reasonable efforts to do so) after your use of commercially reasonable efforts to do so without undue burden or expense, then the provision and/or perfection, as applicable, of any such lien search, insurance certificate or endorsement, and/or Collateral shall not constitute a condition precedent to the availability of the Senior Credit Facilities, but may instead be provided within ninety (90) days after the Closing Date, subject to such extensions as are reasonably agreed by the First Lien Administrative Agent and the Borrower, or pursuant to arrangements to be mutually agreed by the parties hereto acting reasonably. "Specified Representations" means the representations in the Senior Credit Documentation relating to corporate or other organizational existence, organizational power and authority of the Borrower and the Guarantors (as they relate to due authorization, execution, delivery and performance of the Senior Credit Documentation); due authorization, execution, delivery and enforceability, in each case, relating to the entering into and performance of such Senior Credit Documentation by the Borrower and the Guarantors; solvency as of the Closing Date (after giving effect to the Transactions) of the Borrower and its subsidiaries on a consolidated basis (in form and scope consistent with the solvency certificate to be delivered pursuant to paragraph 1 of Exhibit D hereto); no violations or conflicts with organizational documents (as related to the execution and delivery of the Senior Credit Documentation) of the Borrower and the Guarantors; Federal

Reserve margin regulations; the Investment Company Act; the PATRIOT Act; use of proceeds of the First Lien Term Loans and the ABL Loans not violating OFAC or FCPA; and the creation and perfection of the security interests (subject to customary permitted liens) in the Collateral of the Borrower and the Guarantors and subject in all respects to the foregoing provisions of this paragraph. Notwithstanding anything in this Commitment Letter, the Fee Letter, the Senior Credit Documentation or any other letter agreement or other undertaking concerning the financing of the transactions contemplated hereby to the contrary, (a) the commitments of the Initial Senior Lenders hereunder and the Senior Lead Arrangers' agreements to perform the services described herein are subject (i) as to the First Lien Term Facility, to the conditions expressly set forth under the heading "CERTAIN CONDITIONS—Initial Conditions" in the Term Sheet on Exhibit B, (ii) as to the ABL Facility, to the conditions expressly set forth under the heading "CERTAIN CONDITIONS—Initial Conditions" in the Term Sheet on Exhibit C and (iii) to the conditions set forth in this Section 6 and in Exhibit D hereto (collectively, the "Exclusive Funding Conditions"), (b) the only conditions (express or implied) to the availability of the Senior Credit Facilities on the Closing Date are the Exclusive Funding Conditions and (c) to the extent the representations made on the Closing Date with respect to the Company are qualified or subject to "material adverse effect", the definition thereof shall be "Beta Material Adverse Effect" as defined in the Merger Agreement ("Beta Material Adverse Effect") for purposes of any such representations and warranties with respect to the Company made or to be made on or as of the Closing Date. This paragraph, and the provisions herein, shall be referred to as the "Certain Funds Provision." The Initial Senior Lenders will cooperate with you as reasonably requested in coordinating the timing and procedures for the funding of the Senior

7. <u>Indemnification and Expenses</u>

You agree (a) to indemnify and hold harmless the Commitment Parties, their controlled affiliates (other than Excluded Affiliates) and controlling persons and the respective directors, officers, employees, agents and other representatives of each of the foregoing and their respective successors (excluding any Excluded Affiliate, each, an "indemnified person") from and against any and all actual losses, claims, damages, liabilities and expenses, joint or several, to which any such indemnified person may become subject arising out of or in connection with this Commitment Letter, the Fee Letter, the Transactions or the use of proceeds of the Senior Credit Facilities or any claim, litigation, investigation or proceeding (a "Proceeding") relating to any of the foregoing, regardless of whether any indemnified person is a party thereto, whether or not such Proceedings are brought by you, the Company, your or its equity holders, affiliates, creditors or any other person, and to reimburse each indemnified person within thirty (30) days of written demand (together with reasonable backup documentation) for any reasonable out-of-pocket expenses incurred in connection with investigating or defending any of the foregoing (but limited, in the case of legal fees and expenses, to one counsel to such indemnified persons (taken as a whole) and, if reasonably necessary, one local counsel in any relevant material jurisdiction to such indemnified persons (taken as a whole) and, in the case of an actual conflict of interest, one additional counsel to the affected indemnified persons (taken as a whole), in each case excluding allocated costs of in-house counsel); provided that, the foregoing indemnity will not, as to any indemnified person, apply to losses, claims, damages, liabilities or related expenses to the extent they arise from (i) the willful misconduct, bad faith, fraud or gross negligence of such indemnified person (or its controlled affiliates and controlling persons and their respective

directors, officers, employees, agents and other representatives (other than Excluded Affiliates)), (ii) the material breach of the Commitment Letter or Fee Letter by any indemnified person (or its controlled affiliates and controlling persons and their respective directors, officers, employees, agents and other representatives (other than Excluded Affiliates)) (in the case of each of preceding clause (i) and this clause (ii), as determined by a court of competent jurisdiction in a final non-appealable judgment) and (iii) any disputes solely among indemnified persons (other than (x) any claims against the Commitment Parties in their capacity as the Administrative Agent, a Senior Lead Arranger or any similar role under the Senior Credit Facilities unless such claim would otherwise be excluded pursuant to clause (i) above and (y) claims arising out of any act or omission of you or the Company) and (b) if the Closing Date occurs, to reimburse the Commitment Parties and their affiliates (other than any Excluded Affiliate) for all reasonable and documented out-of-pocket expenses (including, but not limited to, due diligence expenses, expenses related to field examinations or appraisals, syndication expenses, travel expenses, and (limited to) reasonable fees, charges and disbursements of (x) one primary counsel to the Senior Lead Arrangers and the Administrative Agent in their capacity as such (which, for the avoidance of doubt, shall not include any fees, charges or disbursements of counsel while acting for any Sale Advisory Person (as defined below)), (y) Milbank LLP (as counsel to the MBD Lenders) and (z) Proskauer Rose LLP (as counsel to Ares) solely from the date hereof to the Closing Date and, if reasonably necessary, one local counsel in any relevant material jurisdiction incurred in connection with the Senior Credit Facilities and any related documentation (including this Commitment Letter, the Fee Letter and the Senior Credit Documentation) or the administration, amendment, modification or waiver of any of the foregoing) within thirty (30) days of written demand (including documentation reasonably supporting such request) (other than with respect to such fees and expenses paid on the Closing Date for which written demand including documentation reasonably supporting such request is provided at least three (3) business days prior to the Closing Date); provided that, such fees and expenses (i) in the case of legal counsel, shall be limited to the reasonable fees and expenses of counsel described in this clause (b) which, in any event, shall exclude allocated costs of inhouse counsel and (ii) in the case of any other advisors and consultants, shall be limited solely to advisors and consultants approved by you. No person a party hereto nor the Company or any indemnified person shall be liable for any damages arising from the use by others of Information or other materials obtained through electronic, telecommunications or other information transmission systems, including, without limitation, SyndTrak, IntraLinks, the internet, email or similar electronic transmission systems, in each case, except to the extent any such damages are found in a final non-appealable judgment of a court of competent jurisdiction to have resulted from the gross negligence, bad faith, fraud or willful misconduct of, or material breach of this Commitment Letter or the Fee Letter by, such person (or its controlled affiliates and controlling persons and their respective directors, officers, employees, agents and other representative (other than Excluded Affiliates)). None of the indemnified persons or you, the Sponsor, Walgreens, the Company or any of your or their respective affiliates or the respective directors, officers, employees, advisors, agents and other representatives of the foregoing shall be liable for any indirect, special, punitive or consequential damages in connection with this Commitment Letter, the Fee Letter, the Senior Credit Facilities or the transactions contemplated hereby; provided that, nothing contained in this sentence shall limit your indemnification and reimbursement obligations to the extent expressly set forth herein in respect of any losses, claims, damages, liabilities and expenses incurred or paid by an indemnified person to a third party unaffiliated with the Commitment Parties. Each indemnified

person (by accepting the benefits hereof) agrees to refund and return any and all amounts paid by you to such indemnified person pursuant to the terms of this paragraph to the extent such indemnified person is not entitled to the payment thereof pursuant to the terms of this paragraph as determined by a court of competent jurisdiction in a final non-appealable judgment. You acknowledge that we may receive a benefit, including without limitation, a discount, credit or other accommodation, from any of such counsel based on the fees such counsel may receive on account of their relationship with us including, without limitation, fees paid pursuant hereto.

You shall not be liable for any settlement of any Proceeding (or expenses solely in respect of such settlement) effected without your consent (which consent shall not be unreasonably withheld, delayed or conditioned), but if settled with your written consent, or if there is a final judgment against an indemnified person in any such Proceeding, you agree to indemnify and hold harmless each indemnified person to the extent and in the manner set forth above. You shall not, without the prior written consent of the affected indemnified person (which consent shall not be unreasonably withheld, conditioned or delayed), effect any settlement of any pending or threatened Proceeding against such indemnified person in respect of which indemnity could have been sought hereunder by such indemnified person unless such settlement (a) includes an unconditional release of such indemnified person from all liability or claims that are the subject matter of such Proceeding, (b) includes customary confidentiality provisions and (c) does not include any statement as to any admission of fault, culpability or failure to act by or on behalf of any indemnified person.

In case any Proceeding is instituted involving any indemnified person for which indemnification is to be sought hereunder by such indemnified person, then such indemnified person will promptly notify you of the commencement of any Proceeding after such indemnified person has actual knowledge of the same; <u>provided</u>, <u>however</u>, that the failure so to notify you will not relieve you from any liability that you may have to such indemnified person pursuant to this <u>Section 7</u>.

Notwithstanding anything to the contrary contained herein, upon the execution of the First Lien Credit Documentation and the ABL Credit Documentation, as applicable, (a) the relevant provisions of such definitive documentation shall supersede the provisions of the preceding paragraphs and (b) your obligation pursuant to this Commitment Letter to reimburse an indemnified person (or its related indemnified persons) for losses, claims, damages, liabilities, expenses, fees or any such indemnified obligations or any other expense reimbursement shall automatically terminate and be replaced in all respects by the relevant provisions set forth in the Senior Credit Documentation.

8. <u>Sharing of Information, Absence of Fiduciary Relationship, Affiliate Activities</u>

You acknowledge that the Commitment Parties (other than the MBD Lenders) and their respective affiliates are, or, in the case of the MBD Lenders and their respective affiliates, are affiliated with, full service securities firms and we may from time to time (a) effect transactions, for our own or our affiliates' account or the account of customers, and hold positions in loans, securities or options on loans or securities of you, the Company or its affiliates and of other companies that may be the subject of the transactions contemplated by this Commitment Letter or with which you or the Company may have commercial or other relationships or adverse

interests or (b) provide debt financing, equity capital, investment banking, financial advisory services, securities trading, hedging, financing and brokerage activities and financial planning and benefits counseling to other companies in respect of which you may have conflicting interests. In addition, consistent with each Commitment Party's policy to hold in confidence the affairs of its customers, the Commitment Parties will not furnish information obtained from you, the Sponsor, the Company or your or their respective affiliates and representatives to any of their other clients (or to clients of its affiliates) or in connection with the performance by the First Lien Lead Arrangers and the First Lien Administrative Agent and their respective affiliates of services for its other clients (or for clients of their affiliates). You also acknowledge that the Commitment Parties and their affiliates have no obligation to use in connection with the transactions contemplated hereby, or to furnish to you, confidential information obtained from other companies or other persons.

You further acknowledge and agree that (a) you are capable of evaluating and understanding, and you understand and accept, the terms, risks and conditions of the transactions contemplated by this Commitment Letter, (b) you have been advised that the Commitment Parties and their affiliates are engaged in a broad range of transactions that may involve interests that differ from your and your affiliates' interests and that the Commitment Parties have no obligation to disclose such interests and transactions to you or your affiliates, (c) you have consulted your own legal, accounting, regulatory and tax advisors to the extent you have deemed appropriate and you are not relying on any Commitment Party for such advice and (d) none of the Commitment Parties nor their affiliates have any obligation to you or your affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein or in any other express writing executed and delivered by such Commitment Party and you.

In addition, please note that an affiliate of the MBD Lenders and MLPFS and/or its affiliates have been retained by you as financial advisors (in such capacity, each a "<u>Financial Advisor</u>") in connection with the Merger. You agree to such retention, and further agree not to assert any claim you might allege based on any actual or potential conflicts of interest that might be asserted to arise or result primarily from, on the one hand, the engagement of each Financial Advisor, and on the other hand, our and our affiliates' relationship with you as described and referred to herein.

Please note that neither the Commitment Parties nor any of their respective affiliates provide tax, accounting or legal advice.

You further acknowledge and agree that (a) no fiduciary, advisory or agency relationship between you and us is intended to be or has been created in respect of any of the transactions contemplated by this Commitment Letter, irrespective of whether we or our affiliates have advised or are advising you on other matters, (b) we, on the one hand, and you, on the other hand, have an arms-length business relationship that does not directly or indirectly give rise to, nor do you rely on, any fiduciary duty on our part, (c) in connection therewith and with the process leading to the Transactions, the Commitment Parties and their respective affiliates (as the case may be) are acting solely as a principal and not as agents or fiduciaries of you, the Sponsor and your and their respective management, stockholders, creditors, affiliates or any other person, (d) you are capable of evaluating and understanding, and you understand and accept, the terms

risks and conditions of the transactions contemplated by this Commitment Letter, (e) you have consulted legal and financial advisors to the extent you deemed appropriate, (f) you have been advised that we and our affiliates are engaged in a broad range of transactions that may involve interests that differ from your interests and that we and our affiliates have no obligation to disclose such interests and transactions to you by virtue of any fiduciary, advisory or agency relationship and (g) you will not claim that any Commitment Party (in its capacity as such) or its applicable affiliates, as the case may be, have rendered advisory services of any nature or respect, or owe a fiduciary or similar duty to you or your affiliates, in connection with the transactions contemplated by this Commitment Letter or the process leading thereto.

9. <u>Confidentiality</u>

This Commitment Letter is delivered to you on the understanding that neither this Commitment Letter nor the Fee Letter nor any of their terms shall be disclosed to any other person, except (a) to the Sponsor, Walgreens and your and their respective officers, directors, employees, affiliates, members, partners, stockholders, actual and potential co-investors, attorneys, accountants, agents and advisors on a confidential basis, (b) to the Company and its officers, directors, employees, affiliates, members, partners, stockholders, attorneys, accountants, agents and advisors on a confidential basis, (c) in any legal, regulatory, judicial or administrative proceeding or as otherwise required by applicable law, rule or regulation or as requested by a governmental authority (including a self-regulatory authority) (in which case you agree, to the extent permitted by law, rule or regulation, to inform us promptly thereof), (d) in connection with the exercise of any remedy or enforcement of any right under this Commitment Letter and the Fee Letter, (e) the existence and contents of the Commitment Letter and the Term Sheets may be disclosed in any proxy, public filing, prospectus, offering memorandum, offering circulation, syndication materials or other materials in connection with the Transactions, (f) the Term Sheets and Exhibit D hereto (but not the Fee Letter or the contents thereof other than the existence thereof and the contents thereof as part of projections, pro forma information and a generic disclosure of aggregate sources and uses) may be disclosed to actual and prospective Lenders and to any rating agency in connection with the Transactions or in any public regulatory filing requirement relating to the Transactions. (g) to the extent any such information becomes publicly available other than by reason of disclosure by you, your controlled affiliates or your representatives in violation of this Commitment Letter, (h) with the Commitment Parties' consent in writing (including via e-mail), (i) to the Initial Second Lien Note Purchasers (as defined below) and their respective officers, directors, employees, affiliates (other than Excluded Affiliates), investors, lenders, limited partners and their respective attorneys, consultants and advisors who, in each case, are involved in the investment of the Second Lien Notes, provided that each of the foregoing persons are bound to maintain the confidentiality hereof and of the Fee Letter on terms substantially similar to those contained herein and (j) after your acceptance hereof, this Commitment Letter and the Fee Letter, including the existence and contents hereof and thereof, may be shared with prospective Additional Agents, in each case on a confidential basis. The foregoing restrictions shall cease to apply after the Senior Credit Documentation shall have been executed and delivered by the parties hereto (other than with respect to any economics (including the Flex Provisions) referenced in the Fee Letter).

The Commitment Parties shall treat confidentially all information received by them from you, the Company, Walgreens, the Sponsor or your or their respective affiliates and

representatives in connection with the Merger and the other Transactions and only use such information for the purposes of providing the commitments and, in the case of the Senior Lead Arrangers and the Administrative Agent, the services contemplated by this Commitment Letter; provided, however, that nothing herein shall prevent the Commitment Parties from disclosing any such information (a) to rating agencies in connection with obtaining ratings for the First Lien Term Facility subject to your prior approval of information to be so disclosed; (b) to any actual or prospective Lenders, participants or derivative counterparties (other than Disqualified Institutions and persons to whom you have affirmatively denied to provide your consent to the assignment or syndication thereto) or, in the case of MBD Lenders, actual or prospective Other MBD Lenders; provided that, the disclosure of any such information to any actual or prospective Lenders, participants or derivative counterparties or, in the case of MBD Lenders, actual or prospective Other MBD Lenders referred to above shall be made subject to the acknowledgment and acceptance by such actual or prospective Lender, participant or derivative counterparty or, in the case of MBD Lenders, actual or prospective Other MBD Lenders that such information is being disseminated on a confidential basis (on substantially the terms set forth in this paragraph or as is otherwise reasonably acceptable to you and the Commitment Parties, including, without limitation, as agreed in any confidential information memorandum or other marketing materials) in accordance with the standard syndication or other transfer processes of the Commitment Parties or customary market standards for dissemination of such type of information, in the event of any electronic access through IntraLinks, another website or similar electronic system or platform, which shall in any event require "click through" or other affirmative action on the part of the recipient to access such information and acknowledge its confidentiality obligations in respect thereof, in each case on terms reasonably acceptable to you; (c) in the case of any MBD Lender or Other MBD Lender, to a rating agency that is involved in the administration or monitoring of such MBD Lender's or Other MBD Lender's investment in the First Lien Credit Facilities without such prior approval); (d) in any legal, judicial or administrative proceeding or other compulsory process or otherwise as required by applicable law, rule or regulations or as requested by a governmental authority (in which case such Commitment Party shall promptly notify you, in advance, to the extent permitted by law, rule or regulation (except with respect to any routine audit or examination conducted by bank accountants or regulatory authority exercising routine examination or regulatory authority)); provided that, notwithstanding the foregoing, no notification shall be required in connection with any filings that may be required by the federal securities laws or the rules, regulations, schedules and/or forms of the Securities and Exchange Commission; (e) upon the request or demand of any governmental or regulatory authority having jurisdiction over such Commitment Party or any of its affiliates or upon the good faith determination by counsel that such information should be disclosed in light of ongoing oversight or review of such Commitment Party by any governmental or regulatory authority having jurisdiction over such Commitment Party or its affiliates (in which case such Commitment Party shall, except with respect to any audit or examination conducted by bank accountants or any governmental bank regulatory authority exercising examination or regulatory authority, promptly notify you, in advance, to the extent lawfully permitted to do so); (f) to the limited partners, lenders, investors, managed accounts, officers, directors, employees, legal counsel, independent auditors, professionals, service providers and other experts or agents of the Commitment Parties (collectively, "Representatives") (provided that, any such Representative is advised of its obligation to retain such information as confidential and agrees to keep information of this type confidential and the Commitment Parties shall remain responsible for such

Representatives compliance with this provision); (g) to any of its affiliates (other than an Excluded Affiliate) and Representatives of its affiliates (other than an Excluded Affiliate) (provided that, any such affiliate or Representative is advised of its obligation to retain such information as confidential, and the Commitment Parties shall be responsible for the compliance of their affiliates and Representatives of their affiliates with this paragraph) solely in connection with the Senior Credit Facilities and the related Transactions (including in connection with the evaluation, monitoring or administration of any MBD Lender's or Other MBD Lender's investment in the Senior Credit Facilities); (h) to the extent any such information becomes publicly available other than by reason of disclosure by the Commitment Parties, their affiliates or Representatives in breach of this Commitment Letter or other obligation of confidentiality owed to you, the Sponsor, Walgreens, the Company or your or their respective affiliates; (i) for purposes of establishing a "due diligence" defense; (j) to the extent that such information is received by such Commitment Party, its affiliates (other than an Excluded Affiliate) or their respective Representatives from a third party that is not known (after due inquiry) by such Commitment Party to be subject to confidentiality obligations to you, the Sponsor, Walgreens, the Company or your or their respective affiliates; (k) to the extent that such information is independently developed by such Commitment Party, its affiliates (other than an Excluded Affiliate) or their respective Representatives without the use of such information; (1) to market data collectors, similar services providers to the lending industry, and service providers to the Senior Lead Arrangers and the Lenders in connection with the administration and management of the Senior Credit Facilities; and (m) to enforce their respective rights hereunder or under the Fee Letter; provided, however, that, no such disclosure shall be made by (i) the Senior Lead Arrangers or the Administrative Agent to any of their affiliates that is engaged as principals primarily in private equity, mezzanine financing or venture capital or any of such affiliate's officers, directors, employees, legal counsel, independent auditors, professionals and other experts or agents other than, in each case, any Over the Wall Person (as defined below) (each of such affiliates and other related persons of a Senior Lead Arranger or the Administrative Agent described in this clause (i), a "Private Equity Affiliate") or (ii) by the Commitment Parties to any of their affiliates' employees, officers, directors, legal counsel, professionals and other experts or agents that are engaged directly or indirectly in a sale of the Company and its subsidiaries as buy-side or sell-side representative and acting in such capacity other than, in each case, any Over the Wall Person (each a "Sale Advisory Person" and, together with the Private Equity Affiliates, the "Excluded Affiliates") or (iii) any Disqualified Institution. As used herein, "Over the Wall Person" means, with respect to any Commitment Party, any directors, officers or senior employees of such Commitment Party or any of its affiliates who are required, in accordance with industry regulations, or the applicable Commitment Party's or its affiliate's internal policies and procedures to act in a supervisory or managerial capacity and the applicable Commitment Party's and its affiliates' internal legal, compliance, risk management, conflicts clearance and other support personnel and credit and investment committee members. The Commitment Parties' obligations under this paragraph shall remain in effect until the earlier of (x) two years from the date hereof and (y) the date the Senior Credit Documentation (which confidentiality provisions shall include exceptions no less permissive than those contained in this Section) is effective, at which time our obligations under this paragraph shall automatically terminate and be superseded by the confidentiality provisions in the Senior Credit Documentation upon the execution and delivery thereof. Notwithstanding anything to the contrary contained herein, nothing in this paragraph shall prohibit any MBD Lender from disclosing any information to any

lender to, or managed account or limited partner of, such MBD Lender to the extent such information is subject to customary confidentiality obligations binding on such lender, managed account or limited partner pursuant to customary investment advisory, fund or loan documentation.

Notwithstanding anything herein to the contrary, any party to this Commitment Letter (and any employee, representative or other agent of such party) may disclose the tax treatment and tax structure of the transactions contemplated by this Commitment Letter and all materials of any kind (including opinions or other tax analyses) that are provided to it relating to such tax treatment and tax structure, except that (i) such disclosures shall not include the identity of any existing or future party (or any affiliate of such party) to this Commitment Letter and (ii) no party shall disclose any information relating to such tax treatment and tax structure to the extent nondisclosure is reasonably necessary in order to comply with applicable securities laws. For this purpose, the tax treatment of the transactions contemplated by this Commitment Letter shall be any purported or claimed U.S. federal income tax treatment of such transactions and the tax structure of such transactions is any fact that may be relevant to understanding the purported or claimed U.S. federal income tax treatment of such transactions.

10. Miscellaneous

Subject to the immediately following paragraph, this Commitment Letter shall not be assignable by any party hereto (except (x) by you to the ultimate Borrower or one or more of your affiliates that is a newly formed domestic "shell" company controlled, directly or indirectly, by you to effect the consummation of the Merger, (y) by us in connection with the appointment of Additional Agents by you as set forth herein or (z) MLPFS may, without notice to the parties hereto, assign its rights and obligations under this Commitment Letter and the Fee Letter to any other registered broker-dealer wholly-owned by Bank of America Corporation to which all or substantially all of Bank of America Corporation's or any of its subsidiaries' investment banking, commercial lending services or related businesses may be transferred following the date of this Commitment Letter) without the prior written consent of the other parties hereto (and any purported assignment without such consent shall be null and void), is intended to be solely for the benefit of the parties hereto and, to the extent set forth in Section 7, the indemnified persons, and is not intended to and does not confer any benefits upon, or create any rights in favor of, any person other than the parties hereto and the indemnified persons to the extent expressly set forth herein. Each Commitment Party shall be liable solely in respect of its own commitment to the Senior Credit Facilities on a several, and not joint, basis with any other Lender. This Commitment Letter may not be amended or waived except by an instrument in writing signed by you and the Commitment Parties. This Commitment Letter may be executed in any number of counterparts, each of which shall be an original, and all of which, when taken together, shall constitute one agreement. Delivery of an executed signature page of this Commitment Letter by facsimile or other electronic transmission (e.g., "pdf" or "tif") shall be effective as delivery of a manually executed counterpart hereof. This Commitment Letter and the Fee Letter are the only agreements that have been entered into among us and you with respect to the Senior Credit Facilities and set forth the entire understanding of the parties with respect thereto. This Commitment Letter shall be governed by, and construed and interpreted in accordance with, the laws of the State of New York without regard to principles of conflicts of law, to the extent that the same are not mandatorily applicable by statute and would require or permit the application of

the law of another jurisdiction; provided, however, that the laws of the state of Delaware which governs the Merger Agreement shall govern in determining (a) the interpretation of a "Beta Material Adverse Effect" and whether a "Beta Material Adverse Effect" has occurred, (b) the accuracy of any Specified Merger Agreement Representation and whether as a result of any inaccuracy thereof you or your applicable affiliate have the right or would have the right (without regard to any notice requirement but taking into account any applicable cure provisions) to terminate your obligations (or to refuse to consummate the Merger) under the Merger Agreement and (c) whether the Merger has been consummated in accordance with the terms of the Merger Agreement. Each of the parties hereto agrees that (i) this Commitment Letter is a binding and enforceable agreement with respect to the subject matter herein, including an agreement to negotiate in good faith the Senior Credit Documentation by the parties hereto in a manner consistent with this Commitment Letter and the applicable Documentation Principles it being acknowledged and agreed that the funding of the Senior Credit Facilities is subject only to the Exclusive Funding Conditions, including the execution and delivery of the Senior Credit Documentation as provided in this Commitment Letter and (ii) the Fee Letter is a binding and enforceable agreement with respect to the subject matter contained therein. Reasonably promptly following the execution of this Commitment Letter, the parties hereto shall proceed with the negotiation in good faith of the Senior Credit Documentation for the purpose of executing and delivering the Senior Credit Documentation substantially simultaneously with the consummation of the Merger. Section headings used herein are for convenience of reference only and are not to affect the construction of, or to be taken into consideration in interpreting, this Commitment Letter.

Notwithstanding anything in this Commitment Letter or in the Fee Letter, each Initial MBD Lender shall have the right to reallocate, sell, resell, assign and/or transfer its respective commitments and/or any closing payment (or any portion thereof) (a) to any other Initial MBD Lender or (b) to any affiliated investment entity and/or other affiliate of Goldman Sachs & Co. LLC or any fund, investor, entity or account that is managed, sponsored or advised by Goldman Sachs & Co. LLC or its affiliates, any such persons described in this clause (b) the "Other MBD Lenders"); provided that, unless you agree in your sole discretion in writing in connection therewith, in each case (i) no such Initial MBD Lender shall be relieved or novated from its respective obligation hereunder with respect to its commitments in connection with any applicable assignment or transfer until the occurrence of the funding of the Initial First Lien Term Loans on the Closing Date, (ii) no assignment or novation shall become effective with respect to all or any portion of any such Initial MBD Lender's commitment in respect of the Initial First Lien Term Loans until the occurrence of the funding of the First Lien Term Loans on the Closing Date and (iii) the Initial MBD Lenders shall retain exclusive control over all rights and obligations with respect to their commitments in respect of the Initial First Lien Term Loans, including all rights with respect to consents, modifications, supplements and amendments, until the occurrence of the funding of the Initial First Lien Term Loans on the Closing Date.

NOTWITHSTANDING ANYTHING CONTAINED HEREIN OR IN THE FEE LETTER TO THE CONTRARY, NO MBD LENDER NOR ARES IS ACTING AS AN UNDERWRITER, ARRANGER, TRUSTEE, AGENT OR IN A SIMILAR ROLE OR OTHERWISE PERFORM ANY SERVICES HEREUNDER AND THE ROLE OF THE MBD LENDERS AND ARES HEREUNDER AND UNDER THE FEE LETTER SHALL BE LIMITED TO THEIR COMMITMENT TO PROVIDE DEBT FINANCING AS A PRINCIPAL.

Each of the parties hereto irrevocably and unconditionally (a) submits, for itself and its property, to the exclusive jurisdiction of any federal court sitting in the Borough of Manhattan in the City of New York or, if that court does not have subject matter jurisdiction, in any state court located in the City and County of New York, and any appellate court from any thereof, over any suit, action or proceeding arising out of or relating to the Transactions or the other transactions contemplated hereby, this Commitment Letter or the Fee Letter or the performance of services hereunder or thereunder or for recognition or enforcement of any judgment and agrees that all claims in respect of any such action or proceeding shall be heard and determined in such New York state or, to the extent permitted by law, in such federal court; provided, however, that the Commitment Parties shall be entitled to assert jurisdiction over you and your property in any court in which jurisdiction may be held over you or your property and (b) agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. You and we agree that service of any process, summons, notice or document by registered mail addressed to any of the parties hereto at the applicable addresses above shall be effective service of process for any suit, action or proceeding brought in any such court. You and we hereby irrevocably and unconditionally waive, to the fullest extent you and we may legally and effectively do so, any objection to the laying of venue of any such suit, action or proceeding brought in any court in accordance with clause (a) of the first sentence of this paragraph and any claim that any such suit, action or proceeding has been brought in any inconvenient forum. YOU AND WE HEREBY IRREVOCABLY WAIVE (TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW) TRIAL BY JURY IN ANY SUIT, ACTION, PROCEEDING, CLAIM OR COUNTERCLAIM BROUGHT BY OR ON BEHALF OF ANY PARTY RELATED TO OR ARISING OUT OF THE TRANSACTIONS, THIS COMMITMENT LETTER OR THE FEE LETTER OR, IN THE CASE OF SENIOR LEAD ARRANGERS AND THE ADMINISTRATIVE AGENT, THE PERFORMANCE OF SERVICES HEREUNDER OR THEREUNDER.

The Commitment Parties hereby notify you that, pursuant to the requirements of the USA PATRIOT Act, Title III of Pub. L. 107-56 (signed into law on October 26, 2001) (as amended, the "PATRIOT Act"), they are required to obtain, verify and record information that identifies the Borrower and each Guarantor, which information includes names, addresses, tax identification numbers and other information that will allow the Commitment Parties or such Lender to identify the Borrower and each Guarantor in accordance with the PATRIOT Act. This notice is given in accordance with the requirements of the PATRIOT Act and is effective for the Commitment Parties and each Lender.

This paragraph, the indemnification, jurisdiction, waiver of jury trial, service of process, venue, governing law, sharing of information, no agency or fiduciary duty and confidentiality provisions contained herein and in the Fee Letter shall remain in full force and effect regardless of whether definitive financing documentation shall be executed and delivered and notwithstanding the termination of this Commitment Letter or the commitments hereunder; provided that, your obligations under this Commitment Letter (but not the Fee Letter) (and other than in respect of your obligations in respect of syndication assistance, your agreements in

respect of no fiduciary or similar duties and your obligations in respect confidentiality (which shall terminate in accordance with Section 9)) shall automatically terminate and be superseded by the provisions of the Senior Credit Documentation upon the execution thereof, and you shall automatically be released from all liability in connection therewith at such time. For the avoidance of doubt, the syndication and information provisions set forth in Section 3 and the second sentence of Section 4 hereof shall survive until the later of (a) the Syndication Period and (b) the Closing Date. You may terminate the Commitment Parties' commitments (or a portion thereof) hereunder at any time subject to the provisions of the preceding sentence; *provided* that in the case of any termination of the commitments under any Senior Credit Facility in part, such termination is applied within such Senior Credit Facility to reduce the commitments of the applicable Commitment Parties thereunder on a pro rata basis.

If the foregoing correctly sets forth our agreement, please indicate your acceptance of the terms of this Commitment Letter and the Fee Letter by returning to us executed counterparts of this Commitment Letter and of the Fee Letter not later than 11:59 p.m., New York time, on March 15, 2019. This offer will automatically expire at such time if we have not received such executed counterparts in accordance with the preceding sentence, unless we shall, in our sole discretion, agree to an extension. Unless we shall, in our sole discretion, agree to an extension, this Commitment Letter and the commitments hereunder shall automatically terminate in the event that (a) in respect of the First Lien Term Facility, if the initial borrowing thereunder does not occur on or before 11:59 p.m., New York City time, on December 20, 2019 (the "Expiration Time"), (b) in respect of the ABL Facility, if the initial borrowing thereunder or effectiveness thereof does not occur on or before the Expiration Time, (c) the Merger closes with or without the use of the Senior Credit Facilities or (d) after execution of the Merger Agreement and prior to the consummation of the Merger, the termination of the Merger Agreement by you (or your affiliates) or with your (or your affiliates') written consent (other than with respect to provisions therein that expressly survive termination); provided that, the termination of any commitment pursuant to this sentence does not prejudice your or our rights and remedies in respect of any breach of this Commitment Letter.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

We are pleased to have been given the opportunity to assist you in connection with this important financing.

Very truly yours,

BANK OF AMERICA, N.A.

By: /s/ Jae Lee Name: Jae Lee Title: Director

MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED

By: /s/ Jae Lee Name: Jae Lee Title: Director

We are pleased to have been given the opportunity to assist you in connection with this important financing.

Very truly yours,

ARES CAPITAL MANAGEMENT LLC, on behalf of one or more affiliated funds, investment vehicles and/or managed accounts,

By: /s/ Mark Affolter Name: Mark Affolter Title: Authorized Signatory

We are pleased to have been given the opportunity to assist you in connection with this important financing.

BROAD STREET LOAN PARTNERS III, L.P.

By: Goldman Sachs & Co. LLC, as Attorney-in-Fact

By: /s/ Kirsten Anthony Name: Kirsten Anthony Title: Managing Director

BROAD STREET LOAN PARTNERS III OFFSHORE, L.P.

By: Goldman Sachs & Co. LLC, as Collateral Servicer and Duly Authorized Agent $\,$

By: /s/ Kirsten Anthony Name: Kirsten Anthony

Title: Managing Director

BROAD STREET LOAN PARTNERS III OFFSHORE — UNLEVERED, L.P.

By: Goldman Sachs & Co. LLC, as Collateral Servicer and Duly Authorized Agent $\,$

By: /s/ Kirsten Anthony Name: Kirsten Anthony Title: Managing Director

BROAD STREET SENIOR CREDIT PARTNERS II, L.P.

By: Goldman Sachs & Co. LLC, as Attorney-in-Fact

By: /s/ Kirsten Anthony Name: Kirsten Anthony Title: Managing Director

Accepted and agreed to as of the date first above written:

HC GROUP HOLDINGS II, INC.

By: /s/ Clifford Berman Name: Clifford Berman Title: Corporate Secretary

Schedule I

Initial MBD Lender Commitments

First Lien Term Facility

MBD Lender	First Lien Term Facility
Broad Street Loan Partners III, L.P.	\$ 12,872,915.93
Broad Street Loan Partners III Offshore, L.P.	\$ 67,951,167.70
Broad Street Loan Partners III Offshore-Unlevered, L.P.	\$ 36,281,298.40
Broad Street Senior Credit Partners II, L.P.	\$ 44,769,617.97
TOTAL	\$ 161,875,000

PROJECT OPAL

Senior Credit Facilities Transaction Summary

Capitalized terms used but not defined in this $\underline{\text{Exhibit A}}$ shall have the meanings set forth in the letter to which this $\underline{\text{Exhibit A}}$ is attached or in $\underline{\text{Exhibit B}}$, \underline{C} or \underline{D} thereto.

HC Group Holdings II, Inc., a Delaware corporation ("Omega"), which is indirectly controlled by Madison Dearborn Partners, LLC ("MDP") and its controlled affiliates and associated funds (together with MDP, collectively, the "Sponsor") and Walgreens, intends to consummate a stock-for-stock exchange, directly or indirectly, with a company previously identified to the Commitment Parties as "Beta" (the "Company"), pursuant to the Merger Agreement defined below. In connection therewith, it is intended that:

- (a) The Company will form Beta Sub, Inc., a Delaware corporation, a direct or indirect wholly-owned domestic subsidiary ("Merger Sub 2") and Beta Sub, LLC, a Delaware limited liability company, a direct or indirect wholly-owned domestic subsidiary ("Merger Sub 2"). On the Closing Date, pursuant to that certain Agreement and Plan of Merger, dated as of the date hereof (together with the exhibits and disclosure schedules thereto, as amended, modified, supplemented or waived, the "Merger Agreement") (A) Merger Sub 1 will merge with and into Omega with Omega as the surviving entity and (B) Omega shall merge with and into Merger Sub 2, with Merger Sub 2 surviving such merger (such mergers collectively referred to herein as the "Merger"). After giving effect to Merger, the Company will own the businesses currently conducted by each of Omega and Beta. Immediately after the Merger, Merger Sub 2 (in such capacity, the "Initial Borrower") will obtain Senior Credit Facilities and issue the Second Lien Notes as described below and immediately thereafter the Company shall assume the obligations of the Initial Borrower as the borrower under the Senior Credit Facilities (the "Debt Assumption").
- (b) The Initial Borrower will obtain senior secured first lien term loan facility in an aggregate principal amount of up to \$925 million, subject to increase pursuant to the Flex Provisions set forth in the Fee Letter and as otherwise provided in Exhibit B hereto, (the "First Lien Term Facility"), as described in Exhibit B to the Commitment Letter.
- (c) The Initial Borrower (and certain of its subsidiaries) will obtain a senior secured asset-based revolving credit facility in an aggregate principal amount of up to \$150 million (the "ABL Facility" and, together with the First Lien Term Loan Facility, collectively, the "Senior Credit Facilities" and each a "Senior Credit Facility"), as described in Exhibit C to the Commitment Letter.
- (d) The Initial Borrower will issue and sell, via a private placement, to the Initial Principal Investors (as defined in the Second Lien Notes Term Sheet) \$400 million in aggregate principal amount of senior secured second lien PIK toggle floating rate notes (the "Second Lien Notes"; and the definitive documentation relating to the Second Lien

Notes, the "Second Lien Notes Documentation"), as described in Exhibit B ("Second Lien Notes Term Sheet") to the commitment letter dated as of the date hereof, among you and the Initial Principal Investors (such letter, as in effect on the date hereof, the "Second Lien Commitment Letter")).

- (e) (A) All existing indebtedness for borrowed money of (I) the Company and its subsidiaries pursuant to: (i) that certain First Lien Note Purchase Agreement, dated as of June 29, 2017, among Beta, as issuer, the purchasers party thereto from time to time and Wells Fargo Bank, National Association, as collateral agent, (ii) that certain Second Lien Note Purchase Agreement, dated as of June 29, 2017, among Beta, as issuer, the purchasers party thereto from time to time and Wells Fargo Bank, National Association, as collateral agent and (iii) that certain Indenture, dated February 11, 2014, by and among Beta, the guarantors named therein and U.S. Bank National Association, as trustee and (II) Omega pursuant to (i) that certain Credit Agreement, dated as of April 7, 2015, by and among HC Group Holdings III, Inc., as borrower, the lenders party thereto, Bank of America, N.A., as administrative agent and the other parties thereto and (ii) that certain Indenture, dated as of April 7, 2015, among HC Group Holdings III, Inc., as issuer, the guarantors party thereto and U.S. Bank National Association, as trustee and collateral agent, will, in each case be repaid or satisfied or discharged, and all related guaranties and security interests with respect thereto will be terminated and released simultaneously concurrently with the initial funding of the Senior Credit Facilities (or arrangements for such termination and release shall have been made) and (B) all outstanding preferred stock issued by the Company will be redeemed for cash (the "Preferred Redemption Cash") and/or converted into common stock of the Company (collectively, the "Refinancing").
- (f) The proceeds of a portion of the ABL Facility (limited as set forth herein), the First Lien Term Facility, the Second Lien Notes and cash on hand at Omega and its subsidiaries and the Company and its subsidiaries on the Closing Date will be applied to consummate the Merger, to effect the Refinancing and to pay the fees, closing payments, premiums, expenses and other transaction costs incurred in connection with the foregoing (the "Transaction Costs").

The transactions described above are collectively referred to herein as the "<u>Transactions</u>." For purposes of the Commitment Letter and the Fee Letter, "<u>Closing Date</u>" shall mean the date of (i) the satisfaction or waiver of the Exclusive Funding Conditions, (ii) the funding of the Senior Credit Facilities, (iii) the issuance, sale and purchase of the Initial Second Lien Notes, (iv) the consummation of the Refinancing and (v) the consummation of the Merger.

PROJECT OPAL

<u>First Lien Term Facility</u> <u>Summary of Terms and Conditions</u>

Set forth below is a summary of the principal terms and conditions for the First Lien Term Facility. Capitalized terms used but not defined in this Exhibit B shall have the meanings set forth in the letter to which this Exhibit B is attached or in Exhibits A, C or D attached thereto. Certain of the terms below are subject to adjustment pursuant to the Flex Provisions contained in the Fee Letter (whether or not such adjustment is referenced below).

Borrower: Initially, the Initial Borrower and immediately following consummation of the Debt Assumption, the Company (such

relevant entity, the "Borrower"); provided that, the Borrower may, in its sole discretion, designate one or more of its direct or indirect domestic wholly-owned subsidiaries as co-borrowers subject to customary conditions including that

such co-borrower be treated as a corporation for U.S. federal income tax purposes.

First Lien Administrative Agent: Bank of America, N.A. ("Bank of America") will act as sole and exclusive administrative agent and collateral agent (in

such capacity, the "First Lien Administrative Agent") for the MBD Lenders (to be defined in a manner consistent with the definition in the body of the Commitment Letter) and a syndicate of banks, financial institutions and institutional lenders reasonably acceptable to the Borrower (such consent not to be unreasonably withheld, delayed or conditioned) (excluding any Disqualified Institutions) (the "First Lien Lenders"), and will perform the duties customarily associated

with such roles.

First Lien Lead Arrangers: Merrill Lynch, Pierce, Fenner & Smith Incorporated ("MLPFS") and each other First Lien Lead Arranger appointed

pursuant to the Commitment Letter will act as joint lead arrangers for the First Lien Term Facility (together with any additional arranger or bookrunner appointed pursuant to Section 2 of the Commitment Letter in such capacity, the "<u>First Lien Lead Arrangers</u>") and as joint lead bookrunners and will perform the duties customarily associated with such roles.

First Lien Term Facility: A senior secured first lien term loan B facility in an aggregate principal amount equal to \$925 million (the "First Lien")

<u>Term Facility</u>"; the loans thereunder, the "<u>First Lien Term Loans</u>") (subject, in each case, to increase at the Borrower's election, to the extent required to account for any original issue discount and/or upfront fees with respect to the First

Lien Term Facility required

pursuant to the Flex Provisions of the Fee Letter; <u>provided</u> that, the increase described above shall be subject to reduction on a dollar-for-dollar basis by any additional amount of the loans advanced under the ABL Facility requested by the Borrower to fund any original issue discount and/or upfront fees with respect to the Senior Credit Facilities required pursuant to the Flex Provisions of the Fee Letter). The First Lien Term Loans shall be made in U.S. Dollars.

The First Lien Credit Documentation shall contain customary provisions for replacing or terminating the commitments of any Defaulting Lender (to be defined in the First Lien Credit Documentation in a manner consistent with the First Lien Documentation Principles).

Incremental Facilities:

The First Lien Credit Documentation will permit the Borrower to (a) add one or more incremental term loan facilities to the First Lien Term Facility and increase commitments under a class of term loans (any such term loan facility or increase, an "Incremental First Lien Term Facility") and/or (b) add one or more incremental revolving credit facilities (any such revolving credit facility or increase, an "Incremental Revolving Facility"; the Incremental First Lien Term Facilities and the Incremental Revolving Facilities are collectively referred to as "Incremental First Lien Facilities") in an aggregate principal amount up to (x) the greater of (I) \$210 million and (II) 1.0x Consolidated EBITDA (determined on a pro forma basis) for the most recent four quarters for which financial statements are available (less all amounts incurred under the corresponding "free and clear" basket of the Second Lien Notes) plus (y) (i) the aggregate principal amount of all voluntary prepayments and debt buybacks of (A) the First Lien Term Facility, (B) Incremental Equivalent Debt that is pari passu with the First Lien Term Facilities, (C) any Incremental Second Lien Notes or Incremental Equivalent Debt that is secured on a junior lien basis or is unsecured to the extent originally incurred under the "free and clear" prong and (ii) permanent voluntary commitment reductions of the ABL Facility in connection with the termination thereof (provided, voluntary permanent commitment reductions with respect to the ABL Facility may only be utilized to incur Incremental Revolving Facilities and not Incremental Term Loans) and any Incremental Revolving Facility except to the extent financed with the proceeds of long-term indebtedness (other than revolving indebtedness) plus (z) an unlimited additional amount; provided that after giving pro forma effect to the incurrence of such additional amount (and after giving effect to all customary pro forma events and adjustments) (i) in the case

of any Incremental Facility secured on a pari passu basis with the existing First Lien Term Facility, the Consolidated First Lien Net Leverage Ratio as of the last day of the most recently ended period of four consecutive fiscal quarters for which financial statements are internally available on a pro forma basis does not exceed the Consolidated First Lien Net Leverage Ratio as of the Closing Date, (ii) in the case of any Incremental Facility secured on a junior lien basis with the existing First Lien Term Facility, the Consolidated Senior Secured Net Leverage Ratio as of the last day of the most recently ended period of four consecutive fiscal quarters for which financial statements are internally available on a pro forma basis does not exceed the Consolidated Senior Secured Net Leverage Ratio as of the Closing Date and (iii) in the case of any unsecured Incremental Facility, either (A) the Fixed Charge Coverage Ratio (as defined below), on a pro forma basis, is not less than 2.00:1.00 or (B) if incurred in connection with a permitted acquisition or other permitted investment, the Fixed Charge Coverage Ratio on a pro forma basis does not decrease (in each case, after giving effect to the incurrence of such amounts (assuming full utilization of the then proposed Incremental Facility and excluding the proceeds of any borrowing under such then proposed Incremental Facility for purposes of determining net debt) and to any related transactions) (such amount under clauses (x), (y) and (z) above, collectively the "Available Incremental Amount"); it being understood that (I) the Borrower may elect to use clause (z) of the Available Incremental Amount prior to clause (x) or (y) and regardless of whether there is capacity under clause (x) or (y), and if clauses (x), (y) and (z) are available and the Borrower does not make an election, the Borrower will be deemed to have elected clause (z); (II) the Borrower may reclassify utilizations among clauses (x), (y) and (z) above if, at the time of such reclassification, the Borrower would be permitted to incur the aggregate principal amount of indebtedness being so reclassified; and (III) any amounts incurred under clauses (x) and (y) of the Available Incremental Amount concurrently with amounts incurred under clause (z) thereof will not count as indebtedness for the purposes of calculating the applicable ratio in clause (z) thereof: provided that (i) no event of default shall have occurred and be continuing or would result therefrom (except in connection with any acquisition or investment (including, without limitation, a Limited Condition Transaction), where such condition shall be no payment or bankruptcy (with respect to the Borrower) event of default), (ii) any Incremental First Lien Facility shall rank pari passu or junior in right of payment with the First Lien Term Facility and will either be secured on a pari passu or junior basis with the First

Lien Term Facility by the same Collateral securing the First Lien Term Facility or be unsecured, (iii) except in the case of a customary bridge loan incurred outside the First Lien Credit Documentation the maturity date of which provides for an automatic extension of the maturity date thereof, subject to customary conditions, to a date that is not earlier than the maturity date of the then existing First Lien Term Facility ("Extendable Bridge Loans"), the Incremental First Lien Term Facilities will have a final maturity no earlier than the final maturity of the initial term loans under the First Lien Term Facility (the "Initial First Lien Term Loans") as of the date of determination; provided, that, Incremental First Lien Term Facilities, Incremental Equivalent Debt, Refinancing Debt, Ratio Debt and indebtedness incurred in connection with a permitted acquisition in an aggregate amount not to exceed the greater of \$52.5 million and 25.0% of Consolidated EBITDA (on a pro forma basis) for the most recent four quarter period for which financial statements are available (the "Inside Maturity Basket") may have a maturity date that is earlier than the then existing First Lien Term Facility and the weighted average life to maturity of any such facility incurred under the Inside Maturity Date Basket may have a weighted life to maturity that is shorter than the then existing First Lien Term Facility, (iv) the weighted average life to maturity of any Incremental First Lien Term Facility (other than any Extendable Bridge Loans incurred outside the First Lien Credit Documentation and First Lien Incremental Term Loans incurred in reliance on the then existing Inside Maturity Basket) shall be no shorter than that of the Initial First Lien Term Loans as originally in effect prior to any extension or amortization or prepayments thereto (without giving effect to any prepayments on Initial First Lien Term Loans except as may be required to achieve fungibility with any existing term loan facility to the extent intended to be fungible), (v) subject to clauses (iii) and (iv) above, the amortization schedule applicable to any Incremental First Lien Term Facility shall be determined by the Borrower and the lenders thereunder, (vi) the All-In Yield (as defined below) applicable to any Incremental First Lien Term Facility will be determined by the Borrower and the lenders providing such Incremental First Lien Term Facility; provided that with respect to any Incremental First Lien Term Facility that is incurred prior to the one year anniversary of the Closing Date and is (a) pari passu in right of payment and security with the Initial First Lien Term Loans, (b) obtained in reliance on clauses (y) and (z) of the Available Incremental Amount, (c) matures within 12-months of the stated maturity of the Initial First Lien Term Loans and (d) not incurred in connection with a permitted acquisition or other permitted

investment, if the All-In Yield of any Incremental First Lien Term Facility exceeds the All-In Yield on the Initial First Lien Term Loans (calculated in the same manner), by more than 50 basis points, the applicable margins for the Initial First Lien Term Loans shall be increased to the extent necessary so that the All-In Yield on the Initial First Lien Term Loans is 50 basis points less than the All-In Yield on such Incremental First Lien Term Facility (it being agreed that any increase in yield to any existing facility required due to the application of an Adjusted LIBOR or ABR floor on any Incremental First Lien Term Facility shall be effected, at the option of the Borrower, through an increase in (or implementation of, as applicable) any Adjusted LIBOR or ABR floor applicable to such existing facility or an increase in the applicable margin or a combination thereof), (vii) any fees payable in connection with any such Incremental First Lien Facility shall be determined by the Borrower and the arrangers providing for such Incremental First Lien Facility, (viii) (A) any Incremental Revolving Facility (other than any Incremental Revolving Facility incurred in reliance on the permanent voluntary commitment reductions of the ABL Facility in clause (y)(ii) of the Available Incremental Amount) will mature no earlier than the maturity of the Initial First Lien Term Loans as of the date of determination and (B) any Incremental First Lien Term Facility may provide for the ability to participate on a pro rata basis or less than or greater than a pro rata basis in any voluntary prepayments of the term loans under other outstanding classes of term loans, and on a pro rata basis or less than a pro rata basis in any mandatory prepayments of the term loans under other outstanding classes of term loans (but not on a greater than pro rata basis except (I) for mandatory prepayments with the proceeds of Refinancing Debt which shall be applied to the class or classes of term loans to be refinanced as selected by the Borrower or (II) as may be directed by the Borrower, ratably to classes of term loans with a maturity date preceding the maturity date of the remaining classes of term loans outstanding (that have agreed to less than a pro rata payment), (ix) there shall be no requirement for the Borrower to satisfy any of the conditions listed under "Conditions Precedent to Ongoing Borrowings" below and (x) except as otherwise required or permitted in clauses (i) through (ix) above, all other terms of such Incremental First Lien Facility, if not consistent with the terms of the existing First Lien Term Facility shall either, (x) not be materially more restrictive to the Borrower (as determined by the Borrower), when taken as a whole, than the terms of the Initial First Lien Term Loans (except for covenants or other provisions applicable only to periods after the latest final scheduled maturity date of the Initial First Lien Term Loans) or (y) be reasonable

satisfactory to the First Lien Administrative Agent (it being understood that (1) covenants or other provisions applicable only to periods after the latest final scheduled maturity date of the Initial First Lien Term Loans need not be reasonably satisfactory to the First Lien Administrative Agent and (2) to the extent that any financial maintenance or other covenant is added for the benefit of any Incremental First Lien Facility, no consent shall be required from the First Lien Administrative Agent or any of the First Lien Lenders to the extent that such financial maintenance or other covenant is also added for the benefit of the existing First Lien Facility). Any Incremental First Lien Facility shall be secured solely by liens on the Collateral (which may be secured on a pari passu or junior lien basis, as selected by the Borrower) or shall be unsecured. Except as otherwise provided above (including with respect to maturity, margin, pricing and fees), the terms of any Incremental Revolving Facility shall be reasonably satisfactory to the First Lien Administrative Agent and the Borrower.

The Borrower may in its sole discretion seek commitments in respect of the Incremental First Lien Facilities from existing First Lien Lenders (each of which shall be entitled to agree or decline to participate in its sole discretion) and additional banks, financial institutions and other institutional lenders in the case of such additional banks, financial institutions and other persons who will become First Lien Lenders in connection therewith; and the Sponsor shall have the right to provide all or a portion of any Incremental Term Facilities subject to the same restrictions that would be applicable to any purchase by or assignment to the Sponsor as set forth under "Assignments and Participations" below.

In addition, the Borrower may, in lieu of adding Incremental First Lien Facilities, utilize any part of the Available Incremental Amount at any time by issuing or incurring Incremental Equivalent Debt (as defined below).

"Incremental Equivalent Debt" means indebtedness in an amount not to exceed the then available Available Incremental Amount consisting of the issuance of senior secured first lien loans or notes, junior lien loans or notes, subordinated unsecured loans or notes or senior unsecured loans or notes, in the case of any securities, issued in a public offering, Rule 144A or other private placement or customary bridge financing in lieu of the foregoing, or senior or subordinated "mezzanine" debt (which "mezzanine" debt can be secured on a junior lien basis), in each case on customary terms and conditions; provided that (x) such

Incremental Equivalent Debt shall not be subject to the requirements set forth in clauses (i), (vi) and (x) of the first paragraph in this "Incremental Facilities" section (other than in the case of clause (vi), Incremental Equivalent Debt consisting of first lien syndicated term loans obtained in reliance on clauses (y) and (z) of the Available Incremental Amount and not otherwise excluded from such clause (vi) pursuant to the terms thereof and (y) clauses (iii) and (iv) shall not apply to any Incremental Equivalent Debt consisting of any Extendable Bridge Loans or any Incremental Equivalent Debt incurred in reliance on the then available Inside Maturity Basket.

"All-In Yield" for purposes of the calculation above shall mean, as to any indebtedness, the yield thereof, whether in the form of interest rate, margin, original issue discount, upfront fees, a LIBOR or Base Rate floor, or otherwise, in each case, incurred or generally payable by the Borrower ratably to all lenders of such indebtedness; provided that (a) original issue discount and upfront fees shall be equated to interest rate assuming a 4-year life to maturity (or, if less, the stated life to maturity at the time of its incurrence of the applicable indebtedness), (b) "All-In Yield" shall not include arrangement fees, structuring fees, commitment fees, underwriting fees, success fees, unused line fees, advisory fees, ticking fees, consent or amendment fees and any similar fees (whether shared or paid, in whole or in part, with or to any or all lenders) and any other fees not generally paid ratably to all lenders of such indebtedness in the initial syndication thereof, (c) if any Incremental First Lien Term Loans include a LIBOR or Base Rate floor that is greater than the LIBOR or Base Rate floor applicable to the initial First Lien Term Loans, such differential between interest rate floors shall be included in the calculation of All-In Yield, but only to the extent an increase in the LIBOR or Base Rate Floor applicable to the Initial First Lien Term Loans would cause an increase in the interest rate then in effect thereunder, and in such case, if the All-In Yield on such Incremental First Lien Term Loans would be more than 0.50% higher than the All-In Yield for the initial First Lien Term Loans, then the LIBOR and Base Rate floors (but not the applicable rate) applicable to the Initial First Lien Term Loans shall be increased to the extent of such differential between interest rate floors, minus 0.50% and (d) shall not take into account fluctuations in the underlying reference rate or fluctuations in currency valuations.

"<u>Fixed Charge Coverage Ratio</u>" shall be defined in the First Lien Credit Documentation in a manner consistent with the First Lien Documentation Principles.

"Consolidated First Lien Net Leverage Ratio" means the ratio of (i) consolidated first lien net debt (consisting of indebtedness for borrowed money, capitalized lease obligations and purchase money debt as reflected on the balance sheet of the Borrower and its restricted subsidiaries in accordance with generally accepted accounted principles ("GAAP"), plus standby letters of credit that have been drawn and not reimbursed within two business days after the date of such drawing, in each case secured by liens on the Collateral that are the liens securing the First Lien Credit Facility or are pari passu with the liens securing the First Lien Term Facility or by liens on the ABL Priority Collateral that rank senior to the liens on the ABL Priority Collateral securing the First Lien Term Facility), minus unrestricted cash and cash equivalents to (ii) Consolidated EBITDA for the most recent four fiscal quarter period for which financial statements are internally available. For the avoidance of doubt, the ABL Facility shall be deemed to be pari passu to the First Lien Term Facility for purposes of calculating the Consolidated First Lien Net Leverage Ratio.

"Consolidated Senior Secured Net Leverage Ratio" means the ratio of (i) consolidated senior secured net debt (consisting of indebtedness for borrowed money, capitalized lease obligations and purchase money debt as reflected on the balance sheet of the Borrower and its restricted subsidiaries in accordance with GAAP, plus standby letters of credit that have been drawn and not reimbursed within two business days after the date of such drawing, in each case secured by liens), minus unrestricted cash and cash equivalents to (ii) Consolidated EBITDA for the most recent four fiscal quarter period for which financial statements are internally available.

"Consolidated Total Net Leverage Ratio" shall mean the ratio of (i) consolidated net debt (consisting of indebtedness for borrowed money, capitalized lease obligations and purchase money debt as reflected on the balance sheet of the Borrower and its restricted subsidiaries in accordance with GAAP), plus standby letters of credit that have been drawn and not reimbursed within two business days after the date of such drawing, minus unrestricted cash and cash equivalents of the Borrower and its restricted subsidiaries to (ii) Consolidated EBITDA for the most recent four fiscal quarter period for which financial statements are internally available.

"Consolidated Net Income" shall be defined in a manner consistent with the First Lien Documentation Principles.

For purposes of the First Lien Credit Documentation,

- "Consolidated EBITDA" shall be defined in a manner consistent with the First Lien Documentation Principles and, in any event, shall:
- (I) include addbacks (not subject to caps) for, without limitation or duplication (including with addbacks to Consolidated Net Income):
- (a) pro forma "run rate" cost savings, operating expense reductions and synergies related to acquisitions, dispositions and other specified transactions, restructurings, cost savings initiatives and other initiatives (including, in each case, as a result of the Transactions) that are reasonably identifiable, factually supportable and projected by the Borrower in good faith to result from actions that have been taken or with respect to which substantial steps have been taken or are expected to be taken within 24 months (in the good faith determination of the Borrower) after such acquisition, disposition or other specified transaction, restructuring, cost savings initiative or other initiative (net of the amount of actual benefits realized);
- (b) such other adjustments and addbacks (i) previously identified and set forth in the Sponsor model delivered to the Initial First Lien Lenders on March 2, 2019 (together with any updates or modifications thereto reasonably agreed between the Sponsor and the Initial First Lien Lenders or as necessary to reflect any exercise of Flex Provisions, the "Sponsor Model") or the quality of earnings report delivered to the Initial First Lien Lenders on March 6, 2019, (ii) evidenced or contained in a due diligence quality of earnings report made available to the First Lien Administrative Agent prepared by (x) a "big-four" nationally recognized accounting firm or (y) any other accounting firm reasonably acceptable to the First Lien Administrative Agent or (iii) consistent with Regulation S-X; and
- (c) other adjustments and add-backs as agreed with the Initial First Lien Lenders;
- and (II) be subject to other customary pro forma adjustments with respect to acquisitions, dispositions and other specified transactions to be set forth in the First Lien Credit Documentation in a manner consistent with the First Lien Documentation Principles.

For purposes of (i) determining compliance with any provision of the First Lien Credit Documentation which requires the calculation of the Fixed Charge Coverage Ratio, the Consolidated

First Lien Net Leverage Ratio, the Consolidated Senior Secured Net Leverage Ratio or the Consolidated Total Net Leverage Ratio (as defined below), (ii) determining compliance with representations, warranties, defaults or events of default or (iii) testing availability under baskets set forth in the First Lien Credit Documentation (including baskets measured as a percentage of total assets or Consolidated EBITDA), in each case, in connection with (1) an acquisition by one or more of the Borrower and its restricted subsidiaries of any assets, business or person permitted to be acquired by the First Lien Credit Documentation, in each case whose consummation is not conditioned on the availability of, or on obtaining, third party financing, (2) an irrevocable debt repurchase or repayment or (3) restricted payments (but in the case of this clause (3), solely to the extent such restricted payments are consummated in connection with a transaction separately subject to clause (1) or (2) above) (any such transaction, a "Limited Condition Transaction"), at the option of the Borrower (the Borrower's election to exercise such option in connection with any Limited Condition Transaction, an "LCT Election"), the date of determination of whether any such action is permitted under the First Lien Credit Documentation, shall be deemed to be the date the definitive agreements for such Limited Condition Transaction are entered into (the "LCT Test Date"), and if, after giving pro forma effect to the Limited Condition Transaction and the other transactions to be entered into in connection therewith as if they had occurred at the beginning of the most recent test period ending prior to the LCT Test Date, the Borrower could have taken such action on the relevant LCT Test Date in compliance with such representation, warranty, ratio, basket or default provision, such representation, warranty, ratio, basket or default provision shall be deemed to have been complied with.

For the avoidance of doubt, if the Borrower has made an LCT Election and any of the ratios or baskets for which compliance was determined or tested as of the LCT Test Date are exceeded as a result of fluctuations in any such ratio or basket (including due to fluctuations of the target of any Limited Condition Transaction, cash or the amount of indebtedness) at or prior to the consummation of the relevant transaction or action, such baskets or ratios will not be deemed to have been exceeded as a result of such fluctuations. If the Borrower has made an LCT Election for any Limited Condition Transaction, then in connection with any subsequent calculation of such ratios or baskets on or following the relevant LCT Test Date and prior to the earlier of (i) the date on which such Limited Condition Transaction is consummated or (ii) the date that the definitive agreement for such Limited Condition Transaction is terminated or expires without

consummation of such Limited Condition Transaction, any such ratio or basket shall be calculated on a pro forma basis assuming such Limited Condition Transaction and other transactions in connection therewith (including any incurrence of debt and the use of proceeds thereof) have been consummated.

Refinancing Facilities:

The First Lien Credit Documentation will permit the Borrower to refinance loans or commitments under any tranche or tranches of loans and/or commitments having the same terms (any such tranche, a "class") under the First Lien Term Facility, as selected by the Borrower (including, for the sake of clarity, any Incremental First Lien Term Facility or any Incremental Revolving Facility) from time to time, in whole or part, in a principal amount not to exceed the principal amount of indebtedness so refinanced (plus any accrued but unpaid interest, existing commitments unutilized under such indebtedness being refinanced to the extent such commitments are permanently reduced, premiums (including tender premiums), penalties and fees payable by the terms of such indebtedness thereon and any defeasance costs and any fees, expenses, original issue discount and upfront payments incurred in connection with such refinancing), with one or more new classes (or an increase to one or more existing classes) of term loans (each, a "Refinancing Term Facility") or new classes (or an increase to one or more existing classes) of revolving credit facilities (each, a "Refinancing Revolving Facility"; the Refinancing Term Facilities and the Refinancing Revolving Facilities are collectively referred to as "Refinancing Facilities"), respectively, under the First Lien Credit Documentation with the consent of the Borrower and the lenders providing such Refinancing Term Facility or Refinancing Revolving Facility or with one or more additional series of unsecured or subordinated notes or loans or senior secured loans or notes that will be secured by the Collateral on a pari passu basis with the First Lien Term Facility or junior lien secured notes or loans that will be secured on a junior basis to the First Lien Term Facility (and such notes or loans, "Refinancing Notes" and, together with the Refinancing Facilities, the "Refinancing Debt"); provided that (i) except in the case of Extendable Bridge Loans and Refinancing Debt incurred in reliance on the then available Inside Maturity Basket, any Refinancing Debt does not mature prior to the final scheduled maturity date of, or have a shorter weighted average life to maturity than, the loans under the class being refinanced, without giving effect to any prepayments of such class being refinanced, (ii) the other terms and conditions of such Refinancing Term Facility, Refinancing Revolving Facility or Refinancing Notes (excluding pricing, interest rate margins, fees, discounts, rate floors and prepayment or redemption terms

which shall be determined by the Borrower) shall either, at the option of the Borrower, (x) reflect market terms and conditions (taken as a whole) at the time of incurrence or issuance (as determined by the Borrower) or (y) if not consistent with the terms of the corresponding class under the First Lien Term Facility, not be materially more restrictive to the Borrower (as determined by the Borrower), when taken as a whole, than the terms of the applicable class under the First Lien Term Facility being refinanced or replaced (except for covenants or other provisions applicable only to periods after the latest final scheduled maturity date of the First Lien Term Facility and revolving credit commitments existing at the time of such refinancing), (iii) any secured Refinancing Debt shall be subject to customary intercreditor documentation, (iv) any Refinancing Debt shall not be guaranteed by any subsidiaries that do not guarantee the First Lien Term Facility, (v) to the extent secured, such Refinancing Debt shall not be secured by any property or assets other than the Collateral, (vi) with respect to any Refinancing Debt that is a Refinancing Revolving Facility and is a refinancing of the existing Incremental Revolving Facility, such Refinancing Debt shall be subject to pro rata payment, borrowing, participation and commitment reduction provisions with the existing Incremental Revolving Facility (other than payments of interest and fees and other than repayments required on the applicable maturity date); provided that the Borrower may permanently repay or terminate commitments in respect of any class of the Revolving Facility (x) other than such existing Revolving Facility, on a pro rata or less than pro rata basis (but not greater than pro rata basis except in connection with any refinancing thereof) with all other classes of the Revolving Facility and (y) on a greater than pro rata basis as compared to any other class of the Revolving Facility that has a later maturity date than such class, (vii) any Refinancing Term Facility may provide for the ability to participate on a pro rata basis or less than or greater than a pro rata basis in any voluntary prepayments of the term loans under other outstanding classes of term loans, and on a pro rata basis or less than a pro rata basis (but not on a greater than pro rata basis except (x) for mandatory prepayments with the proceeds of Refinancing Debt which shall be applied to the class or classes of term loans to be refinanced as selected by the Borrower or (y) as may be directed by the Borrower, ratably to classes of term loans with a maturity date preceding the maturity date of the remaining classes of term loans outstanding) in any mandatory prepayments of the term loans under other outstanding classes of term loans, (viii) the proceeds of such Refinancing Facilities shall be applied, substantially concurrently with the incurrence thereof, to the pro rata

prepayment of outstanding loans under the applicable class being refinanced (and in the case of any class of revolving facility commitments being refinanced, pro rata commitment reductions to such class) and associated interest, premium, fees and expenses and (ix) the First Lien Credit Documentation will contain certain provisions to be agreed to govern the payment, borrowing, participation and commitment reduction of any Refinancing Revolving Facility and any Incremental Revolving Facility then existing.

Purpose:

The proceeds of the First Lien Term Loans will be used by the Borrower on the Closing Date, together with the proceeds of the Second Lien Notes, borrowing under the ABL Facility and certain cash available on the balance sheet of Omega and its subsidiaries and the Company and its subsidiaries, to finance a portion of the Transactions (including working capital and/or purchase price adjustments and the payment of Transaction Costs, upfront fees and original issue discount with respect to the First Lien Term Facility).

Availability:

The First Lien Term Loans will be available in a single drawing on the Closing Date. Repayments and prepayments of the First Lien Term Loans may not be reborrowed.

Interest Rates and Fees:

As set forth on Annex I hereto.

Default Rate:

Any principal payable under or in respect of the First Lien Term Facility not paid when due shall bear interest at the applicable interest rate plus 2% per annum. Other overdue amounts shall bear interest at the interest rate applicable to ABR loans plus 2% per annum. Default interest will be payable on demand.

Final Maturity and Amortization:

The First Lien Term Facility will mature on the date that is seven years after the Closing Date (the "<u>First Lien Term Maturity Date</u>") and will amortize in equal quarterly installments in aggregate annual amounts equal to 1% of the original principal amount of the First Lien Term Facility commencing on the last day of the second full fiscal quarter ended after the Closing Date, with the balance payable on the final maturity date (it being understood and agreed that such amortization payments shall be reduced by application of debt prepayments as provided for herein, including open market debt repurchases); provided that the First Lien Credit Documentation shall provide the right for the Borrower, subject to customary terms and conditions, to extend commitments and/or outstandings pursuant to one or more tranches with only the consent of the respective extending Lenders (it being understood that each First Lien Lender under the tranche that is being

extended shall have been offered the opportunity to participate in such extension on the same terms and conditions as each other Lender under such tranche).

Guarantees:

All obligations of the Borrower (the "Borrower Obligations") under the First Lien Term Facility and any Incremental Revolving Facility and, at the option of the Borrower, under any interest rate protection, currency exchange or other hedging arrangements (other than swap obligations designated as secured obligations under the ABL Facility and any Excluded Swap Obligation (as defined below)) entered into with the First Lien Administrative Agent, a First Lien Lender or any affiliate of a First Lien Lender or the First Lien Administrative Agent at the time of the entering into of such arrangements or, in the case of any such arrangements existing on the Closing Date, on the Closing Date, or with other third party financing sources so long as such third party financing sources agree to provide to the First Lien Administrative Agent a customary letter agreement agreeing to the applicable terms of the Credit Agreement at the time of the entering into of such arrangements ("Hedging Obligations") will, except to the extent such a Guarantee would be prohibited or restricted by applicable law or contract (including any requirement to obtain the consent, approval, license or authorization of any governmental authority or third party), be unconditionally guaranteed jointly and severally on a senior secured basis (the "Guarantees") by each existing and subsequently acquired or organized direct or indirect wholly-owned U.S. subsidiary of the Company (other than any direct or indirect U.S. subsidiary of a direct or indirect non-U.S. subsidiary of the Borrower that is a "controlled foreign corporation" within the meaning of Section 957 of the Internal Revenue Code of 1986, as amended (a "CFC"), any U.S. subsidiary if it has no material assets other than the capital stock and/or indebtedness of one or more foreign subsidiaries of the Borrower that is a CFC (each, a "CFC Holdco"), any broker-dealer subsidiaries, unrestricted subsidiaries, captive insurance subsidiaries, specialty insurers, notfor-profit subsidiaries, special purpose securitization entities, immaterial subsidiaries, and any subsidiary where the Borrower reasonably determines in consultation with the First Lien Administrative Agent that the burden or cost (including any adverse tax consequences to the Borrower or any of its direct or indirect parent companies or subsidiaries) of obtaining a guarantee by such subsidiary would outweigh the practical benefit to the First Lien Lenders afforded thereby, subsidiaries that are acquired pursuant to permitted acquisitions or similar permitted investments that are prohibited from providing a guarantee pursuant to the terms of any permitted indebtedness of such subsidiary (and such prohibition was not incurred in

contemplation of such acquisition) and other subsidiaries as mutually agreed between the Borrower and the First Lien Administrative Agent (collectively, "Excluded Subsidiaries")) (the "Guarantors", and together with the Borrower, the "Loan Parties").

Notwithstanding anything to the contrary contained herein, the requirements of the preceding paragraph shall be, as of the Closing Date, subject to the limitations set forth in the Commitment Letter, including the Certain Funds Provisions.

"Excluded Swap Obligations" means, with respect to any Guarantor, (a) any obligation to pay or perform under any agreement, contract, or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act (each such obligation, a "Swap Obligation") if, and to the extent that, all or a portion of the Guarantee of such Guarantor of, or the grant by such Guarantor of a security interest to secure, such Swap Obligation (or any Guarantee thereof) is or becomes illegal or unlawful under the Commodity Exchange Act or any rule, regulation, or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) (i) by virtue of such Guarantor's failure to constitute an "eligible contract participant," as defined in the Commodity Exchange Act and the regulations thereunder (determined after giving effect to any applicable keepwell, support, or other agreement for the benefit of such Guarantor), at the time the guarantee of (or grant of such security interest by, as applicable) such Guarantor would otherwise have become effective with respect to such Swap Obligation but for such Guarantor's failure to constitute an "eligible contract participant" as such time or (ii) in the case of a Swap Obligation that is subject to a clearing requirement pursuant to section 2(h) of the Commodity Exchange Act, because such Guarantor is a "financial entity," as defined in section 2(h)(7)(C) of the Commodity Exchange Act, at the time the guarantee of (or grant of such security interest by, as applicable) such Guarantor becomes or would become effective with respect to such Swap Obligation or (b) any other Swap Obligation designated as an "Excluded Swap Obligation" of such Guarantor as specified in any agreement between the relevant Loan Parties and hedge bank applicable to such Swap Obligations.

Security:

Subject to the Certain Funds Provision and the limitations set forth in this section, the Borrower Obligations, the Guarantees, and, at the option of the Borrower any Hedging Obligations, will be secured by, subject to permitted liens, a first priority perfected security interest in substantially all of the present and after-

acquired assets of the Borrower and each Guarantor, including but not limited to: (a) a perfected second-priority security interest in the following property of the Loan Parties (subject to permitted liens and other exceptions consistent with the First Lien Documentation Principles and excluding any Excluded Assets (as defined below)): (i) all accounts receivable, cash, deposit accounts and securities accounts (the "Current Asset Collateral"), (ii) all owned and after acquired inventory (the "Inventory Collateral"), (iii) the right to use trademarks, tradenames and other intellectual property in connection with the processing or sale of inventory or the sale or collection on accounts receivable under a royalty fee license agreement or to the extent necessary to sell such Current Asset Collateral or Inventory Collateral, and (iv) all letter of credit rights, commercial tort claims, chattel paper, supporting obligations, general intangibles (including contract rights and customer lists), documents, books, records and instruments relating to such Current Asset Collateral or Inventory Collateral and, in the case of each of clause (i) through (iv), the proceeds thereof (including insurance, indemnity, guaranty and condemnation proceeds) subject to permitted liens and to exceptions and limitations (the foregoing, collectively, but excluding Excluded Assets, the "ABL Priority Collateral") and (b) a perfected first-priority security interest in substantially all other present and after-acquired assets of the Loan Parties (subject to permitted liens and other exceptions consistent with the First Lien Documentation Principles and excluding any Excluded Assets), including, without limitation, (i) a perfected pledge of all the capital stock directly held by the Borrower or any Guarantor in any material wholly-owned restricted subsidiary (which pledge, in the case of the capital stock of any foreign subsidiary of the Borrower that is a CFC or of a CFC Holdco shall not exceed 65% of the stock of such foreign subsidiary or CFC Holdco, as the case may be) and (b) perfected security interests (subject to permitted liens and customary exceptions) in substantially all other tangible and intangible assets of the Borrower and each Guarantor (including equipment, general intangibles (except to the extent constituting ABL Priority Collateral), investment property (except to the extent constituting ABL Priority Collateral), intellectual property (subject to clause (iii) above), intercompany notes and proceeds of the foregoing) (such collateral, excluding ABL Priority Collateral, the "Term Loan Priority Collateral" and together with the ABL Priority Collateral, the "Collateral"), in each case subject to permitted liens and to certain customary exceptions consistent with the First Lien Documentation Principles. In no event shall the First Lien Term Facility be secured by any ABL Priority Collateral that the ABL

Administrative Agent under the ABL Facility has elected to forego perfection in pursuant to its discretionary powers described in the ABL Facility.

Notwithstanding the foregoing, (a) the Collateral shall not include: (i) any fee-owned real property and any real property leasehold rights and interests (it being understood there shall be no requirement to obtain any landlord or other third party waivers, estoppels or collateral access letters), (ii) motor vehicles, airplanes and other assets subject to certificates of title, to the extent a lien thereon cannot be perfected by filing of a UCC financing statement, (iii) commercial tort claims below a threshold to be agreed, (iv) any governmental or regulatory licenses or state or local franchises, charters and authorizations, to the extent a security interest in any such license, franchise, charter or authorization is prohibited or restricted thereby (excluding any prohibition or restriction that is ineffective under the Uniform Commercial Code), (v) pledges and security interests prohibited or restricted by applicable law, rule or regulation (including any requirement to obtain the consent of any governmental authority or regulatory authority), but excluding any prohibition or restriction that is ineffective under the Uniform Commercial Code), (vi) (A) margin stock, (B) equity interests in any person other than the Borrower and wholly-owned restricted subsidiaries (but, in the case of the equity interests of any person that is not a wholly-owned restricted subsidiary, only to the extent the organizational documents or similar agreement with equity holders of such person do not permit the pledge of such equity interests for so long as such prohibition exists), (C) equity interests or debt treated as equity for U.S. federal income tax purposes of first tier foreign subsidiaries that are CFCs and first tier CFC Holdcos in excess of 65% of the issued and outstanding equity interests thereof and (D) equity interests in broker-dealer subsidiaries, unrestricted subsidiaries, captive insurance subsidiaries, not-for-profit subsidiaries or special purpose entities that are not required to guarantee the First Lien Term Facility, (vii) any lease, license or agreement or any property subject to a purchase money security interest, capital lease obligation or similar arrangement. in each case, to the extent that a grant of a security interest therein (A) would violate or invalidate such lease, license or agreement or purchase money or similar arrangement or create a right of termination in favor of any other party thereto other than a Loan Party (after giving effect to the applicable anti-assignment provisions of the Uniform Commercial Code) or (B) would require governmental, regulatory or other third party approval, consent or authorization pursuant to the terms thereof (in each case, after giving effect to the applicable anti-assignment

provisions of the UCC) (other than proceeds and receivables thereof, the assignment of which is expressly deemed effective under the Uniform Commercial Code notwithstanding such prohibition) not obtained (without any requirement to obtain such approval, consent or authorization) (in each case, after giving effect to the applicable anti-assignment provisions of the UCC), in each case of clauses (A) and (B), so long as such requirement is permitted by the negative pledge clause negative covenant, (viii) letter of credit rights, except to the extent perfection of the security interest is therein accomplished by the filing of a Uniform Commercial Code financing statement (it being understood that no actions shall be required to perfect a security interest in letter of credit rights, other than the filing of a Uniform Commercial Code financing statement), (ix) any intent-to-use trademark application prior to the filing, and acceptance by the U.S. Patent and Trademark Office, of a "Statement of Use" or "Amendment to Allege Use" with respect thereto, to the extent, if any, that, and solely during the period, if any, in which, the grant of a security interest therein would impair the validity or enforceability of such intent-to-use trademark application under applicable federal law; (x) assets where the burden or cost (including adverse tax consequences) of obtaining a security interest therein or perfection thereof exceeds the practical benefit to the Lenders afforded thereby as reasonably determined by the Borrower in good faith, in consultation with the First Lien Administrative Agent; and (xi) funds held in a fiduciary capacity for others (the foregoing described in clauses (a)(i) through (xi) are, collectively, the "Excluded Assets"); (b) no actions in any non-U.S. jurisdiction or required by the laws of any non-U.S. jurisdiction shall be required in order to create any security interests in assets located or titled outside of the U.S. or to perfect any security interest in such assets, including any intellectual property registered in any non-U.S. jurisdiction (it being understood that there shall be no security agreements or pledge agreements governed under the laws of any non-U.S. jurisdiction or any requirement to make any filings in any foreign jurisdiction, including with respect to foreign intellectual property); (c) perfection through control agreements or perfection by "control" shall not be required with respect to any Collateral (including deposit accounts and other bank or securities accounts, etc. but other than in respect of certain debt owing to the Loan Parties and certificated equity interests of the Borrower and material wholly-owned restricted subsidiaries otherwise required to be pledged), (d) there shall be no requirement to comply with the Federal Assignment of Claims Act or any similar statute; (e) perfection by possession of immaterial notes and other evidence of immaterial indebtedness shall not be

required and (f) the First Lien Administrative Agent shall not be authorized to enter into any source code escrow arrangement or register any intellectual property.

Notwithstanding the foregoing, the requirements of the preceding paragraphs of this "Security" section shall be, as of the Closing Date, subject to the Certain Funds Provisions.

Mandatory Prepayments:

Loans under the First Lien Term Facility shall be prepaid with (a) 50% of Excess Cash Flow (to be defined in the First Lien Credit Documentation in a manner consistent with the First Lien Documentation Principles) for each fiscal year of the Borrower (commencing with the first full fiscal year completed after the Closing Date) with step-downs to 25% and 0% based upon achievement of Consolidated First Lien Net Leverage Ratios of 0.50x and 1.00x less than the Consolidated First Lien Net Leverage Ratio as of the Closing Date respectively (with such ratios for purposes of determining the applicable Excess Cash Flow percentages, to be recalculated to give pro forma effect to any such paydown or reduction (including paydowns made after year-end and prior to the time such Excess Cash Flow payment is due)); provided that (i) prepayments shall only be required under this clause (i) if the applicable percentage of Excess Cash Flow is greater than an amount to be set forth in the First Lien Credit Documentation (with only amounts in excess thereof being subject to prepayment) and (ii) Excess Cash Flow prepayments required for such fiscal year shall be reduced on a dollar for dollar basis by (x) the amount of voluntary prepayments or repurchases of of (A) Loans or notes under the First Lien Term Facility, the Second Lien Notes, any Incremental First Lien Term Facility, any Incremental Second Lien Notes, any refinancing, replacement or extension of any of the foregoing (in each case, including any debt buyback conducted pursuant to a Dutch Auction or open market purchase based on actual amounts paid) and (B) the ABL Facility and any Incremental Revolving Facility and any refinancing, replacement or extended revolving facilities (in each applicable case to the extent accompanied by a permanent reduction of the relevant commitment), (y) amounts used to repay borrowings of ABL Loans made to account for any additional original issue discount or upfront facility fees that are implemented pursuant to the Flex Provision, in each case made during such fiscal year (to the extent not previously deducted from Excess Cash Flow in the prior year) or after year-end and prior to the time such Excess Cash Flow prepayment is due and (z) cash used for permitted acquisitions and certain other permitted investments (and in each case, related earnouts), certain loans and advances to investment advisors and certain restricted payments made during such fiscal

year (to the extent not previously deducted from Excess Cash Flow in the prior year) or after year-end and prior to the time such Excess Cash Flow prepayment is due, in each case except to the extent financed with long-term indebtedness (other than revolving indebtedness); (b) 100% of the net cash proceeds of all non-ordinary course asset sales or other dispositions of property by the Borrower and its restricted subsidiaries (including casualty insurance and condemnation proceeds, but with exceptions for sales of certain designated business lines, certain sales or other dispositions of ABL Priority Collateral, inventory and other ordinary course dispositions, obsolete or worn-out property, property no longer used or useful in the business and other exceptions to be set forth in the First Lien Credit Documentation in a manner consistent with the First Lien Documentation Principles) in excess of a threshold amount and subject to the right of the Borrower to reinvest 100% of the net cash proceeds in assets used or useful in the business if such proceeds are reinvested (or committed to be reinvested) within 18 months and, if so committed to reinvestment, reinvested no later than 180 days after the end of such 18-month period; and (c) 100% of the net cash proceeds of issuances of debt obligations of the Borrower and its restricted subsidiaries (except the net cash proceeds of any permitted debt other than Refinancing Debt).

Mandatory prepayments shall be applied to the scheduled installments of principal of the First Lien Term Facility as directed by the Borrower (and, absent such direction, in direct order of maturity). Mandatory prepayments shall be applied pro rata among the classes of term loans outstanding except (x) for any proceeds of Refinancing Debt which shall be applied to the class or classes of term loans to be refinanced as selected by the Borrower and (y) as may be directed by the Borrower ratably to classes of term loans with a maturity date preceding the maturity date of the remaining classes of term loans outstanding (that have agreed to less than a pro rata prepayment), provided, further, that Incremental First Lien Term Facilities and Refinancing Term Facilities may participate in mandatory prepayments on a less than pro rata basis. Certain mandatory prepayments under the immediately preceding paragraph may be applied ratably to prepay or offer to purchase any first lien secured indebtedness if required under the terms of the applicable documentation governing such first lien secured indebtedness. Mandatory prepayments in clauses (a) and (b) above shall be subject to customary limitations to the extent attributable to Excess Cash Flow or net cash proceeds of non-U.S. restricted subsidiaries, the repatriation of which would result in adverse tax consequences to the Borrower or any of its direct or indirect parent companies,

direct or indirect shareholders or subsidiaries or would be prohibited or restricted by applicable law or organizational documents of such non-U.S. restricted subsidiaries of the Borrower so long as such restrictions were not implemented for the purpose of avoiding mandatory prepayment requirements.

Any Lender may elect not to accept any mandatory prepayment (except with respect to Term Loans being refinanced with Refinancing Debt or otherwise subject to a refinancing) (such amounts, "<u>Declined Proceeds</u>" and each such Lender, a "<u>Declining Lender</u>"). Any prepayment amount declined by a Declining Lender, subject to any mandatory offer to repurchase requirements of the Second Lien Notes, may be retained by the Borrower and may be applied to increase the cumulative "builder" basket.

Voluntary Prepayments and Reductions in Commitments:

Prepayments of the Term Loans will be permitted at any time (subject to customary notice requirements), in minimum principal amounts to be set forth in the First Lien Credit Documentation, without premium or penalty, subject to reimbursement of the Lenders' redeployment costs in the case of a prepayment of Adjusted LIBOR borrowing prior to the last day of the relevant interest period; provided that if, prior to the date that is six months after the Closing Date, (x) there shall occur any amendment, amendment and restatement or other modification of the definitive documentation for the Initial First Lien Term Loans the primary purpose of which is to reduce the All-In Yield then in effect for the loans thereunder, (y) all or any portion of the Initial First Lien Term Loans is voluntarily prepaid or mandatorily prepaid with the proceeds of, or any conversion of Initial First Lien Term Loans into, any new or replacement tranche of broadly syndicated senior secured first lien term loans in a transaction the primary purpose of which is to reduce the All-In Yield below the All-In Yield in effect for the loans so prepaid or (z) a Lender must assign its Initial First Lien Term Loans as a result of its failure to consent to an amendment, amendment and restatement or other modification of the Initial First Lien Term Loans the primary purpose of which is to reduce the All-In Yield then in effect for the Initial First Lien Term Loans (any of clause (x), (y) or (z), a "Repricing Transaction"), then the aggregate principal amount so prepaid or assigned, or in the case of clause (x), subject to such Repricing Transaction, will be subject to a 1.00% prepayment premium; provided that such prepayment premium shall not apply if such amendment, amendment and restatement or other modification, or such prepayment, refinancing or forced assignment, is in connection with a "change of control" transaction, public equity offering of primary shares or Transformative Acquisition.

Optional commitment reductions of commitments under an Incremental Revolving Facility shall be applied as directed by the Borrower, including to any class of extending or existing revolving commitments (including any revolving commitments under Refinancing Facilities).

All voluntary prepayments of the First Lien Term Facility will be applied to the remaining amortization payments under the First Lien Term Facility as directed by the Borrower (and absent such direction, in direct order of maturity thereof). All voluntary prepayments shall be applied to the class or tranche of term loans as selected by the Borrower.

"Transformative Acquisition" shall mean any acquisition by the Borrower or any other restricted subsidiary that (i) is not permitted by the terms of the First Lien Credit Documentation immediately prior to the consummation of such acquisition, (ii) if permitted by the terms of the First Lien Credit Documentation immediately prior to the consummation of such acquisition, would not provide the Borrower and the other restricted subsidiaries with adequate flexibility under the First Lien Credit Documentation for the continuation and/or expansion of their combined operations following such consummation, as determined by the Borrower acting in good faith based on projections of the combined business (and such projections shall have been delivered to the First Lien Administrative Agent) or (iii) results in a refinancing of the First Lien Term Facility that involves an upsizing in connection with such acquisition.

First Lien Credit Documentation:

The definitive documentation for the First Lien Term Facility (the "First Lien Credit Documentation") shall be initially drafted by counsel to the Borrower, shall be substantially consistent with (and no less favorable to the Borrower and its subsidiaries than) that certain Credit Agreement, dated as of February 28, 2018, among Mayfield Agency Borrower Inc., as administrative borrower, the lenders party thereto, Bank of America, N.A., as administrative agent and the other parties thereto (as amended prior to the date hereof, and adjusted pursuant to terms below, the "First Lien Documentation Principles") and take into account (a) the terms set forth in this Commitment Letter, (b) any additional flexibility provided in any transactions involving affiliates of the Sponsor that is reasonably acceptable to the Borrower and the Commitment Parties, (c) differences related to the Loan Parties, their subsidiaries and their businesses (including (i) as to operational and strategic requirements of the Loan Parties and their subsidiaries (in light of their industries, businesses and business practices, locations, operations, financial accounting and

projections and (ii) the industry and practices of the Loan Parties and their subsidiaries), (d) changes in law or accounting standards or to cure mistakes or defects, (e) the Sponsor Model, and (f) any changes in the operational and agency requirements of the First Lien Administrative Agent and in any event will contain only those conditions to borrowing, mandatory prepayments, representations, warranties, affirmative, negative and financial covenants and events of default expressly set forth in this Term Sheet, in each case applicable to the Borrower and its restricted subsidiaries, in each case, with materiality thresholds, qualifications, exceptions, "baskets" and grace and cure periods to be mutually agreed, and consistent with this Term Sheet (subject to Flex Provisions set forth in the Fee Letter) and the First Lien Documentation Principles; provided that, in any event neither paragraph (24) of the definition of "Permitted Investments" of the precedent credit agreement nor any similar "basket" or exception will be included in the First Lien Credit Documentation. The First Lien Credit Documentation will contain customary Loan Syndications & Trading Association EU Bail-in provisions, customary LSTA provisions regarding ERISA which are solely applicable to the Lenders and customary provisions regarding the division of a limited liability company, and will be no less favorable to the Borrower than each of (i) that certain Credit Agreement, dated as of April 7, 2015, by and among HC Group Holdings III, Inc., as borrower, the lenders party thereto, Bank of America, N.A., as administrative agent and the other parties thereto and (ii) that certain First Lien Note Purchase Agreement, dated as of June 29, 2017, among the Company, as issuer, the purchasers party thereto from time to time and Wells Fargo Bank, National Association, as collateral agent. The definitive terms will be negotiated in good faith to finalize the First Lien Credit Documentation, giving effect to the Certain Funds Provision and the First Lien Documentation Principles, as promptly as reasonably practicable.

Intercreditor Agreements:

The intercreditor agreement (the "Term Loan Intercreditor Agreement") between the First Lien Administrative Agent and the trustee on behalf of the holders of the Second Lien Notes shall be consistent with and substantially similar to that certain Intercreditor Agreement, dated as of February 28, 2018, among Bank of America, N.A., in its capacities as first lien credit agreement administrative agent and second lien credit agreement administrative agent and acknowledged and agreed to by Mayfield Agency Borrower Inc. and the other parties to such acknowledgement; *provided* that (i) an obligation (including any obligation under the First Lien Credit Agreement) will only constitute a "first lien secured obligation" (or similar term) to the

extent such obligation is, at the time of incurrence, permitted to be incurred (and secured) as a first priority obligation under the then extant documentation governing indebtedness constituting first lien secured obligations and the then extant documentation governing indebtedness constituting second lien secured obligations (ii) the Term Loan Intercreditor Agreement will include a customary "DIP cap" that will provide that that the aggregate principal amount of any DIP financing (plus the aggregate principal amount of first lien secured obligations and the aggregate face amount of any letters of credit issued and outstanding under first lien debt facilities in each case that would remain outstanding after giving effect to such DIP financing) does not exceed 115% of the sum of the aggregate outstanding principal amount of first lien secured obligations on the date that the applicable insolvency or liquidation proceeding commenced plus the aggregate face amount of any letters of credit issued and outstanding under first lien debt facilities outstanding on the date that the applicable insolvency or liquidation proceeding commenced (it being understood and agreed that cash management obligations and hedging obligations shall not be subject to such cap amount), (iii) the Term Loan Intercreditor Agreement will permit the second lien secured parties to propose, vote in favor of or otherwise support any plan of reorganization that is not inconsistent with the priorities set forth in the Term Loan Intercreditor Agreement and (iv) the permitted second lien actions will include the ability for a second lien secured party to take actions to prove, preserve, perfect, preserve or protect (but not enforce) its rights in and perfection and priority of its lien on the Collateral.

The intercreditor agreement between the First Lien Administrative Agent, the trustee on behalf of the Second Lien Notes Purchasers and the ABL Administrative Agent shall be (a) consistent with and substantially similar to that certain ABL Intercreditor Agreement, dated as of October 1, 2018, among Bank of America, N.A., in its capacities as ABL collateral agent and first lien collateral agent and acknowledged and agreed to by Specialty Building Products Holdings, LLC and the other parties thereto and (b) otherwise consistent with the Documentation Principles (the "ABL/Term Intercreditor Agreement").

Representations and Warranties:

Limited to the following (to be applicable to the Borrower and its restricted subsidiaries), subject in all respects to the Certain Funds Provision: organization; existence, qualification and power; compliance with laws; due authorization of the First Lien Credit Documentation; no violation of or conflict with laws, organizational documents or agreements; material governmental

approvals; execution, delivery and enforceability of the First Lien Credit Documentation; projections as of the Closing Date; no Material Adverse Effect (after the Closing Date); litigation; labor matters; ownership of property; environmental matters; taxes; ERISA compliance; subsidiaries as of the Closing Date; margin regulations; Investment Company Act; disclosure; intellectual property; creation, validity and perfection of security interests in the Collateral (subject to permitted liens and the Certain Funds Provision); anti-terrorism laws; FCPA; PATRIOT Act; OFAC; accuracy in all material respects of information included in the Beneficial Ownership Certificate (as defined in Exhibit D) as of the Closing Date; and solvency at closing.

"Material Adverse Effect" means (I) on the Closing Date, for the representations with respect to the Company, a Beta Material Adverse Effect and (II) on the Closing Date (other than as described in clause (I)) and after the Closing Date (a) a material and adverse effect on the business, financial condition or results of operations of the Borrower and its restricted subsidiaries, taken as a whole, (b) a material and adverse effect on the rights or remedies, taken as a whole, of the First Lien Administrative Agent or any First Lien Lender under the First Lien Credit Documentation or (c) a material and adverse effect on the ability of the Loan Parties, taken as a whole, to perform their material payment obligations under the First Lien Credit Agreement.

All representations and warranties shall be made on the Closing Date.

Conditions Precedent to Initial Borrowings:

Subject to the Certain Funds Provision, the availability of the First Lien Term Facility on the Closing Date will be subject only to the Exclusive Funding Conditions.

Conditions Precedent to Ongoing Borrowings:

After the Closing Date, except as set forth above for Incremental First Lien Facilities, each extension of credit will be conditioned upon: (a) delivery of notice of borrowing, (b) accuracy of representations and warranties in all material respects (unless such representations relate to an earlier date, in which case, such representations shall have been true and correct in all material respects as of such earlier date); provided that any representation or warranty that is qualified as to materiality shall be true and correct in all respects (after giving effect to such qualification therein), and (c) absence of defaults or events of default at the time of, and after giving effect to the making of, such extensions of credit.

Affirmative Covenants:

Subject to the Certain Funds Provision and limited to the following (to be applicable to the Borrower and its restricted subsidiaries): delivery of annual audited financial statements within 120 days after the end of each fiscal year (but with up to 150 days after the end of the applicable fiscal year for the first such delivery) (accompanied by an audit opinion from a nationally recognized accounting firm that is not subject to qualification or exception as to the scope of such audit or any "going concern" qualification or exception (other than a "going concern" qualification or exception that is due to the impending maturity of any indebtedness or any prospective or actual default or event of default of any financial covenant)) and delivery of quarterly financial statements for the first three fiscal quarters of each fiscal year within 45 days after the end of each such fiscal quarter (but with up to 60 days after the end of each fiscal quarter for the first three such deliveries); compliance certificates within five business days after delivery of the financial statements for the first three fiscal quarters of each fiscal year and the annual audited financial statements; annual budgets within 90 days after the beginning of each fiscal year (which shall, for the avoidance of doubt, only be delivered to private-side lenders); notices of events of default and other ERISA, litigation or environmental events that would reasonably be expected to have a Material Adverse Effect; payment of taxes; maintenance of existence; maintenance of properties; maintenance of insurance; use of commercially reasonable efforts to obtain and maintain public ratings (but not a specific rating) from Moody's Investors Services, Inc. and Standard & Poor's Financial Services LLC, a subsidiary of S&P Global Inc.; compliance with laws (including compliance with environmental laws); books and records; inspection rights; covenant to guarantee obligations and give security (including but not limited to compliance with requests with respect to lien matters and title insurance matters; provided that, in no event shall environmental reports be required); changes in fiscal year without prior notice to the First Lien Administrative Agent; use of proceeds; further assurances as to security; other information (including to comply with Beneficial Ownership Regulation (as defined in Exhibit D)); and designation of subsidiaries.

Negative Covenants:

Limited to the following (to be applicable to the Borrower and its restricted subsidiaries, shall be incurrence-based covenants customary for high yield debt securities): limitations on liens; investments; debt; fundamental changes; dispositions; restricted payments; transactions with affiliates (with an exception for annual management fees, transaction fees, termination fees, expenses and indemnities payable to Sponsor and Walgreens, with annual management fees capped at an amount to be mutually

agreed); restrictions on subsidiary distributions and negative pledge clauses; amending certain material contractually payment or lien subordinated debt ("<u>Junior Debt</u>") in violation of any applicable subordination or intercreditor agreement.

Baskets and exceptions to the foregoing covenants will be set forth in the First Lien Credit Documentation and, in any event, will include, among others:

(i) indebtedness (which shall be defined to exclude any obligation to pay the redemption price for the Company's preferred stock with Preferred Redemption Cash) under Incremental Facilities, Incremental Equivalent Debt, Refinancing Debt, the Second Lien Notes (including incremental facilities, incremental equivalent debt and refinancing debt thereunder as contemplated by Exhibit B to the Second Lien Commitment Letter as in effect on the date hereof) and the ABL Facility (including incremental facilities as contemplated by Exhibit C), (ii) indebtedness subject to compliance, on a pro forma basis after giving effect to such incurrence of indebtedness and any related acquisition, disposition, debt retirement or other specified transactions and customary pro forma adjustments, including pro forma cost savings and synergy addbacks as set forth in the definition of Consolidated EBITDA, with (x) in the case of indebtedness which is secured on a pari passu basis with the First Lien Term Facility, subject to a pro forma Consolidated First Lien Net Leverage Ratio no greater than the Consolidated First Lien Net Leverage Ratio as of the Closing Date, (y) in the case of indebtedness which is secured on a junior lien basis to the First Lien Term Facility, subject to a pro forma Consolidated Senior Secured Net Leverage Ratio no greater than the Consolidated Senior Secured Net Leverage Ratio as of the Closing Date or (z) in the case of indebtedness which is unsecured, subject to a minimum Fixed Charge Coverage Ratio of 2.00:1.00 (such indebtedness in clauses (x), (y) or (z), "Ratio Debt") (computed as of the most recent four quarter period of the Borrower for which financial statements are internally available); provided that (a) any such debt incurred by Loan Parties shall be subject to maturity and weighted average life provisions substantially identical to the Incremental Facility provision and (b) any such debt incurred by a restricted subsidiary that is not a Guarantor shall be subject to a sublimit to be mutually agreed shared with the baskets specified below (the "Shared Non-Loan Party Sublimit"), (iii) unsecured indebtedness in an amount equal to any cash qualified equity contributions (other than in connection with a Specified Equity Contribution and issuance of disqualified equity of the Borrower or any of its restricted subsidiaries) to the extent such equity

contribution shall not be counted for purposes of the builder or other restricted payment baskets set forth below, (iv) indebtedness incurred or assumed in connection with any permitted acquisition so long as, after giving pro forma effect thereto and to any related specified transactions, and customary pro forma adjustments, including pro forma cost savings and synergy addbacks, (A) the Borrower could incur \$1.00 of indebtedness subject to a minimum Fixed Charge Coverage Ratio of 2.00:1.00 or (B) the pro forma Fixed Charge Coverage Ratio would be no less than the Fixed Charge Coverage Ratio immediately prior thereto; provided that (w) any such debt shall be subject to maturity and weighted average life provisions substantially identical to the Incremental Facility provision, (x) such indebtedness incurred by a restricted subsidiary that is not a Guarantor shall be subject to the Shared Non-Loan Party Sublimit and (y) any liens securing debt assumed or incurred pursuant to this clause (iv) shall be permitted under clause (b)(ii), (b)(iii), (b)(iv) or (b)(v) below, (v) earnout and other contingent consideration obligations and deferred purchase price obligations, (vi) to the extent a joint venture constitutes a restricted subsidiary, indebtedness of joint ventures in an amount to be mutually agreed and (vii) indebtedness incurred under baskets to be set forth in the First Lien Credit Documentation, including a general debt basket, a non-Guarantor debt basket, a capital lease and purchase money debt basket, an intercompany debt basket for loans by the Borrower or any Guarantor to any non-Guarantor);

(b) liens (i) on the Collateral securing the Second Lien Notes (including incremental equivalent debt and refinancing debt thereunder as contemplated by Exhibit B to the Second Lien Commitment Letter as in effect on the date hereof), the ABL Facility (including incremental facilities, incremental equivalent debt and refinancing debt thereunder as contemplated by Exhibit C), Incremental Facilities, Incremental Equivalent Debt and Refinancing Debt otherwise permitted pursuant to the terms hereof (which liens may be pari passu with or junior to the then existing First Lien Term Facility, in each case to the extent permitted as set forth in "Incremental Facility" or "Refinancing Facilities" above, as applicable), in each case subject to customary intercreditor arrangements, (ii) securing indebtedness of the Borrower or a Restricted Subsidiary subject to (x) in the case of first lien indebtedness, a maximum pro forma Consolidated First Lien Net Leverage Ratio as of the Closing Date and (y) in the case of junior lien indebtedness, a maximum pro forma Consolidated Senior Secured Net Leverage Ratio not greater than the Consolidated Senior Secured Net Leverage Ratio as of the Closing

Date, in each case subject to (A) customary intercreditor arrangements and (B) substantially similar maturity and weighted average life to maturity as are applicable to Incremental Equivalent Debt as set forth in the "Incremental Facilities" section of this Term Sheet and any indebtedness secured on a first lien basis under this clause (ii) in the form of a first lien term loan incurred by Loan Parties obtained on or prior to the date that is 6 months after the Closing Date shall be subject to the "MFN" requirement in clause (vi) of the "Incremental Facilities" section; (iii) assumed in connection with permitted acquisitions (but not incurred in contemplation thereof) (subject to customary restrictions in scope related to liens securing such indebtedness unless any expansion of such liens is otherwise permitted); (iv) incurred under a general basket to be mutually agreed, provided that indebtedness secured pursuant to the general liens basket secured by the Collateral on a pari passu basis with the First Lien Term Loans will be subject to cap of \$25 million (with no "grower" component), (v) on assets of non-Guarantor subsidiaries securing indebtedness and other obligations of non-Guarantor subsidiaries that are otherwise permitted and (vi) subject to the Specified Post-Closing Undertaking, liens securing obligations owed to suppliers and vendors of inventory;

(c) restricted payments (including dividends, restricted investments and prepayments of Junior Debt, including, for the avoidance of doubt, the Second Lien Notes), (i) from a cumulative "builder" or "growth" basket to be based on either (as selected by the Borrower prior to the launch of general syndication) (x) 50% of cumulative Consolidated Net Income from the first day of the fiscal quarter in which the Closing Date occurs (which will in no event be less than zero) or (y) retained excess cash flow (which in no event shall be less than zero for any fiscal year) plus a starter amount (the "Starter Basket") of the greater of (A) \$75 million and (B) 35.0% of Consolidated EBITDA for the most recently ended four fiscal quarter period (which shall also include the proceeds of equity issuances and contributions (other than excluded equity contributions, Specified Equity Contributions and issuances of disqualified equity of the Borrower or any of its restricted subsidiaries), proceeds from the issue of debt securities by the Borrower that are converted or exchanged for equity interests of the Borrower or any direct or indirect parent thereof (other than excluded equity contributions, Specified Equity Contributions and disqualified equity), Declined Proceeds, and other customary items) (the "Available Amount") which (x) in the case of restricted payments (other than restricted investments) shall be subject to there being no event of default at the time of declaration thereof and, to the extent made in reliance upon the

50% of cumulative Consolidated Net Income or retained excess cash flow prong, the Consolidated Total Net Leverage Ratio on a pro forma basis not exceeding the Consolidated Total Net Leverage Ratio as of the Closing Date and (y) in the case of restricted investments shall be subject to there being no event of default at the time of declaration thereof, (ii) from the proceeds of any excluded equity contributions (other than Specified Equity Contributions and issuances of disqualified equity of the Borrower or any of its restricted subsidiaries) that do not increase the basket in clause (i) above, (iii) for customary permitted investments (including permitted acquisitions consistent with high yield debt securities (but subject to, in the case of acquisitions funded by Borrower and the Guarantors, a basket to be mutually agreed to permit acquisitions by Loan Parties of non-Guarantors (or, in the case of asset acquisitions, acquisitions by Loan Parties of assets which are not Excluded Assets and which do not constitute Collateral)) and intercompany investments and indebtedness (A) in the ordinary course of business among the Borrower and its restricted subsidiaries or (B) otherwise subject to a sublimit or cap in investments by Loan Parties in restricted subsidiaries that are not Guarantors, and other intercompany exceptions to be agreed so long as such debt owed from a Loan Party to a non-Loan Party is subordinated to the First Lien Term Facility, (iv) with unlimited amounts subject (A) in the case of restricted investments to pro forma compliance with a maximum Consolidated Total Net Leverage Ratio of 0.50x less than the Consolidated Total Net Leverage Ratio as of the Closing Date and (B) in the case of other restricted payments, to pro forma compliance with a maximum Consolidated Total Net Leverage Ratio of 1.00x less than the Consolidated Total Net Leverage Ratio as of the Closing Date, provided that no event of default exists or would result therefrom, (v) [reserved], (vi) promissory notes issued by employees or certain other specified persons in connection with any permitted acquisition by the Borrower or any of its restricted subsidiaries in which such employee or specified person purchases equity interests in the target with funds advanced by the Borrower and its subsidiaries, subject to the conditions that (x) no payment or bankruptcy event of default exists or would result therefrom and (y) the acquisition of the target is otherwise permitted; (vii) loans and advances to employees or investment advisors so long as such loan or advance (x) constitutes an advance of one-time payment for the purpose of recruitment or retention or (y) is made for ordinary course business purposes, such as travel and relocation, (viii) cash payments made after the Closing Date in respect of performance-based or time-vested restricted stock units (in each case that are existing (and either vested or unvested) as of the

Closing Date) in an aggregate amount of up to an amount to be set forth in the Senior Facilities Documentation, (ix) investments in Unrestricted Subsidiaries in an amount to be mutually agreed (provided that such amount may be used to increase the general restricted investment baskets), (x) restricted payments in an aggregate amount per annum not to exceed the sum of 6% of market capitalization, (xi) customary AHYDO catch-up payments with respect to any indebtedness of the Borrower or its subsidiaries or any direct or indirect parent of the Borrower, (including an AHYDO payment tied to certain preferred investments) to the extent the net cash proceeds of such indebtedness are received by the Borrower or its subsidiaries and such proceeds are not used to build capacity under the Available Amount or any other basket, (xii) [reserved], (xiii) customary tax distributions, (xiv) the payment of any earn-out in connection with permitted acquisitions and other permitted investments, (xv) [reserved], and (xv) from baskets (including (I) repurchases or other acquisitions of equity interests of the Borrower or any direct or indirect parent thereof resulting from the forgiveness of loans and advances up to an amount to be set forth in the First Lien Credit Documentation, (II) repurchases or other acquisitions of equity interests of the Borrower or any direct or indirect parent thereof from any future, present or former employee, director, officer, consultant or other specified person pursuant to any employee, management or director equity plan, employee, management or director agreement, stock option plan or any other employee, management or director benefit plan or agreement or any shareholder or co-investor agreement in an amount to be agreed, (III) baskets for investments in joint ventures and similar businesses (provided that such amount may be used to increase the general restricted investment baskets) in an amount to be mutually agreed and (IV) a general investments basket and a general restricted payments basket), to be set forth in the First Lien Credit Documentation; and

(d) (i) subject to compliance with the mandatory prepayment requirements for asset sales and other dispositions of property (but in all respects subject to the reinvestment rights set forth therein), asset sales and other dispositions of property on an unlimited basis for fair market value as long as at least 75% of the consideration in excess of a threshold amount consists of cash or cash equivalents (subject to customary exceptions to the cash consideration requirement, including a basket for non-cash consideration that may be designated as cash consideration), (ii) consolidation of billing centers and (iii) other asset dispositions set forth in the First Lien Credit Documentation.

Each covenant (and definitions used therein) shall also (a) include additional customary baskets, exceptions and thresholds and as may otherwise be agreed, including customary specific and general dollar baskets (each with growth components based on a percentage of Consolidated EBITDA), (b) with respect to the indebtedness, liens, investments and restricted payment covenants, permit classification and reclassification from time to time by the Borrower among one or more available baskets within such applicable covenant (but it being understood and agreed that restricted payment capacity can be used to make restricted investments) and (c) permit reliance on one or more available exceptions and baskets at the Borrower's option and if such exceptions and baskets includes a combination of fixed amounts (including any related builder or grower component) and amounts permitted under incurrence-based tests in concurrent transactions, a single transaction or a series of related transactions, any incurrence-based tests shall be calculated without giving effect to the utilization of such fixed amounts.

Financial Covenant:

None.

Unrestricted Subsidiaries:

The First Lien Credit Documentation will contain provisions pursuant to which, subject to the investments covenant in the First Lien Credit Documentation, and, so long as no payment or bankruptcy (with respect to the Borrower) event of default under the First Lien Term Facility exists or would result therefrom, the Borrower will be permitted to designate any existing or subsequently acquired or organized subsidiary as an "unrestricted subsidiary" and subsequently redesignate any such unrestricted subsidiary as a restricted subsidiary; <u>provided</u> that in no event shall any unrestricted subsidiary acquire any material intellectual property from the Borrower or any restricted subsidiary. Unrestricted subsidiaries will not be subject to the mandatory prepayments, representations and warranties, affirmative or negative covenant or event of default provisions of the First Lien Credit Documentation and the results of operations and indebtedness of unrestricted subsidiaries will not be taken into account for purposes of determining any financial ratio or covenant contained in the First Lien Credit Documentation.

Events of Default:

Limited to the following (to be applicable to the Borrower and its restricted subsidiaries: nonpayment of principal, interest, fees or other amounts (with grace period of 5 business days for interest, fees and other amounts); failure to perform negative covenants (and affirmative covenants to provide notice of an event of default (provided that, subject to the proviso to the last sentence of this paragraph, the delivery of such notice shall cure the event of

default resulting from the failure to timely deliver such notice) or maintain the Borrower's corporate existence); failure to perform other covenants subject to a 30-day cure period after written notice from the First Lien Administrative Agent; any representation or warranty is incorrect in any material respect when made or deemed made (subject (i) in the case of all representations made on the Closing Date (other than the Specified Representations) capable of being cured to a 90 day grace period and (ii) in the case of all representations made after the Closing Date capable of being cured to a 30 day grace period); cross-default or cross-acceleration to other indebtedness (other than under the First Lien Credit Documentation) subject to a threshold amount to be set forth in the First Lien Credit Documentation; provided, that, with respect to the ABL Facility, (i) there shall be no cross default except with respect to (x) any payment or bankruptcy event of default under the ABL Facility or (y) any event of default under the ABL Facility that remains unremedied or unwaived for 60 consecutive days after the occurrence thereof or with respect to which the agent or the lenders under the ABL Facility are exercising remedies (including acceleration of obligations or termination of commitments) and (ii) the occurrence of a financial covenant event of default under the ABL Facility shall not constitute an event of default under the First Lien Term Loan Facility until the earlier of (x) the exercise of any remedies by the agent or the lenders under the ABL Facility or (y) acceleration of the obligations or termination of the commitments under the ABL Facility; bankruptcy events or other insolvency events of the Borrower or its material restricted subsidiaries (with a customary grace period for involuntary events); monetary final judgment defaults subject to a threshold amount (to the extent not covered by insurance or indemnity) that have not been satisfied, vacated, discharged or stayed or bonded pending an appeal within a customary grace period; ERISA events subject to Material Adverse Effect; invalidity (actual or asserted in writing by the Borrower or any Guarantor) of the First Lien Credit Documentation or any material guarantees or material portion of the Collateral; and change of control (to be defined in the First Lien Credit Documentation in a manner customary for a public company). Any event of default under the First Lien Credit Documentation, other than any event of default which cannot be waived without the written consent of each Lender directly and adversely affected thereby, shall be deemed not to be "continuing" (and shall be deemed to be "cured") if the events, acts or conditions that gave rise to such event of default have been remedied or cured (including by payment, notice, taking any action or omitting to take any action) or have ceased to exist and

the Borrower is otherwise in compliance with the First Lien Credit Documentation; provided, that the foregoing shall not be applicable with respect to any default or event of default if the Borrower knowingly and willfully fails to give timely notice to the First Lien Administrative Agent and the Lenders of such default or event of default required to be given under the First Lien Credit Documentation.

Voting:

Amendments and waivers of the First Lien Credit Documentation will require the approval of First Lien Lenders holding more than 50% of the aggregate principal amount of the First Lien Term Loans and commitments under the First Lien Term Facility (the "Required First Lien Lenders"), except that the consent of each First Lien Lender directly adversely affected thereby (but not, for the avoidance of doubt, the consent of the Required First Lien Lenders) shall be required, unless otherwise contemplated by the First Lien Credit Documentation, with respect to (I) (a) increases in the commitment of such First Lien Lender (it being understood that a waiver of any condition precedent or the waiver of any condition precedent, default, event of default or mandatory prepayment shall not constitute an increase of any commitment of any First Lien Lender), (b) reductions of principal or interest (other than a waiver of default interest, a default, event of default or mandatory prepayment) or fees (it being understood that any change in the definitions of any ratio used in the calculation of any rate of interest or fees (or the component definitions) shall not constitute a reduction in any rate of interest or fees), (c) extensions of final maturity or the scheduled due date of any amortization principal payment, interest or fee payment and (d) changes in the application of proceeds amongst First Lien Lenders after an event of default and, only following the exercise of remedies, changes in other pro rata sharing provisions and (II) the consent of 100% of the First Lien Lenders (other than Defaulting Lenders) will be required with respect to (a) releases of all or substantially all of the value of the Guarantees or all or substantially all of the Collateral (other than in connection with permitted asset sales) and (b) reductions in voting thresholds. Defaulting Lenders will be subject to the suspension of certain voting rights; provided, that the consent of the First Lien Administrative Agent shall be required with respect to amendments and waivers directly adversely affecting its rights or duties. Notwithstanding the foregoing, (i) any waiver or modification of a condition to an extension of credit under any Incremental First Lien Facility (prior to funding thereof) and any amendments and waivers that affect solely the First Lien Lenders under a class or classes of any Incremental First Lien Facility (prior to funding thereof) and not any other Lender, will, if such amendment or waiver

otherwise require the consent of the Required First Lien Lenders, require only the consent of First Lien Lenders holding more than 50% of the aggregate commitments under such class or classes (in the aggregate), and no other consents or approvals shall be required (it being understood and agreed that Required First Lien Lender consent shall be required for any amendment or waiver of the conditions set forth under the heading "Incremental First Lien Facilities," except as otherwise set forth therein) and (ii) otherwise there shall be no "class" voting under the First Lien Credit Documentation.

The First Lien Credit Documentation will permit amendments thereof without the approval or consent of the Lenders to effect a permitted "repricing transaction" (i.e., a transaction in which any tranche of First Lien Term Loans is refinanced with a replacement tranche of term loans, or is modified with the effect of, bearing a lower All-In Yield) other than any First Lien Lender holding First Lien Term Loans subject to such "repricing transaction" that will continue as a Lender in respect of the repriced tranche of First Lien Term Loans or modified First Lien Term Loans.

Modifications to provisions requiring pro rata payments or sharing of payments (other than as set forth in clause (I)(d) in the second preceding paragraph) shall only require approval of the Required First Lien Lenders and non-pro rata distributions and commitment reductions will be permitted in connection with loan buy back or similar programs, "amend and extend" transactions or the addition of one or more tranches of debt as permitted by the First Lien Credit Documentation.

The First Lien Credit Documentation will permit the First Lien Administrative Agent and the Borrower (i) to enter into one or more amendments thereto to incorporate the provisions of any relevant Incremental Facility, Refinancing Facility or extended tranche of Loans or commitments in connection with "amend and extend" transactions made available without any First Lien Lender's consent, so long as the purpose of such amendment is solely to incorporate the appropriate provisions for such Incremental Facility, Refinancing Facility or extended tranche of Loans or commitments in connection with "amend and extend" transactions in the First Lien Credit Documentation; it being understood that (A) each Lender under the applicable tranche or tranches that are being extended shall have the opportunity to participate in such extension on the same terms and conditions as each other First Lien Lender in such tranche or tranches and (B) such extension shall not be subject to any condition that there does not exist any default or event of default, (ii) to amend the First

Lien Credit Documentation to correct any obvious error or any error or omission of a technical nature in the First Lien Credit Documentation without any further action or consent of any other party with notice given to the First Lien Lenders of any such amendment and (iii) to enter into or amend any applicable intercreditor agreement to give effect thereto or carry out the provisions thereof.

Cost and Yield Protection:

The First Lien Credit Documentation shall contain customary tax gross-up and cost and yield protection provisions consistent with the First Lien Documentation Principles.

Assignments and Participations:

The First Lien Lenders will be permitted to assign (other than to natural persons and any Disqualified Lender (which Disqualified Lenders shall be specified on a schedule that is held with the First Lien Administrative Agent, which may be communicated to a First Lien Lender upon request, but shall not be "posted" or otherwise disclosed by any Lead Arranger or the First Lien Administrative Agent to any other person) and, other than as set forth below, the Sponsor and its affiliates (other than Investment Funds)) loans under the First Lien Term Facility with the consent of the Borrower (not to be unreasonably withheld or delayed); provided that no consent of the Borrower shall be required (i) if such assignment is made to another First Lien Lender or an affiliate or approved fund of a First Lien Lender or (ii) after the occurrence and during the continuance of a payment or bankruptcy (with respect to the Borrower) event of default; provided, further, that if not received within 10 business days of acknowledged receipt of a written request by the First Lien Administrative Agent, the consent of the Borrower with respect to Term Loan assignments only shall be deemed to have been given. All assignments will require the consent of the First Lien Administrative Agent, not to be unreasonably withheld, conditioned or delayed; provided that no consent of the First Lien Administrative Agent shall be required for an assignment (i) of all or any portion of a First Lien Term Loan to another First Lien Lender or an affiliate or approved fund of a First Lien Lender or (ii) of all or any portion of the First Lien Term Loans to the Borrower or any of its subsidiaries or the Sponsor or any of its affiliates as set forth below. Each assignment, other than any assignment made by an MBD Lender to another MBD Lender, will be in an amount of an integral multiple of \$1,000,000 with respect to the First Lien Term Facility or if less, all of such First Lien Lender's remaining loans and commitments of the applicable class or tranche. An assignment fee in the amount of \$3,500 shall be paid by the respective assignor or assignee to the First Lien Administrative Agent (which fee may be waived or reduced in the

sole discretion of the First Lien Administrative Agent); provided that no such fee shall be payable in connection with any assignment by an MBD Lender to any other MBD Lender.

The First Lien Lenders will be permitted to sell participations (other than to natural persons, Disqualified Lenders and the Sponsor and its affiliates (other than Investment Funds)) in loans and commitments without consent being required, subject to customary limitations consistent with the First Lien Documentation Principles. Voting rights of participants shall be limited to matters in respect of (a) increases in commitments participated to such participants (it being understood that a waiver of any condition precedent or the waiver of any default, event of default or mandatory prepayment shall not constitute an increase of any commitment of any First Lien Lender), (b) reductions of principal or interest (other than a waiver of default interest, a default, event of default or mandatory prepayment) or fees (it being understood that any change in the definitions of any ratio used in the calculation of any rate of interest or fees (or the component definitions) shall not constitute a reduction in any rate of interest or fees), (c) extensions of final maturity or the due date of any amortization, interest or fee payment (other than with respect to any extension facility), (d) changes in the application of proceeds amongst First Lien Lenders after an event of default and, only following the exercise of remedies, changes in other pro rata sharing provisions and (e) releases of all or substantially all of the value of the Guarantees or all or substantially all of the Collateral, in each case, with respect to which the affirmative vote of the First Lien Lender from which it purchased a participation would be required.

The First Lien Administrative Agent shall not be responsible or have any liability for, or have any duty to ascertain, inquire into, monitor or enforce, compliance with the provisions hereof relating to Disqualified Lenders. Without limiting the generality of the foregoing, the First Lien Administrative Agent shall not (x) be obligated to ascertain, monitor or inquire as to whether any First Lien Lender or participant or prospective Lender or participant is a Disqualified Lender or (y) have any liability with respect to or arising out of any assignment or participation of commitments or loans, or disclosure of confidential information, to any Disqualified Lender.

The First Lien Credit Documentation shall provide that First Lien Term Loans, loans under any Incremental First Lien Term Facility and loans under any Refinancing Term Facility may be purchased by and assigned to the Sponsor, its affiliates (other than the

(a) open market purchases and/or (b) Dutch auctions open to all Lenders on a pro rata basis in accordance with customary procedures; provided that (i) commitments and loans owned or held by the Sponsor or its affiliates (excluding any Investment Funds) shall be excluded in the determination of any Required First Lien Lender votes (other than any vote which would affect the Sponsor or its applicable affiliate differently from other Lenders of the same class or tranche or deprive the Sponsor or its applicable affiliate of its pro rata share of any payment to which all Lenders of the same class or tranche are entitled), (ii) commitments and loans owned or held by the Sponsor or its affiliates (excluding any Investment Funds) shall not, in the aggregate for all such persons, exceed 25% of the aggregate commitments and loans under the First Lien Term Facility, any Incremental First Lien Term Facility and any Refinancing Term Facility (measured at the time of purchase or assignment and in the aggregate amount of the term facilities (taken as a whole)), (iii) neither the Sponsor nor any of its affiliates (excluding Investment Funds) shall be permitted to attend any "lenderonly" meetings or conference calls or receive any related "lender-only" information or receive advice of counsel to the First Lien Administrative Agent or the Lenders, (iv) the Sponsor and its Non-Debt Fund Affiliates, as applicable, shall agree that the First Lien Administrative Agent shall vote on behalf of the Sponsor or such Non-Debt Fund Affiliate in connection with a plan of reorganization under any insolvency proceeding in the same proportion as allocations of voting with respect to such matter by those First Lien Lenders that are not the Sponsor or a Non-Debt Fund Affiliate unless the plan of reorganization requires the consent of each First Lien Lender or each adversely affected First Lien Lender or affects the Sponsor or such Non-Debt Fund Affiliate, as applicable, in its capacity as a First Lien Lender in a disproportionately adverse manner than its effect on the other Lenders under the First Lien Term Facility and (v) for purposes of determining whether the Required First Lien Lenders have consented to any amendment or waiver under the definitive documentation for the First Lien Credit Documentation, the Investment Funds may not account for more than 49.9% of all loans and commitments of Lenders (whether consenting or not) included in determining whether the Required First Lien Lenders have consented to such amendment or waiver. Notwithstanding the foregoing, the First Lien Credit Documentation shall permit (but not require) the Sponsor or its affiliates to contribute any First Lien Term Loans, loans under any Incremental First Lien Term Facility or loans under any Refinancing Term Facility acquired to

Borrower or any of its subsidiaries and other than natural persons) and Investment Funds on a non-pro rata basis through

the Borrower or any of its subsidiaries for purposes of cancelling such debt, which may include contribution (with the consent of the Borrower) to the Borrower in exchange for debt or equity securities of the Borrower that are otherwise permitted to be issued by the Borrower at such time. For purposes of the foregoing, "Investment Fund" means any bona fide debt fund or an investment vehicle that is engaged in the making, purchasing, holding or otherwise investing in commercial loans, bonds and similar extensions of credit in the ordinary course of business and with respect to which Madison Dearborn Partners, LLC and investment vehicles managed or advised by Madison Dearborn Partners, LLC that are not engaged primarily in making, purchasing, holding or otherwise investing in commercial loans, bonds and similar extensions of credit in the ordinary course have fiduciary duties to the third-party investors in such fund or investment vehicle independent of their duties to the Borrower or the Sponsor and do not make investment decisions for such entities.

In connection with any purchase or assignment permitted under the First Lien Credit Documentation, neither the Sponsor nor its Non-Debt Fund Affiliates will be required to make any representation regarding the possession of material non-public information but may address any purchases with customary "big boy" language in a manner consistent with the First Lien Documentation Principles.

In addition, the First Lien Credit Documentation shall provide that so long as no event of default is continuing First Lien Term Loans, loans under any Incremental First Lien Term Facility and loans under any Refinancing Term Facility may be purchased by and assigned to the Borrower or any of its subsidiaries on a non-pro rata basis through (a) open market purchases and/or (b) Dutch auctions open to all First Lien Lenders on a pro rata basis in accordance with customary procedures; provided that (i) no event of default shall have occurred and be continuing and (ii) any such loans acquired by the Borrower or any of its subsidiaries shall be automatically retired and cancelled promptly upon acquisition thereof (or contribution thereto, including as contemplated by the preceding paragraph).

The First Lien Credit Documentation will contain customary provisions consistent with the First Lien Documentation Principles allowing the Borrower to replace a First Lien Lender or terminate the commitment of a First Lien Lender and prepay that First Lien Lender's outstanding loans in full (in the case of any amendment constituting a Repricing Transaction, along with any "prepayment"

premium", if applicable, that would otherwise be owed in connection therewith) in connection with amendments and waivers requiring the consent of all First Lien Lenders or all First Lien Lenders of an applicable class or tranche or of all Lenders directly adversely affected thereby or all First Lien Lenders of an applicable class or tranche directly adversely affected thereby (so long as the Required First Lien Lenders, or a majority in aggregate principal amount of First Lien Lenders in respect of the applicable class, tranche or facility, have approved the amendment or waiver), increased costs, taxes, etc. and Defaulting Lenders.

Expenses and Indemnification:

If the Closing Date occurs, the Borrower shall pay promptly following written demand (including documentation reasonably supporting such request) (a) all reasonable and documented out-of-pocket expenses of the First Lien Administrative Agent, the Issuing Lender and the Initial First Lien Lenders associated with the syndication of the First Lien Term Facility and the preparation, negotiation, execution, delivery and administration of the First Lien Credit Documentation and any amendment or waiver with respect thereto (but limited, (i) in the case of legal fees and expenses to be paid in connection with the consummation of the Transactions, as provided in Section 7(b) of the body of the Commitment Letter, (ii) in the case of all other legal fees and expenses, to the reasonable fees, disbursements and other charges of one primary counsel to the First Lien Administrative Agent, the Issuing Lender, the Senior Lead Arrangers and the First Lien Lenders, taken as a whole, plus, if reasonably necessary, one local counsel in each applicable material jurisdiction, in each case excluding allocated costs of in-house counsel and (iii) in the case of other consultants and advisers, limited to the fees and expenses of such persons approved by the Borrower) and (b) all reasonable and documented out-of-pocket expenses of the First Lien Administrative Agent, the Issuing Lender and the First Lien Lenders (but limited, (i) in the case of legal fees and expenses, to the fees, disbursements and other charges of one primary counsel to the First Lien Administrative Agent, the Issuing Lender and the First Lien Lenders, taken as a whole (and, if reasonably necessary, one local counsel in each applicable material jurisdiction, and in the case of an actual conflict of interest, one additional counsel to the affected First Lien Lenders, taken as a whole)) and (ii) in the case of other consultants and advisers, limited to the fees and expenses of such persons approved by the Borrower) in connection with the enforcement of the First Lien Credit Documentation or protection of rights thereunder.

If the Closing Date occurs, the First Lien Administrative Agent, the Issuing Lender and the Senior Lead Arrangers and the First

Lien Lenders (and their respective controlled affiliates and their respective directors, officers, employees, agents and other representatives) (excluding Excluded Affiliates and any financial advisor in such capacity) will be indemnified and held harmless against, any actual losses, claims, damages, liabilities or documented expenses (but limited, in the case of legal fees and expenses, to the reasonable fees, disbursements and other charges of one counsel to the indemnified persons taken as a whole, and, if reasonably necessary, one local counsel in any relevant material jurisdiction (and, in the case of an actual conflict of interest, one additional counsel to the affected indemnified persons, taken as a whole), in each case excluding allocated costs of in-house counsel) incurred in respect of the financing contemplated hereby or the use or the proposed use of proceeds thereof, the transactions in connection therewith, or any other claim, litigation, investigation or proceeding relating to any of the foregoing (regardless of whether such indemnified person is a party thereto and regardless of whether such matter is initiated by a third party, the Borrower, the Company or any of their respective affiliates, subsidiaries or equity holders), except, in the case of any indemnified person, to the extent they arise from (i) the gross negligence, bad faith, fraud or willful misconduct of such indemnified person (or respective controlled affiliates and their respective directors, officers, employees, agents and other representatives), (ii) the material breach of the First Lien Credit Documentation by such indemnified person (or respective controlled affiliates and their respective directors, officers, employees, agents and other representatives) (in the case of each of the preceding clause (i) and this clause (ii), as determined by a court of competent jurisdiction in a final non-appealable judgment), (iii) any disputes solely among indemnified persons (other than any claims against an indemnified person in its capacity or in fulfilling its role as the First Lien Administrative Agent, a Senior Lead Arranger or any similar role under the First Lien Term Facility unless such claim would otherwise be excluded pursuant to clause (i) above) and other than any claims arising out of any act or omission of the Borrower or any of its subsidiaries or (iv) settlements effected without the Borrower's prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned), but if settled with the Borrower's written consent, or if there is a final judgment against an indemnified person in any such proceeding, the Borrower shall indemnify and hold harmless each indemnified person to the extent and in the manner set forth above; provided that each indemnified person (by accepting the benefits hereof) agrees to refund and return any and all amounts paid by the Borrower to

such indemnified person to the extent that any of the foregoing items in clause (i) through (iv) above occurs and to the extent that such indemnified person is not entitled to the payment thereof pursuant to the terms of this paragraph as determined by a court of competent jurisdiction in a final non-appealable judgment.

Governing Law and Forum:

New York.

Counsel to the First Lien Administrative Agent and Commitment Parties: Davis Polk & Wardwell LLP.

Interest Rates:

The interest rates under the First Lien Term Facility will be as follows:

At the option of the Borrower, initially, Adjusted LIBOR plus 4.50% or ABR plus 3.50%. From and after the delivery by the Borrower to the First Lien Administrative Agent of the Borrower's financial statements for the period ending one full fiscal quarter following the Closing Date, interest rates under the First Lien Term Facility shall be subject to one 25 basis point reduction based upon achieving either (a) a Consolidated First Lien Net Leverage Ratio of 0.50x less than the Consolidated First Lien Net Leverage Ratio as of the Closing Date as set forth in the applicable officer's certificate or (b) a one notch upgrade of the First Lien Term Loan by both Moody's Investors Services, Inc. and Standard & Poor's Financial Services LLC, a subsidiary of S&P Global Inc.; provided that for the avoidance of doubt, in no event will the applicable margin under the First Lien Term Facility be reduced more than 25 basis points relative to the applicable margin thereunder on the Closing Date.

The Borrower may elect interest periods of 1, 2, 3 or 6 months (or, if agreed by all relevant Lenders, 12 months or a shorter period) for Adjusted LIBOR borrowings.

Calculation of interest shall be on the basis of the actual days elapsed in a year of 360 days (or 365 or 366 days, as the case may be, in the case of ABR loans based on the Prime Rate) and interest shall be payable (i) in the case of Adjusted LIBOR loans, at the end of each interest period and, in any event, at least every 3 months and (ii) in the case of ABR loans, quarterly in arrears.

ABR is the Alternate Base Rate, which is the highest of the First Lien Administrative Agent's Prime Rate, the Federal Funds Effective Rate plus 1/2 of 1.0% and one-month Adjusted LIBOR plus 1.00%.

Adjusted LIBOR is the London interbank offered rate for U.S. dollars, adjusted for customary Eurodollar reserve requirements, if any, subject to a floor of 0.00%.

ABL Facility Summary of Terms and Conditions

Set forth below is a summary of the principal terms and conditions for the ABL Facility. Capitalized terms used but not defined in this <u>Exhibit C</u> shall have the meanings set forth in the letter to which this <u>Exhibit C</u> is attached or in <u>Exhibits A</u>, <u>B</u>, or D attached thereto.

PARTIES

Borrowers:

Initially, the Initial Borrower and immediately following consummation of the Debt Assumption, the Company (such relevant entity, the "Borrower"); provided that, the Borrower may, in its sole discretion, designate one or more of its direct or indirect domestic wholly-owned subsidiaries as co-borrowers subject to customary conditions; provided, further, the Company will act as the borrower representative in accordance with the ABL Credit Documentation (in such capacity, the "Borrower Representative").

Guarantors:

All obligations of the Borrower (the "Borrower Obligations") under the ABL Facility and, at the option of the Borrower Representative, any swap obligation (other than swap obligations designated as secured obligations under the First Lien Term Facility and Excluded Swap Obligations) and cash management obligation owed by the Borrower or any of its restricted subsidiaries to (x) any ABL Lead Arranger, the ABL Administrative Agent, any ABL Lender or any affiliate of any of the foregoing persons (in each case as of the Closing Date or as of the time of entering into the related swap or cash management arrangement) or (y) solely in the case of cash management obligations expressly designated in writing by the Borrower as "Secured Obligations", other third party cash management providers that are providing cash management services to the Borrower, the Company or their respective subsidiaries as of the Closing Date so long as such third party cash management providers agree to provide to ABL Administrative Agent a customary letter agreement agreeing to the applicable terms of the ABL Facility will be unconditionally guaranteed jointly and severally by the same parties that guarantee the Borrower's obligations under the First Lien Term Facility (collectively, the "Guarantors" and, together with the Borrower, collectively, the "Loan Parties" and each a "Loan Party").

Notwithstanding anything to the contrary contained herein, the requirements of the preceding paragraphs shall be, as of the Closing Date, subject to the limitations set forth in the Commitment Letter, including the Certain Funds Provisions.

All the above-described guarantees shall be created on terms, and pursuant to documentation, consistent with the ABL Documentation Principles (as defined below) or otherwise reasonably satisfactory to the ABL Administrative Agent and the Borrower.

ABL Administrative Agent:

Bank of America, N.A. will act as sole and exclusive administrative agent and collateral agent (in such capacity, the "ABL Administrative Agent" and together with the First Lien Administrative Agent and the Second Lien Administrative Agent, the "Administrative Agent") for a syndicate of banks, financial institutions and institutional lenders reasonably acceptable to the Borrower (such consent not to be unreasonably withheld, delayed or conditioned) (excluding any Disqualified Institutions) (the "ABL Lenders" and together with the First Lien Lenders, the "Lenders"), and will perform the duties customarily associated with such roles.

ABL Lead Arranger:

Bank of America, N.A., and any other "ABL Lead Arranger" appointed pursuant to the Commitment Letter (collectively in such capacity, the "ABL Lead Arrangers" and, each an "ABL Lead Arranger").

ABL FACILITY

Type and Amount:

A \$150 million asset-based revolving facility made available to the Borrowers (the "<u>ABL Facility</u>", and the loans thereunder, the "<u>ABL Loans</u>"; the commitments thereunder, the "<u>ABL Commitments</u>"). The ABL Facility will be available in U.S. Dollars and other currencies to be mutually agreed.

Availability and Maturity:

The final maturity date of the ABL Facility shall be five (5) years from the Closing Date (the "ABL Maturity Date").

Subject to Availability (as defined below), ABL Loans may be borrowed, repaid and reborrowed on and after the Closing Date (without premium or penalty) and prior to the ABL Maturity Date in accordance with the terms of the ABL Credit Documentation referred to below.

Borrowing availability will be equal to (x) the lesser of (a) the aggregate amount of the ABL Commitments and (b) the Borrowing Base (as defined below) (such lesser amount, the "Maximum Borrowing Amount") minus (y) the sum of the aggregate outstanding amount of ABL Loans, Swingline Loans (as defined below), unreimbursed drawings under Letters of Credit (as defined below) and, the undrawn amount of outstanding Letters of Credit (collectively, the "ABL Exposure"). "Availability" means as of any date of determination (i) the aggregate Maximum Borrowing Amount minus (ii) the aggregate ABL Exposure.

ABL Loans, Letters of Credit and Swingline Loans will be available on and after the Closing Date and prior to the ABL Maturity Date as specified below under the headings "Use of Proceeds", "Letters of Credit" and "Swingline Loans", respectively, in minimum principal amounts to be mutually agreed.

Borrowing Base:

The "Borrowing Base" at any time shall equal the sum of, in each case, as set forth in the most recently delivered Borrowing Base Certificate (as defined below):

- (a) the lesser of 85% of the NOLV (as defined below) of eligible inventory of the Loan Parties and 75% of the cost of eligible inventory of the Loan Parties (in each case (i) including in-transit eligible inventory up to a percentage of the Borrowing Base to be mutually agreed, subject to customary conditions in accordance with the ABL Documentation Principles and (ii) excluding consigned and slow-moving inventory); provided that inventory availability from this clause (a) shall not account for more than a percentage to be agreed of the Borrowing Base, plus
- (b) 85% of net eligible accounts receivable (including Medicare, Medicaid and private insurance accounts receivable, but excluding self-pay accounts receivable) of the Loan Parties (with net eligible accounts receivable to include (to the extent not otherwise excluded) (i) accounts receivable that remain unpaid until 120 days after the original invoice date and (ii) unbilled accounts receivable for which the last date of service (or a number of days to be mutually agreed prior to the applicable date of determination) that generated such account receivable did not occur more than 30 days prior to the applicable date of determination); provided that (x) Medicare and Medicaid accounts receivable for which the Borrower has not complied with the Federal Assignment of Claims Act with respect thereto shall not account for more than 35% of the Borrowing Base and (y) availability from unbilled accounts receivable shall not account for more than a percentage to be agreed of the Borrowing Base, plus
- (c) 100% of the eligible cash of the Loan Parties, minus
- (d) such reserves as the ABL Administrative Agent may establish in its Permitted Discretion (as described below).

The definitions of eligible inventory, eligible accounts receivable, eligible cash, unrestricted cash, reserves and each other component or defined term utilized in the calculation of the Borrowing Base will be defined in a manner consistent with this Term Sheet and otherwise consistent with the ABL Documentation Principles; provided the definition of eligible cash shall include (x) cash of the Loan Parties held in deposit accounts maintained with the ABL Administrative Agent or its affiliates that is subject to a control agreement and (y) cash of the Loan Parties held in deposit accounts maintained with the ABL Administrative Agent or its affiliates that qualifies for the de minimis exception in the definition of Excluded Accounts if any such deposit account is subject to a control agreement.

Notwithstanding anything to the contrary herein, in no event shall any property of the Company or its subsidiaries who are subsidiaries of the Company prior to the Closing Date (collectively, the "Beta Entities") be included in the Borrowing Base unless and until the Specified Post-Closing Undertaking (as defined below) has been satisfied, at which point the inclusion of such property in the Borrowing Base shall be subject to receipt by the ABL Administrative Agent of a Collateral Audit with respect to such property and to the other applicable terms and conditions of the ABL Credit Documentation; provided that nothing in this paragraph shall affect the Temporary Borrowing Base Amount (as defined below).

The Borrower Representative will use commercially reasonable efforts to deliver a field examination and inventory appraisal (a "Collateral Audit") prior to the Closing Date for the Borrowing Base; provided, that in the event that such Collateral Audit cannot be completed and delivered on or before the Closing Date, for purposes of the Borrowing Base Certificate (as defined below) to be delivered on or prior to the Closing Date and for each subsequent Borrowing Base Certificate that is required to be delivered after the Closing Date and on or prior to the date that is ninety (90) days after the Closing Date (or such later date as may be agreed to by the ABL Administrative Agent), with respect to the Company and its subsidiaries, the Borrowing Base shall be equal to (A) \$125 million (the "Temporary Borrowing Base Amount") and (B) on or after the date that is ninety-one (91) days after the Closing Date (or such later date as may be agreed to by the ABL Administrative Agent), the Borrowing Base shall equal \$0 until the Collateral Audit has been delivered; provided further, that there shall be no default or event of default solely as a result of a failure to complete and deliver the Collateral Audit within the applicable time period. For the avoidance of doubt, the completion of the Collateral Audit for the Company shall not be a condition to the availability of the ABL Facility on the Closing Date.

"Permitted Discretion" means reasonable (from the perspective of a secured asset-based lender) credit judgment exercised in good faith in accordance with customary business practices of the ABL Administrative Agent for comparable asset-based lending transactions.

"NOLV" means the net appraised recovery value of inventory as set forth in each Borrower's stock ledger and each Guarantor's stock ledgers (expressed as a percentage of the cost of such inventory), as applicable, as reasonably determined from time to time by reference to the most recent appraisal received by the ABL Administrative Agent conducted by an independent appraiser reasonably satisfactory to the ABL Administrative Agent.

Prior to the completion of the applicable Collateral Audit, the portion of the Borrowing Base that is attributable to the assets of the target in any permitted acquisition or other similar investment will (solely to the extent such assets contribute an amount in excess of 10% of the total Borrowing Base without giving effect to inclusion of any such acquired assets) be limited to the lesser of (1) 20% of the total Borrowing Base (after giving effect to inclusion of any such acquired assets) and (2) on or prior to the date of closing of such permitted acquisition (the "Acquisition Date") and (i) for each subsequent Borrowing Base Certificate that is required to be delivered after the Acquisition Date and prior to the date that is ninety (90) days after the Acquisition Date, such Borrowing Base shall include the sum of (x) 70% of the eligible accounts receivable of the target entities and (y) 70% of the estimated NOLV (determined in a manner consistent with the most recent inventory appraisal provided to the ABL Administrative Agent) of eligible inventory of the target entities, (ii) thereafter through the date that is 180 days after the Acquisition Date (or such later date as may be agreed to by the ABL Administrative Agent as provided above), such Borrowing Base shall include the sum of (x) 55% of the net book value of eligible accounts receivable of the target entities and (y) 55% of the estimated NOLV of eligible inventory of the target entities and (iii) thereafter, the Borrowing Base shall not include such assets until the applicable Collateral Audit shall have been delivered; provided, for the avoidance of doubt, prior to the Acquisition Date, no amount shall be included in the Borrowing Base pursuant to this paragraph for purposes of determining the Maximum Borrowing Amount that may actually be borrowed at such time.

The Borrowing Base shall be computed on a monthly basis (or more frequently in the Borrower Representative's sole discretion so long as such increased frequency is maintained for the 60-day period following such election) pursuant to a monthly borrowing base certificate (a "Borrowing Base Certificate") to be delivered by the

Borrower Representative to the ABL Administrative Agent not later than the date that is twenty (20) days after the end of each fiscal month; <u>provided</u>, <u>however</u>, that the Borrowing Base may be required to be computed more frequently (but not more frequently than weekly, with the applicable Borrowing Base Certificate to be delivered within three (3) business days after the end of each week) as reasonably determined by the ABL Administrative Agent during a Liquidity Event Period (as defined below).

A "<u>Liquidity Event</u>" shall occur when a Specified Default (as defined below) has occurred or Specified Availability (as defined below) is less than the greater of (a) 10% of the aggregate Maximum Borrowing Amount (calculated on an aggregate basis) and (b) \$10 million for a period of five (5) consecutive business days. A "<u>Liquidity Event Period</u>" shall commence after the occurrence of a Liquidity Event and shall continue until the date that no Specified Default shall have existed and Specified Availability shall have been not less than the greater of (a) 10% of the aggregate Maximum Borrowing Amount (calculated on an aggregate basis) and (b) \$10 million at any time during the preceding twenty (20) consecutive calendar days; <u>provided</u> that a Liquidity Event Period shall only begin upon the written request of the ABL Administrative Agent, which request may be made in its discretion or at the discretion of the Required ABL Lenders (as defined below) after the occurrence of a Liquidity Event.

"Specified Availability" shall mean at any time the sum of (i) without duplication, Availability <u>plus</u> (ii) the amount (if any, and not to be less than zero) by which (1) the Borrowing Base exceeds (2) the ABL Commitments, in each case at such time; <u>provided</u> that the amount attributable to clause (ii) of this definition shall not exceed 2.5% of the aggregate ABL Commitment.

"Specified Default" shall mean any payment or bankruptcy or insolvency event of default, any event of default arising from failure to comply with Borrowing Base requirements (i.e., failure to deliver a Borrowing Base Certificate or a material breach of a representation and warranty therein), event of default for failure to comply with cash management provisions or event of default for failure to comply with the Fixed Charge Coverage Ratio (as defined below) to the extent then in effect.

The establishment or increase of any reserve against the Borrowing Base shall be limited to such reserves against the Borrowing Base as the ABL Administrative Agent from time to time determines in its Permitted Discretion as being necessary (i) to reflect items that could reasonably be expected to adversely affect the value or collectability

of the eligible inventory and eligible accounts receivable or (ii) to reflect items that could reasonably be expected to adversely affect the enforceability or priority of the ABL Administrative Agent's liens on the applicable ABL Priority Collateral (as defined below) (including a customary rent reserve (subject to the below restrictions) in the absence of landlord waivers in jurisdictions where such Collateral is located, to the extent such location is in a landlord lien priming jurisdiction); provided that no reserves may be taken after the Closing Date (or, if later, the date on which the initial Collateral Audit has been delivered to the ABL Administrative Agent) based on circumstances, conditions, events or contingencies known to the ABL Administrative Agent as of such applicable date and for which no reserves were imposed on such applicable date and no reserves taken on such applicable date may be increased, unless, in each case, such circumstances, conditions, events or contingencies shall have changed in any material adverse respect since such applicable date; provided, further, that (i) notwithstanding the foregoing, the ABL Administrative Agent shall only be permitted to impose rent reserves equal to two (2) months' rent (but not to exceed the amount of eligible inventory at such location) if the Borrower Representative does not deliver to the ABL Administrative Agent a reasonably satisfactory landlord lien waiver for any location for which, pursuant to the ABL Credit Documentation, a landlord lien waiver is required to so be delivered (which shall be subject to minimum dollar thresholds for eligible inventory to be mutually agreed) or that is located in a "Landlord Lien State" (for the purposes hereof, "Landlord Lien State" means any jurisdiction in which a landlord's claim for rent has priority by operation of applicable law over the lien of the ABL Administrative Agent in any of the Collateral); (ii) reserves in respect of cash management obligations shall require the consent of the Borrower Representative; (iii) reserves in respect of hedging obligations secured by the Collateral shall require the consent of the Borrower Representative; (iv) notwithstanding anything to the contrary herein, the amount of any such reserve or change shall have a reasonable relationship to the event, condition or other matter that is the basis for such reserve or such change; (v) no reserves or changes shall be duplicative of reserves or changes already accounted for through eligibility criteria (including collection/advance rates); and (vi) no reserves shall be imposed on the first 5% of aggregate dilution of accounts receivable and thereafter no dilution reserve shall exceed 1% for each incremental whole percentage in aggregate dilution over 5%.

After the Closing Date (or, if later, the date on which the initial Collateral Audit has been delivered to the ABL Administrative Agent), the ABL Administrative Agent reserves the right to establish or modify reserves against the Borrowing Base, acting in its

Permitted Discretion, upon at least five (5) business days' prior written notice to the Borrower Representative (which notice shall include a reasonably detailed description of such reserve being established or modified and the basis for such reserve or modification).

During such five (5) business day period, the ABL Administrative Agent shall, if requested, discuss any such reserve or change with the Borrower Representative and the Borrower Representative may take such action as may be required so that the event, condition or matter that is the basis for such reserve or change no longer exists or exists in a manner that would result in the establishment of a lower reserve or result in a lesser change, in each case, in a manner and to the extent reasonably satisfactory to the ABL Administrative Agent; provided that during such five (5) business day period, borrowings that would cause ABL Exposure to exceed the aggregate Maximum Borrowing Amount upon imposition of such reserve shall not be permitted.

Use of Proceeds:

The proceeds of the ABL Loans will be used (a) on the Closing Date, (i) in an aggregate amount not to exceed \$20 million to finance a portion of the Transaction (including the payment of the Transactions Costs), (ii) for working capital purposes, (iii) to replace, backstop or cash collateralize existing letters of credit, (iv) to fund upfront fees in respect of the ABL Facility, and/or (v) to the extent required to account for any original issue discount and/or upfront fees pursuant to the Flex Provisions of the Fee Letter with respect to the Senior Credit Facilities; and (b) after the Closing Date, to finance the working capital needs of the Loan Parties and their subsidiaries and for general corporate purposes of the Loan Parties and their subsidiaries (including for capital expenditures, acquisitions, investments, restricted payments and any other transaction not prohibited by the ABL Credit Documentation).

Letters of Credit:

A portion to be mutually determined of the ABL Facility (but, in no event, less than \$30 million) shall be available for the issuance of stand-by and trade letters of credit (the "Letters of Credit") by the ABL Administrative Agent and/or any other ABL Lender reasonably satisfactory to the Borrower Representative and the ABL Administrative Agent (in such capacity, the "Issuing Lender"). No Letter of Credit shall have an expiration date after the earlier of (a) one year after the date of issuance unless consented to by the Issuing Lender (or 180 days in the case of trade Letters of Credit) and (b) three (3) business days prior to the ABL Maturity Date (unless cash collateralized or backstopped on terms reasonably acceptable to the Issuing Lender and the ABL Administrative Agent); provided that, any standby Letter of Credit with a one-year tenor may provide for the renewal thereof for additional one-year periods (which shall in no

event extend beyond the date referred to in <u>clause (b)</u> above, unless cash collateralized or backstopped on terms reasonably acceptable to the Issuing Lender and the ABL Administrative Agent). Letters of Credit shall be issued in U.S. Dollars and other freely transferable currencies to be mutually agreed from time to time by the Borrower, the ABL Administrative Agent and the Issuing Lender.

Drawings under any Letter of Credit shall be reimbursed by the applicable Borrower (whether with its own funds or with the proceeds of ABL Loans) within one (1) business day after written notice of such drawing is received by the Borrower Representative from the Issuing Lender; provided that, such written notice is given to the Borrower Representative at least two (2) hours prior to the deadline for requesting ABL Loans, otherwise within two (2) business days after written notice of such drawing is received by the Borrower Representative from the Issuing Lender. To the extent that the applicable Borrower does not so reimburse the Issuing Lender, the ABL Lenders under the ABL Facility shall be irrevocably and unconditionally obligated to fund participations in the reimbursement obligations on a pro rata basis (which, subject to the satisfaction (or waiver) of conditions (b) and (c) under "On-Going Conditions after the Closing Date" but without regard to any required borrowing amounts, shall be deemed to be ABL Loans under the ABL Facility which shall initially be ABR Loans and such ABL Loans shall discharge any obligations of such Borrower in respect of such reimbursement).

Letters of Credit may be issued on the Closing Date to replace or provide credit support for any existing letters of credit (including, if applicable, by "grandfathering" such existing letters of credit into the ABL Facility to the extent originally issued by a person that is an ABL Lender on the Closing Date).

If any ABL Lender becomes a "Defaulting Lender", then the Letter of Credit exposure of such Defaulting Lender will automatically be reallocated among the non-defaulting ABL Lenders pro rata in accordance with their ABL Commitments up to an amount such that the ABL Exposure of such non-defaulting ABL Lender does not exceed its ABL Commitment. In the event that such reallocation does not fully cover the Letter of Credit exposure of such Defaulting Lender under the ABL Facility, the Issuing Lender may require the Borrower to cash collateralize such "uncovered" exposure in respect of each outstanding Letter of Credit under the ABL Facility and will have no obligation to issue new Letters of Credit, or to extend or renew existing Letters of Credit to the extent Letter of Credit exposure would exceed the ABL Commitments of the non-defaulting ABL Lenders, unless such "uncovered" exposure is cash collateralized to the Issuing Lender's reasonable satisfaction.

Swingline Loans:

A portion to be mutually determined of the ABL Facility (but, in no event, less than \$20 million) shall be available for swingline loans (the "Swingline Loans") from the ABL Administrative Agent (in such capacity, the "Swingline Lender") on same-day notice in U.S. Dollars. Any Swingline Loans will reduce availability under the ABL Facility on a dollar-for-dollar basis. Each ABL Lender under the ABL Facility shall be irrevocably and unconditionally required to purchase a participation in each Swingline Loan on a pro rata basis.

If any ABL Lender becomes a Defaulting Lender, then the Swingline Loan exposure of such Defaulting Lender will automatically be reallocated among the non-defaulting ABL Lenders in the ABL Facility pro rata in accordance with their ABL Commitment up to an amount such that the ABL Exposure of such non-defaulting ABL Lender does not exceed its ABL Commitment. In the event that such reallocation does not fully cover the Swingline Loan exposure of such Defaulting Lender, the Swingline Lender may require the Borrower to repay such "uncovered" exposure and will have no obligation to make new Swingline Loans under the ABL Facility to the extent Swingline Loan exposure under the ABL Facility would exceed the ABL Commitments of the non-defaulting ABL Lenders under the ABL Facility, unless such "uncovered" exposure is either cash collateralized to the Swingline Lender's reasonable satisfaction or the Borrower obtains a backstop letter of credit from an issuer reasonably satisfactory to the Swingline Lender to support such Defaulting Lender's pro rata share of outstanding Swingline Loans.

Incremental Facility:

The Borrower Representative will have the right, from time to time to increase commitments under the ABL Facility (any such increase, an "<u>Incremental Facility</u>") in an aggregate principal amount not to exceed \$50 million (the "<u>Incremental Amount</u>"); <u>provided</u>, that, at the time of the increase or addition thereof (i) no Specified Default shall

have occurred and be continuing or, in the case of a permitted acquisition or permitted investment (including a Limited Condition Transaction), no payment or bankruptcy Event of Default shall have occurred and be continuing as of the date of the definitive acquisition agreements for such permitted acquisition or permitted investment entered into, (ii) any Incremental Facility will rank pari passu in right of payment and in priority of security interest with the ABL Facility subject to such increase, (iii) the final maturity of any Incremental Facility shall be no earlier than (but may be later than) the existing final maturity of the ABL Facility subject to such increase and (iv) any Incremental Facility shall be on the same terms and pursuant to documentation applicable to the ABL Facility subject to such increase, except with respect to maturity (as set forth above), and structuring, commitment, upfront and arranger fees or other similar fees that may be agreed to among the Borrower Representative and the lenders providing such additional commitments. Borrowings and prepayments shall be made on a ratable basis among the initial ABL Facility and any Incremental Facilities.

The Borrower Representative may, but shall not be obligated to, seek commitments in respect of any Incremental Facility from existing ABL Lenders. No existing ABL Lender will be required to participate in any such Incremental Facility without its consent. The lenders providing any Incremental Facility shall be reasonably satisfactory to the ABL Administrative Agent and, the Issuing Lender and the Swingline Lender, in each case to the extent required under "Assignments and Participations" below.

The use of proceeds of the Incremental Facility will be as agreed by the Borrower Representative and the lenders providing such Incremental Facility.

CERTAIN PAYMENT PROVISIONS

<u>Fees and Interest Rates</u>: As set forth in <u>Annex I</u> attached hereto.

<u>Closing Fees</u>: As set forth in the Fee Letter.

Optional Prepayments and Commitment Reductions:

ABL Loans may be prepaid and ABL Commitments may be reduced, in whole or in part without premium or penalty (subject to reimbursement of the applicable ABL Lenders' actual redeployment costs (but not lost profits) in the case of a prepayment of Eurodollar Loans (as defined below) prior to the last day of the relevant interest period), in minimum amounts to be mutually agreed. Optional prepayments shall be applied: <u>first</u>, to accrued interest on the amount of ABL Loans prepaid, and second, to the outstanding principal amount of any class of ABL Loans as directed by the Borrower

Representative. Optional commitment reductions of any Commitments shall be applied as directed by the Borrower Representative, including to any class of extending or existing Commitments (including any Commitments under Incremental Facilities and extended ABL Facilities) ratably among the ABL Lenders in the class selected.

Mandatory Prepayments:

If at any time the ABL Exposure under the ABL Facility exceeds the aggregate Maximum Borrowing Amount, prepayments of ABL Loans and/or Swingline Loans (and/or cash collateralization of Letters of Credit) shall be required to be made in respect of ABL Loans within one (1) business day after the applicable Borrower receives written notice of such excess from the ABL Administrative Agent in an amount equal to such excess. The application of proceeds from mandatory prepayments shall not reduce the aggregate amount of ABL Commitments and amounts prepaid may be reborrowed, subject to borrowing availability under the ABL Facility.

The ABL Credit Documentation shall contain customary provisions for replacing or terminating the commitments of any Defaulting Lender (to be defined in the ABL Credit Documentation in a manner consistent with the ABL Documentation Principles).

4. <u>COLLATERAL</u>

Collateral:

Subject to the Certain Funds Provision and the provisions of the immediately following paragraph and consistent with the ABL Documentation Principles, the obligations of the Loan Parties in respect of (i) the ABL Facility and any Incremental Facility and (ii) at the option of the Borrower Representative, any swap obligation (other than swap obligations designated as secured obligations under the First Lien Term Facility and Excluded Swap Obligations) and cash management obligation owed by the Borrower or any of its restricted subsidiaries to (x) any ABL Lead Arranger, the ABL Administrative Agent, any ABL Lender or any affiliate of any of the foregoing persons (in each case as of the Closing Date or as of the time of entering into the related swap or cash management arrangement) or (y) solely in the case of cash management obligations expressly designated in writing by the Borrower as "Secured Obligations", other third party cash management providers that are providing cash management services to the Borrower, the Company or their respective subsidiaries as of the Closing Date so long as such third party cash management providers agree to provide to ABL Administrative Agent a customary letter agreement agreeing to the applicable terms of the ABL Facility (provided that any such cash management obligations described in this clause (y) shall be on a

"last out" basis with respect to the application of Collateral proceeds) shall be secured by the following: (a) a perfected first-priority (subject to permitted liens and other exceptions consistent with the ABL Documentation Principles and excluding any Excluded Assets) security interest in the ABL Priority Collateral and (b) a perfected third-priority security interest in the Term Loan Priority Collateral (subject to permitted liens and other exceptions consistent with the ABL Documentation Principles and excluding any Excluded Assets). Notwithstanding anything to the contrary herein, in no event shall the ABL Facility be secured by (i) any real property or (ii) any Term Loan Priority Collateral that the First Lien Administrative Agent under the First Lien Term Facility has elected to forego perfection in pursuant to its discretionary powers described in the First Lien Term Facility.

All the above-described pledges and security interests shall be created on terms, and pursuant to documentation, consistent with the ABL Documentation Principles and subject to exceptions permitted under the ABL Credit Documentation (it being agreed that (a) no actions in any non-U.S. jurisdiction or required by the laws of any non-U.S. jurisdiction shall be required in order to create any security interests in assets located or titled outside of the U.S. or to perfect any security interest in such assets, including any intellectual property registered in any non-U.S. jurisdiction (it being understood that there shall be no security agreements or pledge agreements governed under the laws of any non-U.S. jurisdiction or any requirement to make any filings in any foreign jurisdiction, including with respect to foreign intellectual property); (b) perfection through control agreements or perfection by "control" shall not be required with respect to any Collateral (including deposit accounts and other bank or securities accounts, etc. but other than entry into deposit account control agreements pursuant to "Cash Dominion" below and in respect of certain debt owing to the Loan Parties and certificated equity interests of the Borrower and material wholly-owned restricted subsidiaries otherwise required to be pledged; (c) there shall be no requirement to comply with the Federal Assignment of Claims Act or any similar statute; (d) perfection by possession of immaterial notes and other evidence of immaterial indebtedness shall not be required; and (e) the ABL Administrative Agent shall not be authorized to enter into any source code escrow arrangement or register any intellectual property. Notwithstanding the foregoing, the requirements of the preceding paragraphs of this "Security" section shall be, as of the Closing Date, subject to the Certain Funds Provisions.

Intercreditor Agreements:

The ABL/Term Intercreditor Agreement.

Cash Dominion:

Within 90 days after the Closing Date, unless otherwise agreed to by the ABL Administrative Agent in its sole discretion, the Loan Parties will implement cash management procedures, to be mutually agreed and otherwise consistent with the ABL Documentation Principles, including (but not be limited to) entering into customary deposit account control agreements which will provide for the ABL Administrative Agent to have control as defined in the UCC (subject to the limitations set forth below on directing application of funds prior to a Liquidity Event Period) of certain deposit accounts (other than payroll accounts, employee benefit accounts, withholding tax and other fiduciary accounts, a segregated account to be set up to hold Preferred Redemption Cash, escrow accounts in respect of arrangements with non-affiliated third parties, workers' compensation accounts, customs accounts, trust and tax withholding accounts that are funded by the Loan Parties in the ordinary course of business or as required by any requirement of law and, in each case, used solely for such purposes, certain cash collateral accounts subject to liens expressly permitted under the ABL Credit Documentation, zero balance accounts, petty cash accounts that contain an average daily balance of less than \$5,000,000 for any one account (with an aggregate cap for all such accounts of \$10,000,000) and accounts held by non-Loan Parties and other exceptions consistent with the ABL Documentation Principles) (collectively, the "Excluded Accounts"); provided that (i) Excluded Accounts shall not include any deposit account that is used for the deposit or remittance of collections of accounts receivables that constitute eligible accounts receivables and that are included in the Borrowing Base (for the avoidance of doubt, the immediately preceding proviso shall not apply to any deposit account that is used for the deposit or remittance of collections of account receivables which constitute Excluded Assets), (ii) the Loan Parties will cause or direct all cash (subject to (x) exceptions described above and other exceptions to be mutually agreed between the Borrower Representative and the ABL Administrative Agent and (y) a customary "double lockbox" structure for proceeds of Medicare and Medicaid accounts receivable, with such proceeds being subject to the provisions of this clause (ii) following their initial deposit into an account over which the ABL Administrative Agent does not have control) to be transferred daily to, or otherwise maintained in, accounts subject to an account control agreement, and during a Liquidity Event Period the ABL Administrative Agent may require that all such cash be swept on a daily basis to an account of the ABL Administrative Agent maintained at a financial institution to be determined to be applied by the ABL Administrative Agent to repay or cash collateralize outstanding ABL Exposure. The ABL Administrative Agent shall be obligated to release cash control upon the termination of the Liquidity Event Period and (iii) until the

Specified Post-Closing Undertaking is satisfied, control agreements with respect to deposit accounts of any Beta Entity shall not be required.

5. <u>CERTAIN CONDITIONS</u>

Initial Conditions:

Subject to the Certain Funds Provision, the availability of the ABL Loans on the Closing Date will be subject only to the Exclusive Funding Conditions.

On-Going Conditions after the Closing Date:

After the Closing Date, the making of each ABL Loan or the issuance of a Letter of Credit (except in connection with certain incurrences, including under any Incremental Facility) shall be conditioned solely upon (a) delivery of a notice of borrowing or credit extension, (b) the accuracy in all material respects of all representations and warranties in the ABL Credit Documentation (although any representation or warranty which expressly relates to a given date or period shall be required only to be true and correct in all material respects as of the respective date or for the respective period, as the case may be), (c) there being no default or event of default in existence at the time of, or immediately after giving effect to the making of, such extension of credit and (d) availability under the Maximum Borrowing Amount.

6. DOCUMENTATION

ABL Credit Documentation:

The definitive documentation for the ABL Facility (the "ABL Credit Documentation" and, collectively with the First Lien Credit Documentation, the "Credit Documentation") shall be based on the First Lien Credit Documentation and, with respect to asset-based lending provisions and loan and letter of credit mechanics only, that certain ABL Credit Agreement, dated as of October 1, 2018, among Specialty Building Products Holdings, LLC, as parent borrower, Bank of America, N.A., as administrative and collateral agent and the other parties thereto (as amended prior to the date hereof, the "Specified ABL Precedent", and adjusted pursuant to terms below, the "ABL Documentation Principles") and take into account (a) the terms set forth in this Commitment Letter, (b) any additional flexibility provided in any transactions involving affiliates of the Sponsor that is reasonably acceptable to the Borrower and the Commitment Parties, (c) differences related to the Loan Parties, their subsidiaries and their businesses (including (i) as to operational and strategic requirements of the Loan Parties and their subsidiaries (in light of their industries, businesses and business practices, locations, operations, financial accounting and projections), (ii) the industry and practices of the Loan Parties and their subsidiaries and (iii) the nature of the accounts receivable and inventory that are eligible for inclusion in the Borrowing Base, the eligibility criteria with respect to which

shall be customary for ABL facilities to borrowers in the healthcare industry), (d) changes in law or accounting standards or to cure mistakes or defects, (e) the Sponsor Model, and (f) any changes in the operational and agency requirements of the Administrative Agent and in any event will contain only those conditions to borrowing, mandatory prepayments, representations, warranties, affirmative, negative and financial covenants and events of default expressly set forth in this Term Sheet, in each case applicable to the Borrower and its restricted subsidiaries, in each case, with materiality thresholds, qualifications, exceptions, "baskets" and grace and cure periods to be mutually agreed, and consistent with this Term Sheet and the Documentation Principles. The ABL Credit Documentation will contain customary Loan Syndications & Trading Association EU Bail-in provisions, customary LSTA provisions regarding ERISA which are solely applicable to the Lenders and customary provisions regarding the division of a limited liability company. The definitive terms will be negotiated in good faith to finalize the ABL Credit Documentation, giving effect to the Certain Funds Provision and the Documentation Principles, as promptly as reasonably practicable. The ABL Documentation Principles are collectively with the First Lien Documentation Principles".

Representations and Warranties:

Consistent with the ABL Documentation Principles, limited to the representations and warranties set forth in the section titled "Representations and Warranties" on Exhibit B and substantially identical to such representations and warranties included in the First Lien Credit Documentation; provided that the ABL Credit Documentation will also include the representations in Sections 3.22 and 3.23 of the Specified ABL Precedent.

Limited Condition Transaction:

The Limited Condition Transaction provisions shall be the same as those contained in the First Lien Credit Documentation except as modified to reflect that (i) on the relevant LCT Test Date, the Payment Conditions shall be required to be satisfied and (ii) on the closing date of the Limited Condition Transaction, sufficient Availability exists for purposes of the incurrence of any extension of credit under the ABL Facility in connection with such Limited Condition Transaction (if ABL Loans are to be made or Letters of Credit are to be issued) on such date.

Affirmative Covenants:

Consistent with the ABL Documentation Principles, limited to the affirmative covenants set forth in the section titled "Affirmative Covenants" on Exhibit B and substantially identical to such affirmative covenants in the First Lien Credit Documentation:

provided that, in addition, the Borrowers shall be required to (a) deliver (x) Borrowing Base Certificates pursuant to "Borrowing Base" above and (y) a quarterly compliance certificate in a manner consistent with the ABL Documentation Principles, (b) maintain cash management as required under "Cash Dominion" above, and (c) permit annual third-party Collateral Audits with respect to the ABL Priority Collateral consistent with the ABL Documentation Principles (provided that the applicable trigger for additional Collateral Audits shall be changed to the greater of (i) 15% of the Maximum Borrowing Amount and (ii) \$15 million), in each case in a manner consistent with the ABL Documentation Principles.

On or prior to the date that is six months after the Closing Date (or such later date as agreed to by the ABL Administrative Agent in its Permitted Discretion), the Borrower shall either (i) cause to be terminated those financing statements with respect to the Beta Entities agreed by the Borrower and the ABL Lead Arranger on or prior to the date hereof as being subject to the obligations in this paragraph and which name any Beta Entity as a "debtor" and any vendor or similar contractual counterparty of any Beta Entity as a "secured party" (the "Specified Beta Vendor Financing Statements") and terminate (or amend to remove any lien grant) any prime vendor agreement or other similar agreement between any one or more Beta Entities and any vendor or similar contractual counterparty thereof the obligations under which are secured by any collateral described in any Specified Beta Vendor Financing Statement (each such agreement, a "Specified Beta Vendor Agreement" and such obligations, the "Specified Beta Vendor Obligations") or (ii) cause the liens securing the Specified Beta Vendor Obligations to be subordinated to the liens securing the Borrower Obligations pursuant to one or more intercreditor agreements in form and substance satisfactory to the ABL Administrative Agent in its permitted discretion (the undertaking in this paragraph, the "Specified Post-Closing Undertaking"). The Specified Post-Closing Undertaking shall be satisfied if (A) (x) the aggregate amount of the Specified Beta Vendor Obligations owed to any vendor does not exceed an amount to be mutually agreed at any time outstanding and (y) the total value of all assets of the applicable Beta Entities subject to such liens does not exceed an amount to be mutually agreed at any time outstanding and (B) the Borrower uses commercially reasonable efforts to cause such liens to be terminated or subordinated to the liens securing the Borrower Obligations pursuant to one or more intercreditor agreements in form and substance satisfactory to the ABL Administrative Agent in its permitted discretion; provided that, while the assets of the applicable Beta Entities shall be eligible for inclusion in the Borrowing Base (subject to the customary eligibility criteria

other than those relating to the existence of liens securing the Specified Beta Vendor Obligations), there shall be a reserve equal to the amount of the obligations owed to the relevant vendor deducted from the Maximum Borrowing Amount.

Financial Covenant:

None, as long as (a) Specified Availability is not less than the greater of (i) 10% of the aggregate Maximum Borrowing Amount (calculated on an aggregate basis) and (ii) \$10 million (the "<u>Availability Trigger</u>"). If Specified Availability falls below such threshold, then the Borrower Representative shall be required to comply on a quarterly basis with a minimum Fixed Charge Coverage Ratio of at least 1.0 to 1.0 for the most recent period of four consecutive fiscal quarters for which financial statements are available (the "<u>Financial Covenant</u>") and shall continue until the date that Specified Availability shall have been not less than the greater of (a) 10% of the aggregate Maximum Borrowing Amount (calculated on an aggregate basis) and (b) \$10 million at any time during the preceding twenty (20) consecutive calendar days.

"Fixed Charge Coverage Ratio" means, as of any date of determination, the ratio of (a) Consolidated EBITDA minus taxes paid in cash minus capital expenditures paid in cash (other than to the extent financed with proceeds of asset sales, indebtedness (other than indebtedness under the ABL Facility) or non-disqualified equity and subject to other exceptions consistent with the ABL Documentation Principles) to (b) Fixed Charges (as defined below), in each case for the most recently ended four fiscal quarter period for which financial statements are available.

"Fixed Charges" means the sum of (a) consolidated interest expense paid or payable currently in cash, but in any event to (i) exclude to the extent not added back to Consolidated EBITDA as interest expense (A) fees and expenses associated with the Transactions and any annual agency fees, (B) costs associated with obtaining or breakage costs in respect of swap agreements, (C) fees and expenses associated with any asset sales, acquisitions, investments, equity issuances or debt issuances (in each case, whether or not consummated) and (D) amortization of deferred financing costs and (ii) be net of interest income, (b) scheduled principal amortization payments of long-term debt for borrowed money (including principal payments in respect of capital leases to the extent allocated to principal, but excluding payments in respect of intercompany debt and earnouts), and (c) solely for the purpose of calculating the Fixed Charge Coverage Ratio for additional restricted payments and dividends made in reliance on the Payment Conditions (as defined below), any such restricted payments and dividends made in cash.

Any cash equity contribution (which shall be common equity or other equity not constituting "disqualified stock" on terms reasonably acceptable to the ABL Administrative Agent) made to the Borrower Representative (or any direct or indirect parent thereof and then contributed to the Borrower Representative) on or prior to the day that is fifteen (15) calendar days after the day on which a compliance certificate is required to be delivered for such fiscal quarter (or, if later, fifteen (15) calendar days after the day on which the Availability Trigger occurs) (the "Cure Period") will, at the request of the Borrower Representative, be included in the calculation of Consolidated EBITDA solely for the purposes of determining compliance with the Financial Covenant at the end of such fiscal quarter and applicable subsequent periods (any such equity contribution so included in the calculation of Consolidated EBITDA, a "Specified Equity Contribution"); provided that, (a) no more than two Specified Equity Contributions may be made in any consecutive four fiscal quarter period, (b) no more than five Specified Equity Contributions may be made during the term of the ABL Facility, (c) a Specified Equity Contribution shall not be greater than the amount required to cause the Borrower to be in compliance with the Financial Covenant, and (d) Specified Equity Contributions may not be relied on for any purpose other than increasing Consolidated EBITDA for purposes of the Financial Covenant. Upon the delivery by the Borrower Representative of written notice that it intends to make a Specified Equity Contribution, any resultant event of default or potential event of default shall be deemed retroactively not to have occurred; provided that, the Borrower Representative shall not be permitted to borrow new ABL Loans or Swingline Loans or request new issuances of Letters of Credit unless and until the Specified Equity Contribution is made or all events of default are waived; provided, further, that if the Specified Equity Contribution is not made before the expiration of the Cure Period, such event of default or potential event of default shall be deemed reinstated. Neither the ABL Administrative Agent nor any ABL Lender shall exercise the right to accelerate the ABL Loans or terminate the ABL Commitments and none of the ABL Administrative Agent, any ABL Lender or any other secured party shall exercise any right to foreclose on or take possession of the Collateral or exercise any other remedy prior to the expiration of the Cure Period solely on the basis of an event of default having occurred and being continuing with respect to the Financial Covenant (except to the extent that the Borrower Representative has confirmed in writing that it does not intend to provide such Specified Equity Contribution).

Negative Covenants:

Consistent with the ABL Documentation Principles, limited to the negative covenants set forth in the section titled "Negative Covenants" on Exhibit B and substantially identical to such negative

covenants included in the First Lien Credit Documentation but modified to (i) permit the indebtedness and liens contemplated by the First Lien Term Facility and Second Lien Notes (including incremental facilities, incremental equivalent debt and refinancing debt thereunder as contemplated by Exhibit B and Exhibit B to the Second Lien Commitment Letter) and any extension or refinancing thereof and (ii) reflect the nature of the ABL Facility as asset based loans in a manner consistent with the ABL Documentation Principles; provided, that (i) the baskets allowing unlimited restricted payments, investments and Junior Debt payments subject to compliance with a leverage ratio and the Available Amount basket shall be replaced with baskets subject to the Payment Conditions and Junior Debt shall be defined in the ABL Credit Documentation to also include certain unsecured indebtedness consistent with the Specified ABL Precedent, (ii) in connection with a disposition of Borrowing Base assets in excess of an amount to be mutually agreed (but, which amount in no event shall be less than, \$7.5 million), the Borrower shall deliver to the Administrative Agent a pro forma Borrowing Base Certificate, (iii) permitted liens on the ABL Priority Collateral shall be junior to the liens securing the ABL Facility (it being understood and agreed that all indebtedness permitted to be secured by all or a portion of the Term Loan Priority Collateral shall be permitted to be secured on a senior lien basis to the liens over the Term Loan Priority Collateral securing the ABL Facility), (iv) a basket for unsecured, non-amortizing debt with a maturity of at least 91 days later than the ABL Facility if the Payment Conditions are satisfied shall be included and (v) accounts receivable factoring facilities and securitizations shall not be permitted.

In addition, (i) the permission for liens existing on the Closing Date and evidenced by any Specified Beta Vendor Financing Statement shall (A) limit the amount of Specified Beta Vendor Obligations that can be secured by such liens to an amount to be agreed and (B) be subject to the Specified Post-Closing Undertaking and (ii) until the Specified Post-Closing Undertaking has been satisfied, (x) Omega and its subsidiaries shall not transfer any Borrowing Base assets (other than cash) to any Beta Entity in an aggregate amount in excess of \$2.5 million in any calendar month (but subject to an overall cap to be agreed) (and as a result of any such transfer, the ABL Agent may institute a reserve in its permitted discretion), and there shall be a prohibition on any merger of Omega or any of its subsidiaries into any Beta Entity (it being understood and agreed that cash may be transferred among the Beta Entities and Omega and its subsidiaries), (y) any inventory of any Beta Entity shall be physically separate from, and not commingled with, any inventory of Omega or any of its subsidiaries and (z) the Borrower shall not permit the remittance of

any collections of accounts receivable of any Beta Entity, or the deposit of any proceeds of any inventory of any Beta Entity, into any deposit account that is used for the deposit or remittance of collections of accounts receivables that constitute eligible accounts receivables and that are included in the Borrowing Base.

Notwithstanding the foregoing, so long as (a) no Specified Default exists immediately before or immediately after the action or proposed action and (b) (1) pro forma (for any of the actions to be taken in clauses (i) through (iv) below) Specified Availability as of the date of such transaction and pro forma (for any of the actions to be taken in clauses (i) through (iv) below) Specified Availability for the 20 consecutive day period immediately preceding the date of such transaction, in each case, is equal to or greater than the greater of (A) in the case of clauses (i) and (iv) below, (x) 15% of the aggregate Maximum Borrowing Amount and (y) \$15 million, (2) in the case of clause (ii) and (iii) below or the designation of an Unrestricted Subsidiary, (x) 12.5% of the aggregate Maximum Borrowing Amount and (y) \$12.5 million, and (3) pro forma (for any of the actions to be taken in clauses (i) through (iv) below) the Fixed Charge Coverage Ratio is at least 1.00 to 1.00 or (2) pro forma (for any of the actions to be taken in clauses (i) through (iv) below) Specified Availability (as of the date of such transaction and pro forma (for any of the actions to be taken in clauses (i), through (iv) below) Specified Availability for the 20 calendar days preceding the date of such transaction) is equal to or greater than the greater of (I) in the case of clauses (i) and (iv) below, (x) 20.0% of the aggregate Maximum Borrowing Amount and (y) \$20 million, and (II) in the case of clauses (ii) and (iii) below or the designation of an Unrestricted Subsidiary (x) 17.5% of the aggregate Maximum Borrowing Amount and (y) \$17.5 million, the ABL Credit Documentation will, among other exceptions permit (i) dividends and other restricted payments in respect of capital stock, (ii) acquisitions and other investments, (iii) the incurrence of unsecured, non-amortizing indebtedness maturing not earlier than 91 days later than the ABL Facility and (iv) the voluntary or optional repurchase, redemption, defeasance or prepayment of all or a portion of Junior Debt (for the avoidance of doubt there will be no restrictions on prepaying First Lien Term Loans, Second Lien Notes and other senior debt except as set forth above) (the "Payment Conditions").

Unless a Specified Default under the ABL Facility has occurred and is continuing or would result therefrom, the Borrower and its restricted subsidiaries may make acquisitions (each, a "Permitted Acquisition"), subject solely to the following terms and conditions: (i) after giving effect thereto, the Borrower and its restricted subsidiaries are in compliance with the permitted lines of business covenant, (ii) pro forma compliance with the Payment Conditions and (iii) if the

Borrower or any of its restricted subsidiaries acquires the majority of the equity interests of any person in connection with such acquisition such person will, subject to the right of the Borrower to designate an unrestricted subsidiary and (subject to a cap on amounts invested by Loan Parties in entities that do not become (or assets that do not become owned by) Loan Parties) become a Loan Party.

Unrestricted Subsidiaries:

The ABL Credit Documentation will contain provisions pursuant to which, subject to the investments covenant in the ABL Credit Documentation, and, so long as (x) no Specified Default exists or would result therefrom and (y) the Payment Conditions for an investment are satisfied on a pro forma basis or the investment resulting from such designation is otherwise permitted, the Borrower will be permitted to designate any existing or subsequently acquired or organized subsidiary as an "unrestricted subsidiary" and subsequently re-designate any such unrestricted subsidiary as a restricted subsidiary; provided that in no event shall any unrestricted subsidiary acquire any material intellectual property from the Borrower or any restricted subsidiary. Unrestricted subsidiaries will not be subject to the mandatory prepayments, representations and warranties, affirmative, negative or financial covenants or event of default provisions of the ABL Credit Documentation and the cash held by, results of operations and indebtedness and interest expense of unrestricted subsidiaries will not be taken into account for purposes of determining any financial ratio or covenant contained in the ABL Credit Documentation.

Events of Default:

Consistent with the ABL Documentation Principles, limited to the events of default set forth in the section titled "Events of Default" on Exhibit B and substantially identical to such events of default included in the First Lien Credit Documentation and, in addition, shall include maintenance of cash management as required under "Cash Dominion" above (subject to a five (5) business day grace period prior to a Liquidity Event), failure to deliver compliance certificates or Borrowing Base Certificates (subject to a five (5) business day grace period for failure to deliver such certificates or, in the case of delivery of Borrowing Base Certificates, two (2) business day grace when weekly Borrowing Base Certificates are required to be delivered) and failure to perform the Financial Covenant if then in effect (subject to the Cure Period); provided, with respect to the cross-default to other indebtedness, solely with respect to the First Lien Term Loan Facility, there shall be no cross default except with respect to (x) any payment or bankruptcy event of default under the First Lien Term Loan Facility that remains unremedied, unwaived or uncured for 60 consecutive days after the occurrence thereof or with respect to which the agent or the lenders under the First Lien Term Loan Facility are exercising remedies (including acceleration of obligations or termination of commitments).

Voting:

Amendments and waivers of the ABL Credit Documentation will require the approval of ABL Lenders (other than Defaulting Lenders) holding more than 50% of the aggregate principal amount of the ABL Loans and commitments under the ABL Facility (the "Required ABL Lenders"), except that the consent of each ABL Lender directly adversely affected thereby (but not, for the avoidance of doubt, the consent of the Required ABL Lenders) shall be required, unless otherwise contemplated by the ABL Credit Documentation, with respect to (I) (a) increases in the commitment of such ABL Lender (it being understood that a waiver of any condition precedent or the waiver of any condition precedent, default, event of default or mandatory prepayment shall not constitute an increase of any commitment of any ABL Lender), (b) reductions of principal or interest (other than a waiver of default interest, a default, event of default or mandatory prepayment) or fees, (c) extensions of final maturity or the scheduled due date of any principal payment, interest or fee payment and (d) changes in the application of proceeds amongst ABL Lenders after an event of default and, only following the exercise of remedies, changes in other pro rata sharing provisions; (ii) the consent of all ABL Lenders (other than Defaulting Lenders) shall be required with respect to (a) releases of all or substantially all of the value of the Guarantees or all or substantially all of the Collateral (other than in connection with permitted asset sales) and (b) reductions in voting thresholds; (iii) the consent of ABL Lenders (other than Defaulting Lenders) holding more than 66 2/3% of the aggregate principal amount of ABL Loans and ABL Commitments under the ABL Facility (the "Supermajority Lenders") shall be required with respect to changes to the definition of Borrowing Base or the constituent definitions in a manner that would have the effect of increasing availability thereunder (including changes in eligibility criteria and advance rates), other than changes in reserves implemented by the ABL Administrative Agent in its Permitted Discretion; and (iv) the consent of the Issuing Lender, the Swingline Lender and the ABL Administrative Agent, respectively, shall be required with respect to amendments directly and adversely affecting their respective rights and duties. It being understood that (A) additional extensions of credit permitted under the ABL Credit Documentation shall not require the consent of all ABL Lenders (or the Required ABL Lenders) but instead shall only require the consent of each ABL Lender extending such credit, (B) any applicable intercreditor agreement may be amended solely with the consent of the ABL Administrative Agent to give effect thereto or to carry out the purposes thereof and (C) subject to the next paragraph there shall

be no "class" voting requirement for amendments, modifications or supplements to the ABL Credit Documentation. In the event that there are less than three (3) unaffiliated Lenders party to the ABL Credit Documentation, the Required ABL Lenders and the Supermajority Lenders shall be all Lenders.

Modifications to provisions requiring pro rata payments or sharing of payments (other than as set forth in clause (I)(d) in the second preceding paragraph) shall only require approval of the Required ABL Lenders and non-pro rata distributions and commitment reductions will be permitted in connection with "amend and extend" transactions or the addition of one or more tranches of debt as permitted by the ABL Credit Documentation.

The ABL Credit Documentation shall contain a mechanism to permit the Borrower Representative (a) with the consent of each directly and adversely affected ABL Lender, but without the consent of any other ABL Lender or the Required ABL Lenders, to extend the final expiration date of such ABL Lender's Commitment and to provide for different interest rates and fees and voluntary prepayments for the ABL Lender providing such extended ABL Commitment and (b) with the consent of each directly and adversely affected ABL Lender under the ABL Facility (but no other ABL Lender or the ABL Administrative Agent), to provide for a "re-pricing" amendment which reduces the interest rate accruing in respect of the ABL Loans held by such ABL Lenders.

The ABL Credit Documentation will permit the ABL Administrative Agent and the Borrower (i) to amend the ABL Credit Documentation to correct any obvious error or any error or omission of a technical nature in the ABL Credit Documentation without any further action or consent of any other party with notice given to the ABL Lenders of any such amendment and (ii) to enter into or amend any applicable intercreditor agreement to give effect thereto or carry out the provisions thereof.

Assignments and Participations:

The ABL Lenders will be permitted to assign (a) ABL Loans and ABL Commitments with the consent of the Borrower (not to be unreasonably withheld or delayed); <u>provided</u> that, no consent of the Borrower shall be required (i) after the occurrence and during the continuance of a payment or bankruptcy event of default or (ii) if such assignment is made to another ABL Lender or an affiliate, branch or approved fund of any ABL Lender; <u>provided</u>, that no assignments shall be made to natural persons, the Sponsor, any affiliate of the Sponsor and any Disqualified Institution (so long as the list of Disqualified Institutions is made available to all ABL Lenders); <u>provided</u>, <u>further</u>, that if not received within ten (10)

business days of written request by the ABL Administrative Agent, the consent of the Borrower shall be deemed to have been given. Subject to the preceding sentence, all assignments will also require the consent of the ABL Administrative Agent, the Issuing Lender and the Swingline Lender, not to be unreasonably withheld or delayed. Each assignment will be in an amount of an integral multiple of \$5,000,000 or, if less, all of such ABL Lender's remaining ABL Loans and ABL Commitments, and shall be accompanied with a processing and recordation fee of \$3,500 to the ABL Administrative Agent. Assignments will be by novation.

The ABL Lenders will have the right to participate their commitments and ABL Loans to other persons (other than any natural persons, the Sponsor, any affiliate of the Sponsor and any Disqualified Institution (so long as the list of Disqualified Institutions is made available to all ABL Lenders)). Participants shall have the same benefits as the ABL Lenders with respect to yield protection and increased cost provisions, subject to customary limitations and restrictions. Voting rights of participants shall be limited to matters in respect of (a) increases in commitments participated to such participants (it being understood that a waiver of any condition precedent or the waiver of any default, event of default or mandatory prepayment shall not constitute an increase of any commitment of any ABL Lender), (b) reductions of principal or interest (other than a waiver of default interest, a default, event of default or mandatory prepayment) or fees, (c) extensions of final maturity or the due date of any interest or fee payment (other than with respect to any extension facility), (d) changes in the application of proceeds amongst ABL Lenders after an event of default and, only following the exercise of remedies, changes in other pro rata sharing provisions and (e) releases of all or substantially all of the value of the Guarantees or all or substantially all of the Collateral, in each case, with respect to which the affirmative vote of the ABL Lender from which it purchased a participation would be required. Pledges of ABL Loans in accordance with applicable law shall be permitted without restriction.

The ABL Administrative Agent shall be permitted to make the list of Disqualified Institutions available to any ABL Lender who specifically requests a copy thereof, but such list shall not be "posted" or otherwise disclosed by any ABL Lead Arranger or the ABL Administrative Agent to any other person. The ABL Administrative Agent shall not have any liability for assignments or participations made to Disqualified Institutions (regardless of whether the consent of the ABL Administrative Agent is required thereto), and no Loan Party, ABL Lender or any of their respective affiliates will bring any claim to such effect.

<u>Yield Protection</u>: The ABL Credit Documentation shall contain yield protection provisions substantially identical to the yield protection provisions included in the First Lien Credit Documentation but modified to reflect the nature of the ABL Facility as asset

based loans, in each case in accordance with the ABL Documentation Principles.

Expenses and Indemnification: The ABL Credit Documentation shall contain expense and indemnification provisions substantially identical to the

expense and indemnification provisions included in the First Lien Credit Documentation but modified to reflect the nature of the ABL Facility as asset based loans (including, without limitation, reimbursement provisions with respect to

Collateral Audits), in each case in accordance with the ABL Documentation Principles.

Governing Law and Forum: New York.

Counsel to the ABL Davis P

Administrative Agent:

Davis Polk & Wardwell LLP.

INTEREST AND CERTAIN FEES

Interest Rate Options:

The Borrower Representative may elect that the ABL Loans comprising each borrowing to the Borrowers bear interest at a rate per annum equal to (a) the ABR <u>plus</u> the Applicable Margin or (b) the Eurodollar Rate <u>plus</u> the Applicable Margin; <u>provided</u>, that all Swingline Loans shall bear interest at a rate per annum equal to the ABR <u>plus</u> the Applicable Margin.

As used herein:

"ABR" means the highest of the Administrative Agent's Prime Rate, the Federal Funds Effective Rate plus 1/2 of 1.0% and one-month adjusted LIBOR plus 1.00%.

"ABR Loans" means ABL Loans to the Borrowers bearing interest based upon the ABR.

"Applicable Margin" means (a) until the pricing certificate for the first full quarter after the Closing Date has been delivered, (i) in the case of Eurodollar Loans, 1.50% or (ii) in the case of ABR Loans, 0.50%, and (b) thereafter, subject to the below pricing grid based on average daily Availability for the fiscal quarter:

Availability as a		
percentage of the		
applicable Maximum	Eurodollar	
Borrowing Amount	Loans	ABR Loans
Greater than or equal to 66.66%	1.25%	0.25%
Less than 66.66% but greater than or equal to 33.33%	1.50%	0.50%
Less than 33.33%	1.75%	0.75%

"Eurodollar Rate" means the London interbank offered rate for U.S. dollars, adjusted for customary Eurodollar reserve requirements, if any, subject to a floor of 0.00%.

"Eurodollar Loans" means ABL Loans to the Borrowers bearing interest based upon the Eurodollar Rate.

Interest Payment Dates:

In the case of ABR Loans, quarterly in arrears.

In the case of Eurodollar Loans, on the last day of each relevant interest period and, in the case of any interest period longer than three months, on each successive date three months after the first day of such interest period.

Commitment Fees:

The Borrowers shall pay a commitment fee (the "Commitment Fee") calculated at a rate per annum equal to 0.375% on the average daily unused portion of the aggregate commitments in respect of the ABL Facility of non-defaulting Lenders, payable quarterly in arrears. Following delivery of financial statements for the first full fiscal quarter after the Closing Date, the Commitment Fee rate shall be subject to the below pricing grid based on daily average utilization of the aggregate commitments in respect of the ABL Facility for the previous fiscal quarter. For purposes of the commitment fee calculations, the applicable ABL Exposure (other than Swingline Loans) shall be deemed to be a utilization of the aggregate ABL Facility.

Level	Average Daily Utilization	Commitment Fee
I	≥ 50%	0.25%
II	< 50%	0.375%

Letter of Credit Fees:

The Borrower shall pay a fee (a) with respect to trade Letters of Credit issued under the ABL Facility, a per annum rate equal to the Applicable Margin for ABL Loans which are Eurodollar Loans, and (b) with respect to standby Letters of Credit, a per annum rate equal to the Applicable Margin for ABL Loans which are Eurodollar Loans. Such fee shall be shared ratably among the applicable ABL Lenders (other than Defaulting Lenders) and shall be payable quarterly in arrears.

A fronting fee in an amount to be agreed (but in any event not to exceed 0.125% per annum) on the face amount of each Letter of Credit shall be payable quarterly in arrears to the Issuing Lender for

its own account. The Borrower shall also pay to the Issuing Lender for its own account such Issuing Lender's customary and reasonable issuance and administration fees.

<u>Default Rate</u>: Any principal payable under or in respect of the ABL Facility not paid when due shall bear interest at the applicable

interest rate plus 2% per annum. Other overdue amounts shall bear interest at the interest rate applicable to ABR loans

plus 2% per annum. Default interest will be payable on demand.

Rate and Fee Basis: All per annum rates shall be calculated on the basis of a year of 360 days (or 365/366 days, in the case of ABR Loans,

the interest rate payable on which is then based on the ABR).

PROJECT OPAL

Conditions

The availability of the Senior Credit Facilities shall be subject solely to the satisfaction or waiver by (x) with respect to the First Lien Term Facility, the Initial First Lien Lenders with commitments in respect of the First Lien Term Facility and (y) with respect to the ABL Facility, the Initial ABL Lenders with commitments in respect of the ABL Facility, in each case, of the following conditions (subject to the Certain Funds Provision). Capitalized terms used but not defined in this Exhibit D have the meanings set forth in the letter to which this Exhibit D is attached or in Exhibits A, B or C thereto.

- 1. Subject to the Certain Funds Provision, the Borrower and each Guarantor shall have executed and delivered the Senior Credit Documentation, in each case, consistent with the Commitment Letter and Term Sheets, to which they are parties and the Commitment Parties shall have received:
 - (a) a customary notice of borrowing;
 - (b) customary closing officer's (certifying as to resolutions, organizational documents and incumbency, as to the accuracy of the Specified Representations in all material respects and no amendments or modifications to the provisions of the Merger Agreement that are materially adverse to the interests of the Initial Senior Lenders) and good standing (of the jurisdiction of organization of the Borrower and Guarantors) certificates and legal opinions;
 - (c) a certificate (substantially in the form of <u>Annex I</u> to this <u>Exhibit D</u>) from the chief financial officer (or other officer with reasonably equivalent duties) of the Borrower certifying that the Borrower and its subsidiaries, on a consolidated basis after giving effect to the Transactions, are solvent; and
 - (d) in the case of the ABL Facility only, a Borrowing Base Certificate (which may state the Temporary Borrowing Base Amount if a Collateral Audit has not been completed prior to the Closing Date).
- 2. Prior to, or substantially concurrently with the initial fundings contemplated by the Commitment Letter, the Refinancing shall have been consummated.
- 3. The Merger shall be consummated in all material respects pursuant to the Merger Agreement (for the avoidance of doubt, the Initial Senior Lenders are satisfied with the Merger Agreement dated the date hereof) substantially concurrently with the initial funding of the Senior Credit Facilities without giving effect to any amendments or modifications to the provisions thereof or express waivers or consents thereto that, in each case, are materially adverse to the interests of the Initial Senior Lenders without the consent of the Initial Senior Lenders, such consent not to be unreasonably withheld, conditioned or delayed (it being understood and agreed that (i) any change in the Merger Consideration (as defined in the Merger Agreement) shall be deemed not to be adverse to the interest of the Initial Senior Lenders and (ii) any adverse

modification to the definition of Beta Material Adverse Effect (or adverse express waiver or express consent in respect of the definition of Beta Material Adverse Effect) without the prior written consent of the Initial Senior Lenders (such consent not to be unreasonably withheld, delayed or conditioned) shall be deemed to be materially adverse to the interests of the Initial Senior Lenders); provided that in each case the Initial Senior Lenders shall be deemed to have consented to such modification, amendment, waiver or consent unless they shall object thereto within 3 business days of receipt of written notice of such modification, amendment, consent or waiver.

- 4. Since the date of the Merger Agreement there shall not have been a Beta Material Adverse Effect (as defined in the Merger Agreement as of the date hereof).
- 5. The Commitment Parties shall have received (1) (i) the audited consolidated statements of operations, shareholders' equity and cash flows of HC Group Holdings III, Inc. for the fiscal years ended December 31, 2016 and December 31, 2017, and the related audited consolidated balance sheets as of the end of such fiscal years and (ii) the audited consolidated statements of operations, shareholders' equity and cash flows of HC Group Holdings III, Inc. for, and the related audited consolidated balance sheets as of the end of, each fiscal year of HC Group Holdings III, Inc. ended at least 120 days prior to the Closing Date (each Commitment Party acknowledges receipt of such audited statements of operations, shareholders' equity and cash flows and the related balance sheets in clause (1)(i) hereof), (2) (i) the unaudited consolidated statement of operations of HC Group Holdings III, Inc. for the fiscal quarters ending March 31, 2018, June 30, 2018 and September 30, 2018, and the related unaudited consolidated balance sheet as of the end of such fiscal quarters and (ii) the unaudited consolidated statement of operations of HC Group Holdings III, Inc. for, and the related unaudited consolidated balance sheet as of the end of, each fiscal quarter of HC Group Holdings III, Inc. (other than the fourth fiscal quarter) ended after the close of the fiscal quarter ending September 30, 2018 and at least 45 days prior to the Closing Date (each Commitment Party acknowledges receipt of unaudited consolidated statements of operations and the related balance sheet in clause (2)(i) hereof), (3) (i) the audited consolidated statements of operations, shareholders' equity and cash flows of the Company for the fiscal years ended December 31, 2016 and December 31, 2017, and the related audited consolidated balance sheets as of the end of such fiscal years and (ii) the audited consolidated statements of operations, shareholders' equity and cash flows of the Company for, and the related audited consolidated balance sheets as of the end of, each fiscal year of the Company ended at least 120 days prior to the Closing Date (each Commitment Party acknowledges receipt of such audited statements of operations, shareholders' equity and cash flows and the related balance sheets in clause (3)(i) hereof), (4) (i) the unaudited consolidated statement of operations of the Company for the fiscal quarters ending March 31, 2018, June 30, 2018 and September 30, 2018, and the related unaudited consolidated balance sheet as of the end of such fiscal quarters and (ii) the unaudited consolidated statement of operations of the Company for, and the related unaudited consolidated balance sheet as of the end of, each fiscal quarter of the Company (other than the fourth fiscal quarter) ended after the close of the fiscal quarter ending September 30, 2018 and at least 45 days prior to the Closing Date (each Commitment Party acknowledges receipt of unaudited consolidated statements of operations and the related balance sheet in clause (4)(i) hereof) and (5) a pro forma consolidated balance sheet of the Company as of the last day of the most recently completed four-fiscal quarter period ended at least 45 days (or 120 days in case such period is the end of the Company's fiscal year) prior to

the Closing Date, prepared after giving effect to the Transactions as if the Transactions had occurred as of such date and any other adjustments as agreed by the Sponsor and the Commitment Parties (which need not be prepared in compliance with Regulation S-X of the Securities Act of 1933, as amended, or include adjustments for purchase or recapitalization accounting (including adjustments of the type contemplated by Financial Accounting Standards Board Accounting Standards Codification 805, Business Combinations (formerly SFAS 141R))).

- 6. So long as requested at least ten (10) business days prior to the Closing Date, the Administrative Agents shall have received, at least three (3) business days prior to the Closing Date, all documentation and other information with respect to the Borrower and the Guarantors that is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including, without limitation, the PATRIOT Act. At least three (3) business days prior to the Closing Date (to the extent requested by the Administrative Agents or any Commitment Party at least ten (10) business days prior to the Closing Date), the Borrower or any Guarantor that qualifies as a "legal entity customer" under 31 C.F.R. § 1010.230 (the "Beneficial Ownership Regulation") shall deliver beneficial ownership certificates substantially similar in form and substance to the form of Certification Regarding Beneficial Owners of Legal Entity Customers published jointly, in May 2018, by the Loan Syndications and Trading Association and Securities Industry and Financial Markets Association (a "Beneficial Ownership Certificate") in relation to the Borrower or such Guarantor to each Administrative Agent or Commitment Party that so requests.
- 7. Payment of all fees, closing payments and expenses earned, due and payable to the Commitment Parties and the Lenders required to be paid on the Closing Date under the Senior Credit Facilities for which invoices have been received at least three (3) business days in advance (which amounts may be offset against the proceeds of the applicable Facility) shall have been made (or shall be made substantially contemporaneously with funding).
- 8. Subject to the Certain Funds Provision, with respect to the Senior Credit Facilities, all actions necessary to establish that the Administrative Agent will have a perfected security interest in the Collateral of the Borrower and each Guarantor under the Senior Credit Facilities shall have been taken (or shall be taken substantially contemporaneously with funding).
- 9. The Specified Merger Agreement Representations and the Specified Representations shall be true and correct in all material respects.
- 10. The Senior Lead Arrangers shall have been afforded a period (the "Marketing Period") of at least fifteen (15) consecutive business days (unless a shorter period is acceptable to the Senior Lead Arrangers) following the date of delivery of the financial statements required under clause (1)(i), (2)(i), (3)(i), (4)(i) and (5) (and, if applicable, clauses (1)(ii), (2)(ii), (3)(ii) and (4)(ii)) of paragraph 5 above (the "Required Financial Information") to seek to syndicate the Senior Credit Facilities; provided, that in no event shall such Marketing Period be restarted or cease to continue if additional financial information as required pursuant to clauses (1)(ii), (2)(ii), (3)(ii) and (4)(ii) of paragraph 5 of this Exhibit D are delivered after such Marketing Period has commenced (which commencement, for the avoidance of doubt, begins at such time that all required information under clauses (1), (2), (3), (4) and (5) of paragraph 5 above, excluding information that is not then required by clause (1)(ii), (2)(ii), (3)(ii) and (4)(ii) (determined as if

the date of the commencement of the Marketing Period were the Closing Date), has been delivered; provided that (A) the Marketing Period shall not commence prior to the date the proxy statement for the Merger is mailed to Beta's shareholders, (B) May 27, 2019, July 4, 2019, July 5, 2019 and November 27, 2019 through December 1, 2019 shall not be considered business days for purposes of such Marketing Period (provided, however, that such exclusion shall not restart such period) and (C) if such 15 consecutive business day period shall not have fully elapsed on or prior to August 16, 2019, then such period shall not commence any earlier than September 3, 2019. If the Borrower in good faith reasonably believes that it has delivered the Required Financial Information, it may deliver to the Senior Lead Arrangers written notice to that effect (stating when it believes it completed any such delivery), in which case the Borrower shall be deemed to have delivered the Required Financial Information on the date specified in such notice and the fifteen (15) consecutive business day period described above shall be deemed to have commenced on the date specified in such notice, in each case unless the Senior Lead Arrangers in good faith reasonably believe that the Borrower has not completed delivery of the Required Financial Information and, within two business days after its receipt of such notice from the Borrower, the Senior Lead Arrangers deliver a written notice to the Borrower to that effect (stating with specificity which Required Financial Information is required).

FORM OF SOLVENCY CERTIFICATE

[·],

This Solvency Certificate is being executed and delivered pursuant to Section [:] of that certain [·](1) (the "First Lien Credit Agreement") and Section [:] of that certain [·](2) (the "ABL Credit Agreement" and, together with the First Lien Credit Agreement, collectively, the "Credit Agreements" and each a "Credit Agreement"); the terms defined therein being used herein as therein defined.

I, [·], the [chief financial officer/equivalent officer] of the Borrower, solely in such capacity and not in an individual capacity, hereby certify that I am the [chief financial officer/equivalent officer] of the Borrower and that I am generally familiar with the businesses and assets of the Borrower and its Subsidiaries (taken as a whole), I have made such other investigations and inquiries as I have deemed appropriate and I am duly authorized to execute this Solvency Certificate on behalf of the Borrower pursuant to each of the Credit Agreements.

I further certify, solely in my capacity as [chief financial officer/equivalent officer] of the Borrower, and not in my individual capacity, as of the date hereof and after giving effect to the Transactions and the incurrence of the indebtedness and obligations being incurred in connection with the Credit Agreements and the Transactions on the date hereof, that, (a) the sum of the debt (including contingent liabilities) of the Borrower and its Subsidiaries, taken as a whole, does not exceed the present fair saleable value (on a going concern basis) of the assets of the Borrower and its Subsidiaries, taken as a whole; (b) the capital of the Borrower and its Subsidiaries, taken as a whole, contemplated as of the date hereof; and (c) the Borrower and its Subsidiaries, taken as a whole, do not intend to incur, or believe that they will incur, debts including current obligations beyond their ability to pay such debt as they mature in the ordinary course of business. For the purposes hereof, the amount of any contingent liability at any time shall be computed as the amount that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

[Remainder of page intentionally left blank]

- (1) Describe First Lien Credit Agreement.
- (2) Describe ABL Credit Agreement.

IN WITNESS WHEREOF, I have executed this Solvency Certificate on the date first written above.

By: Name: [·]
Title: [·]

EXECUTION VERSION

GS MEZZANINE PARTNERS VII, L.P.
GS MEZZANINE PARTNERS VII OFFSHORE, L.P.
GS MEZZANINE PARTNERS VII OFFSHORE TREATY, L.P.
BROAD STREET DANISH CREDIT PARTNERS, L.P.
BROAD STREET CREDIT HOLDINGS LLC

200 West Street New York, New York 10282

ARES MANAGEMENT LLC ARES CAPITAL MANAGEMENT LLC

245 Park Avenue, 44th Floor New York, New York 10167

March 14, 2019

HC Group Holdings II, Inc. c/o Option Care Infusion Services, Inc. 3000 Lakeside Dr. #300N Bannockburn, IL 60015 Attention: Mike Shapiro

> Project Opal \$400 Million Senior Secured Second Lien PIK Toggle Floating Rate Notes <u>Commitment Letter</u>

Ladies and Gentlemen:

You have advised the principal investors listed on Schedule I(a) hereto (each, an "Initial GS Principal Investor" and, collectively, the "Initial GS Principal Investors," and, together with any Other GS Principal Investors (as defined below) to which the Initial GS Principal Investors have reallocated, sold, resold, assigned and/or transferred commitments pursuant to, and in accordance with, this Commitment Letter (as defined below), collectively, the "GS Principal Investors") and the principal investors listed on Schedule I(b) hereto (each, an "Initial Ares Principal Investor, and collectively, the "Initial Ares Principal Investors" and, together with any Other Ares Principal Investors (as defined below) to which the Initial Ares Principal Investors have reallocated, sold, resold, assigned and/or transferred commitments pursuant to, and in accordance with, this Commitment Letter, the "Ares Principal Investors" and, together with the GS Principal Investors, the "Principal Investors") that you intend to consummate the transactions described in Exhibit A hereto. Capitalized terms used but not defined herein are used with the meanings assigned to them on the exhibits attached hereto (such exhibits, together with this letter, collectively, the "Commitment Letter"). The Initial GS Principal Investors, together with Other Ares Principal Investors are herein referred to as "Other Principal Investors."

1. <u>Commitments; Closing Payment</u>

In connection with the Transactions, the Initial Principal Investors are pleased to advise you of their respective several, but not joint, commitments to purchase for cash 100% of the entire aggregate principal amount of the senior secured PIK toggle second lien floating rate notes in an aggregate principal amount of \$400 million to be issued by the Issuer on the Closing Date (the "Second Lien Notes"), upon the terms expressly set forth in this Commitment Letter (including, without limitation, in the Summary of Terms and Conditions attached hereto as Exhibits B (the "Second Lien Notes Term Sheet"), in the respective principal amounts set forth opposite each such Initial Principal Investor's name on Schedules I(a) and I(b) hereto, and subject solely to the Exclusive Funding Conditions (as defined below).

As consideration for the commitments of the Initial Principal Investors hereunder, you agree to pay (or cause to be paid) the closing payment set forth in the closing payment letter dated the date hereof (the "Closing Payment Letter") among you and us and delivered herewith with respect to the Second Lien Notes, on the terms and subject to the conditions set forth therein. Once paid, the closing payment shall not be refundable under any circumstances.

2. <u>Provision of Information and Access</u>

You agree, until the execution and delivery of the Second Lien Notes Documentation, (a) to provide all information with respect to the business, operations, assets, liabilities and financial projections of you and the Company and your and its respective subsidiaries, the Transactions and the other transactions contemplated hereby and such other information as was or is provided under the First Lien Commitment Letter to the Commitment Parties (as defined in the First Lien Commitment Letter), including, any financial statements and (b) to make available at mutually agreeable times and to a reasonable extent your and the Sponsor's respective officers and employees, and to use your commercially reasonable efforts to cause, to the extent practical, appropriate and reasonable and in all instances subject to, and not in contravention of, the terms of the Merger Agreement, the Company to make available, the officers and employees of the Company and its subsidiaries to discuss with the Initial Principal Investors, their representatives and legal and accounting advisors, the business and affairs of the Company and its subsidiaries and of you and your subsidiaries, such discussion in each case to be undertaken prior to the execution of the Second Lien Notes Documentation. Notwithstanding anything to the contrary contained in this Commitment Letter, compliance with this Section 2 shall not constitute a condition precedent to the Initial Principal Investors' respective commitments hereunder or to the Closing Date.

You agree to use commercially reasonable efforts to deliver to the Initial Principal Investors written notice of the date anticipated to be the Closing Date at least 3 business days prior to such date.

3. <u>Information</u>

You hereby represent and warrant (with respect to Information (as defined below) relating to the Company and its subsidiaries and businesses, to your knowledge) that (a) all written

information, other than the projections, including financial estimates, budgets and forecasts (the "Projections"), forward looking statements and information of a general economic or industry-specific nature, concerning you, the Issuer, the Company and your and their respective subsidiaries, the Merger and the other Transactions contemplated hereby (the "Information"), that has been or will be made available to us by you or your representatives in connection with the transactions contemplated hereby, taken as a whole and as supplemented, does not contain (or, in the case of Information furnished after the date hereof, will not contain), as of the time it was (or hereafter is) furnished, any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statements are made, as supplemented and updated as provided below and (b) the Projections that have been or will be made available to us by or on behalf of you in connection with the transactions contemplated hereby have been or will be prepared in good faith based upon assumptions believed by you to be reasonable at the time furnished (it being recognized by the Initial Principal Investors that (i) such Projections are not to be viewed as facts or a guarantee of performance and are subject to significant uncertainties and contingencies many of which are beyond your control and (ii) no assurance can be given that any particular financial projections will be realized, and that actual results during the period or periods covered by any such Projections may differ from the projected results, and such differences may be material). You agree that if, at any time prior to the Closing Date, you become aware that any of the representations and warranties in the preceding sentence would be (with respect to Information relating to the Company and its subsidiaries and businesses, to your knowledge) incorrect in any material respect if the Information or Projections were being furnished and such representations and warranties were being made at such time, then you will use commercially reasonable efforts to promptly supplement the Information and the Projections so that (with respect to Information relating to the Company and its businesses, to your knowledge) such representations are correct, in all material respects, under those circumstances. The making or accuracy of the foregoing representations and warranties, whether or not cured, shall not be a condition to the obligations of the Initial Principal Investors.

4. Conditions

Each Initial Principal Investor's commitments hereunder are subject solely to the satisfaction (or waiver) of the Exclusive Funding Conditions, and upon the satisfaction (or waiver by the Initial Principal Investors) of the Exclusive Funding Conditions, the purchase of the Second Lien Notes shall occur.

Notwithstanding anything in this Commitment Letter, the Closing Payment Letter, the Second Lien Notes Documentation or any other letter agreement or other undertaking concerning the financing of the transactions contemplated hereby to the contrary, (a) the only representations the accuracy of which shall be a condition to the purchase of the Second Lien Notes on the Closing Date shall be (i) such of the representations made by or with respect to the Company in the Merger Agreement as are material to the interests of the Initial Principal Investors, but only to the extent that you or your applicable affiliates have the right (determined without regard to any notice provisions but taking into account any applicable cure provisions) to terminate your (or their) obligations under the Merger Agreement or decline to consummate the Merger as a result of a breach of such representations in the Merger Agreement (the "Specified Merger Agreement Representations") and (ii) the Specified Representations (as defined below) and (b) the terms of

the Second Lien Notes Documentation shall be in a form such that they do not impair the purchase of the Second Lien Notes on the Closing Date if the Exclusive Funding Conditions are satisfied (or waived by the Initial Principal Investors), it being understood that, to the extent any lien search, insurance certificate or endorsement or Collateral (including the creation or perfection of any security interest) is not or cannot be provided on the Closing Date (other than the pledge and perfection of Collateral of the Issuer and the Guarantors with respect to which a lien may be perfected solely by (A) the filing of financing statements under the Uniform Commercial Code ("UCC") and (B) the delivery to the First Lien Administrative Agent (as gratuitous bailee for perfection in accordance with the Term Loan Intercreditor Agreement) of stock certificates or other certificates, if any, representing equity interests of the Issuer or the Guarantors required to be pledged pursuant to the Second Lien Notes Term Sheet to the extent (x) possession of such certificates perfects a security interest therein and (y) other than in the case of stock certificates or other certificates representing equity interests of the Initial Issuer, such stock certificates have been received from the Company after your use of commercially reasonable efforts to do so) after your use of commercially reasonable efforts to do so without undue burden or expense, then the provision and/or perfection, as applicable, of any such lien search, insurance certificate or endorsement and/or Collateral shall not constitute a condition precedent to the purchase of the Second Lien Notes, but may instead be provided within ninety (90) days after the Closing Date, subject to such extensions as are reasonably agreed by the Issuer and the First Lien Administrative Agent or the ABL Administrative Agent in accordance with the First Lien Documentation or the ABL Documentation, as applicable, or pursuant to arrangements to be mutually agreed by the parties hereto acting reasonably. "Specified Representations" means the representations in the Second Lien Notes Documentation relating to corporate or other organizational existence, organizational power and authority of the Issuer and the Guarantors (as they relate to due authorization, execution, delivery and performance of the Second Lien Notes Documentation); due authorization, execution, delivery and enforceability, in each case, relating to the entering into and performance of such Second Lien Notes Documentation by the Issuer and the Guarantors; solvency as of the Closing Date (after giving effect to the Transactions) of the Issuer and its subsidiaries on a consolidated basis (in form and scope consistent with the solvency certificate to be delivered pursuant to paragraph 1 of Exhibit C hereto); no violations or conflicts with organizational documents (as related to the execution and delivery of the Second Lien Notes Documentation) of the Issuer and the Guarantors; Federal Reserve margin regulations; the Investment Company Act; the PATRIOT Act; use of proceeds of the Second Lien Notes not violating OFAC or FCPA; and the creation and perfection of the security interests (subject to customary permitted liens) in the Collateral of the Issuer and the Guarantors and subject in all respects to the foregoing provisions of this paragraph. Notwithstanding anything in this Commitment Letter, the Closing Payment Letter, the Second Lien Notes Documentation or any other letter agreement or other undertaking concerning the financing of the transactions contemplated hereby to the contrary, (a) the commitments of the Initial Principal Investors hereunder are subject to (i) the conditions expressly set forth under the heading "Conditions Precedent to Purchase of Second Lien Notes" in the Second Lien Notes Term Sheet on Exhibit B and (ii) to the conditions set forth in this Section 4 and in Exhibit C hereto (collectively, the "Exclusive Funding Conditions"), (b) the only conditions (express or implied) to the purchase of the Second Lien Notes on the Closing Date are the Exclusive Funding Conditions and (c) to the extent the representations made on the Closing Date with respect to the Company are qualified or subject to "material adverse effect", the definition thereof shall be "Beta Material Adverse Effect" as defined in the Merger Agreement ("Beta

Material Adverse Effect") for purposes of any such representations and warranties with respect to the Company made or to be made on or as of the Closing Date. This paragraph, and the provisions herein, shall be referred to as the "Certain Funds Provision." The Initial Principal Investors will cooperate with you as reasonably requested in coordinating the timing and procedures for the purchase of the Second Lien Notes in a manner consistent with the Merger Agreement on the Closing Date.

5. <u>Indemnification and Expenses</u>

You agree (a) to indemnify and hold harmless the Principal Investors, their controlled affiliates (other than Excluded Affiliates) and controlling persons and the respective directors, officers, employees, agents and other representatives of each of the foregoing and their respective successors (excluding any Excluded Affiliate, each, an "indemnified person") from and against any and all actual losses, claims, damages, liabilities and expenses, joint or several, to which any such indemnified person may become subject arising out of or in connection with this Commitment Letter, the Closing Payment Letter, the Transactions or the use of proceeds of the Second Lien Notes or any claim, litigation, investigation or proceeding (a "Proceeding") relating to any of the foregoing, regardless of whether any indemnified person is a party thereto, whether or not such Proceedings are brought by you, the Company, your or its equity holders, affiliates, creditors or any other person, and to reimburse each indemnified person within thirty (30) days of written demand (together with reasonable backup documentation) for any reasonable out-of-pocket expenses incurred in connection with investigating or defending any of the foregoing (but limited, in the case of legal fees and expenses, to one counsel to such indemnified persons (taken as a whole) and, if reasonably necessary, one local counsel in any relevant material jurisdiction to such indemnified persons (taken as a whole) and, in the case of an actual conflict of interest, one additional counsel to the affected indemnified persons (taken as a whole), in each case excluding allocated costs of in-house counsel); provided that, the foregoing indemnity will not, as to any indemnified person, apply to losses, claims, damages, liabilities or related expenses to the extent they arise from (i) the willful misconduct, bad faith, fraud or gross negligence of such indemnified person (or its controlled affiliates and controlling persons and their respective directors, officers, employees, agents and other representatives (other than Excluded Affiliates)), (ii) the material breach of the Commitment Letter or the Closing Payment Letter by any indemnified person (or its controlled affiliates and controlling persons and their respective directors, officers, employees, agents and other representatives (other than Excluded Affiliates)) (in the case of each of preceding clause (i) and this clause (ii), as determined by a court of competent jurisdiction in a final non-appealable judgment) and (iii) any disputes solely among indemnified persons (other than claims arising out of any act or omission of you or the Company) and (b) if the Closing Date occurs, to reimburse the Principal Investors and their affiliates (other than any Excluded Affiliate) for all reasonable and documented out-of-pocket expenses (including, but not limited to, due diligence expenses, travel expenses, and (limited to) reasonable fees, charges and disbursements of (x) Milbank LLP (as counsel to the GS Principal Investors) and (y) Proskauer Rose LLP (as counsel to the Ares Principal Investors) and, if reasonably necessary, one local counsel in any relevant material jurisdiction incurred in connection with the Second Lien Notes and any related documentation (including this Commitment Letter and the Second Lien Notes Documentation) or the administration, amendment, modification or waiver of any of the foregoing) within thirty (30) days of written demand (including documentation reasonably supporting such request) (other than with respect to such fees and expenses paid on the Closing Date for which written demand

including documentation reasonably supporting such request is provided at least three (3) business days prior to the Closing Date); provided that, such fees and expenses (i) in the case of legal counsel, shall be limited to the reasonable fees and expenses of counsel described in this clause (b) which, in any event, shall exclude allocated costs of in-house counsel and (ii) in the case of any other advisors and consultants, shall be limited solely to advisors and consultants approved by you. No person a party hereto nor the Company or any indemnified person shall be liable for any damages arising from the use by others of Information or other materials obtained through electronic, telecommunications or other information transmission systems, including, without limitation, SyndTrak, IntraLinks, the internet, email or similar electronic transmission systems, in each case, except to the extent any such damages are found in a final non-appealable judgment of a court of competent jurisdiction to have resulted from the gross negligence, bad faith, fraud or willful misconduct of, or material breach of this Commitment Letter or the Closing Payment Letter by, such person (or its controlled affiliates and controlling persons and their respective directors, officers, employees, agents and other representatives (other than Excluded Affiliates)). None of the indemnified persons or you, the Sponsor, Walgreens, the Company or any of your or their respective affiliates or the respective directors, officers, employees, advisors, agents and other representatives of the foregoing shall be liable for any indirect, special, punitive or consequential damages in connection with this Commitment Letter, the Closing Payment Letter, the Second Lien Notes or the transactions contemplated hereby; provided that, nothing contained in this sentence shall limit your indemnification and reimbursement obligations to the extent expressly set forth herein in respect of any losses, claims, damages, liabilities and expenses incurred or paid by an indemnified person to a third party unaffiliated with the Principal Investors. Each indemnified person (by accepting the benefits hereof) agrees to refund and return any and all amounts paid by you to such indemnified person pursuant to the terms of this paragraph to the extent such indemnified person is not entitled to the payment thereof pursuant to the terms of this paragraph as determined by a court of competent jurisdiction in a final non-appealable judgment. You acknowledge that we may receive a benefit, including without limitation, a discount, credit or other accommodation, from any of such counsel based on the fees such counsel may receive on account of their relationship with us including, without limitation, fees paid pursuant hereto.

You shall not be liable for any settlement of any Proceeding (or expenses solely in respect of such settlement) effected without your consent (which consent shall not be unreasonably withheld, delayed or conditioned), but if settled with your written consent, or if there is a final judgment against an indemnified person in any such Proceeding, you agree to indemnify and hold harmless each indemnified person to the extent and in the manner set forth above. You shall not, without the prior written consent of the affected indemnified person (which consent shall not be unreasonably withheld, conditioned or delayed), effect any settlement of any pending or threatened Proceeding against such indemnified person in respect of which indemnity could have been sought hereunder by such indemnified person unless such settlement (a) includes an unconditional release of such indemnified person from all liability or claims that are the subject matter of such Proceeding, (b) includes customary confidentiality provisions and (c) does not include any statement as to any admission of fault, culpability or failure to act by or on behalf of any indemnified person.

In case any Proceeding is instituted involving any indemnified person for which indemnification is to be sought hereunder by such indemnified person, then such indemnified person will promptly notify you of the commencement of any Proceeding after such indemnified

person has actual knowledge of the same; <u>provided</u>, <u>however</u>, that the failure so to notify you will not relieve you from any liability that you may have to such indemnified person pursuant to this <u>Section 5</u>.

Notwithstanding anything to the contrary contained herein, upon the execution of the Second Lien Notes Documentation (a) the relevant provisions of such definitive documentation shall supersede the provisions of the preceding paragraphs and (b) your obligation pursuant to this Commitment Letter to reimburse an indemnified person (or its related indemnified persons) for losses, claims, damages, liabilities, expenses, fees or any such indemnified obligations or any other expense reimbursement shall automatically terminate and be replaced in all respects by the relevant provisions set forth in the Second Lien Notes Documentation.

6. <u>Sharing of Information, Absence of Fiduciary Relationship, Affiliate Activities</u>

You acknowledge that the Initial Principal Investors (or their affiliates, certain of which are full service securities firms (the "Investment Banks")) may from time to time (a) effect transactions, for our own or our affiliates' account or the account of customers, and hold positions in loans, securities or options on loans or securities of you, the Company or its affiliates and of other companies that may be the subject of the transactions contemplated by this Commitment Letter or with which you or the Company may have commercial or other relationships or adverse interests or (b) provide debt financing, equity capital or, in the case of the Investment Banks, investment banking, financial advisory services, securities trading, hedging, financing and brokerage activities and financial planning and benefits counseling to other companies in respect of which you may have conflicting interests. In addition, consistent with each Initial Principal Investor's policy to hold in confidence the affairs of its customers, the Initial Principal Investors will not furnish information obtained from you, the Sponsor, the Company or your or their respective affiliates and representatives to any of their other clients (or to clients of its affiliates) or in connection with the performance by affiliates of the Initial Principal Investors of services for its other clients (or for clients of their affiliates). You also acknowledge that the Initial Principal Investors and their affiliates have no obligation to use in connection with the transactions contemplated hereby, or to furnish to you, confidential information obtained from other companies or other persons.

You further acknowledge and agree that (a) you are capable of evaluating and understanding, and you understand and accept, the terms, risks and conditions of the transactions contemplated by this Commitment Letter, (b) you have been advised that the Initial Principal Investors, the Investment Banks, and their affiliates are engaged in a broad range of transactions that may involve interests that differ from your and your affiliates' interests and that the Initial Principal Investors have no obligation to disclose such interests and transactions to you or your affiliates, (c) you have consulted your own legal, accounting, regulatory and tax advisors to the extent you have deemed appropriate and you are not relying on any Initial Principal Investor for such advice and (d) none of the Initial Principal Investors, Investment Banks nor their affiliates have any obligation to you or your affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein or in any other express writing executed and delivered by such Initial Principal Investor and you.

Please note that neither the Initial Principal Investors nor any of their respective affiliates provide tax, accounting or legal advice.

You further acknowledge and agree that (a) no fiduciary, advisory or agency relationship between you and us is intended to be or has been created in respect of any of the transactions contemplated by this Commitment Letter, irrespective of whether we or our affiliates have advised or are advising you on other matters, (b) we, on the one hand, and you, on the other hand, have an arms-length business relationship that does not directly or indirectly give rise to, nor do you rely on, any fiduciary duty on our part, (c) in connection therewith and with the process leading to the Transactions, the Initial Principal Investors and their respective affiliates (as the case may be) are acting solely as a principal and not as agents or fiduciaries of you, the Sponsor and your or their respective management, stockholders, creditors, affiliates or any other person, (d) you are capable of evaluating and understanding, and you understand and accept, the terms, risks and conditions of the transactions contemplated by this Commitment Letter, (e) you have consulted legal and financial advisors to the extent you deemed appropriate, (f) you have been advised that we and our affiliates are engaged in a broad range of transactions that may involve interests that differ from your interests and that we and our affiliates have no obligation to disclose such interests and transactions to you by virtue of any fiduciary, advisory or agency relationship and (g) you will not claim that any Initial Principal Investor (in its capacity as such), Investment Bank or their respective applicable affiliates, as the case may be, have rendered advisory services of any nature or respect, or owe a fiduciary or similar duty to you or your affiliates, in connection with the transactions contemplated by this Commitment Letter or the process leading thereto.

In addition, please note that an affiliate of the Initial GS Principal Investors has been retained by you as a financial advisor (in such capacity, the "<u>Financial Advisor</u>") in connection with the Merger. You agree to such retention, and further agree not to assert any claim you might allege based on any actual or potential conflicts of interest that might be asserted to arise or result primarily from, on the one hand, the engagement of the Financial Advisor, and on the other hand, our and our affiliates' relationships with you as described and referred to herein.

7. <u>Confidentiality</u>

This Commitment Letter is delivered to you on the understanding that neither this Commitment Letter nor the Closing Payment Letter nor any of their terms shall be disclosed to any other person, except (a) to the Sponsor, Walgreens and your and their respective officers, directors, employees, affiliates, members, partners, stockholders, actual and potential co-investors, attorneys, accountants, agents and advisors on a confidential basis, (b) to the Company and its officers, directors, employees, affiliates, members, partners, stockholders, attorneys, accountants, agents and advisors on a confidential basis, (c) in any legal, regulatory, judicial or administrative proceeding or as otherwise required by applicable law, rule or regulation or as requested by a governmental authority (including a self-regulatory authority) (in which case you agree, to the extent permitted by law, rule or regulation, to inform us promptly thereof), (d) in connection with the exercise of any remedy or enforcement of any right under this Commitment Letter and the Closing Payment Letter, (e) the existence and contents of the Commitment Letter and the Second Lien Notes Term Sheet may be disclosed in any proxy, public filing, prospectus, offering memorandum, offering circulation, syndication materials or other materials in connection with the Transactions, (f) the Second Lien Notes Term Sheet and Exhibit C hereto (but not the Closing

Payment Letter or the contents thereof other than the existence thereof and the contents thereof as part of projections, pro forma information and a generic disclosure of aggregate sources and uses but without disclosing the closing payment set forth therein specifically on a standalone basis) may be disclosed to any rating agency in connection with the Transactions or any public regulatory filing requirement relating to the Transactions, (g) a marketing version of the Second Lien Notes Term Sheet, in form and substance reasonably satisfactory to the Initial Principal Investors, may be disclosed in any syndication or other marketing materials in connection with the First Lien Term Facility, (h) to the extent any such information becomes publicly available other than by reason of disclosure by you, your controlled affiliates or your representatives in violation of this Commitment Letter, (i) with the Initial Principal Investors' consent in writing (including via e-mail) and (j) to the Commitment Parties (as defined in the First Lien Commitment Letter) (provided that, any disclosure of the Closing Payment Letter or its terms and substance under this clause (j) shall be redacted in respect of the closing payment set forth therein) and their respective officers, directors, employees, affiliates (other than Excluded Affiliates), investors, lenders, limited partners, and their respective attorneys, consultants and advisors who, in each case, are involved in the investment of the Second Lien Notes, provided that each of the foregoing persons are bound to maintain the confidentiality hereof and of the Closing Payment Letter on terms substantially similar to those contained herein. The foregoing restrictions shall cease to apply after the Second Lien Notes Documentation shall have been executed and delivered by the parties hereto (other than with respect to the closing payment referenced in the Closing Payment Letter).

Each Initial Principal Investor shall treat confidentially all information received by them from you, the Company, Walgreens, the Sponsor or your or their respective affiliates and representatives in connection with the Merger and the other Transactions and only use such information for the purposes of providing the commitments contemplated by this Commitment Letter; provided, however, that nothing herein shall prevent the Initial Principal Investors from disclosing any such information (a) to rating agencies that are involved with the evaluation, administration and/or monitoring of an Initial Principal Investor's investment in the Second Lien Notes; (b) to any actual or prospective Principal Investors or actual or prospective purchasers of the Second Lien Notes (or beneficial interests therein) (other than Disqualified Institutions and persons to whom you have affirmatively denied to provide your consent to the assignment or syndication thereof); provided that, the disclosure of any such information to any such person referred to above shall be made subject to the acknowledgment and acceptance by such person that such information is being disseminated on a confidential basis (on substantially the terms set forth in this paragraph or as is otherwise reasonably acceptable to you and the Initial Principal Investors) in accordance with the standard sell-down/transfer processes of the Initial Principal Investors or customary market standards for dissemination of such type of information; (c) in any legal, judicial or administrative proceeding or other compulsory process or otherwise as required by applicable law, rule or regulations or as requested by a governmental authority (in which case such Initial Principal Investor shall promptly notify you, in advance, to the extent permitted by law, rule or regulation (except with respect to any routine audit or examination conducted by bank accountants or regulatory authority exercising routine examination or regulatory authority)); provided that, notwithstanding the foregoing, no notification shall be required in connection with any filings that may be required by the federal securities laws or the rules, regulations, schedules and/or forms of the Securities and Exchange Commission; (d) upon the request or demand of any governmental or regulatory authority having jurisdiction over such Initial Principal Investor or any of its affiliates or upon the good faith determination by counsel that such information should be disclosed in light

of ongoing oversight or review of such Initial Principal Investor by any governmental or regulatory authority having jurisdiction over such Initial Principal Investor or its affiliates (in which case such Initial Principal Investor shall, except with respect to any audit or examination conducted by bank accountants or any governmental bank regulatory authority exercising examination or regulatory authority, promptly notify you, in advance, to the extent lawfully permitted to do so); (e) to such Initial Principal Investor's affiliates (other than Excluded Affiliates) and its and their respective limited partners, lenders, investors, managed accounts, service providers and rating agencies and to the respective officers, directors, employees, legal counsel, independent auditors, professionals and other experts or agents of each of the foregoing (collectively, "Representatives"), in each case who are involved in the investment of the applicable Initial Principal Investor in the Second Lien Notes (provided that any such Representative is advised of its obligation to retain such information as confidential and agrees to keep information of this type confidential and the Initial Principal Investors shall remain responsible for such Representatives' compliance with this provision); (f) to the extent any such information becomes publicly available other than by reason of disclosure by the Initial Principal Investors, their affiliates or Representatives in breach of this Commitment Letter or other obligation of confidentiality owed to you, the Sponsor, Walgreens, the Company or your or their respective affiliates; (g) for purposes of establishing a "due diligence" defense; (h) to the extent that such information is received by such Initial Principal Investor, its affiliates (other than an Excluded Affiliate) or their respective Representatives from a third party that is not known (after due inquiry) by such Initial Principal Investor to be subject to confidentiality obligations to you, the Sponsor, Walgreens, the Company or your or their respective affiliates; (i) to the extent that such information is independently developed by such Initial Principal Investor, its affiliates (other than an Excluded Affiliate) or their respective Representatives without the use of such information; and (i) to enforce their respective rights hereunder or under the Closing Payment Letter; provided, however, that, no such disclosure shall be made by the Initial Principal Investors to (i) any of their affiliates' employees, officers, directors, legal counsel, professionals and other experts or agents that are engaged directly or indirectly in a sale of the Company and its subsidiaries as buy-side or sell-side representative and acting in such capacity other than, in each case, any Over the Wall Person (each a "Sale Advisory Person" or an "Excluded Affiliates") or (ii) any Disqualified Institution. As used herein, "Over the Wall Person" means, with respect to any Initial Principal Investor, any directors, officers or senior employees of such Initial Principal Investor or any of its affiliates who are required, in accordance with industry regulations, or the applicable Initial Principal Investor's or its affiliate's internal policies and procedures to act in a supervisory or managerial capacity and the applicable Initial Principal Investor's and its affiliates' internal legal, compliance, risk management, conflicts clearance and other support personnel and credit and investment committee members. The Initial Principal Investors' obligations under this paragraph shall remain in effect until the earlier of (x) two years from the date hereof and (y) the date the Second Lien Notes Documentation is effective (the confidentiality provisions of which shall include exceptions no less permissive than those contained in this <u>Section 7</u>), at which time our obligations under this paragraph shall automatically terminate and be superseded by the confidentiality provisions in the Second Lien Notes Documentation upon the execution and delivery thereof. Notwithstanding anything to the contrary contained herein, nothing in this paragraph shall prohibit any Principal Investor from disclosing any information to any lender to, or any managed account or limited partner of, such Principal Investor to the extent such information is subject to customary confidentiality obligations binding on such lender, managed account or limited partner pursuant to customary investment advisory, fund or loan documentation.

Notwithstanding anything herein to the contrary, any party to this Commitment Letter (and any employee, representative or other agent of such party) may disclose the tax treatment and tax structure of the transactions contemplated by this Commitment Letter and all materials of any kind (including opinions or other tax analyses) that are provided to it relating to such tax treatment and tax structure, except that (i) such disclosures shall not include the identity of any existing or future party (or any affiliate of such party) to this Commitment Letter and (ii) no party shall disclose any information relating to such tax treatment and tax structure to the extent nondisclosure is reasonably necessary in order to comply with applicable securities laws. For this purpose, the tax treatment of the transactions contemplated by this Commitment Letter shall be any purported or claimed U.S. federal income tax treatment of such transactions and the tax structure of such transactions is any fact that may be relevant to understanding the purported or claimed U.S. federal income tax treatment of such transactions.

"Disqualified Institutions" means (a) those persons that are competitors of you, the Company and its subsidiaries and Walgreens Co. ("Walgreens") or the Sponsor to the extent separately identified by you or the Sponsor to us (or, after the Closing Date, by the Issuer or the Sponsor to the Initial Principal Investors) in writing from time to time, in each case including any such entity's affiliates that are identified as such pursuant to this clause (a) or those that are clearly identifiable on the basis of their name (in each case, other than bona fide diversified debt funds) (other than those excluded pursuant to clauses (b) of this paragraph)) or (b) those banks, financial institutions and other persons to the extent separately identified by you or the Sponsor to us in writing prior to the date hereof or as you and we shall mutually agree on or after such date, in each case including any such entity's affiliates that are identified as such pursuant to this clause (b) or are clearly identifiable as such on the basis of their name.

8. Miscellaneous

Subject to the immediately following paragraph, this Commitment Letter shall not be assignable by any party hereto (except by you to the ultimate Issuer or one or more of your affiliates that is a newly formed domestic "shell" company controlled, directly or indirectly, by you to effect the consummation of the Merger) without the prior written consent of the other parties hereto (and any purported assignment without such consent shall be null and void), is intended to be solely for the benefit of the parties hereto and, to the extent set forth in Section 5, the indemnified persons, and is not intended to and does not confer any benefits upon, or create any rights in favor of, any person other than the parties hereto and the indemnified persons to the extent expressly set forth herein. Each Initial Principal Investor shall be liable solely in respect of its own commitment to purchase the Second Lien Notes on a several, and not joint, basis with any other Initial Principal Investor. This Commitment Letter may not be amended or waived except by an instrument in writing signed by you and the Initial Principal Investors. This Commitment Letter may be executed in any number of counterparts, each of which shall be an original, and all of which, when taken together, shall constitute one agreement. Delivery of an executed signature page of this Commitment Letter by facsimile or other electronic transmission (e.g., "pdf" or "tif") shall be effective as delivery of a manually executed counterpart hereof. This Commitment Letter and the Closing Payment Letter are the only agreements that have been entered into among us and you with respect to the Second Lien Notes and set forth the entire understanding of the parties with respect thereto. This Commitment Letter shall be governed by, and construed and interpreted in accordance with, the laws of the State of New York without regard to principles of conflicts of

law, to the extent that the same are not mandatorily applicable by statute and would require or permit the application of the law of another jurisdiction; provided, however, that the laws of the state of Delaware which governs the Merger Agreement shall govern in determining (a) the interpretation of a "Beta Material Adverse Effect" and whether a "Beta Material Adverse Effect" has occurred, (b) the accuracy of any Specified Merger Agreement Representation and whether as a result of any inaccuracy thereof you or your applicable affiliate have the right or would have the right (without regard to any notice requirement but taking into account any applicable cure provisions) to terminate your obligations (or to refuse to consummate the Merger) under the Merger Agreement and (c) whether the Merger has been consummated in accordance with the terms of the Merger Agreement. Each of the parties hereto agrees that (i) this Commitment Letter is a binding and enforceable agreement with respect to the subject matter herein, including an agreement to negotiate in good faith the Second Lien Notes Documentation by the parties hereto in a manner consistent with this Commitment Letter and the Second Lien Notes Documentation Principles, it being acknowledged and agreed that the purchase of the Second Lien Notes is subject only to the Exclusive Funding Conditions, including the execution and delivery of the Second Lien Notes Documentation as provided in this Commitment Letter and (ii) the Closing Payment Letter is a binding and enforceable agreement with respect to the subject matter contained therein. Reasonably promptly following the execution of this Commitment Letter, the parties hereto shall proceed with the negotiation in good faith of the Second Lien Notes Documentation for the purpose of executing and delivering the Second Lien Notes Documentation substantially simultaneously with the consummation of the Merger. Section headings used herein are for convenience of reference only and are not to affect the construction of, or

Notwithstanding anything to the contrary contained in the immediately preceding paragraph or otherwise in this Commitment Letter or in the Closing Payment Letter, (I) each Initial GS Principal Investor shall have the right to reallocate, sell, resell, assign and/or transfer its Second Lien Notes (including, for the avoidance of doubt, its beneficial interests therein) and/or its commitments in respect thereof and/or any closing payment (or any portion thereof) (a) to any other Initial GS Principal Investor or (b) to any affiliated investment entity and/or other affiliate of Goldman Sachs & Co. LLC or any fund, investor, entity or account that is managed, sponsored or advised by Goldman Sachs & Co. LLC or its affiliates (any such persons described in this clause (b) who acquires any Second Lien Notes and/or commitments in respect thereof pursuant to this clause (I), "Other GS Principal Investors") and (II) each Initial Ares Principal Investor shall have the right to reallocate, sell, resell, assign and/or transfer its Second Lien Notes (including, for the avoidance of doubt, its beneficial interests therein) and/or its commitments in respect thereof and/or the closing payment (or any portion thereof) (a) to any other Initial Ares Principal Investor or (b) to any affiliated investment entity and/or other affiliate of Ares Management LLC or Ares Capital Management LLC or any fund, investor, entity or account that is managed, sponsored or advised by Ares Management LLC or Ares Capital Management LLC or their affiliates, any such persons described in this clause (b) who acquires any Second Lien Notes and/or commitments in respect thereof pursuant to this clause (II), "Other Ares Principal Investors"); provided that, unless you agree in your sole discretion in writing in connection therewith, in each case (i) no such Initial Principal Investor shall be relieved or novated from its respective obligation hereunder with respect to the Second Lien Notes in connection with any applicable assignment or transfer until the o

Principal Investor's commitment in respect of the Second Lien Notes until the occurrence of the purchase and sale of the Second Lien Notes on the Closing Date and (iii) the Initial Principal Investors shall retain exclusive control over all rights and obligations with respect to their commitments in respect of the Second Lien Notes, including all rights with respect to consents, modifications, supplements and amendments, until the occurrence of the purchase and sale of the Second Lien Notes on the Closing Date.

NOTWITHSTANDING ANYTHING TO THE CONTRARY HEREIN OR IN THE EXHIBITS HERETO OR IN THE CLOSING PAYMENT LETTER, NO INITIAL PRINCIPAL INVESTOR OR OTHER PRINCIPAL INVESTOR IS ACTING AS AN UNDERWRITER, ARRANGER, TRUSTEE, AGENT OR IN A SIMILAR ROLE OR IS OTHERWISE PERFORMING ANY SERVICES HEREUNDER AND THE ROLE OF EACH INITIAL PRINCIPAL INVESTOR AND OTHER PRINCIPAL INVESTOR HEREUNDER AND UNDER THE SECOND LIEN NOTES TERM SHEET AND THE CLOSING PAYMENT LETTER SHALL BE LIMITED TO ITS COMMITMENT TO PROVIDE DEBT FINANCING AS A PRINCIPAL.

Each of the parties hereto irrevocably and unconditionally (a) submits, for itself and its property, to the exclusive jurisdiction of any federal court sitting in the Borough of Manhattan in the City of New York or, if that court does not have subject matter jurisdiction, in any state court located in the City and County of New York, and any appellate court from any thereof, over any suit, action or proceeding arising out of or relating to the Transactions or the other transactions contemplated hereby, this Commitment Letter or the Closing Payment Letter or for recognition or enforcement of any judgment and agrees that all claims in respect of any such action or proceeding shall be heard and determined in such New York state or, to the extent permitted by law, in such federal court; provided, however, that the Initial Principal Investors shall be entitled to assert jurisdiction over you and your property in any court in which jurisdiction may be held over you or your property and (b) agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. You and we agree that service of any process, summons, notice or document by registered mail addressed to any of the parties hereto at the applicable addresses above shall be effective service of process for any suit, action or proceeding brought in any such court. You and we hereby irrevocably and unconditionally waive, to the fullest extent you and we may legally and effectively do so, any objection to the laying of venue of any such suit, action or proceeding brought in any court in accordance with clause (a) of the first sentence of this paragraph and any claim that any such suit, action or proceeding has been brought in any inconvenient forum. YOU AND WE HEREBY IRREVOCABLY WAIVE (TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW) TRIAL BY JURY IN ANY SUIT, ACTION, PROCEEDING, CLAIM OR COUNTERCLAIM BROUGHT BY OR ON BEHALF OF ANY PARTY RELATED

The Initial Principal Investors hereby notify you that, pursuant to the requirements of the USA PATRIOT Act, Title III of Pub. L. 107-56 (signed into law on October 26, 2001) (as amended, the "PATRIOT Act"), they are required to obtain, verify and record information that identifies the Issuer and each Guarantor, which information includes names, addresses, tax identification numbers and other information that will allow the Initial Principal Investors to

identify the Issuer and each Guarantor in accordance with the PATRIOT Act. This notice is given in accordance with the requirements of the PATRIOT Act and is effective for the Initial Principal Investors.

This paragraph, the indemnification, jurisdiction, waiver of jury trial, service of process, venue, governing law, sharing of information, no agency or fiduciary duty, confidentiality and reimbursement (if applicable) provisions contained herein and the provisions of the Closing Payment Letter shall remain in full force and effect regardless of whether definitive financing documentation shall be executed and delivered and notwithstanding the termination of this Commitment Letter or the commitments hereunder; provided that, your obligations under this Commitment Letter (but not under the Closing Payment Letter, and other than in respect of your agreements in respect of no fiduciary or similar duties and your obligations in respect of confidentiality (which shall terminate in accordance with Section 7)) shall automatically terminate and be superseded by the provisions of the Second Lien Notes Documentation upon the execution thereof, and you shall automatically be released from all liability in connection therewith at such time.

If the foregoing correctly sets forth our agreement, please indicate your acceptance of the terms of this Commitment Letter and the Closing Payment Letter by returning to us executed counterparts of this Commitment Letter and the Closing Payment Letter not later than 11:59 p.m., New York time, on March 15, 2019. This offer will automatically expire at such time if we have not received such executed counterparts in accordance with the preceding sentence, unless we shall, in our sole discretion, agree to an extension, this Commitment Letter and the commitments hereunder shall automatically terminate in the event that (a) if the purchase of the Second Lien Notes does not occur on or before 11:59 p.m., New York City time, on December 20, 2019 (the "Expiration Time"), (b) the Merger closes or (c) after execution of the Merger Agreement and prior to the consummation of the Merger, the termination of the Merger Agreement by you (or your affiliates) or with your (or your affiliates') written consent (other than with respect to provisions therein that expressly survive termination) (the earliest of (a), (b) and (c), the "Commitment Termination Date"); provided that, the termination of any commitment pursuant to this sentence does not prejudice your or our rights and remedies in respect of any breach of this Commitment Letter.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

We are pleased to have been given the opportunity to assist you in connection with this important financing.

Very truly yours,

ARES CAPITAL MANAGEMENT LLC, on behalf of one or more affiliated funds, investment vehicles and/or managed accounts,

By: /s/ Mark Affolter Name: Mark Affolter Title: Authorized Signatory

ARES MANAGEMENT LLC, on behalf of one or more affiliated funds, investment vehicles and/or managed accounts,

By: /s/ Naseem Sagati Aghili Name: Naseem Sagati Aghili Title: Authorized Signatory

[Project Opal - Commitment Letter Signature Page]

We are pleased to have been given the opportunity to assist you in connection with this important financing.

GS MEZZANINE PARTNERS VII, L.P.

By: GS Mezzanine Partners VII GP, L.L.C., its General Partner

By: /s/ Kirsten Anthony

Name: Kirsten Anthony Title: Managing Director

GS MEZZANINE PARTNERS VII OFFSHORE, L.P.

By: GS Mezzanine Partners VII GP, L.L.C., its General Partner

By: /s/ Kirsten Anthony Name: Kirsten Anthony Title: Managing Director

GS MEZZANINE PARTNERS VII OFFSHORE TREATY, L.P.

By: Goldman Sachs & Co. LLC, Collateral Servicer and Duly Authorized

Agent

By: /s/ Kirsten Anthony Name: Kirsten Anthony Title: Managing Director

BROAD STREET DANISH CREDIT PARTNERS, L.P.

By: Goldman Sachs & Co. LLC, Duly Authorized

By: /s/ Kirsten Anthony Name: Kirsten Anthony Title: Managing Director

BROAD STREET CREDIT HOLDINGS LLC

By: /s/ Kirsten Anthony Name: Kirsten Hagen Title: Vice President

[Project Opal - Commitment Letter Signature Page]

Accepted and agreed to as of the date first above written:

HC GROUP HOLDINGS II, INC.

By: /s/ Clifford Berman
Name: Clifford Berman Title: Corporate Secretary

[Project Opal - Commitment Letter Signature Page]

<u>Project Opal</u> \$400 Million Senior Secured Second Lien PIK Toggle Floating Rate Notes Commitments of Initial Principal Investors

Schedule I(a)

	Second Lien Notes			
Initial GS Principal Investors	Commitment Amounts			
GS Mezzanine Partners VII, L.P.	\$	15,395,000		
GS Mezzanine Partners VII Offshore, L.P.	\$	91,822,000		
GS Mezzanine Partners VII Offshore Treaty, L.P.	\$	34,986,000		
Broad Street Danish Credit Partners, L.P.	\$	5,200,000		
Broad Street Credit Holdings LLC	\$	52,597,000		
TOTAL	\$	200,000,000		

Schedule I(b)

Initial Ares Principal Investors	Second Lien Notes ommitment Amounts
Ares Capital Management LLC (on behalf of one or more affiliated funds, investment vehicles and/or managed	
accounts)	\$ 100,000,000
Ares Management LLC (on behalf of one or more affiliated funds, investment vehicles and/or managed accounts)	\$ 100,000,000
TOTAL	\$ 200,000,000

PROJECT OPAL

Second Lien Notes Transaction Summary

Capitalized terms used but not defined in this <u>Exhibit A</u> shall have the meanings set forth in the letter to which this <u>Exhibit A</u> is attached or in <u>Exhibits B</u> or C thereto.

HC Group Holdings II, Inc., a Delaware corporation ("Omega"), which is indirectly controlled by Madison Dearborn Partners, LLC ("MDP") and its controlled affiliates and associated funds (together with MDP, collectively, the "Sponsor") and Walgreens, intends to consummate a stock-for-stock exchange, directly or indirectly, with a company previously identified to the Initial Principal Investors as "Beta" (the "Company"), pursuant to the Merger Agreement defined below. In connection therewith, it is intended that:

- (a) The Company will form Beta Sub, Inc., a Delaware corporation, a direct or indirect wholly-owned domestic subsidiary ("Merger Sub 1") and Beta Sub, LLC, a Delaware limited liability company, a direct or indirect wholly-owned domestic subsidiary ("Merger Sub 2"). On the Closing Date, pursuant to that certain Agreement and Plan of Merger, dated as of the date hereof (together with the exhibits and disclosure schedules thereto, as amended, modified, supplemented or waived, the "Merger Agreement") (A) Merger Sub 1 will merge with and into Omega with Omega as the surviving entity and (B) Omega shall merge with and into Merger Sub 2, with Merger Sub 2 surviving such merger (such mergers collectively referred to herein as the "Merger"). After giving effect to Merger, the Company will own the businesses currently conducted by each of Omega and Beta. Immediately after the Merger, Merger Sub 2 (in such capacity, the "Initial Issuer") will obtain Senior Credit Facilities and issue the Second Lien Notes as described below and immediately thereafter the Company shall assume the obligations of the Initial Issuer as the issuer under the Second Lien Notes (the "Debt Assumption").
- (b) The Issuer will obtain senior secured first lien term loan facility in an aggregate principal amount of up to \$925 million, subject to increase pursuant to the Flex Provisions set forth in First Lien Fee Letter (as defined below) (the "First Lien Term Facility" and the loans thereunder, the "First Lien Term Loans"), as described in Exhibit B to that certain commitment letter (such term sheet, as in effect on the date hereof, the "First Lien Term Sheet"), dated as of the date hereof, among you and the Commitment Parties (as defined therein) party thereto (such commitment letter, together with the term sheet attached thereto, as in effect on the date hereof, the "First Lien Commitment Letter"). The lenders under the First Lien Term Facility are herein referred to as the "First Lien Lenders"; the administrative agent and collateral agent thereunder is herein referred to as the "First Lien Administrative Agent"; the definitive documentation relating to the First Lien Term Facility is herein referred to as the "First Lien Documentation"; that certain fee letter dated as of the date hereof, as in effect on the date hereof, by and among you and the Commitment Parties (as defined in the First Lien Commitment Letter) party thereto accompanying the First Lien Commitment Letter is herein referred to as the "First Lien Fee Letter"; the First

Lien Commitment Letter, together with the First Lien Fee Letter, are herein referred to as the "<u>First Lien Commitment Papers</u>." The provisions under the heading "Market Flex" in the First Lien Fee, as such letter may be amended, amended and restated or otherwise modified, are herein referred to as the "<u>Flex Provisions</u>."

- (c) The Issuer (and certain of its subsidiaries) will obtain senior secured asset-based revolving credit facility in an aggregate principal amount of up to \$150 million (the "ABL Facility" and, together with the First Lien Term Facility, the "First Lien Credit Facilities"), as described in Exhibit C to First Lien Commitment Letter (the "ABL Term Sheet"). The lenders under the ABL Facility are herein referred to as the "ABL Lenders"; the administrative agent and collateral agent thereunder is herein referred to as the "ABL Administrative Agent"; the definitive documentation relating to the ABL Facility is herein referred to as the "ABL Documentation."
- (d) The Issuer will issue and sell to the Initial Principal Investors senior secured second lien PIK toggle floating rate notes in an aggregate principal amount of up to \$400 million (the "Second Lien Notes"), as described in Exhibit B to the Commitment Letter.
- (e) (A) All existing indebtedness for borrowed money of (I) the Company and its subsidiaries pursuant to: (i) that certain First Lien Note Purchase Agreement, dated as of June 29, 2017, among Beta, as issuer, the purchasers party thereto from time to time and Wells Fargo Bank, National Association, as collateral agent, (ii) that certain Second Lien Note Purchase Agreement, dated as of June 29, 2017, among Beta, as issuer, the purchasers party thereto from time to time and Wells Fargo Bank, National Association, as collateral agent and (iii) that certain Indenture, dated February 11, 2014, by and among Beta, the guarantors named therein and U.S. Bank National Association, as trustee and (II) Omega pursuant to (i) that certain Credit Agreement, dated as of April 7, 2015, by and among HC Group Holdings III, Inc., as borrower, the lenders party thereto, Bank of America, N.A., as administrative agent and the other parties thereto and (ii) that certain Indenture, dated as of April 7, 2015, among HC Group Holdings III, Inc., as issuer, the guarantors party thereto and U.S. Bank National Association, as trustee and collateral agent, will, in each case be repaid or satisfied or discharged, and all related guaranties and security interests with respect thereto will be terminated and released simultaneously concurrently with the initial funding of the First Lien Credit Facilities and the purchase of the Second Lien Notes (or arrangements for such termination and release shall have been made) and (B) all outstanding preferred stock issued by the Company will be redeemed for cash (the "Preferred Redemption Cash") and/or converted into common stock of the Company (collectively, the "Refinancing").
- (f) The proceeds of a portion of the ABL Facility, the First Lien Term Facility, the Second Lien Notes and cash on hand at Omega and its subsidiaries and the Company and its subsidiaries on the Closing Date will be applied to consummate the Merger, to effect the Refinancing and to pay the fees, closing payments, premiums, expenses and other transaction costs incurred in connection with the foregoing (the "<u>Transaction Costs</u>").

The transactions described above are collectively referred to herein as the "<u>Transactions</u>." For purposes of the Commitment Letter and the Closing Payment Letter, "<u>Closing Date</u>" shall mean the date of (i) the satisfaction or waiver of the Exclusive Funding Conditions, (ii) the funding under the First Lien Credit Facilities, (iii) the issuance, sale and purchase of the Second Lien Notes, (iv) the consummation of the Refinancing and (v) the consummation of the Merger.

PROJECT OPAL

\$400 million Senior Secured Second Lien PIK Toggle Floating Rate Notes Summary of Terms and Conditions

Set forth below is a summary of the principal terms and conditions for the Second Lien Notes. Capitalized terms used but not defined in this Exhibit B shall have the meanings set forth in the letter to which this Exhibit B is attached or in Exhibits A or C attached thereto.

Issuer:

Initially, the Initial Issuer and immediately following consummation of the Debt Assumption, the Company (such relevant entity, the "Issuer"); provided that, the Issuer may, in its sole discretion, designate one or more of its direct or indirect domestic wholly-owned subsidiaries as co-issuers subject to customary conditions; including that such co-issuer be treated as a corporation for U.S. federal income tax purposes.

Principal Investors and Purchasers:

"GS Principal Investors" means (a) each Initial GS Principal Investor, (b) each affiliated investment entity and/or other affiliate of Goldman Sachs & Co. LLC and (c) each fund, investor, entity or account that is managed, sponsored or advised by Goldman Sachs & Co. LLC or its affiliates.

"Ares Principal Investors" means (a) each Initial Ares Principal Investor, (b) each affiliated investment entity and/or other affiliate of Ares Management LLC or Ares Capital Management LLC and (c) each fund, investor, entity or account that is managed, sponsored or advised by Ares Management LLC or Ares Capital Management LLC or their affiliates.

"GS Purchasers" means each (a) each Initial GS Principal Investor and (b) each other GS Principal Investor that executes or otherwise becomes a party to the Note Purchase Agreement (as defined below) (or otherwise becomes a beneficial owner of Second Lien Notes (as defined below)) or to which any Second Lien Notes (or beneficial interests therein) are transferred or assigned.

"Ares Purchasers" means each (a) each Initial Ares Principal Investor and (b) each other Ares Principal Investor that executes or otherwise becomes a party to the Note Purchase Agreement (or otherwise becomes a beneficial owner of Second Lien Notes (as defined below)) or to which any Second Lien Notes (or beneficial interests therein) are transferred or assigned.

"Purchasers" means (a) the GS Purchasers and (b) the Ares Purchasers.

"GS Disposition Date" means the first date occurring after the Closing Date on which the GS Principal Investors (in the aggregate) cease to beneficially own (within the meaning of Rule 13d-3 promulgated under the Securities Exchange Act of 1934, as amended) more than 25% of the aggregate principal amount of the then outstanding Second Lien Notes.

"Ares Disposition Date" means the first date occurring after the Closing Date on which the Ares Principal Investors (in the aggregate) cease to beneficially own (within the meaning of Rule 13d-3 promulgated under the Securities Exchange Act of 1934, as amended) more than 25% of the aggregate principal amount of the then outstanding Second Lien Notes.

"Controlling Parties" means (i) the GS Purchasers so long as the GS Disposition Date has not occurred and (ii) the Ares Purchasers so long as the Ares Disposition Date has not occurred; *provided* that from and after the Sell-Down Date (as defined below), the Controlling Party shall be Holders (as defined below) of Second Lien Notes holding more than 50% of the aggregate principal amount of the then outstanding Second Lien Notes.

"Sell-Down Date" means the first date after the Closing Date after which both the GS Disposition Date and the Ares Disposition Date has occurred.

Trustee, Calculation Agent and Collateral Agent:

Ankura Trust Company, LLC, subject to the receipt of customary approvals and the completion of onboarding procedures, is expected to act as trustee (the "<u>Trustee</u>"), calculation agent and collateral agent (in such capacity, the "<u>Collateral Agent</u>") for and on behalf of the holders of the Second Lien Notes (the "<u>Holders</u>"). The Trustee will perform the duties customarily associated with such roles, including, in the case of the calculation agent, the determination of interest rates. The fees and out-of-pocket expenses of the Trustee (including the fees of outside counsel to the Trustee) shall be borne by the Issuer in a manner to be agreed between the Issuer and the Trustee.

Second Lien Notes:

Senior secured second lien PIK toggle floating rate notes of the Issuer in an aggregate principal amount of \$400 million to be issued under the Indenture (as defined below) and purchased by the applicable Purchasers under the Note Purchase Agreement on the Closing Date (the "Second Lien Notes"). Other than customary

provisions for replacement notes, no additional Second Lien Notes or other notes will be issued under the Indenture.

First Lien Indebtedness:

The Indenture will include a single basket in the indebtedness covenant and a single corresponding liens exception with respect to (a) the incurrence of indebtedness under the First Lien Term Facility and the ABL Facility and (b) other indebtedness that is secured by the Collateral (as defined below) on a pari passu basis with the First Lien Term Facility or the ABL Facility (any such indebtedness under (a) and (b), "First Lien Indebtedness"), which shall permit only the following (i) the First Lien Term Loans, (ii) the ABL Facility (as the commitments thereunder may be increased by up to \$50.0 million), (iii) Additional First Lien Indebtedness (as defined below), (iv) such additional First Lien Indebtedness that the Initial Principal Investors and the Issuer working together in good faith mutually agree is permitted to be incurred under the First Lien Documentation and the ABL Documentation and (v) any permitted refinancing indebtedness in respect of the foregoing; *provided* that up to \$25.0 million of the general liens basket in the Second Lien Notes Documentation will also be available to secure indebtedness otherwise permitted to be incurred under the Second Lien Notes Documentation as First Lien Indebtedness. There will not be other capacity to incur First Lien Indebtedness under the Second Lien Notes Documentation. First Lien Indebtedness may not have any obligors that are not Note Parties, may only be secured by the Collateral (as defined below) and must be subject to the Intercreditor Agreements (as defined below).

"Additional First Lien Indebtedness" means indebtedness in an aggregate principal amount up to (x) the greater of (I) \$210 million and (II) 1.0x Consolidated EBITDA (as defined below) (determined on a pro forma basis) for the most recent four quarters for which financial statements are available (less all amounts of Incremental Equivalent Debt incurred under the corresponding "free and clear" basket for Incremental Equivalent Debt described below) plus (y) the aggregate principal amount of all voluntary prepayments and debt buybacks of (i) First Lien Indebtedness (with respect to any revolving facility, to the extent accompanied by a permanent reduction of the revolving commitments) and (ii) any Incremental Equivalent Debt to the extent originally incurred under the "free and clear" prong, in each case of clauses (i) and (ii), except to the extent financed with the proceeds of long-term indebtedness (other than revolving indebtedness) plus (z) an unlimited additional amount; provided that after giving pro forma effect to the incurrence of such additional amount (and after giving effect to all customary pro forma events and adjustments), the Consolidated First Lien Net Leverage Ratio (to be defined in a manner consistent

with the Second Lien Notes Documentation Principles) as of the last day of the most recently ended period of four consecutive fiscal quarters for which financial statements are internally available on a pro forma basis does not exceed the Consolidated First Lien Net Leverage Ratio as of the Closing Date (after giving effect to the incurrence of such amounts (assuming full utilization of the then proposed Additional First Lien Indebtedness and excluding the proceeds of any borrowing under such then proposed Additional First Lien Indebtedness for purposes of determining net debt) and to any related transactions); it being understood that (I) the Issuer may elect to use clause (z) above prior to clause (x) or (y) and regardless of whether there is capacity under clause (x) or (y), and if clauses (x), (y) and (z) are available and the Issuer does not make an election, the Issuer will be deemed to have elected clause (z); (II) the Issuer may reclassify utilizations among clauses (x), (y) and (z) above if, at the time of such reclassification, the Issuer would be permitted to incur the aggregate principal amount of indebtedness being so reclassified; and (III) any amounts incurred under clauses (x) and (y) above concurrently with amounts incurred under clause (z) thereof will not count as indebtedness for the purposes of calculating the applicable ratio in clause (z) thereof.

Incremental Equivalent Debt:

The Second Lien Notes Documentation will permit the Issuer to raise one or more second lien, junior lien or unsecured term loan or note facilities the terms and conditions of which are customary pursuant to documentation other than the Second Lien Notes Documentation (i.e., outside the Second Lien Notes Documentation) ("Incremental Equivalent Debt") in an aggregate principal amount up to (x) the greater of (I) \$210 million and (II) 1.0x Consolidated EBITDA (determined on a pro forma basis) for the most recent four quarters for which financial statements are available (less all amounts of Additional First Lien Indebtedness incurred under the corresponding "free and clear" basket for Additional First Lien Indebtedness described above) plus (y) the aggregate principal amount of all voluntary prepayments and debt buybacks of (i) First Lien Indebtedness (with respect to any revolving facility (including the ABL Facility), to the extent accompanied by a permanent reduction of the revolving commitments), (ii) Incremental Equivalent Debt that is pari passu with the Second Lien Notes and (iii) any Incremental Equivalent Debt that is secured on a junior lien basis to the Second Lien Notes or unsecured to the extent originally incurred under the "free and clear" prong, in each case of (i) through (iii), except to the extent financed with the proceeds of long-term indebtedness (other than revolving indebtedness) plus (z) an unlimited additional amount; provided that after giving pro forma effect to the incurrence of such additional amount (and after giving effect to all customary pro

forma events and adjustments) (i) in the case of any Incremental Equivalent Debt secured on a pari passu basis with the Second Lien Notes or on a junior lien basis to the Second Lien Notes, the Consolidated Senior Secured Net Leverage Ratio (to be defined in a manner consistent with the Second Lien Notes Documentation Principles) as of the last day of the most recently ended period of four consecutive fiscal quarters for which financial statements are internally available on a pro forma basis does not exceed the Consolidated Senior Secured Net Leverage Ratio as of the Closing Date and (ii) in the case of any unsecured Incremental Equivalent Debt, either (A) the Fixed Charge Coverage Ratio (to be defined in a manner consistent with the Second Lien Notes Documentation Principles but which shall include PIK interest expense, if any), on a pro forma basis, is not less than 2.00:1.00 or (B) if incurred in connection with a permitted acquisition or other permitted investment, the Fixed Charge Coverage Ratio on a pro forma basis does not decrease (in each case, after giving effect to the incurrence of such amounts (assuming full utilization of the then proposed Incremental Equivalent Debt and excluding the proceeds of any borrowing under such then proposed Incremental Equivalent Debt for purposes of determining net debt) and to any related transactions) (such amount under clauses (x), (y) and (z) above, collectively the "Available Incremental Amount"); it being understood that (I) the Issuer may elect to use clause (z) of the Available Incremental Amount prior to clause (x) or (y) and regardless of whether there is capacity under clause (x) or (y), and if clauses (x), (y) and (z) are available and the Issuer does not make an election, the Issuer will be deemed to have elected clause (z); (II) the Issuer may reclassify utilizations among clauses (x), (y) and (z) above if, at the time of such reclassification, the Issuer would be permitted to incur the aggregate principal amount of indebtedness being so reclassified; and (III) any amounts incurred under clauses (x) and (y) of the Available Incremental Amount concurrently with amounts incurred under clause (z) thereof will not count as indebtedness for the purposes of calculating the applicable ratio in clause (z) thereof; provided that (i) any Incremental Equivalent Debt shall rank pari passu or junior in right of payment with the Second Lien Notes and will either be secured on a pari passu basis with the Second Lien Notes by the same Collateral securing the Second Lien Notes (the "Parity Lien Debt"), secured on a junior lien basis to the Second Lien Notes by the same Collateral securing the Second Lien Notes or be unsecured, (ii) any Incremental Equivalent Debt that is secured on a pari passu basis with the Second Lien Notes shall be subject to the "MFN" requirement in clause (vi) of the "Incremental Facilities" section of the First Lien Term Sheet (but excluding, for the avoidance of doubt, the "sunset"

and other exceptions set forth in such section), (iii) except in the case of a customary bridge loan incurred outside the Second Lien Notes Documentation the maturity date of which provides for an automatic extension of the maturity date thereof, subject to customary conditions, to a date that is not earlier than the Maturity Date (as defined below) ("Extendable Bridge Loans"), the Incremental Equivalent Debt will have a final maturity no earlier than the Maturity Date, (iv) the weighted average life to maturity of any Incremental Equivalent Debt shall be no shorter than that of the Second Lien Notes as originally in effect prior to any extension or any redemptions of the Second Lien Notes), (v) any fees payable in connection with any such Incremental Equivalent Debt shall be determined by the Issuer and the arrangers providing for such Incremental Equivalent Debt, (vi) except as otherwise required or permitted in clauses (i) through (v) above, all other terms of such Incremental Equivalent Debt, if not consistent with the terms of the Second Lien Notes shall either (x) not be materially more restrictive to the Issuer (as determined by the Issuer), when taken as a whole, than the terms of the Second Lien Notes (except for covenants or other provisions applicable only to periods after the Maturity Date) or (y) be reasonably satisfactory to each applicable Controlling Party (it being understood that (1) covenants or other provisions applicable only to periods after the Maturity Date need not be reasonably satisfactory to any Controlling Party and (2) to the extent that any financial maintenance or other covenant is added for the benefit of any Incremental Equivalent Debt, no consent shall be required from any Controlling Party to the extent that such financial maintenance or other covenant is also added for the benefit of the Second Lien Notes). Incremental Equivalent Debt may not have any obligors that are not Note Parties, may only be secured (to the extent secured) by the Collateral and must be subject to customary intercreditor arrangements that are reasonably satisfactory to each applicable Controlling Party.

Limited Condition Transactions:

The Limited Condition Transaction provisions shall be the same as those contained in the First Lien Documentation subject to appropriate modifications to reflect the second lien status of the Second Lien Notes.

Refinancing Second Lien Debt:

The Second Lien Notes Documentation will permit the Issuer to refinance the Second Lien Notes from time to time, in whole or part, pursuant to documentation other than the Second Lien Notes Documentation (i.e., outside the Second Lien Notes Documentation) ("Refinancing Second Lien Debt") in a principal amount not to exceed the principal amount of indebtedness so refinanced (plus any accrued but unpaid interest, existing commitments unutilized under such indebtedness being refinanced

to the extent such commitments are permanently reduced, premiums (including tender premiums), penalties and fees payable by the terms of such indebtedness thereon and any defeasance costs and any fees, expenses, original issue discount and upfront payments incurred in connection with such refinancing), with one or more additional series of unsecured or subordinated notes or loans or senior secured loans or notes incurred by the Issuer that will be secured by the Collateral on a pari passu basis with the Second Lien Notes or on a junior basis to the Second Lien Notes; provided that (i) except in the case of Extendable Bridge Loans, any Refinancing Second Lien Debt does not mature prior to the Maturity Date, or have a shorter weighted average life to maturity than, that of the Second Lien Notes, (ii) the other terms and conditions of any such Refinancing Second Lien Debt (excluding pricing, interest rate margins, fees, discounts, rate floors and prepayment or redemption terms which shall be determined by the Issuer) shall, if not consistent with the terms of the Second Lien Notes, not be materially more restrictive to the Issuer (as determined by the Issuer), when taken as a whole, than the terms and conditions of the Second Lien Notes Documentation (except for covenants or other provisions applicable only to periods after the latest final scheduled maturity date of the Second Lien Notes remaining outstanding after giving effect to the incurrence or issuance of such Refinancing Second Lien Debt), (iii) any secured Refinancing Second Lien Debt shall be subject to customary intercreditor documentation that is reasonably acceptable to each applicable Controlling Party, (iv) any Refinancing Second Lien Debt shall not have obligors that are not obligors of the Second Lien Notes, (v) to the extent secured, such Refinancing Second Lien Debt shall not be secured by any property or assets other than the Collateral and (vi) the proceeds of such Refinancing Second Lien Debt shall be applied, substantially concurrently with the incurrence thereof, to the redemption of outstanding Second Lien Notes and associated interest, premium, fees and expenses.

Purpose:

The proceeds of the Second Lien Notes will be used by the Issuer on the Closing Date, together with the proceeds of the First Lien Term Facility, borrowing under the ABL Facility (if any) and certain cash available on the balance sheet of Omega and its subsidiaries and the Company and its subsidiaries, to finance a portion of the Transactions (including working capital and/or purchase price adjustments and the payment of Transaction Costs, upfront fees and original issue discount with respect to the First Lien Term Facility).

Availability:

The entire amount of the Second Lien Notes will be issued, sold and purchased on the Closing Date. Repayments and redemptions of the Second Lien Notes may not be reissued.

Interest Rates:

As set forth on Annex I hereto.

Default Rate:

Any principal payable under or in respect of the Second Lien Notes not paid when due shall bear interest at the applicable interest rate plus 2.00% per annum. Other overdue amounts shall bear interest at the applicable interest rate on the principal to which such overdue amount relates plus 2.00%. Default interest will be payable on demand.

Final Maturity and Amortization:

The Second Lien Notes will mature on the date that is eight years after the Closing Date (the "Maturity Date") and will have no interim amortization.

Guarantees:

The same parties that guarantee the Issuer's obligations under the First Lien Term Facility (collectively, the "<u>Guarantors</u>" and, together with the Issuer, collectively, the "<u>Note Parties</u>" and each a "<u>Note Party</u>") will unconditionally guarantee jointly and severally on a senior second lien basis (the "<u>Second Lien Guarantees</u>") all obligations of the Issuer under the Second Lien Notes (the "<u>Second Lien Obligations</u>"). The guarantee release provisions will be substantially the same as those in the First Lien Documentation and will also provide for the release of a Second Lien Guarantee as described in the last sentence of "Security" below.

Security:

Subject to the Certain Funds Provision and consistent with the Second Lien Notes Documentation Principles, the obligations of the Issuer and the Guarantors in respect of the Second Lien Notes will be secured (subject to exceptions consistent with the Second Lien Notes Documentation Principles) on a second-priority basis by the same Term Loan Priority Collateral (as defined in the First Lien Commitment Letter) and on a third-priority basis by the ABL Priority Collateral (as defined in the First Lien Commitment Letter) (the "Collateral"). The collateral release provisions will be substantially the same as those in the First Lien Documentation; provided that, there shall be an automatic release (pursuant to the Indenture (in the case of clause (x) below) and the Intercreditor Agreements (as defined below) (in the case of clause (y) below) of the lien on any asset or property constituting Collateral securing the Second Lien Notes if such asset or property is disposed of (other than to a Note Party) (x) in connection with a transaction permitted by the Second Lien Notes Documentation or (y) in connection with the exercise of remedies by the First Lien Administrative Agent (or other applicable designated representative) or the ABL

Administrative Agent (or other applicable designated representative) so long as the liens on such asset or property securing all first lien obligations (including ABL obligations) are also released (*provided* that such release shall not extend to, or affect the rights of the second lien secured parties in, the proceeds of such disposition). The Term Loan Intercreditor Agreement shall also provide that if the equity interests of a Guarantor are disposed of as described in the proviso to the prior sentence, then in addition to the lien on such equity interests securing the Second Lien Notes being released, the Second Lien Guarantee of such Guarantor shall also be released so long such Guarantor's guarantee of all first lien obligations (including ABL obligations) is also released (*provided* that such release shall not extend to, or affect the rights of the second lien secured parties in, the proceeds of such disposition).

Intercreditor Agreements:

The intercreditor agreement (the "Term Loan Intercreditor Agreement") between the First Lien Administrative Agent and the Trustee shall be consistent with and substantially similar to that certain Intercreditor Agreement, dated as of February 28, 2018, among Bank of America, N.A., in its capacities as first lien credit agreement administrative agent and second lien credit agreement administrative agent and acknowledged and agreed to by Mayfield Agency Borrower Inc. and the other parties to such acknowledgement; provided that (i) an obligation (including any obligation under the First Lien Documentation) will only constitute a "first lien secured obligation" (or similar term) to the extent such obligation is, at the time of incurrence, permitted to be incurred (and secured) as a first priority obligation under the then extant documentation governing indebtedness constituting first lien secured obligations and the then extant documentation governing indebtedness constituting second lien secured obligations, (ii) the Term Loan Intercreditor Agreement will include a customary "DIP cap" that will provide that that the aggregate principal amount of any DIP financing (plus the aggregate principal amount of first lien secured obligations and the aggregate face amount of any letters of credit issued and outstanding under first lien debt facilities in each case that would remain outstanding after giving effect to such DIP financing) does not exceed 115% of the sum of the aggregate outstanding principal amount of first lien secured obligations on the date that the applicable insolvency or liquidation proceeding commenced plus the aggregate face amount of any letters of credit issued and outstanding under first lien debt facilities outstanding on the date that the applicable insolvency or liquidation proceeding commenced (it being understood and agreed that cash management obligations and hedging obligations shall not be subject to such cap amount), (iii) the Term Loan Intercreditor Agreement will permit

the second lien secured parties to propose, vote in favor of or otherwise support any plan of reorganization that is not inconsistent with the priorities set forth in the Term Loan Intercreditor Agreement and (iv) the permitted second lien actions will include the ability for a second lien secured party to take actions to prove, preserve, perfect, preserve or protect (but not enforce) its rights in and perfection and priority of its lien on the Collateral.

The intercreditor agreement between the First Lien Administrative Agent, the Trustee and the ABL Administrative Agent shall be (a) consistent with and substantially similar to that certain ABL Intercreditor Agreement, dated as of October 1, 2018, among Bank of America, N.A., in its capacities as ABL collateral agent and first lien collateral agent and acknowledged and agreed to by Specialty Building Products Holdings, LLC and the other parties thereto and (b) otherwise consistent with the First Lien Documentation Principles (as defined in the First Lien Commitment Letter), ABL Documentation Principles (as defined in the First Lien Commitment Letter) and the Second Lien Notes Documentation Principles (the "ABL/Term Intercreditor Agreement" and, together with the Term Loan Intercreditor Agreement, the "Intercreditor Agreements").

Offer to Purchase Upon a Change of Control:

The Issuer will be required to make an offer to repurchase all of the Second Lien Notes following the occurrence of a Change of Control (to be defined in a manner consistent with the Second Lien Notes Documentation Principles) at a price in cash equal to (x) 101% of the outstanding principal amount thereof, *plus* (y) accrued and unpaid interest to, but not including, the date of repurchase unless the Issuer shall redeem all of the Second Lien Notes pursuant to the "Optional Redemption" section below.

Offer to Purchase from Asset Sale Proceeds:

The Issuer will be required to make an offer to repurchase the Second Lien Notes, without premium or penalty, and, to the extent required thereunder, any Parity Lien Debt that is pari passu in right of payment with the Second Lien Notes on a pro rata basis, which offer shall be at 100% of the principal amount thereof plus accrued and unpaid interest to, but not including, the date of repurchase with an amount equal to 100% of the net cash proceeds of all non-ordinary course asset sales or other dispositions of property by the Issuer and its restricted subsidiaries (including casualty insurance and condemnation proceeds, but with exceptions for sales of certain designated business lines, certain sales or other dispositions of ABL Priority Collateral (as defined in the First Lien Commitment Letter), inventory and other ordinary course dispositions, obsolete or worn-out property, property no longer used or useful in the business and other exceptions to be set forth in the Second Lien Notes

Documentation in a manner consistent with the Second Lien Notes Documentation Principles) in excess of a threshold amount and subject to the right of the Issuer to (i) repay or prepay First Lien Indebtedness with 100% of the net cash proceeds (so long as, in the case of any repayment or prepayment of revolving borrowings, the underlying commitments are terminated) or (ii) reinvest 100% of the net cash proceeds in assets used or useful in the business if such proceeds are reinvested (or committed to be reinvested) within 18 months and, if so committed to reinvestment, reinvested no later than 180 days after the end of such 18-month period.

Offer to Purchase from Proceeds of Indebtedness:

The Issuer will be required to make an offer to repurchase the Second Lien Notes with 100% of the net cash proceeds of issuances of non-permitted debt obligations of the Issuer and its restricted subsidiaries consistent with the terms of the First Lien Term Sheet; <u>provided</u> that the Issuer shall only be required to make such offer to the extent the proceeds are not required to be applied to the First Lien Term Loans.

Optional Redemption:

The Second Lien Notes will not be redeemable at the option of the Issuer except as follows:

- · In the case of any optional redemption occurring on or after the Closing Date but prior to the first anniversary of the Closing Date, the Second Lien Notes shall be redeemable at par plus a call premium equal to 3% of the principal amount of the Second Lien Notes being so redeemed plus accrued and unpaid interest.
- In the case of any optional redemption occurring on or after the first anniversary of the Closing Date but prior to the second anniversary of the Closing Date, the Second Lien Notes shall be redeemable at par plus a call premium equal to 2% of the principal amount of the Second Lien Notes being so redeemed plus accrued and unpaid interest.
- In the case of any optional redemption occurring on or after the second anniversary of the Closing Date but prior to the third anniversary of the Closing Date, the Second Lien Notes shall be redeemable at par plus a call premium equal to 1% of the principal amount of the Second Lien Notes being so redeemed plus accrued and unpaid interest.

On and after the third anniversary of the Closing Date, the Second Lien Notes shall be redeemable at par plus accrued and unpaid interest.

Any redemption or repurchase of the Second Lien Notes shall be accompanied by accrued and unpaid interest to, but not including, the applicable redemption or repurchase date.

Second Lien Notes
Documentation Principles:

The definitive documentation for the Second Lien Notes (the "Second Lien Notes Documentation") shall be initially drafted by counsel to the Issuer (other than the Note Purchase Agreement, which shall initially be drafted by counsel to the Initial Principal Investors), be based on the First Lien Documentation, (I) with appropriate changes (a) to reflect the nature of the Second Lien Notes as debt securities (instead of loans) issued pursuant to (i) a note purchase agreement (the "Note Purchase Agreement"), among the Issuer, the Guarantors and the Initial Principal Investors, (ii) an indenture (the "Indenture"), among the Issuer, the Guarantors and the Trustee and Collateral Agent, providing for the issuance of the Second Lien Notes, (iii) security documentation, instruments or agreements evidencing liens and security interests in the Collateral in favor of the Collateral Agent equivalent in scope (including with respect to the exceptions therefrom) to the collateral granted to the lenders under the First Lien Term Facility, but modified to reflect customary differences in roles between a trustee and/or collateral agent under an indenture on the one hand and an active administrative agent under a credit facility on the other hand (provided discretionary determinations that under the First Lien Documentation are made by the First Lien Administrative Agent will generally be made under the Second Lien Notes Documentation by the Controlling Parties subject to exceptions with respect to provisions directly affecting the Trustee and others to be mutually agreed) and (iv) the Intercreditor Agreement, (b) to reflect that the dollar "baskets" for the representations. warranties, affirmative covenants and negative covenants under the Second Lien Notes Documentation and the dollar thresholds for the events of default in the Second Lien Notes Documentation will be sized with cushions that are 25% greater than the corresponding dollar "baskets" and thresholds (provided that in no event will such cushion apply to any ratio-based baskets, first lien debt/liens capacity, thresholds relating to collateral and guarantee requirements or the "starter amount" under the "Available Amount Basket" (to be defined in a manner consistent with the Second Lien Notes Documentation Principles) (such items, the "Non-Cushion Items")), (c) to include in the Note Purchase Agreement representations, warranties and conditions precedent consistent with the descriptions under "Representations and Warranties" and

"Conditions Precedent to Purchase of Second Lien Notes" below, (d) to include in the Indenture the affirmative and negative covenants and events of default consistent with the descriptions under "Affirmative Covenants," "Negative Covenants" and "Events of Default" below and (e) to reflect that discretionary determinations that under the First Lien Documentation are made by the First Lien Administrative Agent will generally be made under the Second Lien Notes Documentation by the Controlling Parties subject to exceptions with respect to provisions directly affecting the Trustee and others to be mutually agreed (as adjusted pursuant to terms below, the "Second Lien Notes Documentation Principles"), and (II) in any event will contain only those conditions to purchase, mandatory offers to purchase, representations, warranties, affirmative and negative covenants (but no financial covenants) and events of default expressly set forth or referenced in this Second Lien Notes Term Sheet, in each case applicable to the Issuer and its restricted subsidiaries, in each case, with materiality thresholds, qualifications, exceptions, "baskets" and grace and cure periods to be mutually agreed, and consistent with this Second Lien Notes Term Sheet and the Second Lien Notes Documentation Principles. The definitive terms will be negotiated in good faith to finalize the Second Lien Notes Documentation Principles, as promptly as reasonably practicable.

Notwithstanding anything to the contrary in the foregoing otherwise in this Commitment Letter, to the extent the amount of any dollar-based basket and/or related grower percentage (including the "free and clear" amount), the amount of any dollar-based threshold, the level of any ratio-based basket or the definition of "Consolidated EBITDA" in the First Lien Term Sheet or the proposed First Lien Documentation is modified or removed pursuant to the Flex Provisions or is otherwise modified in the course of the syndication by the First Lien Lead Arrangers (as defined in the First Lien Commitment Letter) of the First Lien Term Facility pursuant the First Lien Commitment Papers (whether in a manner more favorable or less favorable to the First Lien Lenders), such term in this Exhibit B and, if applicable, in the proposed Second Lien Notes Documentation shall be deemed to be modified or removed in an identical manner subject to the application of any applicable cushion (subject to the Non-Cushion Items); *provided* that (i) in no event will the "starter" amount for the Available Amount Basket be greater than the greater of \$75 million and 35% of Consolidated EBITDA for the most recently ended four fiscal quarter period, (ii) in no event will the add-backs to Consolidated EBITDA expressly identified in the First Lien Term Sheet be modified in a manner that is more favorable to the Issuer and (iii)

nothing in this sentence will result in a modification to Applicable Margin (as defined below) or the Applicable Rate (as defined below) or of any redemption premiums.

"Consolidated EBITDA" shall be defined in a manner consistent with the Second Lien Documentation Principles; provided, however that any add-back for anticipated cost savings and "run rate" synergies shall be subject to an aggregate cap of up to 30% of Consolidated EBITDA for any period; provided that such cap shall not apply to any add-back to the extent related to, or applicable from, the Transactions and/or identified in the Sponsor Model (as defined in the First Lien Commitment Letter) or any quality of earnings report, or that is otherwise in compliance with financial statements prepared in accordance with Regulation S-X.

Other Key Documentation Matters:

In addition, the Second Lien Notes Documentation will include terms, conditions and provisions to the following effect:

- 1. prior to the Sell-Down Date, the Second Lien Notes Documentation will not permit (or include the ability for) "contribution indebtedness" (or any similar concept permitting indebtedness to be incurred as a result of an equity contribution) and, from and after such date, the contribution debt basket included in the Second Lien Notes Documentation shall be limited to an amount equal to 100% of any cash common equity contribution;
- 2. the Sponsor shall not be permitted to acquire or hold, collectively, whether acquired in primary or secondary transactions, any First Lien Indebtedness, except (x) in the case of the First Lien Term Facility, as otherwise provided opposite the heading "Assignments and Participations" in the First Lien Term Sheet and (y) in the case of any other First Lien Indebtedness, as may be provided in the documentation governing such First Lien Indebtedness; *provided* that the provisions thereof are substantially identical to, or less favorable to the Sponsor than, the provisions described in the foregoing clause (x);
- 3. the Issuer shall not, and shall not permit any Guarantor to, directly or indirectly, create, incur, assume or suffer to exist (x) indebtedness (including indebtedness acquired or assumed as part of a permitted acquisition) that is contractually subordinated or junior in right of payment to any indebtedness of the Issuer or such Guarantor, as the case may be, unless such indebtedness is expressly subordinated in right of payment to the Second Lien Notes or the

applicable Guarantor's Second Lien Guarantee to the extent and in the same manner as such indebtedness is subordinated in right of payment to other indebtedness of the Issuer or such Guarantor, as the case may be, or (y) any indebtedness that is secured and which is, by its express terms, subordinated as to rights to receive, or subject to turnover of, payments or proceeds of collateral to any other indebtedness of the Issuer or a Guarantor secured in whole or in part by the same collateral (including any "first-loss" or "last-out" tranche under the First Lien Term Facility or any other first lien facilities), unless the liens securing such indebtedness rank *pari passu* or junior to the liens securing the Second Lien Notes;

- 4. the permitted refinancing provisions of the Second Lien Notes Documentation will not permit the refinancing of unsecured indebtedness with secured indebtedness or the refinancing of indebtedness secured by a Lien on the Collateral with indebtedness secured by a more senior Lien on the Collateral; and
- 5. at the Issuer's option, the Second Lien Notes Documentation shall contain customary "AHYDO catch-up" payment provisions, providing that the Issuer will make payments on the Second Lien Notes, before the close of any accrual period ending after the fifth anniversary of the issue date, an amount sufficient to ensure that the Second Lien Notes will not be "applicable high yield discount obligations" within the meaning of Section 163(i)(1) of the Internal Revenue Code.

Representations and Warranties:

Subject to the Second Lien Notes Documentation Principles, the representations and warranties shall be the same as those representations and warranties contained in the First Lien Documentation subject to appropriate modifications: (a) to reflect the second lien status of the Second Lien Notes, (b) to the extent there are any dollar "baskets" set forth in the representations and warranties that are not Non-Cushion Items, including a 25% cushions to such dollar "baskets" relative to the corresponding dollar "baskets" in the First Lien Documentation and (c) to reflect the status of the Second Lien Notes as debt securities under an indenture (rather than loans under a credit agreement); it being further understood that the Note Purchase Agreement will also include representations and warranties that would be customary in an offering of securities pursuant to Rule 144A and/or Regulation S with respect to applicable securities law matters (e.g., no general

solicitation). All representations and warranties shall be made on the Closing Date.

Conditions Precedent to Purchase of Second Lien Notes: Subject to the Certain Funds Provision, the availability of the Second Lien Notes on the Closing Date will be subject only to the Exclusive Funding Conditions.

Affirmative Covenants:

Subject to the Second Lien Notes Documentation Principles, the affirmative covenants shall be the same as those affirmative covenants contained in the First Lien Documentation subject to appropriate modifications (a) to reflect the second lien status of the Second Lien Notes, (b) to the extent there are any dollar "baskets" set forth in the affirmative covenants that are not Non-Cushion Items, including a 25% cushions to such dollar "baskets" relative to the corresponding dollar "baskets" in the First Lien Documentation and (c) to reflect the status of the Second Lien Notes as debt securities under an indenture (rather than loans under a credit agreement); it being understood that the Note Purchase Agreement will include affirmative covenants that would be customary in an offering of securities pursuant to Rule 144A and/or Regulation S with respect to applicable securities law matters (e.g., DTC eligibility and using commercially reasonable efforts to provide Rule 144A information so long as the Second Lien Notes constitute "restricted securities"); it being further understood that the Purchasers shall receive customary annual projections and budget reports for each fiscal year (with such delivery required, with respect to any fiscal year, within 90 days of the end of the prior fiscal year).

Negative Covenants:

Subject to the Second Lien Notes Documentation Principles, the negative covenants shall the same as those negative covenants contained in the First Lien Documentation, except that (a) appropriate modifications will be made to reflect (i) the second lien status of the Second Lien Notes and (ii) the status of the Second Lien Notes as debt securities under an indenture (rather than loans under a credit agreement), (b) the dollar "baskets" for the negative covenants under the Second Lien Notes Documentation will be sized with cushions that are 25% greater than the corresponding dollar "baskets" (subject to the Non-Cushion Items) under the First Lien Documentation and (c) any Ratio Debt secured on a pari passu basis with the Second Lien Notes shall be subject to the "MFN" requirement in clause (vi) of the "Incremental Facilities" section of the First Lien Term Sheet (but excluding for the avoidance of doubt, the "sunset" and other exceptions set forth in such section); it being understood that the incurrence of First Lien Indebtedness will be limited as provided under the caption "First Lien Indebtedness" above; it being further understood that the Note Purchase

Agreement will include negative covenants that would be customary in an offering of securities pursuant to Rule 144A and/or Regulation S with respect to applicable securities law matters (e.g., restrictions on conducting any offering that would be integrated with the sale of the Second Lien Notes in a manner that would require registration of the Second Lien Notes under the Securities Act of 1933, as amended).

Assignments and Participations:

On and after the Closing Date, the Second Lien Notes will be held in book entry form, in the form of one or more global notes, through The Depository Trust Company

The Second Lien Notes (including any commitments in respect thereof or beneficial interests therein) may be sold, pledged, assigned or otherwise transferred at any time under the Indenture without restriction in accordance with applicable securities laws; *provided* that the Note Purchase Agreement will provide that (other than with respect to participations which are governed by the immediately following paragraph) (A) the Purchasers will not sell, pledge, assign or otherwise transfer any Second Lien Notes to any natural persons or Disqualified Institutions and (B) a Purchaser will only be permitted to sell, pledge, assign or otherwise transfer any Second Lien Notes with the consent of the Issuer (not to be unreasonably withheld or delayed); *provided* that no consent of the Issuer shall be required (i) if such sale, pledge, assignment or other transfer is made to another beneficial owner of the Second Lien Notes or an affiliate or approved fund of a beneficial owner of Second Lien Notes or (ii) after the occurrence and during the continuance of a payment or bankruptcy (with respect to the Issuer) event of default; *provided, further*, that if not received within 10 business days of acknowledged receipt of a written request by a beneficial owner of Second Lien Notes, the consent of the Issuer shall be deemed to have been given and (C) a Purchaser may not sell, pledge, assign or otherwise transfer any Second Lien Notes unless the purchaser, pledgee, assignee or other transferee agrees to the provisions in (A) and (B) of this paragraph and the provisions of the immediately following paragraph.

The Note Purchase Agreement will also provide that the Purchasers will be permitted to sell participations (other than to natural persons, Disqualified Institutions and the Sponsor and its affiliates (other than Investment Funds) (to be defined in a manner consistent with the Second Lien Notes Documentation Principles) in Second Lien Notes without consent being required, subject to customary limitations consistent with the corresponding provisions of the First Lien Documentation.

Financial Covenant:

None.

Unrestricted Subsidiaries:

The unrestricted subsidiary provisions shall be the same as those contained in the First Lien Credit Documentation subject to appropriate modifications to reflect the second lien status of the Second Lien Notes.

Events of Default:

Subject to the Second Lien Notes Documentation Principles, the Second Lien Notes Documentation will contain events of default the same as those contained in the First Lien Documentation, including as to exceptions and qualifications, provided the Second Lien Notes Documentation shall (i) include thresholds that are 25% higher than those in the First Lien Documentation subject to the Non-Cushion Items, (ii) include a "cross acceleration" and a "cross-payment default" at final maturity to the First Lien Term Facility, the ABL Facility and other First Lien Indebtedness (and not a cross default to the First Lien Term Facility, the ABL Facility or such First Lien Indebtedness), and (iii) include cross-default to continuing defaults under other material indebtedness of a Note Party or a restricted subsidiary.

Voting:

Amendments, supplements and waivers of the Second Lien Notes Documentation will require the approval of the Holders holding more than 50% of the aggregate principal amount of the Second Lien Notes (the "Required Holders"), except that the consent of each Holder affected shall also be required with respect to certain matters customary for high yield debt securities, including any reduction of principal or change in maturity. The amendment provisions shall otherwise be customary for high yield debt securities; provided that, prior to the Sell-Down Date there shall be no ability to make any amendments or supplements to the Second Lien Notes Documentation without the consent of the Required Holders (except in respect of the addition or release of a Guarantor and/or Collateral in accordance with the terms of the Indenture), it being understood that from and after such date certain customary administrative and technical amendments to the Second Lien Notes Documentation (including to the collateral security documents) may be effected without the consent of any Holder of Second Lien Notes.

At any time after the Sell-Down Date and (x) prior to the discharge of the obligations under the First Lien Term Facility, the First Lien Administrative Agent and (y) prior to the discharge of the obligations under the ABL Facility (solely with respect to the ABL Priority Collateral), the ABL Administrative Agent, shall be entitled to extend any deadline or requirement in connection with compliance with the guarantee and security provisions under, and

in accordance with, the First Lien Documentation and such extensions shall automatically apply to the corresponding requirements under the Second Lien Notes Documentation.

Cost and Yield Protection: Customary for transactions of this type.

Confidentiality: Customary confidentiality provisions consistent with the Second Lien Notes Documentation Principles, which shall

include, in any event, exceptions consistent with those set forth in the Commitment Letter.

Expenses and Indemnification: The Second Lien Notes Documentation will contain provisions for expenses and indemnification the same as those

provisions for expenses and indemnification contained in the First Lien Documentation. The Second Lien Notes Documentation will also include indemnity and expense reimbursement for the Trustee in a manner agreed to between

the Issuer and the Trustee.

Governing Law and Forum: New York.

Counsel to the GS Principal

Investors:

Milbank LLP.

Counsel to the Ares Principal

Investors:

Proskauer Rose LLP

B-19

INTEREST

Applicable Rate:

The Second Lien Notes will bear interest at a rate per annum, reset quarterly, equal to the sum of (i) Adjusted LIBOR plus (ii) the Applicable Margin (such sum, the "Applicable Rate").

For any interest period ending on or prior to the first anniversary of the Closing Date, the Issuer may elect to pay 100% of the interest with respect to such interest period in kind (the "100% PIK Option" and, such amount of interest to be paid in kind, the "100% PIK Interest Amount"). In the event that the Issuer determines to exercise the 100% PIK Option with respect to any interest period, the interest rate for such period will be the Applicable Rate for such interest period plus an additional 100 basis points.

For any interest period ending on or prior to the second anniversary of the Closing Date, the Issuer may elect to pay 50% of the interest with respect to such interest period in kind (the "50% PIK Option" and, such portion of the interest to be paid in kind, the "50% PIK Interest Amount"), with the remainder of the interest for such period to be paid in cash. In the event that the Issuer determines to exercise the 50% PIK Option, the interest rate for such period will be the Applicable Rate for such interest period plus an additional 50 basis points.

In order to exercise the 100% PIK Option or the 50% PIK Option with respect to any interest period, the Issuer shall be required to provide a written notice indicating the same to Trustee and the Purchasers in a manner to be mutually agreed.

"Applicable Margin" means 8.75% per annum.

"Adjusted LIBOR" means, with respect to any interest period when used in reference to any Second Lien Note, (a) the rate of interest (rounded upwards, if necessary, to the nearest 1/100th) appearing on Reuters Screen LIBOR01 Page (or otherwise on the Reuters screen) (or any successor or substitute page of such service, or any successor to such service as determined by the calculation agent) as the London interbank offered rate for deposits in U.S. dollars for a term of three months, at approximately 11:00 a.m. (London time) on the applicable determination date (but if more than one rate is specified on such page, the rate will be an arithmetic average of all such rates) or (b) if such rate described in clause (a) of this definition is not available, the rate of interest (rounded upwards, if necessary, to the nearest 1/100th) appearing in the money markets section of the Wall Street Journal (or any successor or substitute page of such service, or any successor to

such service as determined by the calculation agent) as the London interbank offered rate for deposits in U.S. dollars for a term of three months on the applicable determination date; provided that, notwithstanding anything to the contrary, in no event shall Adjusted LIBOR be less than 0.00% per annum (the "<u>LIBOR Floor</u>").

The Second Lien Notes Documentation will contain provisions, to be agreed between you and the Initial Principal Investors, providing for replacement of the Adjusted LIBOR Rate in the event of LIBOR discontinuation.

Interest Payment Dates:

Quarterly on each date corresponding to the Closing Date and the three-month, six-month and nine-month anniversary of such date.

Rate Basis:

Interest will be computed on the basis of a 360-day year for actual days elapsed.

PROJECT OPAL

Conditions

The purchase of the Second Lien Notes shall be subject solely to the satisfaction or waiver by the Initial Principal Investors of the following conditions (subject to the Certain Funds Provision). Capitalized terms used but not defined in this <u>Exhibit C</u> have the meanings set forth in the letter to which this <u>Exhibit C</u> is attached or in <u>Exhibits A</u> or <u>B</u> thereto.

- 1. Subject to the Certain Funds Provision, the Issuer and each Guarantor shall have executed and delivered the Second Lien Notes Documentation, in each case, consistent with the Commitment Letter and the Second Lien Notes Term Sheet, to which they are parties and the Initial Principal Investors shall have received:
 - (a) a customary notice of borrowing;
 - (b) customary closing officer's (certifying as to resolutions, organizational documents and incumbency, as to the accuracy of the Specified Representations in all material respects and no amendments or modifications to the provisions of the Merger Agreement that are materially adverse to the interests of the Initial Principal Investors) and good standing (of the jurisdiction of organization of the Issuer and Guarantors) certificates and legal opinions; and
 - (c) a certificate (substantially in the form of <u>Annex I</u> to this <u>Exhibit C</u>) from the chief financial officer (or other officer with reasonably equivalent duties) of the Issuer certifying that the Issuer and its subsidiaries, on a consolidated basis after giving effect to the Transactions, are solvent.
- 2. Prior to, or substantially concurrently with the purchase of the Second Lien Notes, the Refinancing shall have been consummated.
- 3. The Merger shall be consummated in all material respects pursuant to the Merger Agreement substantially concurrently with the purchase of the Second Lien Notes without giving effect to any amendments or modifications to the provisions thereof or express waivers or consents thereto that, in each case, are materially adverse to the interests of the Initial Principal Investors without the consent of the Initial Principal Investors, such consent not to be unreasonably withheld, conditioned or delayed (it being understood and agreed that (i) any change in the Merger Consideration (as defined in the Merger Agreement) shall be deemed not to be adverse to the interest of the Initial Principal Investors and (ii) any adverse modification to the definition of Beta Material Adverse Effect (or adverse express waiver or express consent in respect of the definition of Beta Material Adverse Effect) without the prior written consent of the Initial Principal Investors (such consent not to be unreasonably withheld, delayed or conditioned) shall be deemed to be materially adverse to the interests of the Initial Principal Investors); provided that in each case the Initial Principal Investors shall be deemed to have consented to such modification, amendment,

waiver or consent unless they shall object thereto within 3 business days of receipt of written notice of such modification, amendment, consent or waiver.

- 4. Since the date of the Merger Agreement there shall not have been a Beta Material Adverse Effect (as defined in the Merger Agreement as of the date hereof).
- The Initial Principal Investors shall have received (1) (i) the audited consolidated statements of operations, shareholders' equity and cash flows of HC Group Holdings III, Inc. for the fiscal years ended December 31, 2016 and December 31, 2017, and the related audited consolidated balance sheets as of the end of such fiscal years and (ii) the audited consolidated statements of operations, shareholders' equity and cash flows of HC Group Holdings III, Inc. for, and the related audited consolidated balance sheets as of the end of, each fiscal year of HC Group Holdings III, Inc. ended at least 120 days prior to the Closing Date (each Initial Principal Investor acknowledges receipt of such audited statements of operations, shareholders' equity and cash flows and the related balance sheets in clause (1)(i) hereof), (2) (i) the unaudited consolidated statement of operations of HC Group Holdings III, Inc. for the fiscal quarters ending March 31, 2018, June 30, 2018 and September 30, 2018, and the related unaudited consolidated balance sheet as of the end of such fiscal quarters and (ii) the unaudited consolidated statement of operations of HC Group Holdings III, Inc. for, and the related unaudited consolidated balance sheet as of the end of, each fiscal quarter of HC Group Holdings III, Inc. (other than the fourth fiscal quarter) ended after the close of the fiscal quarter ending September 30, 2018 and at least 45 days prior to the Closing Date (each Initial Principal Investor acknowledges receipt of unaudited consolidated statements of operations and the related balance sheet in clause (2)(i) hereof), (3) (i) the audited consolidated statements of operations, shareholders' equity and cash flows of the Company for the fiscal years ended December 31, 2016 and December 31, 2017, and the related audited consolidated balance sheets as of the end of such fiscal years and (ii) the audited consolidated statements of operations, shareholders' equity and cash flows of the Company for, and the related audited consolidated balance sheets as of the end of, each fiscal year of the Company ended at least 120 days prior to the Closing Date (each Initial Principal acknowledges receipt of such audited statements of operations, shareholders' equity and cash flows and the related balance sheets in clause (3)(i) hereof), (4) (i) the unaudited consolidated statement of operations of the Company for the fiscal quarters ending March 31, 2018, June 30, 2018 and September 30, 2018, and the related unaudited consolidated balance sheet as of the end of such fiscal quarters and (ii) the unaudited consolidated statement of operations of the Company for, and the related unaudited consolidated balance sheet as of the end of, each fiscal quarter of the Company (other than the fourth fiscal quarter) ended after the close of the fiscal quarter ending September 30, 2018 and at least 45 days prior to the Closing Date (each Initial Principal Investor acknowledges receipt of unaudited consolidated statements of operations and the related balance sheet in clause (4)(i) hereof) and (5) a pro forma consolidated balance sheet of the Company as of the last day of the most recently completed four-fiscal quarter period ended at least 45 days (or 120 days in case such period is the end of the Company's fiscal year) prior to the Closing Date, prepared after giving effect to the Transactions as if the Transactions had occurred as of such date and any other adjustments as agreed by the Sponsor and the Initial Principal Investors (which need not be prepared in compliance with Regulation S-X of the Securities Act of 1933, as amended, or include adjustments for purchase or recapitalization accounting (including adjustments of the type

contemplated by	v Financial	Accounting	Standards	Board	Accounting	Standards	Codification	805.	Business	Combinations	(formerly	SFAS	141R	(((

- 6. So long as requested at least ten (10) business days prior to the Closing Date, the Initial Principal Investors and the Trustee shall have received, at least three (3) business days prior to the Closing Date, all documentation and other information with respect to the Issuer and the Guarantors that is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including, without limitation, the PATRIOT Act. At least two (2) business days prior to the Closing Date (to the extent requested by the Trustee or any Initial Principal Investor at least ten (10) business days prior to the Closing Date), the Issuer or any Guarantor that qualifies as a "legal entity customer" under 31 C.F.R. § 1010.230 shall deliver one beneficial ownership certificate to the Trustee and the Initial Principal Investors, which certification shall be substantially similar in form and substance to the form of Certification Regarding Beneficial Owners of Legal Entity Customers published jointly, in May 2018, by the Loan Syndications and Trading Association and Securities Industry and Financial Markets Association (a "Beneficial Ownership Certificate"), in relation to the Issuer or such Guarantor.
- 7. Payment of all closing payments and expenses earned, due and payable to the Initial Principal Investors required to be paid on the Closing Date under the Second Lien Notes Documentation for which (in the case of expenses) invoices have been received at least three (3) business days in advance (which amounts may be (in the case of expenses) and shall be (in the case of the closing payment) netted against the proceeds of the Second Lien Notes) shall have been made (or shall be made substantially contemporaneously with funding).
- 8. Subject to the Certain Funds Provision, with respect to the Second Lien Notes, all actions necessary to establish that the Trustee will have a perfected security interest in the Collateral of the Issuer and each Guarantor under the Second Lien Notes Documentation shall have been taken (or shall be taken substantially contemporaneously with funding).
- 9. The Specified Merger Agreement Representations and the Specified Representations shall be true and correct in all material respects.

FORM OF SOLVENCY CERTIFICATE

[·],

This Solvency Certificate is being executed and delivered pursuant to Section [:] of that certain Note Purchase Agreement dated $[\cdot]$, 2019 among $[\cdot]$ (the "Note Purchase Agreement"); the terms defined therein being used herein as therein defined or in the indenture, dated as of $[\cdot]$, 2019, among $[\cdot]$ (the "Indenture").

I, [·], the [chief financial officer/equivalent officer] of the Issuer, solely in such capacity and not in an individual capacity, hereby certify that I am the [chief financial officer/equivalent officer] of the Issuer and that I am generally familiar with the businesses and assets of the Issuer and its Subsidiaries (taken as a whole), I have made such other investigations and inquiries as I have deemed appropriate and I am duly authorized to execute this Solvency Certificate on behalf of the Issuer pursuant to each of the Note Purchase Agreement.

I further certify, solely in my capacity as [chief financial officer/equivalent officer] of the Issuer, and not in my individual capacity, as of the date hereof and after giving effect to the Transactions and the incurrence of the indebtedness and obligations being incurred in connection with the Note Purchase Agreement, the Indenture and the Transactions on the date hereof, that, (a) the sum of the debt (including contingent liabilities) of the Issuer and its Subsidiaries, taken as a whole, does not exceed the present fair saleable value (on a going concern basis) of the assets of the Issuer and its Subsidiaries, taken as a whole; (b) the capital of the Issuer and its Subsidiaries, taken as a whole, contemplated as of the date hereof; and (c) the Issuer and its Subsidiaries, taken as a whole, do not intend to incur, or believe that they will incur, debts including current obligations beyond their ability to pay such debt as they mature in the ordinary course of business. For the purposes hereof, the amount of any contingent liability at any time shall be computed as the amount that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

[Remainder of page intentionally left blank]

 $IN\ WITNESS\ WHEREOF, I\ have\ executed\ this\ Solvency\ Certificate\ on\ the\ date\ first\ written\ above.$

By:

Name: [·]
Title: [·]

JOINT FILING AGREEMENT

In accordance with Rule 13d-1(k)(1) promulgated under the Securities Exchange Act of 1934, as amended, the undersigned hereby agree to the joint filing with each of the Reporting Persons (as such term is defined in the Schedule 13D referred to below) on behalf of each of them of a statement on Schedule 13D (including amendments thereto) with respect to the common stock, par value \$0.0001 per share, of BioScrip, Inc., a Delaware corporation, and that this agreement may be included as an exhibit to such joint filing.

IN WITNESS WHEREOF, the undersigned hereby execute this agreement as of March 15, 2019.

ASSF IV AIV B HOLDINGS, L.P.

By: ASSF Operating Manager IV, L.P.,

Its: Manager

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ASSF OPERATING MANAGER IV, L.P.

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT HOLDINGS L.P.

By: ARES HOLDCO LLC

Its: General Partner

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES HOLDCO LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES HOLDINGS INC.

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT CORPORATION

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES MANAGEMENT GP LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES VOTING LLC

By: ARES PARTNERS HOLDCO LLC

Its: Sole Member

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory

ARES PARTNERS HOLDCO LLC

/s/ Naseem Sagati Aghili

By: Naseem Sagati Aghili Its: Authorized Signatory